Donner Summit Public Utility District Board of Directors Special Meeting Agenda

Wednesday, July 27, 2022 - 6:00 P.M. DSPUD Office, 53823 Sherritt Lane, Soda Springs California

For the Special Meeting (Section 54956), July 27, 2022 at 6:00 P.M., the meeting will be conducted via Zoom due to the COVID-19 pandemic and in accordance with Governor Newsom's Executive Order N-29-20 and Assembly Bill 361. The Zoom Webinar ID 875 8385 6467 Password 063617 If your computer does not have audio or visual capabilities, you may participate in the meeting by phone 1 669-900-6833.

Any member of the public desiring to address the Board on any matter within the Jurisdictional Authority of the District or on a matter on the Agenda before or during the Boards consideration of that item may do so **by logging into Zoom and using the meeting ID and password or dial in only number referenced above**. After receiving recognition from the Board President, please give your Name and Address (City) and your comments or questions.

Unless specifically noted, the Board of Directors may act upon all items on the agenda.

- 1. Call to Order
- 2. Roll Call
- 3. Clear the Agenda
- **4. Public Participation** This is time set aside for the public to address the Board on any matter not on the agenda. Comments related to any item already on the agenda should be addressed at the time that that item is considered. Each speaker will be limited to five minutes, but speaker time may be reduced at the discretion of the Board President if there are a large number of speakers on any given subject.
- 5. Action Items
 - A. Adopt a Resolution to Amend and Restate the Money Purchase Pension Plan
- **6. Director Reports:** In accordance with Government Code Section 54954.2(a), Directors may make brief announcements or brief reports on their own activities (concerning the District). They may ask questions for clarification make a referral to staff or take action to have staff place a matter of business on a future agenda.
- 7. Adjournment

We certify that on July 26, 2022, a copy of this agenda was posted in public view at 53823 Sherritt Lane Soda Springs, California, and the Soda Springs/Norden Post Office.

Deleane Mehler Administrative Assistant Steven Palmer General Manager

July 27, 2022 Agenda Item: 5A



STAFF REPORT

TO: Board of Directors

PREPARED BY: Steven Palmer, PE, General Manager

APPROVED BY: Steven Palmer, PE, General Manager

SUBJECT: Adopt a Resolution Amending and Restating the Money Purchase Pension

Plan

RECOMMENDATION

Adopt a Resolution Amending and Restating the Money Purchase Pension Plan.

BACKGROUND

The Money Purchase Pension Plan (MPP) Administrator, NH Hicks, has requested that the Board adopt the attach resolution to meet the requirements of IRS Notice 2017-37 as required in IRS Revenue Procedure 2016-37. This action is required in order for the District to continuing offering the MPP as a qualified retirement plan.

DISCUSSION

This action is a restatement of existing MPP terms and conditions, and no changes are made. The MPP Plan Documents are included as Attachment 1.

ANALYSIS

If the Board does not adopt this resolution, then the IRS will no longer consider the MPP as a qualified plan.

FISCAL IMPACT

The District will be able to continue offering MPP to employees and results in no change to current fiscal impact. Fiscal impact of not continuing the MPP has not been quantified but is likely significant.

CEQA ASSESSMENT

This is not a CEQA Project

ATTACHMENTS

- 1. MPP Cycle 3 Plan Documents (electronic only)
- 2. MPP Cycle 3 Summary Plan Description
- 3. Resolution 2022-28

DONNER SUMMIT PUBLIC UTILITY DISTRICT MONEY PURCHASE PENSION PLAN

CYCLE 3 PLAN DOCUMENTS



Table of Contents

NH Hicks Contacts

Service Agreement

Power of Attorney Form 2848

Resolution Adopting Plan

Adoption Agreement

Basic Plan Document

Trust Agreement

IRS Opinion Letter

Loan Policy, if applicable

Thank you for choosing NH HICKS for your plan administration

For any administration questions, please contact:

Kim Hedrick, your Plan Administrator khedrick@nhhicks.com



Or

For local service, meetings and legal consulting, please contact:

Tom K. Hicks, Attorney at Law tomhicks@nhhicks.com



We also invite you to visit our website, www.nhhicks.com, for general FAQ's and to keep up to date on IRS and DOL regulations that may impact your plan.

SERVICE AGREEMENT

This Service Agreement ("Agreement") is entered into this ______ day of ______ 2022 by and between <u>DONNER SUMMIT PUBLIC UTILITY DISTRICT</u> ("Employer") and NH Hicks with respect to <u>DONNER SUMMIT PUBLIC UTILITY DISTRICT MONEY PURCHASE PENSION PLAN</u> ("Plan"). For all purposes of the Agreement, the Employer is the "Plan Sponsor", and the Plan Sponsor is deemed to be the "Plan Administrator" unless specified otherwise.

- 1. <u>NH HICKS Administrative Functions.</u> NH Hicks agrees to perform the following administrative functions for the Employer for fees to be paid:
 - 1.1 Preparation of IRS pre-approved Plan & Trust Documents (including Summary Plan Description) for defined contribution and/or defined benefit plans and for approval by the Employer and review by Employer's legal counsel. NH Hicks will also prepare restatement of the Employer's document when required;
 - 1.2 Verification of the rules for determining eligibility for participation in or benefits under the Plan;
 - 1.3 Performance of the required testing and presentation of correction methods so that Employer can take action to maintain the Plan's annual compliance;
 - 1.4 Calculation of Plan benefits and/or contribution allocations for the review and approval of the Employer;
 - 1.5 Preparation of participants' annual benefit and/or account statements and the Summary Annual Report;
 - 1.6 Annual reconciliation of distributions;
 - 1.7 Preparation of information required to be communicated to employee/participants by Employer;
 - 1.8 Preparation of reports required by Federal and State governmental agencies for review, execution and filing by the Employer, including Form 5500 and 1099R when applicable;
 - 1.9 Correspondence work with the IRS, DOL or other governmental agencies regarding plan audits or inquiries. Any time spent over two hours will be charged at current hourly rate of \$200.00.
 - 1.10 Upon Employer request, NH Hicks will prepare documents needed to terminate Plan
- 2. <u>Employer's Responsibilities.</u> It is understood that NH Hicks will assist the Employer in sustaining compliance, but is not responsible for the compliance of the Plan. The Employer understands that they have a duty to perform functions necessary for the Plan such as the following:
 - 2.1 Timely submission of information requested by NH Hicks. NH Hicks will generally state the deadline for returning information in written form. If NH Hicks does not receive all requested information within 8 months after the Plan year end, Employer recognizes that NH Hicks may not have sufficient time to prepare the Annual Return/Report forms (i.e. Form 5500). NH Hicks cannot be responsible for any late filings, penalties, fines, or taxes that result from Employer failure to provide NH Hicks with complete information on a timely basis.
 - 2.2 Timely depositing funds as needed. As regulations on deadlines may change, NH Hicks will assist Employer in determining any current deadlines upon request. Some of the deadlines under current regulation include:
 - Deductibility by the due date of Employers corporate return including extensions;
 - Minimum funding standards for pension plans within 8 ½ months of the end of the Plan year; and 401(k) salary deferrals and loan repayments should be made as soon as possible, but depending on the number of participants, generally within 7 days after the relevant payroll date.
 - 2.3 Meeting IRS filing deadlines. This includes the Employer being responsible for following NH Hicks' instructions to review, sign and file reports required by Federal and State governmental agencies including Form 5500 and 1099-R. Please note that some forms must be filed electronically.
 - 2.4 Ensuring NH HICKS is provided with copies of all necessary investment statements.
 - 2.5 Maintaining an ERISA/Fidelity bond as needed. NH Hicks will assist Employer in determining the required bond face amount upon request.
 - 2.6. Ensuring participants receive all required notices, elections, and reports provided to Employer by NH Hicks, including enrollment forms prior to meeting entry date, Summary Plan Description (SPD) within 90 days of plan entry, and any required Blackout Participant Notice (typically provided by Financial Institution and not NH Hicks). NH Hicks does not communicate directly with participants except when specifically authorized by Employer to do so. NH Hicks is not authorized to, and will not, independently initiate communications with participants. NH Hicks is not responsible for advising participants regarding their Plan options, any rights they have under the Plan, or any issues in relation to the Plan and its operations of which they should be aware.
 - 2.7 Notifying NH Hicks in writing of any participant terminations, loans or hardship requests. Employer is responsible to obtain participants' and, if required, spouses' signatures on all benefit distribution and loan forms. While NH

Hicks will assist Employer in preparing loan documentation and amortization schedules for participant loans, Employer is responsible for setting up procedures relating to making loans to participants from the Plan, including uploading the loan repayment schedule on Employer's payroll system, and obtaining timely repayment of the loan. While NH Hicks will assist Employer in preparing distribution documentation for hardship withdrawals, Employer is responsible for establishing procedures relating to approving and disbursing hardship distributions, including the evaluation of the participant's hardship to determine if it qualifies for a distribution, maintaining the documentation needed from the participant to support the claim that a hardship has occurred, ensuring any cessation of deferrals upon receipt of the hardship distribution, as well as notifying the participant when deferrals may resume.

- 2.8 Retention of a certified public accountant to audit the Plan if required. In that event, NH Hicks must await the completion of the audit before preparation of the Form 5500 can be finalized.
- As a fiduciary, Employer is responsible for monitoring the performance of any party providing services to the Plan. Employer must review the reports or other items that NH Hicks prepares for Employer on a regular basis, and notify NH Hicks immediately of any errors or inconsistencies that Employer identifies on any report, form, or other communication from NH HICKS. Employer is also responsible for ensuring that the fees paid with Plan assets for services are reasonable. Therefore, Employer is responsible for reviewing this Agreement and the contracts into which Employer enters on behalf of the Plan to ensure that Employer understands what is being paid to all of Employers service providers and that the amount being paid is reasonable. If NH Hicks were to recommend other service providers or financial institutions for the Plan, NH Hicks is not endorsing the performance or amount of any compensation paid to those providers or institutions, and should not be considered to be a substitute for Employer's own judgment as a plan fiduciary.
- 2.10 Maintenance of signed and dated copies of all plan documents at all times. These documents must be made available for inspection by participants and beneficiaries at Employer's principal office. In addition, Employer must furnish copies of these documents in writing if a participant so requests. Although NH Hicks keeps copies of documents, these copies are for NH Hicks' files.
- 2.11 Maintenance of information related to the preparation of the Forms 5500 and 1099-R, generally for at least six years after the forms are due. In addition, the DOL requires Employer to retain sufficient information to determine the benefits of the participants and beneficiaries. Although NH Hicks keeps copies of work performed, these copies are for NH Hicks' files. Therefore, Employer must retain copies of the work performed and the information sent by NH Hicks. Failure to do so can result in a civil penalty payable to the DOL.
- 3. <u>NH Hicks and Employer Acknowledgements and Representations.</u> It is understood and acknowledged by and between NH Hicks and Employer that:
 - 3.1 Neither this nor any other agreement shall relieve Employer or any other designated fiduciaries or other responsible persons providing services to the Plan of any responsibilities or liabilities imposed by the Employee Retirement Income Security Act of 1974 ("ERISA") and the Internal Revenue Code of 1986, as amended from time to time
 - 3.2 NH Hicks is not a fiduciary and shall not at any time under this Agreement or otherwise act in any capacity that is or may be construed as that of a fiduciary, or investment counselor to the Employer's Retirement Plan or Trust, or handle Plan assets in any manner. NH Hicks does not provide legal services or advice and makes no representations or warranties that its services have been reviewed or approved by legal counsel. NH HICKS does not provide tax advice. NH Hicks is not responsible for ensuring that Employer is in compliance with the requirements under ERISA Sections 404(c) or Labor Regulations Section 2550.404a-5. If the feature is in the Plan, NH Hicks is also not responsible for ensuring that the Employer's participants have been automatically enrolled, including necessary adjustments or for any qualified default investment alternative (QDIA) requirements.
 - 3.3 NH Hicks will annually request information from Employer, including employee census, investment statements, and updated employer information. Employer understands that NH Hicks will rely upon the information provided by the Employer for accuracy and shall not have any duty to question any action or direction of the Employer or, at Employer's direction, Employer's other advisors, and will have no responsibility to independently verify the accuracy of that information, including the value of trust investments and earnings. NH Hicks assumes no responsibility to acquire information other than to request it from the Employer, and will not be liable for any errors or omissions made as a result of incomplete or incorrect information that Employer furnishes to NH Hicks. Employer understands that NH Hicks will act in all matters only upon the express written direction of the Employer and will at no time exercise any discretion or independent authority with respect to the information received or the performance of the administrative services described in this Agreement, no such independent authority or discretionary rights having been granted to NH Hicks by the Employer under this or any other agreement.
 - 3.4 If the Plan offers to participants the option of self-directed brokerage accounts, NH Hicks is not responsible for monitoring the access to any such accounts, the Plan assets invested in such accounts, or the risks of violating ERISA that may arise, including but not limited to.

- Verifying that the self-directed brokerage account option has been made available in a manner that is nondiscriminatory;
- Monitoring such accounts for, or identifying prohibited transactions that may occur through the use of such accounts;
- Advising you regarding additional bonding requirements that may result from investments in anything other than "qualifying assets"; and
- Identifying and preparing tax forms in relation to Unrelated Business Taxable Income that may result from certain investments in such accounts.
- 3.5 Employer recognizes that if NH Hicks has taken over or is taking over as service provider of Employer's Plan as of the date of this Agreement, that NH Hicks is not responsible, nor will Employer hold NH HICKS responsible, for any prior administration of Employer's Plan or any other plans Employer may have maintained at any time before, during, or after NH Hicks's representation of Employer. Employer further recognizes that NH Hicks can only endeavor to provide administration services based on the information provided and cannot guarantee or advise on the current or future qualification status of this or any prior plans maintained by the Employer. Further, if Employer maintains other qualified plans not being administered by NH Hicks, Employer recognizes that NH Hicks cannot be held responsible for any combined limitations or top- heavy requirements that may affect the qualified plan status of the Plan or Plans being administered by NH Hicks.
- 3.6 Employer is responsible for informing NH Hicks in writing of other plans and any other entities which are in a controlled or affiliated service group with Employer, entities which may be subsequently acquired, of any leased or shared employees, and of any change in the entity type of the Employer. NH Hicks is not responsible for any penalties, plan disqualification or any other liability due to the failure of the Employer's fiduciaries to submit said information on a timely basis or due to acts or omissions to act by said fiduciaries.
- 3.7 Neither NH Hicks, nor any of its officers, employees or agents are liable for:
 - any inability to perform services or any costs or losses due to loss or corruption of data or other technology as the result of any causes outside NH Hicks' reasonable control or force majeure;
 - any direct or indirect losses or damages of any kind arising out of or in connection with this Agreement
 (whether general, special, consequential, incidental, exemplary or otherwise, regardless of the form
 of action and regardless if an authorized representative of NH Hicks has been advised or should have
 known of the possibility of such damages).

If any part of this limitation of liability is found to be invalid, inapplicable or unenforceable for any reason, then the aggregate liability of NH Hicks in such circumstances for liabilities that otherwise would have been limited shall not exceed the amount of fees that Employer paid NH Hicks for the plan year in which the damages occurred

3.8 Employer agrees to indemnify, defend and hold NH Hicks and each of its officers, employees, and agents harmless from and against all taxes, claims, losses, damages (including punitive), liabilities, attorney fees, costs (including court or outside CPA audit) and any other expenses of an kind whatsoever which may be incurred by or imposed upon Employer or NH Hicks, any participant under any of the Employer's Plans or upon any third party acting in any capacity in connection with the Employer's Plans.

4. <u>Fees and Terms of Payment</u>

- 4.1 The fee schedule is currently available at www.nhhicks.com or upon request. Employer also signed NH Hicks' Authorization Form which provides an estimate of the fees that will be incurred in the first plan year, based on the fee schedule and facts known about the Employer's plan. In addition, prior to each annual administration, NH Hicks will submit an invoice per the current fee schedule showing the amount of fees for the period. Once the annual administration is complete, NH Hicks will submit a final invoice (including any miscellaneous fees such as trust accounting or rush fees) if additional payment is due.
- 4.2 Plan termination fees, which are in addition to annual administration charges, will be quoted on an individual basis.
- 4.3 NH Hicks' fees are the joint obligation of the Employer and the Plan (to the extent permitted by ERISA).

5. <u>Terms of this Agreement</u>

- 5.1 Either NH Hicks or Employer shall have the right to terminate this Agreement at any time, without cause or notice; otherwise, this Agreement shall be in effect until terminated with written notice. The termination is effective upon receipt of notice.
- 5.2 NH Hicks may modify the terms of this Agreement upon thirty (30) days' prior notice to Employer. These modifications may not require a signature from Employer. Employer shall be bound by any such modification unless Employer terminates this Agreement in accordance with its terms.

- The rights and obligations of the Employer under this agreement shall insure to the benefit of, and shall be binding upon, the successors and assigns of the Employer.
- 5.4 This Agreement is entered into, or to be performed, in Chico, California, and shall be interpreted, construed, and enforced in accordance with California law.

The parties have duly executed this Service Agreement as of the date first written above.

EMPLOYER:	DONNER SUMMIT PUBLIC UTILITY DISTRICT
	Signature:
	Print Name:
	Title:
PLAN:	DONNER SUMMIT PUBLIC UTILITY DISTRICT MONEY PURCHASE PENSION PLAN
	Signature:
	Print Name:
	Title:
NH HICKS:	NH Hicks
	Tom K. Hicks, President

(Rev. February 2020) Department of the Treasury

Internal Revenue Service

and Declaration of Representative

Power of Attorney

▶ Go to www.irs.gov/Form2848 for instructions and the latest information.

For IRS Use Only Received by: Name Telephone

July 27, 2022 Agenda Item: 5A ATTACHMENT 1

Part I Power of Attorney Caution: A separate Form			ch taxpa	yer. Form 284	8 will not be he	onored	Telephone _ Function	
for any purpose other than	•		0 1	_			Date /	
1 Taxpayer Information. Taxpayer	must sign and date this	torm on pa	ge 2, iine		- 4:£: 4:			
Taxpayer name and address Donner Summit Public Utility	District			. ,	ntification numb	er(s)		
53823 Sherritt Lane	DISCITCE				1602556	T _D	, , , , , , , , , , , , , , , , , , ,	
Soda Springs	CA 95728			Daytime telep (530) 426-		001	umber (if applic	able)
hereby appoints the following representat	tive(s) as attorney(s)-in-f	fact:						
2 Representative(s) must sign and	d date this form on page	2, Part II.						
Name and address				CAF No.	0900)5-69207F	8	
Tom K Hicks				PTIN				
1294 East First Ave. Suite 11	.0			Telephone	No. (530)) 891-49	75	
Chico	CA 95926	_		Fax No.	(530)) 891-41	L85	
Check if to be sent notices and commi	unications		Check i	f new:Address				. Ц
Name and address					••••••			
					••••••			
					No		••••••	
				Fax No.	······································		······································	
Check if to be sent notices and commi	unications		Check	f new:Address				<u> </u>
Name and address					••••••			
				PIIN		•••••••	••••••	
					No			
(Note. IRS sends notices and communications to	only two representatives)		Check i	f new:Address	Telenh	none No.		
	only two representatives.		Officer					<u> </u>
Name and address					••••••			
					No			
					•••••••••••••••			
(Note. IRS sends notices and communications to	only two representatives.)		Check i	f new:Address	Teleph	one No.	Fax No.	
to represent the taxpayer before the Inter	nal Revenue Service an	nd perform t	the follov	ving acts:			-	
3 Acts authorized (you are required to inspect my confidential tax information a shall have the authority to sign any agree	and to perform acts that I can	n perform with	n respect to	the tax matters d	lescribed below. F	or example, m	ny representative(s	
Description of Matter (Income, Employr				Tax Form Numb	oor	Voor(a) 6- 1	Pariod(s) (if annliss	ablo)
Whistleblower, Practitioner Discipline 4980H Shared Responsibility Pay			(1040,	941, 720, etc.) (if			Period(s) (if applica ee instructions)	able)
Donner Summit Public Utility	District Money		5500,	945, 1099R		2022-202	24	
Purchase Pension Plan								
4 Specific use not recorded on C check this box. See Line 4. Spec					-			AF,
·								• F _
5a Additional acts authorized. In a instructions for line 5a for more in X Authorize disclosure to third p	nformation): Access	my IRS rec	cords via		e Service Pròvi		tne following act	is (see
Authorize disclosure to third p	arues, <u>A</u> Supsillu	ale or add f	epresent	auve(s), <u>←</u> Si	gir a returri,	-		
Other acts authorized:								
			_					

	b Specific acts not authorized. My representative(s) is (are) not authorized to endorse or otherwise negotiate a accepting payment by any means, electronic or otherwise, into an account owned or controlled by the represer entity with whom the representative(s) is (are) associated) issued by the government in respect of a federal tax List any other specific deletions to the acts otherwise authorized in this power of attorney (see instructions for I	ntative(s) or any firm or other liability.
6	Retention/revocation of prior power(s) of attorney. The filing of this power of attorney automatically revokes attorney on file with the Internal Revenue Service for the same matters and years or periods covered by this do to revoke a prior power of attorney, check here	ocument. If you do not want
7	Signature of taxpayer. if a tax matter concerns a year in which a joint return was filed, each spouse must file a even if they are appointing the same representative(s). If signed by a corporate officer, partner, guardian, tax in representative (or designated individual, if applicable), executor, receiver, administrator, or trustee on behalf of the legal authority to execute this form on behalf of the taxpayer.	natters partner, partnership
	▶ IF NOT COMPLETED, SIGNED, AND DATED, THE IRS WILL RETURN THIS POWER OF ATT	ORNEY TO THE TAXPAYER.
•••••	Signature Date Title	e (if applicable)
•••••	Print Name Print name of taypayer f	rom line 1 if other than individual

Part II Declaration of Representative

Under penalties of perjury, by my signature below,I declare that:

- •I am not currently suspended or disbarred from practice, or ineligible for practice, before the Internal Revenue Service;
- •I am subject to regulations contained in Circular 230 (31 CFR, Subtitle A, Part 10), as amended, concerning practice before the Internal Revenue Service;
- ■I am authorized to represent the taxpayer identified in Part I for the matter(s) specified there; and
- I am one of the following:
- a Attorney--a member in good standing of the bar of the highest court of the jurisdiction shown below.
- b Certified Public Accountant--a holder of an active license to practice as a certified public accountant in the jurisdiction shown below.
- c Enrolled Agent--enrolled as an agent by the IRS per the requirements of Circular 230.
- d Officer--a bona fide officer of the taxpayer organization.
- e Full-Time Employee -- a full-time employee of the taxpayer.
- f Family Member--a member of the taxpayer's immediate family (spouse, parent, child, grandparent, grandchild, step-parent, step-child, brother, or sister).
- g Enrolled Actuary--enrolled as an actuary by the Joint Board for the Enrollment of Actuaries under 29 U.S.C. 1242 (the authority to practice before the IRS is limited by section 10.3(d) of Circular 230).
- h Unenrolled Return Preparer--Authority to practice before the IRS is limited. An unenrolled return preparer may represent, provided the preparer (1) prepared and signed the return or claim for refund (or prepared if there is no signature space on the form); (2) was eligible to sign the return or claim for refund; (3) has a valid PTIN; and (4) possesses the required Annual Filing Season Program Record of Completion(s). See Special Rules and Requirements for Unenrolled Return Preparers in the instructions for additional information.
- k Qualifying Student--receives permission to represent taxpayers before the IRS by virtue of his/her status as a law, business, or accounting student working in an LITC or STCP. See instructions for Part II for additional information and requirements.
- r Enrolled Retirement Plan Agent--enrolled as a retirement plan agent under the requirements of Circular 230 (the authority to practice before the Internal Revenue Service is limited by section 10.3(e)).

▶ IF THIS DECLARATION OF REPRESENTATIVE IS NOT COMPLETED, SIGNED, AND DATED, THE IRS WILL RETURN THE POWER OF ATTORNEY. REPRESENTATIVES MUST SIGN IN THE ORDER LISTED IN PART I, LINE 2.

Note: For designations d-f, enter your title, position, or relationship to the taxpayer in the "Licensing jurisdiction" column.

Designation Insert above letter (a-r).	Licensing jurisdiction (State) or other licensing authority (if applicable).	Bar, license, certification, registration, or enrollment number (if applicable).	Signature	Date
a	CA	159625		01/01/2022

ADOPTION AGREEMENT FOR THE DATAIR MASS-SUBMITTER PRE-APPROVED DOCUMENT 20-001

.....

DEFINED CONTRIBUTION PRE-APPROVED ADOPTION AGREEMENT NON-STANDARDIZED ALTERNATIVE THREE: PROFIT SHARING PLAN

Note: Section references are to the Plan or the Trust. All capitalized terms are defined in the Plan or Trust. All dates are entered in the MM/DD or MM/DD/YYYY format, as applicable.

The DATAIR Pre-Approved Adoption Agreement Defined Contribution Plan is hereby adopted by

Donner Summit Public Utility District

(the "Employer").

The Plan shall be known as:

Donner Summit Public Utility District Money Purchase Pension Plan the "Plan".

A. General Plan Information

Emplo	oyer	Info	ormation				
A1.			er Address: 53823 Sherritt Lane , Soda Springs, CA 95728				
A2.	Employer Telephone: (530) 426-3456						
A3.			ver Tax ID (EIN): 94-1602556				
A4.			yer Fiscal Year End: 06/30				
A5.			Organization of Employer:				
			Sole Proprietorship				
	b.		Partnership				
	C.		C Corporation				
	d.		S Corporation				
			Limited Liability Company (LLC)				
	f.		Limited Liability Partnership (LLP)				
	g.		Not for Profit Corporation				
	h.		Professional Service Corporation				
	i.	X	Other (Must be legal entity recognized under federal income tax laws.): Tax Exempt				
			Government				
A6.	Bu	sine	ss Code (as used on Form 5500; 6-digit NAICS): 221100				
A7.			f Legal Construction: California				
A8.			usiness Commenced: January 1, 1947				
A9.	Ме	mbe	rs of a Controlled Group or Affiliated Service Group (select all that apply):				
	a.		Controlled Group (List controlled group members.):				
	b.		Affiliated Service Group (List affiliated service group members.):				
	No	te: T	his list is optional and for informational purposes only. Controlled Group and Affiliated Service				
	Gro	oup r	members who are also Participating Employers will sign this Adoption Agreement, a separate				
	Add	optio	n Agreement, or a Participating Employer Agreement.				
DI I							

A10. Three Digit Plan Number: 001

A11.	Plan Age	
	Name:	
		53823 Sherritt Lane
	City: State:	Soda Springs CA
	Zip:	95728
		(530) 426-3456
A12.	Plan Adr	ministrator:
	Name:	
		53823 Sherritt Lane
	City: State:	Soda Springs CA
	Zip:	95728
	Phone:	(530) 426-3456
A13.	Plan Adr	ministrator ID Number: 94-1602556
		ermination Letter Date (Leave blank for a new plan.): 06/30/2020
A15.	IRS File	Folder Number (Leave blank for a new plan.): 31799858520-001
	Informati	
A16.	Plan Tru	• •
		pplicable, the Plan is funded solely by Insurance Contract(s).
	Name:	Cathy A. Preis 53823 Sherritt Lane
	City:	Soda Springs
	State:	CA
		95728-
	Phone:	(530) 426-3456
	Namai	Philip Comish
	Name:	Philip Gamick 53823 Sherritt Lane
	City:	Soda Springs
	State:	CA
	Zip:	<u>95728-</u>
	Phone:	(530) 426-3456
	Name:	
	Address:	
	City:	
	State:	
	Zip:	
A 4 7	Phone:	Number of coords
		Number: 68-6090198 tee(s) must sign the Adoption Agreement, discretionary amendments, and interim
A 10.	amendm	
		No No
	. =	Yes
	_	executed copy of the Trust Agreement must be attached to this Plan. The Plan and Trust
		ent must be read and construed together. The powers, rights, and responsibilities of the
		shall be those specified in the Trust Agreement.
A19.		are two or more Trustees appointed, they will be bound by the following (select one).
		Not applicable.
	. —	The act of a majority of the Trustees.
	_	The act of 1 individuals act in the capacity of Trustee.
	<u>ப</u>	The dot of _ marriadals dot in the capacity of Trustee.
Partic	ipating E	imployers
400	Daniel alia	etion Franciscom and Multiple Franciscon Dieu (autort aus)

A20. Participating Employers and Multiple Employer Plan (select one).

a. \boxtimes This Plan is not a Multiple Employer Plan and no Related Employers are adopting the Plan.

	b.		The Employer is the Lead Sponsor of a Plan that includes Participatir are Participating Employers who are not Related Employers, this Plan Plan under Code sec. 413(c).						
	C.		Note		ployers	s will need	d to sign this	Adoption Agre	Sponsor is eement, a Participating
В. (Gen	eral	Plai	n Features					
		Date							
В1.	Effe a.	ctive The		e s ctive Date of the Pla	an is .	Tanuary	1. 2022		
		Note	: Ma	y not be earlier tha	n the i	first day o	f the initial P	lan Year or for	restatement of an existing
	L C			first day of the curi			المحمامية الثر	aa Effaatiya De	ata far tha Dlan pravision
	D. S			ective Dates. The f (select one).	Ollowir	ng dates v	viii repiace ti	he Ellective Da	ate for the Plan provision
		1.		Not Applicable					
		2.		The following spec	cial Eff	fective Da	tes will apply	y (select all tha	at apply):
				☐ Safe Harbor	Contri	butions:		/	′/
		Note	B. The	date(s) entered n	_: 12V no	t he earlie	r than the F	ffective Date of	'/ of the Plan
	c.			ective Dates <i>(selec</i>			i ilian ilio L	noonvo Bato o	r trio i lari.
		1.		Not applicable, thi	s is a	new Plan.			
		2.		The Plan was initi	ally eff	ective: <u>J</u> a	nuary 1,	<u>1981</u> .	
	d.	Rest		ent (select one).	o io o	now Dian			
		2.		Not applicable, thi This Plan Docume				nt of a nre-exis	ting plan
		3.	\boxtimes	This Plan Docume		•		•	• .
	e.		en P	lan			,	'	51
		1.		Not Applicable					
		2.		The Plan is/was fr new Participants v					uo to any ovietina
				nt, on or after the o					de to arry existing
Con	tribu	ıtions							
				(Select one for ea	ach co	ntribution	source.)		
					Not D	ermitted	Permitted	Formerly	Date Removed
	a.	Non-	-Elec	tive Contributions	NOLF		⊠	_	
	b.	Volu	ntarv	After-Tax Contribu	utions		_		//
	c.		•	oor Contributions		⊠			//
	d.			y Wage Contributio	ne				//
	u.	1 160	annıç	, wage continuation	113	\boxtimes			//
В3.	Dee	med							
	a.		_	IRAs are permitted	(sele	ct one).			
		1. 2.	\square	No.					
		4.		Yes.					

C. Eligibility

C1.	Excluded	Classes	of Em	ployees
-----	----------	---------	-------	---------

The o	class	es of	f Employees specified below will not be eligible to be a Participant in the Plan (select
one).			
a.		No E	Exclusions
b.	\times	The	following exclusions will apply (select all that apply):
	1.	×	Union Employees
:	2.	\times	Non-Resident Aliens
	3.	\times	Employees acquired in a Code section 410(b)(6)(C) transaction
	4.		HCEs
	5.		Key Employees
	6.		Leased Employees
,	7.		HCEs who are Key Employees
	8.		Self-Employed Individuals
!	9.		Compensation is based solely on commissions
	10.		Employees not covered by a CBA with:
	11.		Other exclusion:
Note.	: The	exc	lusions selected cannot result in the group of NHCEs participating under the Plan being

Note: The exclusions selected cannot result in the group of NHCEs participating under the Plan being only those NHCEs with the lowest amount of compensation and/or the shortest periods of Service and who may represent the minimum number of these Employees necessary to satisfy coverage under Code section 410(b).

Note: Selection of any options other than a, b1, b2, b3, or b4 will result in the plan having to satisfy coverage testing under Code section 410(b).

Note: The definition of "other exclusion" provided must be objectively determinable, may not be specified in a manner that is subject to Plan Administrator discretion, and may not discriminate in favor of HCEs. The Plan Administrator will make all determinations in connection with such issues in a uniform, nondiscriminatory manner.

Note: The definition of "other exclusion" cannot result in an exclusion based on age or length of service or create an indirect service requirement.

C2. Age Requirement for Eligibility (Select one.)

- **b.** \boxtimes Employees must be at least $\underbrace{18.0}$ years of age to be eligible to participate in the Plan. Note: Age cannot be greater than age $\underbrace{21.}$ If the Plan is maintained exclusively for employees of a Code section 170(b)(a)(A)(ii) educational institution and Participants are 100% vested no later than upon attainment of one Year of Vesting Service, the age can be no greater than 26.

C3.	Emp	ploye	Requirement for Eligibility es who meet the service requirements specified below during the Eligibility Computation ill be eligible to participate in the specified portion of the Plan (select one). No service required Service required (select one):
		1.	☐ One Year of Eligibility Service
		2.	☐ Two Years of Eligibility Service
		3.	months of Eligibility Service
		4. -	consecutive months of Eligibility Service
		5. 6.	Years of Eligibility Service with Hours of Eligibility Service in each year
		7.	months of Eligibility Service with Hours of Eligibility Service in each month Other(e.g., 6 consecutive months of service with 500 hours of service earned over the 6-
		7.	Other(e.g.,6 consecutive months of service with 500 hours of service earned over the 6-month time frame):
	requ	uired,	e Years of Eligibility Service required cannot exceed 2. If more than 1 year of service is the Participant must be 100% vested in their Non-Elective Contributions at two Years of Service.
	Not requ	e: Th uired,	e months of service required cannot exceed 24. If more than 12 months of service are the Participant must be 100% vested in their Non-Elective Contributions at 24 months of Service.
	Not be s	e: Th specif or of I	e definition of "other service requirement" provided must be objectively determinable, may not fied in a manner that is subject to Plan Administrator discretion, and may not discriminate in HCEs. The Plan Administrator will make all determinations in connection with such issues in a nondiscriminatory manner.
	Not the thai	e: Ho hours n the	ours of service failsafe: Regardless of the election made above, if eligibility is determined using sof service method, an Employee will be deemed to meet the eligibility requirement no later end of an Eligibility Computation Period during which the Employee completes 1,000 Hours of Service; provided, that the individual is an Eligible Employee on the applicable entry date.
C4.			ce requirements that include an hours requirement a Participant will have met the service ent at the following time (select one).
	a.	\times	As soon as they have met the service requirement regardless if the entire period has passed.
	b.		At the end of the measurement period in which the service requirement was met.
C5.	Pre	vailin es. Th	ng Wage Eligibility - Excluded Classes g Wage Contributions will have no age or service requirements and will have immediate entry ne following Participants will be excluded from receiving Prevailing Wage Contributions (select
	a.		No excluded classes.
	b.		The following Participants will be excluded (select all that apply):
		1.	HCEs.
		2 .	☐ Other: e: The "Other" excluded classes must be objectively determinable, may not be specified in a
		mar HCl	e: The Other excluded classes must be objectively determinable, may not be specified in a nner that is subject to Plan Administrator discretion, and may not discriminate in favor of Es. The Plan Administrator will make all determinations in connection with such issues in a orm, nondiscriminatory manner.

C6.	. One-Time Election				
					nake a one-time election pursuant to Plan Section 2.1.6 to not become a ne Plan (select one).
	a.		No.		
	b.	X	Yes	•	
		ng S			
C7.					Service
	a.				asuring a Year of Eligibility Service is the following <i>(select one).</i> cable, there are no service requirements for any source under the Plan.
	b.	⊠	Hou	ırs of	service method. A Year of Eligibility Service is <u>1000</u> hours (not more than 1,000) gibility Computation Period.
	c.		Elap	sed	time method.
	Not	e: If c	ption	b is	not selected skip questions C8 – C9. If option a is selected skip question C10.
C8.	Elig	ibilit	y Co	mput	ation Period
	If th mor	e hou oth pe	irs of eriod	serv endir	ce method is selected above, the first Eligibility Computation Period is the 12- g on the first anniversary of the Employee's Employment Commencement Date. ity Computation Periods will be measured as follows (select one).
	a.				nain a 12-month period based on the anniversary of the Employee's Employment cement Date.
	b.	X			tch the Plan Years beginning with the first Plan Year commencing prior to the first ary of the Employee's Employment Commencement Date.
C9.	Rı	reak i	in Fli	aibil	ty Service
33.	If t fai <i>No</i> wi	the he ils to o <i>te: It</i>	ours o comp f actu define	of sei plete pal ho ed as	vice method is selected above, a Break in Eligibility Service occurs if an Employee (not more than 500) Hours of Service in an Eligibility Computation Period. Turns are being used and no Hours of Service are completed then a Break in Service the lesser of 500 hours or the number of Hours of Service needed to earn a Year
C10	. Ес	quiva	lenci	ies	
	a.	Ηοι			
	If the hours of service method is selected above, when records of hours are not maintained the following equivalency will apply (select one).				
		1.		-	applicable, actual hours will be used for all Employees.
		2.	\times		following equivalency will apply (select one).
			A.	\times	Days worked (10 hours)
			В.		Weeks worked (45 hours)
			С.		Semi-Monthly or Bi-weekly Payroll period worked (95 hours)
			D.		Months worked (190 hours)
	b.	Elap			time method is selected above, eligibility service will be measured using the
					ect one).
		1.		•	exact date in years.
		2.			exact date in months.
		3.			endar months with Employee granted a month of service if they work at least one
			_		r of Service in that month.
		4. 5			nearest calendar month.
		5.		Con	pleted calendar months.

C11.	Service with Predecessor and Prior Employers
	Service with Predecessor Employers must be treated as service for the
	maintains the plan of the Predecessor Employer. In all other cases, serv
	Employer or a Prior Employer is granted as specified below.

	ma	aintai	ns th	Predecessor Employers must be treated as service for the Employer if the Employer e plan of the Predecessor Employer. In all other cases, service with a Predecessor a Prior Employer is granted as specified below.		
		a. Predecessor or Prior Employer Service (Select one.)				
		1.	\times	No service with a Predecessor or Prior Employer will be credited.		
		2.	□ A .	Service with a Predecessor or Prior Employer is credited as specified below. Service with the following entities shall be credited as service under this Plan:		
			В.	Service with the above entities has been determined under the terms of the following documents, if any:		
	b.	Serv	/ice v	vith a Predecessor or Prior Employer will be credited for the following purposes (select		
				oply):		
		1.		Eligibility Service.		
		2.		Vesting Service.		
		3.		Contribution Allocations.		
		4.		Attainment of Early or Normal Retirement Age.		
	C.	Ser	/ice v	vith a Predecessor or Prior Employer will be limited as follows:		
C12	. Aç	ge an		rvice Waiver		
	a.			age and service requirements will be waived for Employees Employed on //		
	b.	The	follo	wing requirements will be waived (select all that apply).		
		1.		The age requirement.		
		2.		The service requirement.		
	C.		oloye	es subject to the waiver will enter the Plan on <i>(select one):</i>		
		1.		The Effective Date of this document.		
		2.		The next Plan Entry Date for the applicable contribution source.		
		3.		Other (e.g. specific date): (date must be prior to next Plan Entry Date).		
	d.			wing limitations will apply to the age and/or service waiver (select one).		
		1.		No limitations.		
		2.		Other (e.g., only Employees hired due to merger with Acme Inc. will be subject to the waiver):		
Plar	n Ent	ry				
	. Er	ntry C				
				who meet the service requirements specified during the Eligibility Computation Period		
			-	le to participate in the Plan <i>(select one).</i>		
		\times	Dail	•		
				ekly on the (e.g., first) day of the week		
				nthly on the (e.g., last) day of the month		
				arterly starting on/		
	е.	_	Sen	ni-annually starting on/		
	f.		First	t Day of the Plan Year		
	_		Last	t Day of the Plan Year		
			Ann	iversary Date of the Employee's hire date		
	i.			er Entry Date(s) (e.g. payroll date):		
	No	ote: If	optic	on a is selected skip question C14.		

C14.	Entry Time Employees will be eligible to participate in the specified portion of the Plan on the Entry Date the date they meet the eligibility requirements specified for the contribution source (select one). a. coincident with or next following							
	b. next following c. nearest to							
C15.	Special Entry Dates In addition to the Entry Dates specified above the following dates will also be Entry Dates (select one). a. ☑ Not Applicable b. ☐ Special Entry Date(s) (e.g., 2/1/2018): Note: The special Entry Dates must be objectively determinable, may not be specified in a manner that is subject to Plan Administrator discretion, and may not discriminate in favor of HCEs. The Plan Administrator will make all determinations in connection with such issues in a uniform, nondiscriminatory manner.							
D.	Dates and Definitions							
Gene D1.	ral Dates Plan Year a. The Plan Year (must be 12 consecutive months.) 07/01 to 06/30. b. ☐ The Plan has a short Plan Year of:/ to/							
D2.	Limitation Year The Limitation Year is the (select one): a. ☑ Plan Year. b. ☐ calendar year coinciding with or ending within the Plan Year. c. ☐ 12 consecutive month period ending/ d. ☐ Employer's Fiscal Year ending with or within Plan Year. e. ☐ 12 consecutive month period ending (e.g., the last Friday in February) with or within Plan Year.							
D3.	Anniversary Date The Anniversary Date is (select one): a. ☑ the last day of the Plan Year. b. ☐ the first day of the Plan Year. c. ☐/ of each Plan Year. d. ☐ other (must be at least annually):							

D4.	Valuation Date								
		The Valuation Date is the (select all that apply):							
	_	Last day of the Plan Year							
	· 	Semiannually							
		Quarterly							
		Monthly							
	e. 🗆	·							
	. П	Other (i.e., assets held in XYZ fund will be valuated on the 15th day of each calendar month, must be at least annually):							
Retire	ement A	ge							
D5.		Retirement Age (NRA)							
		h Participant the Normal Retirement Age is <i>(select all that apply):</i>							
	a. 🗆	The later of age 65 or the 5th anniversary of participation in the Plan (statutory).							
		Age <u>65</u> .							
		Age and Years of Service.							
		Age and Years of Participation.							
		Age and the anniversary of actual participation in the Plan.							
	т. 📙	Other: , but in no event later than the later of age 65 or the 5th anniversary of							
	Note: T	participation. he age specified must be at least 55 and cannot exceed 65. If an age less than 62 is							
	specified, the IRS Opinion Letter cannot be used to show that the age is reasonably representative								
	of the typical retirement age for the industry in which the Participants work. Note: The service specified cannot be greater than 5 years.								
	Note: If more than one option is selected, NRA is attained on the first date the requirements of any								
	option a For this	are met, but in no event later than age 65 and the 5th anniversary of participation in the Plan. purpose only, participation is assumed to commence as of the first day of the first Plan Year							
		n the Employee became a Participant. he definition of "Normal Retirement Age" provided must be objectively determinable, may not							
	be spec	rified in a manner that is subject to Plan Administrator discretion, and may not discriminate in HCEs. The Plan Administrator will make all determinations in connection with such issues in							
		m, nondiscriminatory manner.							
D6.	Normal	Retirement Date							
		rmal Retirement Date is (select one):							
	a. ⊠	The actual date NRA is attained.							
	b. 🔲	The first day of the month in which NRA is attained.							
	c. 🔲	The first day of the month nearest the date NRA is attained.							
	d. 🔲	The first day of the month coincident with or next following the date NRA is attained.							
	е. 🔲	Anniversary Date nearest the date NRA is attained.							
	f. 🔲	Anniversary Date coincident with or next preceding the date NRA is attained.							
	g. 🔲	The last day of the month in which NRA is attained.							
	h. 🔲	Other (e.g., The last day of the month nearest the date NRA is attained):							

D7.		arly Retirement Age (ERA)						
				rticipant, the Early Retirement Age is (select all that apply):				
				Plan does not provide an ERA.				
				·				
				and Years of Service.				
				and Years of Participation.				
				and the anniversary of actual participation in the Plan.				
	f.		Oth	er (e.g., Age 55 and the fifth anniversary of employment):				
			•	on a is selected skip question D8.				
				event shall ERA exceed NRA.				
				e than one option is selected, the Participant attains ERA at the earliest age when any of requirements are satisfied.				
				efinition of "Early Retirement Age" provided must be objectively determinable, may not				
				in a manner that is subject to Plan Administrator discretion, and may not discriminate in				
				Es. The Plan Administrator will make all determinations in connection with such issues in				
	aι	ınifo	rm, no	ondiscriminatory manner.				
Do	- -	whe E) a tiva	mont Data				
D8.				ement Date etirement Date is (select one):				
			-	actual date ERA is attained.				
		_		first day of the month in which ERA is attained.				
				first day of the month nearest the date ERA is attained.				
				first day of the month coincident with or next following the date ERA is attained.				
				iversary Date nearest the date ERA is attained.				
				iversary Date coincident with or next following the date ERA is attained.				
			The	last day of the month in which ERA is attained.				
	h.		Oth	er (e.g., The last day of the month nearest the date ERA is attained):				
Cana	" -	Dafi.	. 141	•				
Gene D9.				pensated Employee (HCE) Determination				
Da.				I be determined using the top-paid group election (select one).				
	۵.	1.	_5 v.	No				
		2.		Yes				
	h			I be determined using the calendar year data election <i>(select one).</i>				
	ν.	1.	_3 w	No				
		2.		Yes				
	_		_	termining HCEs, Compensation will be measured over the following period <i>(select one).</i>				
	C.	1.	⊠	The preceding Plan Year.				
		2.	_	·				
				The calendar year beginning within the preceding Plan Year (non-calendar year plans only).				
		3.		The 12-month period ending/ (Select this option when using the Plan Year of				
			ш	another plan of the Employer.)				

D10.		Disability Disability									
			-	e or Participant has a "Disability" and is "Disabled" if (select all that apply):							
	a.			disability benefits are provided and there are no disability-related vesting or waiver visions.							
	b.	X	may	Participant suffers from a medically determinable physical or mental impairment that be expected to result in death or to last for a continuous period of not less than 12 on this and that renders him incapable of performing his duties.							
	C.	\times		Social Security Administration has determined that the Participant is eligible to receive ial Security disability benefits.							
	d.			Participant has begun to receive payments under the long-term disability program or a parable disability program maintained by the Employer.							
	e.		Othe	er:							
			more	e than one option is selected, a Participant is Disabled on the first date the requirements are met.							
	No sp	te: T ecifie	he de d in a	efinition of "Disability" provided must be objectively determinable and may not be a manner that is subject to Plan Administrator discretion. The Plan Administrator will be serminations in connection with such issues in a uniform, nondiscriminatory manner.							
D11.	De	sign	ated	Beneficiary							
		a. Default Beneficiary									
				ent the Participant does not have a valid Designated Beneficiary, the Designated ary will be <i>(select one):</i>							
				determined under Plan Section 3.6.2.							
		2.		the Participant's spouse; if no spouse, the Participant's estate.							
		3.	×	other (e.g., the Participant's spouse; if no spouse the Participant's children; if no children the Participant's estate): The participant's spouse; if no							
				spouse, the participant's children; if no children, the							
				participant's estate.							
	Note: If the Plan is a REA safe harbor plan the Participant's spouse must be the first defa Designated Beneficiary.										
	b.			on of Beneficiary Designation							
		1.		pant's valid beneficiary designation will expire when (select one):							
		2.		designations will never expire.							
		۷.	X	upon divorce the ex-spouse shall be treated as having predeceased the Participant unless a beneficiary designation is completed post-divorce naming the ex-spouse as a Beneficiary.							
		3.		other (e.g., marriage of the Participant):							
		Note		Beneficiary designation cannot expire upon legal separation.							
	c.			r Marriage Rule							
				Year Marriage rule will apply (select one):							
		1.	\boxtimes	No.							
		2.		Yes.							

E. Compensation

	· · · · · · ·	, on out on						
E1.	A Partio a. ⊠ b. □ c. □	refinition cipant's Compensation is based on (select one): W-2 Compensation 3401(a) Compensation 415(c)(3) Compensation Simplified 415 Compensation						
Exclu E2.	b. S. A.	rom Compensation ion of Deferrals ected deferrals will be excluded from the definition of Compensation for the purposes ad (select one). All of the items listed below. The following items (select all that apply): 125 (Cafeteria Plan) 132(f)(4) (Transportation) 402(e)(3) (401(k) and 403(b) deferrals) 402(h)(1)(B) (SEP deferrals) 408(p) (Simple Retirement Account 402(k) deferrals) 457(b) deferrals Il salary deferrals must be included or excluded to maintain a Code sec. 414(s) safe harbor on of Compensation. Exclusion of less than the full list will necessitate testing under Code 414(s) to ensure the definition of Compensation is not discriminatory. Ithe item is not selected it will be included in the definition of Compensation for the applicable ution source.						
E3.	Non-Participating Non-Resident Aliens Provided the compensation is excluded from gross income and not effectively connected with a U.S. trade or business, compensation paid to nonresident aliens who are not Participants will be excluded from the definition of Compensation (select one). a. No. No. Yes.							

E4. Other Exclusions

The selected items will be excluded from the definition of Compensation for the purposes specified (select one).

a.	×	No other exclusions						
b.		The	following items (select all that apply):					
	1.		Overtime					
	2.		Commissions					
	3.		Discretionary bonuses					
	4.		Bonuses					
	5.		Taxable employee benefits					
	6.		Compensation paid by an Employer who is not a Participating Employer					
	7.		In excess of \$					
	8.		Other exclusion (e.g. compensation earned under a prevailing wage agreement):					

Note: The definition of "other exclusion" provided must be objectively determinable, may not be specified in a manner that is subject to Plan Administrator discretion, and may not discriminate in favor of HCEs. The Plan Administrator will make all determinations in connection with such issues in a uniform, nondiscriminatory manner.

Note: Any exclusion in this section aside from taxable employee benefits will necessitate testing under Code section 414(s) to ensure the definition of Compensation is not discriminatory.

Inclusions in Compensation

E5. Other Inclusions

The selected items will be included in the definition of Compensation for the purposes specified (select one).

_	_				
а.		INIA	other	incli	icione
~ .		HAO	OHIGH	пп	เอเบเอ

b.	\times	The following ite	ems (select all that apply):

- 1.

 Earned before entry date
- 2. Deemed 125 Compensation
- 4.

 Post-Severance Compensation
- 5.

 Differential Wage Payments
- 6. Military Continuation Payments
- 7. Unfunded Deferred Compensation Plan

Note: The definition of "other inclusions" provided must be objectively determinable, may not be specified in a manner that is subject to Plan Administrator discretion, and may not discriminate in favor of HCEs. The Plan Administrator will make all determinations in connection with such issues in a uniform, nondiscriminatory manner.

Note: Elections made other than items a, b1, b2, or b3 will necessitate testing under Code section 414(s) to ensure the definition of Compensation is not discriminatory.

Other Compensation Salary Continuation for Disabled Participants Salary paid to Participants who are disabled is (select one): a. 🗵 Excluded. **b.** \square Included for all Participants. **c.** \square Included for all Participants who are not HCEs. **Compensation Computation Period** The Compensation Computation Period is the following period that ends with or within the Plan Year (select one). a. Plan Year **b.** \square Limitation Year c. Calendar year d. Pay period e. Monthly period f. Quarterly period **9**⋅ □ Semi-Annually h. 🛮 Bi-Weekly i. Weekly period The 12-consecutive month period ending on____/_ Note: The Compensation Computation Period for Safe Harbor Non-Elective Contributions is the Plan

Note: If option j is selected, for Employees whose Employment Commencement Date is less than 12 months before the end of the 12-month period designated, Compensation will be determined over the Plan Year.

Year.

Code Section 415 and 416 Compensation E8.

E8.	Co	ode S	ectio	on 415 and 416 Compensation					
	a.	For t	the p	urposes of Code section 415 and 416, Compensation will be defined as (select one):					
		1.	\times	W-2 Compensation.					
		2.		3401(a) Compensation.					
		3.		415(c)(3) Compensation.					
		4.		Simplified 415 Compensation.					
	b. For the purposes of Code section 415 and 416, the selected items will be included in the								
	de		n of (Compensation (select all that apply).					
		1.		Deemed 125 Compensation					
		2.		Post Year End Compensation					
		3.	\times	Post-Severance Compensation					
		4.	\times	Differential Wage Payments					
		5.	\times	Military Continuation Payments					
		6.		Unfunded Deferred Compensation Plan					
		7.		Nonparticipating Nonresident Aliens salary					
		8.		Other Inclusion (e.g., PTO Payments):					
				e definition of "other inclusions" provided must be objectively determinable, may not be					
				in a manner that is subject to Plan Administrator discretion, and may not discriminate in					
				HCEs. The Plan Administrator will make all determinations in connection with such a uniform, nondiscriminatory manner.					
	_			urposes of Code section 416, salary paid to Participants who are disabled is <i>(select one):</i>					
	٥.	1 01 1	шо р	arposos of Gode Goden Tro, salary para to Fartisiparito wile are disabled to (66/66).					
		1.	X	Excluded.					
		2.		Included for all Participants.					
		3.		Included for all Participants who are not HCEs.					
F.	V	olun	tary	After-Tax Contributions and Safe Harbor Contributions					
			•						
Volur	ıtar	y Aft	er-Ta	ax Contributions					
F1.	Vo	lunta	ary A	fter-Tax Contributions are limited to (select one):					
	a.		No F	Plan limitations.					
	b.		The	following limitations (select all that apply):					
		1.		Voluntary After-Tax Contributions cannot exceed% of Compensation per Plan					
				Year.					
		2.		Voluntary After-Tax Contributions cannot exceed \$ per Plan Year.					
Safe				ributions arbor Contributions					
ΓΖ.			-	I make the following ADP Test Safe Harbor Non-Elective Contribution of at least%					
				ation (must be at least 3%) to Participants eligible to receive Safe Harbor Contributions					
				lowing plan:					
	No	ote: T	he fu	Il name of the other Employer plan must be completed. The other Employer plan must					
	ha	ve th	e sar	me Plan Year and safe harbor contributions as this Plan.					

F3.	Allocation Dates Safe Harbor Contributions will be allocated as of the (select one): a.						
G.	N	on-E	Elective Contributions				
G1.	 Allocation Conditions In order to be eligible to receive a Non-Elective Contribution, the Participant must (select one): a. □ No allocation conditions. b. ☑ The following allocation conditions will apply (select all that apply): 1. □ be employed on the last day of the Plan Year 2. □ have received Compensation since the prior Allocation Date for Non-Elective Contributions 3. □ must be employed on the Allocation Date for Non-Elective Contributions 4. ☑ must complete the following service requirement: A. ☑ 1000 (no more than 1,000) Hours of Credited Service in the Plan Year. B. □ (no more than 365) days using the Elapsed Time method of counting service. C. □ (no more than 12) months using the Elapsed Time method of counting service. Note: If option a is selected, skip question G2.					lective s Year. counting	
G2.	Th	Allocation Condition Waivers The allocation conditions for Non-Elective Contributions will be waived if, during the Plan Year, a Participant (select one for each applicable purpose): All Allocation Last Day Service					
	a.	No a	allocation condition waivers	Conditions ⊠	Requirement	Requirement	
	b.	The	e following waivers apply (select all that appl				
		1.	dies.				
		2.	becomes Disabled.				
		3.	terminates from service after reaching NR	A. 🗆			
		4.	is employed on the last day of the Plan Ye	ar. N/A	N/A		
		5.	other event:				
			A. All Allocation Conditions:	ided must be ob an Administrator dministrator will	jectively determin discretion and, m make all determin	ay not	

G3.	Non-Elective Contribution Amount							
	Non-Elective Contributions will be made in the following amount (select one):							
	a. ⊠	a discretionary amount determined by the Employer each Plan Year.						
	b. 🗆	an amount necessary to meet the requirements of the Non-Elective Contribution allocation formula specified below.						
		% <i>(no more than 25%)</i> of eligible Plan Compensation.						
	d. 🔲	\$ per Plan Year, but no more than 25% of eligible Plan Compensation.						
	Note: C	the amount required under any applicable collectively bargained for agreement (CBA). Options b – e are mandatory contributions that must be made every Plan Year. Option b cannot be chosen unless option G4a is chosen or G4i is chosen and the formula						
	Note: If	ed includes a dollar or percentage amount per Participant. Toption e is chosen, the CBA provisions related to the Non-Elective Contribution amount must uded as an addendum to the Plan.						
G4.	Non-Fl	ective Contribution Allocation Formula						
04.	Non-Ele	ective Contributions will be allocated to Participants eligible to receive a Non-Elective ution according to the following formula (select one):						
	а. 🔲	Flat dollar amount. The allocation will be based on the (select one):						
	1.	☐ Plan Year.						
	2.	☐ Compensation Computation Period.						
	b. 🔲	Pro-rata. The allocation will be based on the (select one):						
	1.	☐ Plan Year.						
	2.	☐ Compensation Computation Period.						
	c. 🗆	% of Compensation per Participant.						
	d. 🗖	Age Weighted.						
	1. 2.	Pre-Retirement Interest Rate: (must between 7.5% and 8.5%). Post-Retirement Interest Rate: (must between 7.5% and 8.5%).						
	2. 3.	Post-Retirement Mortality (e.g., UP-84 Unisex):						
	e. 🗆	Integrated Formula. (Complete Question G5.)						
	f. 🗆	Uniform Points. (Complete Question G6.)						
	g. 🗵	Class Allocation based on each Participant in an individual group.						
	h. 🔲	Class Allocation based on defined groups. (Complete Question G7.)						
	i. 🔲	Other (e.g., the greater of \$500 or 1% of Plan Year Compensation):						
		Options a - e are design based safe harbor allocation formulas. Options f - i will require testing						
		Code section 401(a)(4). Inless option e is selected, skip question G5. Unless option f is selected, skip question G6.						
		option h is selected, skip question G7.						
		The other Non-Elective Contribution Allocation Formula specified must be objectively						
	may no	inable, may not be specified in a manner that is subject to Plan Administrator discretion, and t discriminate in favor of HCEs. The Plan Administrator will make all determinations in						
		tion with such issues in a uniform, nondiscriminatory manner.						
	allocation	n order to be covered by the IRS pre-approval letter, the other Non-Elective Contribution on formula must be a combination of Non-Elective Contribution allocation formulas available Plan Document. Any other formulas will not be covered by the IRS pre-approval letter.						

G5.	Int	tegrated Non-Elective Contribution Allocation Formula				
		If an integrated allocation formula is selected above the Plan will allocate the Non-Elective				
				tion in a (select one):		
		1.		2-step integrated allocation as described in Plan Section 2.3.5.		
				The Base Contribution Percentage will be%		
		2.		The Excess Contribution Percentage will be%		
		۷.	Ш	4-step integrated allocation as described in Plan Section 2.3.5. The Base Contribution Percentage will be%		
				Permitted Disparity will be limited to%		
		3.		Pro-rata integrated allocation as described in Plan Section 2.3.5		
			Ā.	Will the pro-rata integrated allocation formula satisfy the requirements under Code		
				section 401(I) (select one)?		
				1. No.		
			_	2.		
			В.	If the Plan is Top-Heavy, the Non-Elective Contribution allocation formula will (select		
				one): 1. □ remain the same with the Minimum Top-Heavy Allocation determined last.		
	h	lf on	into	2. □ change to the 4-step integrated allocation as described in Plan Section 2.3.5 grated formula is selected above the Integration Level will be (select one):		
	D.	11.		The Taxable Wage Base (TWB) under the Social Security Act.		
		2.		% of the TWB (if more than 100% testing under Code section 401(I) will be		
			ш	required)		
		3.		The greater of \$ or% of the TWB (if more than 100% testing under Code		
				section 401(I) will be required)		
		4.		The lesser of \$ or the TWB (the dollar amount must be greater than 80% of the		
				TWB or testing under Code section 401(I) will be required)		
		5.		80% of the TWB plus \$ (not more than \$1,000 or testing under Code section		
		6.	_	401(I) will be required)		
		0.		Other (e.g., 80% of the TWB rounded up to the next \$3,000):		
G6.	Ur	niforr	n Po	ints Non-Elective Contribution Allocation Formula		
				points allocation formula is selected above the Plan will allocate the Non-Elective		
				as specified below.		
	a.			-Elective Contribution will be allocated (select one):		
		1.		pro-rata by based on all points awarded for the Plan Year.		
		2.		\$ for each point awarded to the Participant.		
		3.		% of the Participant's Compensation for each point awarded to that Participant.		
	b.		_	ill be awarded to each Participant following the formula below (select all that apply):		
		1.		points for each year of age		
		2.		points for each Year of Credited Service		
		3. 4		points for each \$ (not more than \$200) of Compensation		
	_	4.		points for (e.g., 1 point for each Year of Credited Service at NRA)		
	C.		I ota	al points for each Participant will be limited to points.		

G7.	Group Allocation Non-Elective Contribution Allocation Formula If the group allocation formula is selected above the Plan will allocate the Non-Elective Contribution							
	as specified below.							
	a. For	Group A: the Non-Elective Contributions will be allocated (select one):						
	1.	pro-rata based on Compensation to all Participants in the group.						
	2.	the same dollar amount to all Participants in the group.						
	3.	cents per Hour of Credited Service .						
	predete The all lowest section	The allocation groups must be objectively determinable and must not violate the definite ermined allocation formula requirement of Treasury Regulation section 1.401-1(b)(1)(ii).Note: ocation groups cannot be designed in a manner to benefit only those non-HCEs with the Compensation, shortest period of service, or who are needed to pass testing under Code 410(b).						
		For Self-Employed Individuals, the allocation method must not result in a cash or deferred						
	Note: T of the N provide	n under Treasury Regulation section 1.401(k)-1(a)(6). The Employer must provide the Plan Administrator or Trustee, if applicable, written notification whon-Elective Contribution to be allocated to each allocation group. The instructions must be ad no later than the Employer's tax return due date, including extensions, of the year for which exaction is made.						
G8.	Limitat	ione						
G0.		ective Contributions will be limited by (select one):						
	a. ⊠	No Plan imposed limitations						
	b. 🗆	The following limitations apply (select all that apply):						
	1.	☐ Minimum Non-Elective Contribution for each Participant for a Plan Year is (select one):						
		A. \$						
		B.						
	2.	Maximum Non-Elective Contribution for each Participant for a Plan Year is <i>(select one)</i> :						
		inaximum Non-Elective Contribution for each Farticipant for a Fian Teal is (select one).						
		A . \$						
		B.						
	3.	other (e.g., maximum of \$1,500 per HCE):						
	Note: 7	The other limitations on Non-Elective Contributions must be objectively determinable, may not						
	be specified in a manner that is subject to Plan Administrator discretion, and may not discriminate in							
	favor of HCEs. The Plan Administrator will make all determinations in connection with such issues in							
	a unitoi	rm, nondiscriminatory manner.						
G9.	Allocat	tion Dates						
		-Elective Contributions will be allocated as of the <i>(select one):</i>						
	1.	□ last day of the Plan Year						
	2.	☐ last day of each pay period						
	3.	☐ last day of each Plan quarter						
	4.	☐ last day of each calendar month						
	5.	other (e.g., last day of each calendar quarter):						
	b. 🔲	If Non-Elective Contributions are not allocated on the last day of the Plan Year, any						
	_	allocation conditions will be pro-rated.						

G10.	410(b) Coverage Failure Participants who normally would not be given an allocation of Non-Elective Contributions may be given an allocation in order to satisfy the requirements of Code section 410(b) (select one):							
	a. □ b. ⊠	No. Yes	The . The	Plan must be amended in the case of a failure under Code section 410(b). a Code section 410(b) fail safe will be used in the case of a failure under Code 110(b).				
G11.	a. Nor Election1.2.3.	n-Elective (ctive (Contr not all [all [Parti	Contributions to Disabled Participants Contributions will be allocated to the following Participants eligible to receive a Non-ribution without regard to any allocation conditions (select one): applicable. Disabled Participants who are not HCEs. Disabled Participants. icipants eligible to receive a Non-Elective Contribution under this section will (select				
	2.	A.B.	rece	at the end of the Plan Year immediately preceding the Plan Year in which the Participant attains ERA.				
	The Me a. ⊠ b. □ c. □	of Cred ethod Not Hou the Elap	dited of M Appl rs of Cred osed	Service leasuring a Year of Credited Service is (select one):				
G13.		ours one): the l	of sei : Plan 12-m calen	Year. Independent of the computation of the employment anniversary date. Independent of the employment anniversary date. Indep				

G14. Ec

G14	Equivalencies						
∪	a. Hours of Service						
		If the hours of service method is selected above, when records of hours are not maintained the following equivalency will apply (select one).					
		1.		Not applicable, actual hours will be used for all Employees.			
		2.		Days worked (10 hours).			
		3.		Weeks worked (45 hours).			
		4.		Semi-Monthly or Bi-weekly Payroll period worked (95 hours).			
		5 .		Months worked (190 hours).			
	b.	Elapsed Time					
		If the elapsed time method is selected above, credited service will be measured using (select					
		one)					
		1.		Exact date in years.			
		2.		Exact date in months.			
		3.		Calendar months with Employee granted a month of service if they work at least one Hour of Service in that month.			
		4.		Nearest calendar month.			
		5.		Completed calendar months.			
G15.	Break in Credited Service If the hours of service method is selected above, a Break in Credited Service occurs if an Employe fails to complete (not more than 500) Hours of Service in a Credited Service Computation Period. Note: If actual hours are being used and no Hours of Service are completed then a Break in Service will be defined as the lesser of 500 hours or the number of Hours of Service needed to earn a Year						
G16.	Q ι In	ualifie the ca	ed Mi	Service. Ilitary Service If death or Disability resulting from Qualified Military Service, the Participant will be a allocation as if they returned to employment with the Employer immediately prior to			
	de			ability (select one):			

Н. **Other Contributions**

b. 🛛 Yes.

Prevailing Wage Contributions

Designation of Contribution

Prevailing Wage Contributions will be treated as the following contribution source (select one):

- a.

 Non-Elective Contributions. **b. \Boxed** Qualified Non-Elective Contributions.
- The determination between Non-Elective Contribution and Qualified Non-Elective Contribution will be made at the time the contribution is allocated to the Participant's Account.

H2.	Offset Prevaili	ing Wage Contributions will offset the following Employer Contributions <i>(select one):</i> Not applicable; Prevailing Wage Contributions will supplement other Employer Contributions.			
	b. 🛘 1. 2.	the following Employer Contributions (select all that apply): Non-Elective Contributions.			
	۷.	Other (e.g., Qualified Non-Elective Contributions):			
H3.	Prevaili a. □ b. □ c. □ d. □	ling Wage Allocation Formula ling Wage Contributions will be allocated to Participants' Account (select all that apply): % of Participant's Compensation per payroll period. \$ per payroll period. pursuant to the following collective bargaining agreement: % of Participant's fringe benefits per payroll period. as described in the Prevailing Wage Addendum. Other (e.g., \$.25 per Hour of Service):			
Quali H4.	Qualified Non-Elective Contributions (QNECs) 14. QNECs The Employer may always make QNECs on behalf of Participants in order to correct any tes operational failures. In addition, the Employer may make additional discretionary QNECs (se one).				
	a. ⊠ b. □ <i>Note: It</i>	No. Yes. f option a is selected, skip questions H5 – H7.			
H5.		lity for QNECs lowing Participants will be eligible for the additional discretionary QNEC (select one): all Participants. the following Participants (select all that apply): non-HCE Participants. Participants employed on the last day of the Plan Year. other (e.g., Participants covered under a collective bargaining agreement):			
H6.		Allocation Formula ditional discretionary QNEC will be allocated to Participants eligible for such contributions one): in proportion to a Participant's Compensation. as a uniform dollar amount. other (e.g., 3% of Compensation):			
Н7.		% (no more than 25%) of eligible Plan Compensation. \$ per Plan Year, but no more than 25% of eligible Plan Compensation.			

	Rollove Rollove	Per Contributions or Contributions may be contributed to the Plan (select one): No. Yes (select all that apply): Direct Rollover Contributions are allowed. In-Direct Rollover Contributions are allowed.			
		option a is selected, skip questions H9 – H10.			
		er Contribution Eligibility owing individuals will be permitted to make Rollover Contributions to the Plan (select one): all Employees. all Employees who do not fall into an excluded class. all Participants who are Employees, have met eligibility requirements, and passed an entry date. other (e.g., all Participants):			
	In addit that app	er Contribution Types ion to pre-tax assets, Rollover Contributions of the following types will be permitted (select all oly): non-Roth after-tax (direct rollover only). Participant loans (direct rollover only). Rollover of in-kind assets will be permitted for (select all that apply): Participant loans. other plans of the Employer. all Rollover Contributions.			
l.	Тор-Н	leavy and Other Testing			
I1.	Top-He	p-Heavy Allocation eavy Minimum Allocation event the Plan is Top-Heavy, if necessary, the Employer will (select one): make an additional contribution to meet the Top-Heavy Minimum Allocation requirements. first satisfy the Top-Heavy Contribution requirements then allocate the remaining Employer Contribution.			
I2.		p-Heavy Minimum Eligibility p-Heavy Minimum allocation will not be given to (select one): not applicable. the following Participants (select all that apply): Participants who are Key Employees. Participants who are Key Employees will be eligible at the discretion of the Plan Administrator. Participants who are covered by a collectively bargained agreement.			

13.		Top-Heavy Minimum Allocation Formula The Top-Heavy Minimum Allocation will be <i>(select one):</i>						
			The lesser of 3% or the highest percentage allocated to any Key Employee.					
			% (at least 3%).					
		_						
-			s Top-Heavy Provisions					
l4.			eavy Minimum Allocation					
	The Employer that maintains a qualified defined benefit plan in which any Participant in the Plan is, was, or could become a Participant adds the following optional provisions that it deems necessary to satisfy Code section 416 because of the required aggregation of multiple plans (select one):							
	a.	X	Not applicable - No other plan or all other plans terminated prior to the Effective Date of this Adoption Agreement.					
	b.		A minimum contribution allocation of 5% of each eligible Non-Key Employee's total Compensation shall be provided in a defined contribution plan of the Employer.					
			A minimum benefit of the lesser of 2% times Years of Service or 20% of each eligible Non- Key Employee's Average Compensation shall be provided in a defined benefit plan of the Employer.					
	d.		A minimum benefit of the lesser of 2% times Years of Service or 20% of each eligible Non-Key Employee's Average Compensation shall be provided in a defined benefit plan of the Employer but offset by the amount contributed on such eligible Non-Key Employee's behalf under any defined contribution plan of the Employer.					
	e.		other (e.g., 6% of each eligible Non-Key Employee's total Compensation shall be provided in a defined contribution plan of the Employer):					
	Note: If option a is selected skip question I5.							
I 5.	Top-Heavy Assumptions The interest rate and mortality table used to establish the present value of accrued benefits in order to calculate the Top-Heavy Ratio under Code section 416 will be (select one): a. □ same as the following defined benefit plan:							
	b.		other assumptions.					
			Interest Rate:% Mortality table:					
10	.	!!-						
16.	Top-Heavy Duplications If Employer maintains two or more defined contribution plans, the Employer has determined that a Minimum Top-Heavy Allocation will be provided as follows (select one):							
	a.	X	Not applicable - No other plan or other plan terminated prior to the Effective Date of this Adoption Agreement.					
	b.		A minimum contribution of % of each Non-Key Participant's Compensation shall be provided by:					
		1.	☐ this Plan.					
		2.	☐ The following defined contribution plan:					
		3.	☐ Employees who will receive the minimum contribution under such other plan:					
	Note: Satisfying the Minimum Top-Heavy Allocation in another plan for some but not all of the Participants may cause the Plan to fail to satisfy the uniformity requirement of Treasury Regulations section 1.401(a)(4)-2(b)(2)(ii) for plans using a design-based safe harbor, even though all other requirements of the safe harbor are met.							

J. Vesting and Forfeitures

All Voluntary After-Tax Contributions, Prevailing Wage Contributions, Traditional Safe Harbor ADP Contributions and QNECs will be 100% vested and nonforfeitable at all times.

J1.	A Particle a. □ b. ⊠ c. □ d. □ 1. 2. 3. 4. 5. 6. 7. Note: C	dules ective Contribution Vesting Schedule Expant's Non-Elective Contributions will vest according to the following schedule (select one): 100% immediate vesting. 6-year tiered vesting schedule. 100% vested after (no more than 3) Years of Vesting Service. another vesting schedule % at less than 1 Year of Vesting Service % at more than 1 but less than 2 Years of Vesting Service % at more than 2 but less than 3 Years of Vesting Service % at more than 3 but less than 4 Years of Vesting Service % at more than 4 but less than 5 Years of Vesting Service % at more than 5 but less than 6 Years of Vesting Service. 100% at more than 6 Years of Vesting Service.
	Note: If	option d is chosen and d4 is not 100%, d3 cannot be less than 20%, d4 cannot be less than 5 cannot be less than 80%.
J2.	Top-He Particip a. □ b. □ c. ☑ d. □ e. □ 1. 2. 3. 4. 5. 6. 7. Note: If	Paray Contribution Vesting Schedule ant's Top-Heavy Contributions will vest according to the following schedule (select one): not applicable. 100% immediate vesting. 6-year tiered vesting schedule.
J3.	A Partio	rated Vesting sipant will become 100% vested in all their accounts if they are still employed when they all that apply): reach ERA. die. become Disabled. other (e.g., terminated from service due to closing of factory):
	_	(1.0) 11

Ν

Meas	suring V	estin	g Service								
J4.		Year of Vesting Service									
	The Me	ethod	of Measuring a Year of Vesting Service is (select one):								
	а. 🔲	Not	Applicable.								
	b. ⊠		rs of service method. A Year of Vesting Service is $\underline{1000}$ hours (not more than 1,000) in Vesting Service Computation Period.								
	c. 🔲	Elap	osed time method.								
J5.	Vestin	g Ser	vice Computation Period								
	If the h (select		of service method is selected above, the Vesting Service Computation Period will be								
	a. ⊠	the	Plan Year.								
	b. 🔲	the	12-month period ending on the employment anniversary date.								
	c. 🔲	the	calendar year ending with or within the Plan Year.								
	d. □	the	Employee's Eligibility Computation Period.								
J6.	Equiva	alenci	ies								
	-		f Service Service								
			irs of service method is selected above, when records of hours are not maintained the equivalency will apply (select one).								
	1.	\times	Not applicable, actual hours will be used for all Employees.								
	2.		Days worked (10 hours).								
	3.		Weeks worked (45 hours).								
	4.		Semi-Monthly or Bi-weekly Payroll period worked (95 hours).								
	5.		Months worked (190 hours).								
	b. Ela	psed									
	If th	ie ela	psed time method is selected above, credited service will be measured using (select								
	one	e).									
	1.		Exact date in years.								
	2.		Exact date in months.								
	3.		Calendar months with Employee granted a month of service if they work at least one Hour of Service in that month.								

J7. **Break in Vesting Service**

■ Nearest calendar month. ☐ Completed calendar months.

4.

If the hours of service method is selected above, a Break in Vesting Service occurs if an Employee fails to complete 500 (not more than 500) Hours of Service in a Vesting Service Computation

Note: If actual hours are being used and a no Hours of Service are completed then a Break in Service will be defined as the lesser of 500 hours or the number of Hours of Service needed to earn a Year of Vesting Service.

J8.	The fol Account a. \boxtimes	sions from Years of Vesting Service lowing Years of Vesting Service will be excluded when determining a Participant's Vesting at Balance (select one): all Years of Vesting Service are counted.
	b. 🗆	the following years will be excluded (select all that apply):
	1.	years before the Participant attains age 18.
	2.	years before the Plan's Original Effective Date.
	3.	years before a predecessor plan's original effective date. The predecessor plan's original effective date is//
J9.		,
	b. ⊠	
J10.	If a Pa	Dioyment rticipant has been rehired their Years of Vesting Service will be calculated using the following select one):
	a. □	not applicable.
	b. ⊠	• •
		one-year hold out rule.
	d. 🗆	·
	eitures	
J11.		Forfeitures ures may be used to (select one):
	a. ⊠	offset Employer Contributions at the Plan Administrator's discretion in a nondiscriminatory and uniform manner.
	b. 🗆	provide a supplemental Employer Contributions at the Plan Administrator's discretion in a nondiscriminatory and uniform manner.
	forfeite	Forfeitures may always be used to reduce administrative expense of the Plan or to restore ad account balances of rehired Employees. f option a is selected skip question J12 and J13.
J12.		cure Eligibility Requirements
		ures will be used to supplement Employer Contributions (select one):
	a.	Participants eligible to receive an allocation of the respective type of contribution
	b. □ c. □	All Participants
	. =	Participant who are employed on the date the Forfeitures are determined
	a. 🗆 e. 🗆	Participant who are employed on the date the Forfeitures are allocated
	Ÿ. ⊔	Other:(e.g. Participants who are employed on the last day of the Plan Year):

J13.		ture Allocation method situres will be used to supplement Employer Contributions (select one):
	a. 🗆	In the same manner as the respective Employer Contribution for the Plan Year
	b. 🗆	pro-rata for the Plan Year
	c. 🗆	pro-rata for the Compensation Computation Period
	d. 🗆	Other: (e.g., following the Non-Elective Allocation Formula).
J14.		ture Determination Period
	a. ⊠	ures will be determined <i>(select one):</i> during the Plan Year.
	b. □	since the prior Valuation Date.
	_	since the prior valuation bate.
	d. □	other (e.g., for the period before the prior Valuation Date):
	Note: specifi favor o	The "Other" forfeiture determination period must be objectively determinable, may not be ed in a manner that is subject to Plan Administrator discretion, and may not discriminate in of HCEs. The Plan Administrator will make all determinations in connection with such issues in arm, nondiscriminatory manner.
J15.	Forfeit	ture Allocation Date
		ures will be allocated on (select all that apply):
		each allocation date for Non-Elective Contributions
		each Valuation Date
	C.	other (e.g., each Anniversary Date):
	a man The Pl	The "Other" forfeiture allocation date must be objectively determinable, may not be specified in ner that is subject to Plan Administrator discretion, and may not discriminate in favor of HCEs. an Administrator will make all determinations in connection with such issues in a uniform, criminatory manner.
J16.	Deeme	ed Forfeiture Date
	of (sele	on-vested portion of Participant's Account Balance will be deemed distributed and forfeited as ect one):
	a. 🗆	'
	b	,
	C. ⊠	the last day of the Plan Year that contains the Participant's fifth consecutive break in service.
	manne The Pl	other (e.g., the Valuation Date next following the Participant's date of termination): The "Other" deemed forfeiture date must be objectively determinable, may not be specified in a er that is subject to Plan Administrator discretion, and may not discriminate in favor of HCEs. an Administrator will make all determinations in connection with such issues in a uniform, criminatory manner.
K.	Distr	ibutions
RE <i>A</i>	Require	ements
K1.	REA S	afe Harbor
	(select	·
	a. □	No.
	b. 🔲	Yes, REA will apply only to assets merged in from plans subject to the REA requirements.
	C. ⊠	Yes.

K2.	If the F Survivo a b Note: I Particip Survivo	led Joint and Survivor Annuity (QJSA) Plan does not meet the requirement to be or Annuity percentage is (select one). not applicable. "" (no less than 50 and no more that addition to the percentage provided for pant must be permitted to elect an optional or Annuity percentage is less than 75%, the elected percentage is 75% or more, the Quite process and survival an	a REA safe harbo in 100). the Qualified Joir al form of annuity. he Qualified Optio	nt and Survivor If the selected onal Survivor A	⁻ Annuity, a d Qualified Joint and Annuity will be 75%. If
Form K3.	Availa Particip	stribution ble Forms of Distributions cants eligible to take a distribution can take	se the distribution	in the followin	g format(s) <i>(select all</i>
	that ap a. ⊠	lump sum distributions with the followin	g limitations <i>(e.g.,</i>	not to exceed	\$10,000):
	b. 🗆	partial, non-periodic distributions with the at least \$200):	e following limitat	tions (e.g., eac	ch distribution must be
	c. 🗆	installment payments with the following \$200):	limitations (e.g.,	each distributio	on must be at least
	d. 🔲	,	e.g., each distribu	ution must be a	nt least \$200):
	Note: I	If a distribution form is selected and no lin	nitation is complet	ted, no further	limitation will apply.
K4.	Minim	um Distribution Amount			
	a. 🗖	The Plan will provide the lesser of the a \$100) to a Participant with no vested ba		r \$ <i>(an</i>	nount cannot exceed
K5.		oution due to Disability butable event due to Disability occurs (se	elect all that apply	·)·	
	a. 🗆	Termination Date. (Distributable event of employment.)		•	rmination of
	b . ⊠	The Participant's actual date of Disabilit	y, as determined	by the Plan Ad	dministrator.
	c. 🗆	months after the Participant's actu Administrator.	al date of Disabili	ty, as determir	ned by the Plan
	d. □	The date the Plan Administrator determ	ines the Participa	int to be Disab	led.
K6.	For dis	oution Determination Date tribution purposes due to a distributable e e shall be determined as of (select one for	r each applicable		s vested account Voluntary After-Tax and
		Other Th		Retirement	Rollover Accounts
	wit	e last day of the Plan Year coinciding h or next following the date of the tributable event.			
	b. The	e Valuation Date coinciding with or next owing the date of the distributable event			
	c. The	e Valuation Date coinciding with or mediately preceding the date of the			
	d. As foll	tributable event. soon as administratively feasible owing the date of the distributable ent, based on the preceding Valuation te.	⊠	⊠	⊠

	e.		nluation Date preceding the pant's Normal or Early Retirement						
	f.	Other:							
	Sp	 Te De 	e other determination date <i>(e.g. last da</i> rmination for Other Reasons: eath, Disability, or Retirement: ective Deferrals, etc.:	y of the Pl	an Year)).			
K7.	For	r distribu icated b	tions at termination for reasons other t made within a reasonable period fol made within a reasonable period fol	han death lowing the lowing the	, Disabili Distribu Anniver	ity, or re tion Def sary Da	etirement termination	will be (s	
	b.	Distribu	tions to Participants who have a distrik select one): made within a reasonable period fol	outable eve lowing the	ent due t Distribu	o death	ı, Disabili terminatio		ement
	c.	Distribu 1. ⊠ 2. □	'	lowing the	Distribu	tion De	terminatio	on Date.	
K8.	In ((se a. b.	order to elect one No \$2	limit	tribution, t	he total o	distribut	ion must	be at leas	st
In-Se K9.	In- In- a. b.	Service □ No ☑ Ye	Distributions Distributions are available under the P o.		t one):				
K10.	In-		Distribution Event Distributions will be permitted from the ect one):	· ·		•	· ·	·	
	a.	All Southe follows: 1. No. 2. QN. 3. Ro. 4. Tra		Age 59½ □ □ □ □ □ □ □ □ □ □ □ □ □ □ □ □ □ □	NRA	ERA I	Disability	Other Age	Specify age

	6.		er (e.g. Money Purchase tributions)						
	before Note: F	the Partici	dless of what is selected above, In- articipant has reached age 59 ½ or pants will be eligible for an In-Servi s selected above.	incurred	a disabilit	y under	Code sed	ction 22(e	e).
K11.	In-Serv	ice D	rvice Distribution Events istribution will be permitted from the at apply):	Non-Ele	ective Acco	ount for t	the speci	fied reas	on
	a. 🔲	Amo	ounts have been allocated for	years. (Must be at	least 2.,)		
	b. 🛚	Req	uire participation for at least y	ears. (N	lust be at l	east 5.)			
	c. 🗆		ounts have been allocated for it least years. <i>(Must be at lea</i>		/lust be at l	least 2.)	AND req	uire part	icipation
	d. 🛚		ounts have been allocated for it least years. <i>(Must be at lea</i>		/lust be at l	least 2.)	OR requ	ire partic	ipation
			pants will be eligible for an In-Servi s selected above. These events are						
K12.	In-Serv that ap a. ⊠	rice D ply): Rolle	Distributions at Any-Time istributions are permitted under the over Contributions. Intary After-Tax Contributions.	Plan at a	any-time fo	or the fol	lowing so	ources <i>(s</i>	elect all
K13.			Distributions Limitations						
			e Distributions will be limited by the	following	(select or	ne):			
	a. ⊠	No I	mitations.						
	b. 🔲	The	following limitations (select all that	apply):					
	1.		Participant must be 100% vested i	n all Plai	n Accounts	i.			
	2.		Participant may request In-Se	ervice Di	stributions	during e	each Plar	ı Year.	
	3.		each In-Service Distribution must be \$1,000) or the total value of the ve						
	4.		Distribution.	.//-					
	Note: 0		other (e.g., five distributions in total equest, regardless of the number of d a single In-Service Distribution.		_• ts from whi	ich the d	listributio	n is to be	taken,
Hards	ship Dis	tribu	tions						
			stributions						
			stributions will be permitted from the nardship reasons is defined under F					reasons	s (the
					Harbor Iship Defin	ition		afe Harbo	
	a. ⊠	Nor	ne	пагс	isnip Delin	ILIOIT	пагиы	nip Defini	lion
	b. □		Accounts Listed Below				ı		
	c. □	The	following Accounts (select all that	apply):	Ä			_ 	
	1.		Non-Elective Contributions	,					
	2.		Rollover Contributions						
	3.		Transfer Contributions						
	4.		Voluntary After-Tax Contributions	;					

	5.		Other (e.g.,	Top-Heavy	Contribution	ns):		
	Note: If	optic	on a is selecte	ed, skip ques	tions K15 –	K16.		
K15.	If Hards (select a. □	ship [one): A ha lack in a	ardship will be s other availa uniform and r	re allowed for defined as in ble resource non-discrimin	or a non-safe mmediate a s. The dete natory mann	nd heav	y financia	definition, the definition will be all need where the Participant made by the Plan Administrator
	b. 🗌	Othe	er (e.g., medi	cal expenses	s):			
K16.	Hardsh	ip Dis	on Hardship			ng (sele	ct one):	
	a. 🗆		mitations.					
	1.	_	following limit				Handa	kin Distrikastians in saak Disa
	١.		Year.	can receive	no more tha	an	_ Hards	hip Distributions in each Plan
	2.		each Hardsh Balance ava					100% of the Participant's Account
	3.		other (e.g., r		•			n):•
	Distrib Particip (select a.	ution ants <i>one):</i> Non	e.	ualified Milita	ary Service v	·		o take the following distributions
			, the following			ted (sele	ect all tha	t apply).
	1.	×	Deemed Sev					
	2.	×	Qualified Re	servist Distri	butions.			
K18.	Distribu or after	itions the c by the	late a Domes Plan Adminis	te Payee are tic Relations	e permitted v Order is de			ant continues to be employed on Qualified Domestic Relations
K19.	Distribu	itions defin Not attai	ed benefit pla applicable. nment of age	alances tran an can be dis 62.	stributed upc	on <i>(sele</i> d	ct one):	ase Pension Plan, Target Benefit earlier than age 62).

Required Minimum Distributions K20. Required Beginning Date The Required Beginning Date for Participants who are not a 5% Owner will be (select one): April 1st of the calendar year following the calendar year in which the Participant attains age b. ⊠ the later of April 1st of the calendar year following the calendar year in which the Participant attains age 70 1/2, or April 1st of the calendar year following the calendar year in which the Participant retires. the Participant has the ability to choose between (1) April 1st of the calendar year following the calendar year in which the Participant attains age 70 ½ or. (2) if later the April 1st of the calendar year following the calendar year in which the Participant retires. K21. Distributions to Beneficiaries Beneficiaries of deceased Participants (select one): **a.** \boxtimes may not elect to apply the 5-year rule. **b.** may elect to apply the 5-year rule. **Mandatory Cash-Out and Automatic Direct Rollover** K22. Mandatory Cash-Out Threshold The Mandatory Cash Out threshold is (select one): a. no Mandatory Cash Out. **b.** ⊠ \$5000. Note: If option a is selected, skip questions K23 – K25. **K23.** Mandatory Cash Out Timing The Mandatory Cash Out will occur _____ following the Participant's termination from service (select all that apply). **a.** \square as soon as administratively feasible **b.** □ as soon as administratively feasible after the next Valuation Date **c.** ⊠ as soon as administratively feasible after the last day of the Plan Year K24. Rollover Contributions Included in Mandatory Cash Out Rollover Contributions will be included when determining if the Mandatory Cash Out threshold has been exceeded (select one): a. \times No. b. Yes. K25. Automatic Rollover If a Mandatory Cash-Out distribution takes place (select one). All Mandatory Cash-Out Distributions will be paid directly to the Participant or Beneficiary. b. Mandatory Cash-Out Distributions not in excess of \$1000 (cannot exceed \$1.000) will be paid directly to the Participant or Beneficiary. Mandatory Cash-Out Distributions in excess of the specified amount will be directly rolled over into an IRA. All Mandatory Cash-Out Distributions will be directly rolled over into an IRA. Note: In order to select option a, the Mandatory Cash-Out Level must be \$1,000 or less and Rollover Accounts must be included in determining the Mandatory Cash-Out Level.

L. Administrative Provisions

Loans L1.	Loans Participants will be permitted to take loans from their Account balance (select one): a. □ No. b. ☑ Yes.
Earni L2.	Designation of Contribution If the Plan does not value Accounts on a daily basis, for Participants who receive distributions on any date other than a Valuation Date, earnings will be (select one): a. □ not applicable. b. ☑ credited solely as of the immediately preceding Valuation Date. c. □ the actual earnings on the date of distribution.
L3.	Earnings on Forfeiture Accounts Forfeiture Account holding the Plan's aggregate Forfeitures be subject to Trust earnings (select one): a. ☑ No. b. ☐ Yes.
Inves L4.	Participant Direction Participants may direct the investment of their following Accounts (select one): a. None b. All Accounts C. The following Accounts (select all that apply): 1. Non-Elective Contributions 2. QNECs 3. Rollover Contributions 4. Transfer Contributions 5. Voluntary After-Tax Contributions 6. Other (e.g., Top-Heavy Contributions): Note: If option a is selected skip questions L5 – L6.
L5.	Limitations on Participant Direction Participant direction of investments will be limited by the following (select one): a. □ no limitations. b. □ the following limitations (select all that apply): 1. □ must be 100% vested in directed Accounts. 2. □ other (e.g., XXXXXX):
L6.	404(c) The Plan intends to comply with ERISA section 404(c) (select one): a. □ No. b. □ Yes.

L7. Life Insurance	L7.	Life Insu	ırance
--------------------	-----	-----------	--------

a. 🗵 No.

M. Authorizations and Signatures

The name, address and telephone number of the Document Sponsor and the document provider are:

Document Sponsor

DATAIR Employee Benefit Systems, Inc. 735 N. Cass Avenue Westmont, IL 60559-1100 (630) 325-2600

Document Provider

NH Hicks 1294 E. 1st Ave., Ste. 110 Chico, CA 95926 (530) 891-4975

This Plan must be registered with the Document Sponsor within 60 days of adoption of this document, and the Document Serial Number assigned by the Document Sponsor shall be affixed to this signature page. The adopting Employer must notify the Document Sponsor through the document provider if the Plan is terminated, merged, or of any changes in the name, address, or EIN of the adopting employer at least annually, and within 30 days of any request of the Document Sponsor. If the adopting Employer terminates its relationship with the document provider, its Plan will no longer be considered a pre-approved plan sponsored by the Document Sponsor.

Unregistered use of this document will result in the Plan no longer participating in this pre-approved plan, and the document will be considered an individually designed plan, without reliance on the opinion letter of the Document Sponsor, which could result in the disqualification of the Plan.

If the Employer's Plan fails to attain or retain qualification, such Plan will no longer participate in this preapproved plan and will be considered an individually designed plan.

The Document Sponsor will inform the Employer of any amendments made to the Plan or of the discontinuance or abandonment of the Plan.

The adopting Employer may rely on an opinion letter issued by the Internal Revenue Service as evidence that the Plan is qualified under Code section 401 except to the extent provided in Revenue Procedure 2011-49.

The Employer may not rely on the opinion letter in certain other circumstances or with respect to certain qualification requirements that are specified in the opinion letter issued with respect to the Plan and in Revenue Procedure 2011-49.

In order to have reliance in such circumstances or with respect to such qualification requirements, application for a determination letter must be made to Employee Plans Determinations of the Internal Revenue Service.

This Adoption Agreement may be used only in conjunction with Base Plan Document #20 (the DATAIR Pre-Approved Defined Contribution Plan) Revised 06/30/2020.

* * *

The Employer hereby adopts the Plan as evidenced by the foregoing Adoption Agreement on this 31st day of July, 2022.

Employer:
Donner Summit Public Utility District
Cathy Preis
President
Plan Serial Number:

DATAIR DEFINED CONTRIBUTION PRE-APPROVED BASE PLAN DOCUMENT #20

DATAIR DEFINED CONTRIBUTION PRE-APPROVED BASE PLAN DOCUMENT #20

	1
INTRODUCTION Adoption and Title. Effective Date. Purpose.	1 1 1
DEFINITIONS	2
	18
PARTICIPATION Eligibility Requirements. Commencement of Participation. Participation Upon Reemployment. Termination of Participation. Plan Administrator's Determination. One-time Election Not to Participate. Change in Status. Existing Participants. Service Crediting Method.	20 20 20 20 20 21 21 21 22 22
CONTRIBUTIONS	23
Elective Deferrals Contributed by the Employer on	23 24
Automatic Compensation Reduction (ACR), Eligible Automatic Contribution Arrangements (EACA) and Qualified Automatic Contribution Arrangements QACA) Employee Contributions. Return of Contributions. Required Diversification For Plans That Hold Publicly Traded Securities.	25 32 32 33
ALLOCATIONS	36
Basic Allocation. Profit Sharing, Money Purchase Pension Plans. Cash or Deferred Profit Sharing Plans. Minimum Top-Heavy Allocation. Integration with Social Security - Profit Sharing Plans. Integration with Social Security - Money Purchase Plans. Fail-Safe Allocation. Contributions on Behalf of Disabled Participants. Treatment in the Case of Death or Disability Resulting from Active Military Service.	36 36 38 39 39 40 40 42 42
VESTED BENEFITS	43
	43
	44 44
Forfeitures.	45
	Adoption and Title. Effective Date. Purpose. DEFINITIONS PARTICIPATION Eligibility Requirements. Commencement of Participation. Participation Upon Reemployment. Termination of Participation. Plan Administrator's Determination. One-time Election Not to Participate. Change in Status. Existing Participants. Service Crediting Method. CONTRIBUTIONS Employer Contributions. Elective Deferrals Contributed by the Employer on Behalf of Electing Employees. Automatic Compensation Reduction (ACR), Eligible Automatic Contribution Arrangements (EACA) and Qualified Automatic Contribution Arrangements QACA) Employee Contributions. Required Diversification For Plans That Hold Publicly Traded Securities. ALLOCATIONS Basic Allocation. Profit Sharing, Money Purchase Pension Plans. Cash or Deferred Profit Sharing Plans. Minimum Top-Heavy Allocation. Integration with Social Security - Profit Sharing Plans. Integration with Social Security - Money Purchase Plans. Fail-Safe Allocation. Contributions on Behalf of Disabled Participants. Treatment in the Case of Death or Disability Resulting from Active Military Service. VESTED BENEFITS Vesting. Leave of Absence. Reemployment.

ARTICLE V	DISTRIBUTIONS	46
2.5.1	Distributable Benefit.	46
2.5.2	Forms of Distribution.	46
2.5.3	Methods of Distribution.	47
2.5.4	Commencement of Distributions.	48
2.5.5	General Distributions Requirements.	50
2.5.6		51
	Annuity and Consent Requirements.	
2.5.7	Required Minimum Distributions.	55
2.5.8	Nature of Distributions.	60
2.5.9	Qualified Military Service.	61
2.5.10	Hardship Distributions of Elective Deferrals.	61
2.5.11	Distributions in the Event of Incapacity.	62
2.5.12	In Service Distributions.	62
ARTICLE VI	CONTINGENT TOP HEAVY PROVISIONS	65
2.6.1	Top-Heavy Requirements.	65
2.6.2	Top-Heavy Definitions.	66
ARTICLE VII	SPECIAL CODA LIMITATIONS	70
2.7.1	Limitation on Deferral Percentage for Highly	70
	Compensated Employees.	
2.7.2	Limitation on Matching Contributions.	71
2.7.3	Multiple Plan Limitations For ADP and ACP Testing.	72
2.7.4	Different Testing Methods for Elective Deferrals and	73
2.7.4		13
0.7.5	Matching Contributions.	70
2.7.5	Distribution of Excess Elective Deferrals.	73
2.7.6	Distribution of Excess Contributions.	74
2.7.7	Distribution of Excess Aggregate Contributions.	75
2.7.8	Limitation on Distributions.	76
2.7.9	Forfeiture of Matching Contributions.	77
ARTICLE VIII	SIMPLE 401(k) LIMITATIONS	78
2.8.1	Establishing a SIMPLE 401(k) Plan.	78
2.8.2	Definitions.	78
2.8.3	Salary Reduction Contributions.	78
2.8.4	Other Contributions.	79
2.8.5	Limitation on Other Contributions.	79
2.8.6	Section 415 Limitations.	79
2.8.7	Election and Notice Requirements.	79
		80
2.8.8	Vesting Requirements.	
2.8.9	Top-Heavy Rules.	80
2.8.10	Nondiscrimination Tests.	80
2.8.11	Revocation.	80
ARTICLE IX	SAFE HARBOR CODA PROVISIONS	81
2.9.1	Safe Harbor CODA Rules.	81
2.9.2	Definitions.	83
2.9.3	ADP Test Safe Harbor Contributions.	84
2.9.4	ACP Test Safe Harbor Matching Contributions.	84
ARTICLE X	PARTICIPATING EMPLOYERS	86
2.10.1	Adoption.	86
2.10.2	Plan Maintenance.	86
2.10.3	Operational Rules.	86
2.10.4	Duties.	87
2.10.5	Withdrawal of an Employer.	87

PART III		88
ARTICLE I 3.1.1 3.1.2 3.1.3 3.1.4 3.1.5 3.1.6	ACCOUNTING Accounts. Valuation Adjustments. Allocation of Earnings, Gains, and Losses. Interim Valuations. Earnings on Forfeitures. Plan Expenses.	88 89 90 90 90
ARTICLE II 3.2.1 3.2.2	LIMITATIONS Limitations on Annual Additions. Determination of Annual Additions and Maximum Permissible Amount.	91 91 91
3.2.3 3.2.4 3.2.5 3.2.6	Excess Annual Additions. Participation in Certain Other Plans. Definitions. Controlled Businesses.	92 92 93 96
ARTICLE III 3.3.1 3.3.2 3.3.3 3.3.4 3.3.5 3.3.6 3.3.7 3.3.8 3.3.9 3.3.10 3.3.11	FIDUCIARIES Standard of Conduct. Individual Fiduciaries. Disqualification from Service. Bonding. Prior Acts. Insurance and Indemnity. Expenses. Agents, Accountants, and Legal Counsel. Investment Manager. Finality of Decisions or Acts. 404(c) Election.	97 97 97 97 97 97 97 98 98 98
ARTICLE IV 3.4.1 3.4.2 3.4.3 3.4.4 3.4.5 3.4.6 3.4.7 3.4.8 3.4.9	PLAN ADMINISTRATOR Administration of Plan. Disclosure Requirements. Information Generally Available. Statement of Account. Explanation of Rollover Treatment. Electromechanical Communications. Elections on Behalf of an Incapacitated Person. Operational Rules for Related Employer Groups. Use of Electronic Medium.	99 99 100 100 100 100 100 100 101
ARTICLE V 3.5.1 3.5.2 3.5.3 3.5.4 3.5.5 3.5.6	PARTICIPANT LOANS Authorization. Spousal Consent. Limitations. Availability. Default. Qualified Military Service.	102 102 102 102 103 103 103
ARTICLE VI 3.6.1 3.6.2 3.6.3 3.6.4	BENEFICIARIES Designation of Beneficiaries. Absence or Death of Beneficiaries. Surviving Spouse Election. Right to Direct on Behalf of a Participant or Beneficiary.	104 104 104 104 105

Appendix C EXPANDED HARDSHIP DISTRIBUTIONS AND REQUIRED MINIMUM DISTRIBUTION ADDENDUM

139

PART I

ARTICLE I

INTRODUCTION

- **1.1.1 Adoption and Title.** The Employer(s) whose signatures appear on the Adoption Agreement, hereby adopt a Plan to be known by the names set forth in the Adoption Agreement.
- **1.1.2 Effective Date.** The provisions of this Plan will be effective as of the Effective Date set forth in the Adoption Agreement.
- **1.1.3 Purpose.** This Plan is established for the purpose of providing retirement benefits to Eligible Employees in accordance with the Plan and the Adoption Agreement. In the event of any conflict between the terms of this plan and any conflicting provision contained in any associated trust, custodial account document or any other document that is incorporated by reference, the terms of this plan will govern.

ARTICLE II

DEFINITIONS

As used in this Plan and the Adoption Agreement, the following terms will have the following meanings:

- **"Account":** One or more accounts, as listed in Section 3.1.1, established as necessary by the Plan Administrator and maintained to account for assets held for the benefit of a Participant.
- "Average Contribution Percentage" (ACP): The Actual Contribution Percentage will mean, for a specified group of Participants (either Highly Compensated Employees or Non-highly Compensated Employees) for a Plan Year, the average of the Contribution Percentages of the Eligible Participants in the group.
- "ACP Test Safe Harbor Account": An Account established and maintained for a Participant for accounting purposes with respect to their share of ACP Test Safe Harbor Contributions.
- "ACP Test Safe Harbor Contributions": Matching Contributions described in Section 2.9.4.
- "Act": The Employee Retirement Income Security Act of 1974, as amended from time to time.
- "Average Deferral Percentage" (ADP): For a specified group of participants (either Highly Compensated Employees or Non-highly Compensated Employees) for a Plan Year, the average of the ratios (calculated separately for each participant in such group) of (1) the amount of Employer Contributions actually paid over to the Trust on behalf of such Participant for the Plan Year to (2) the Participant's Compensation for such Plan Year. Employer Contributions on behalf of any Participant will include: (1) any Elective Deferrals (other than Catch-up Contributions) made pursuant to the Participant's deferral election (including Excess Elective Deferrals of Highly Compensated Employees), but excluding (a) Excess Elective Deferrals of Non-highly Compensated Employees that arise solely from Elective Deferrals made under the Plan or plans of this Employer and (b) Elective Deferrals that are taken into account in the Actual Contribution Percentage test (provided the ADP test is satisfied both with and without exclusion of these Elective Deferrals); and (2) if elected by the Employer in the Adoption Agreement, Qualified Non-Elective Contributions and Qualified Matching Contributions. For purposes of computing Actual Deferral Percentages, an Employee who would be a Participant but for the failure to make Elective Deferrals will be treated as a Participant on whose behalf no Elective Deferrals are made.
- "ADP Test Safe Harbor Account": An Account established and maintained for a Participant for accounting purposes with respect to their share of ADP Test Safe Harbor Contributions.
- "ADP Test Safe Harbor Contributions": Matching Contributions and Non-Elective Contributions described in Section 2.9.3.
- "Affiliated Employer": An employer that is not related to the Employer who is the Plan Sponsor of a Multiple Employer Plan. The Affiliated Employer may be a single entity or a group of employers treated as a single employer when they are required to be aggregated pursuant to Code sections 414(d), 414(c), 414(m) or 414(o) (a required aggregation group). The Affiliate Employer, with the consent of the Plan Sponsor may participate in a Multiple Employer Plan by executing a separate Adoption Agreement or the Adoption Addendum to the Plan. In the case of an Affiliated Employer that is a member of a required aggregation group, each member of the group must affirmatively elect to participate in a Multiple Employer Plan.
- "Alternate Payee": Any spouse, former spouse, child, or other dependent of a Participant who is recognized by a Qualified Domestic Relations Order as having a right to receive all, or a portion of, the benefits payable under this or any other plan of the Employer with respect to such Participant.

- **"Anniversary Date":** Unless otherwise specified in the Adoption Agreement, the last day of each Plan Year.
- "Annuity Starting Date": The first day of the first period for which an amount is paid as an annuity or any other form.
- "Automatic Compensation Reduction" (ACR): An arrangement in a cash or deferred plan as defined in Section 2.2.3(a) of this Plan under which an Employee that has not affirmatively elected to defer Compensation into the Plan will have an amount deferred on their behalf by the Plan Administrator.
- **"Beneficiary":** The person or persons entitled hereunder to receive the benefits that may be payable upon or after a Participant's death pursuant to Sections 3.6.1 and 3.6.2.
- "Break in Service": The failure of an Employee to complete more than 500 Hours of Service or such lesser number specified in the Adoption Agreement during any 12 consecutive month computation period. The computation periods will be specified by the Employer in the Adoption Agreement. The computation period used for a Year of Service will be used to determine a Break in Service for that purpose, and the same computation period used for a Year of Vesting Service will be used to determine a Break in Service for purposes of vesting service. Under the Elapsed Time Method of service counting, the term "one year Period of Severance" will be substituted for the term "1 year Break in Service." An individual reemployed pursuant to section 414(u) of the Code will be treated with respect to the Plan as not having incurred a Break in Service with the Employer maintaining the Plan by reason of such individual's period of Qualified Military Service.
- "Catch-up Contributions": Elective Deferrals made to the Plan that are in excess of an otherwise applicable Plan limit and that are made by Participants who are age 50 or over by the end of the applicable taxable year. See Section 2.2.2(b) for further discussion of Catch-up Contributions.
- "Code": The Internal Revenue Code of 1986, as amended from time to time.
- **"Compensation":** The compensation as defined in Section 3.2.5 and as specified in the Adoption Agreement (or Earned Income in the case of a Self-Employed Individual) that is actually paid to the Employee by the Employer during a Compensation Computation Period.

In the case of a Self-Employed Individual, Compensation means Earned Income during such period. Compensation for a Participant's initial year of participation will be Compensation from the Participant's Entry Date or for the Plan Year, as elected in the Adoption Agreement.

For Plan Years beginning on or after January 1, 1994 and before January 1, 2002, the annual Compensation of each Participant taken into account for determining all benefits provided under the Plan for any Plan Year will not \$150,000, as adjusted for increases in the cost-of-living in accordance with Code section 401(a)(17)(B). The cost-of-living adjustment in effect for a calendar year applies to any Compensation Computation Period beginning in such calendar year.

For any Plan Year beginning after December 31, 2001, the annual Compensation of each Participant taken into account in determining allocations will not exceed \$200,000, as adjusted for cost-of-living increases in accordance with Code section 401(a)(17)(B). However, solely for purposes of determining a participant's allocations for Plan Years beginning after December 31, 2001, the annual Compensation limit in effect for determination periods beginning before January 1, 2002 is \$200,000. Annual Compensation means compensation during a Compensation Computation Period, which is either the Plan Year or such other consecutive 12 month period over which compensation is otherwise determined under the Plan. If Compensation for any prior Compensation Computation Period is taken into account in determining a Participant's allocations for the current Plan Year, the Compensation for such prior Compensation Computation Period is subject to the applicable annual compensation limit in effect under Code section 401(a)(17) for that prior period. The cost-of-living adjustment in effect for a calendar year applies to annual Compensation for the Compensation Computation Period that begins with or within such calendar year.

If a Compensation Computation Period consists of fewer than 12 months, the annual Compensation limit is an amount equal to the otherwise applicable annual Compensation limit multiplied by a fraction, the numerator of which is the number of months in the short Compensation Computation Period, and the denominator of which is 12.

If Compensation for any prior Compensation Computation Period is taken into account in determining a Participant's allocations for the current Plan Year, the Compensation for such prior Compensation Computation Period is subject to the applicable annual compensation limit in effect for that prior period.

For purposes of Code sections 403(b)(3), 415(c)(3), and 457(e)(5), an Employee who is in Qualified Military Service will be treated as receiving Compensation from the Employer during such period of Qualified Military Service equal to: (A) the Compensation the Employee would have received during such period if the Employee were not in Qualified Military Service, determined based on the rate of pay the Employee would have received from the Employer but for absence during the period of Qualified Military Service, or (B) if the Compensation the Employee would have received during such period was not reasonably certain, the Employee's average Compensation from the Employer during the 12-month period immediately preceding the Qualified Military Service (or, if shorter, the period of employment immediately preceding the Qualified Military Service).

Differential Wage Payment For Retirement Plan Purposes. The Employer who elects to make Differential Wage Payments to active duty members in Qualified Military Service will treat the Differential Wage Payments as a payment of wages by the Employer to an Employee.

Treatment of Differential Wage Payments. For the purpose of the Plan:

- (i) an individual receiving a Differential Wage Payment will be treated as an Employee of the Employer making the payment;
- (ii) the Differential Wage Payment will be treated as Compensation; and
- (iii) the Plan will not be treated as failing to meet the requirements of any plan qualification requirements under Code section 401(a) where a contribution or benefit is based on the Differential Wage Payment.

"Compensation Computation Period": The period specified as the Compensation Computation Period in the Adoption Agreement.

"Contribution Percentage": The ratio (expressed as a percentage) of the participant's Contribution Percentage Amounts to the Participant's Compensation for the Plan Year.

"Contribution Percentage Amounts": The sum of the Employee Contributions, Matching Contributions, and Qualified Matching Contributions (to the extent not taken into account for purposes of the ADP test) made under the Plan on behalf of the participant for the Plan Year. Such Contribution Percentage Amounts will not include Matching Contributions that are forfeited either to correct Excess Aggregate Contributions or because the contributions to which they relate are Excess Deferrals, Excess Contributions, or Excess Aggregate Contributions. If so elected in the adoption agreement the Employer may include Qualified Non-Elective Contributions in the Contribution Percentage Amounts. The Employer also may elect to use Elective Deferrals in the Contribution Percentage Amounts so long as the ADP test is met before the Elective Deferrals that are used to meet the ACP test.

"Controlled Account": An Account established and maintained for a Participant for accounting purposes over which they are permitted, under the terms of the Plan and the Plan's investment policy, to exercise investment control.

"Deductible Voluntary Account": An Account established and maintained for a Participant for accounting purposes with respect to their Voluntary Employee Contributions that were deductible by the Participant at the time they were made.

"Deemed IRA Account": (Deemed IRA) An Account established and maintained for a Participant for accounting purposes that meets the applicable requirements for an Individual Retirement Account under Sections 3.12.1 (Deemed Pre-tax IRA), 3.12.2 (Deemed Roth IRA) and Code sections 408(q) and 408A to which they may make Deemed IRA Contributions.

"Deemed IRA Contribution": A Participant contribution designated as an IRA contribution to a Deemed IRA Account. Deemed IRA Contributions include Pre-tax IRA Contributions and Roth IRA Contributions.

"Deemed Section 125 Compensation": An amount not available to a Participant in cash in lieu of group health coverage (a forced reduction in pay) because the Participant is unable to certify that they have other health coverage. Therefore, the Employee is included in the Employer's health plan. The amount withheld will be treated as Deemed Code section 125 Compensation only if the Employer does not request or collect information regarding the Participant's other health coverage as part of the enrollment process for the health plan.

"Designated Beneficiary": The individual or trust that is designated as the Beneficiary under the Plan and who is a designated beneficiary under Code section 401(a)(9) and Treasury Regulations section 1.401(a)(9)-4.

"Differential Wage Payment": For purposes of the Plan, the term "Differential Wage Payment" means any payment which: (1) is made by an Employer to an individual with respect to any period during which the individual is performing service in the uniformed services (as defined in chapter 43 of title 38, United States Code) while on active duty for a period of more than 30 days; and (2) represents all or a portion of the wages the individual would have received from the Employer if the individual were performing services for the Employer. The Differential Wage Payment provisions will apply to wages paid after December 31, 2008.

"Direct Rollover": A payment by the Plan to the Eligible Retirement Plan specified by the Distributee.

"Disability": Notwithstanding anything contained in the Plan to the contrary, an Employer may elect in the Adoption Agreement the manner in which a Participant is determined to have a Disability. Such determination required to be made by the Plan Administrator in connection with the permanence and degree of a physical or mental impairment will be made in a uniform and nondiscriminatory manner based on medical evidence.

Generally, a person is Disabled if they suffer from a medically determinable physical or mental impairment that may be expected to result in death or to last for a continuous period of not less than 12 months and that renders them incapable of performing their duties.

"Disabled": An Employee is Disabled if they have a Disability.

"Distributable Benefit": The benefit to which a Participant is entitled following termination of their employment or the date they are determined to be Disabled, if so elected in the Adoption Agreement.

"Distributee": An Employee or former Employee. In addition, the Employee's or former Employee's surviving spouse and the Employee's or the former Employee's spouse or former spouse who is an Alternate Payee under a Qualified Domestic Relations Order, are Distributees with regard to the interest of the spouse or former spouse. For distributions after December 31, 2006, a Distributee includes the Employee's or former Employee's non-spouse Designated Beneficiary, in which case, the distribution can only be transferred to a traditional or Roth IRA established on behalf of the non-spouse Designated Beneficiary for the purpose of receiving the distribution.

"Distribution Determination Date": The date as of which the Distributable Benefit of a Participant is determined, as discussed in Section 2.5.1(d).

"Early Retirement Age": The age specified as the Early Retirement Age, if any, in the Adoption Agreement.

"Early Retirement Date": The date specified as the Early Retirement Date, if any, in the Adoption Agreement.

"Earned Income": The net earnings from self-employment in the trade or business with respect to which the Plan is established for which personal services of the Participant are a material income-producing factor. Net earnings will be determined without regard to items not included in gross income and the deductions allocable to such items but with regard to the deduction allowed to the taxpayer by Code section 164(f). Net earnings will be reduced by contributions to a qualified plan to the extent deductible under Code section 404.

"Elapsed Time Method": A method of measuring service for purposes of eligibility to participate, vesting, and contribution allocations that determines service by reference to the total period of time that elapses while the Employee is employed (i.e., while the employment relationship exists) with the Employer or Employers maintaining this Plan. The Adoption Agreement may specify different methods to determine service for purposes of vesting, participation, and benefit accrual. Treasury Regulation section 1.410(a)-7 provides additional detail about the application of this method.

- (a) General rule. For purposes of eligibility to participate and vesting, an Employee's service begins on their Employment Commencement Date, and service is counted until their Severance From Service Date, as such terms are defined in Subsection (f) below.
- (b) "Service spanning" The following rules apply for purposes of eligibility to participate and vesting.
 - (1) The Period of Severance from the date an Employee severs employment due to a quit, discharge, or retirement to the date the Employee again performs an Hour of Service, within the meaning of Labor Regulations section 2530.200b-2(a)(1), counts as a Period of Service unless such Period of Severance is 12 months or longer.
 - (2) If an Employee is absent from service for less than 12 months for any other reason (e.g., layoff) and then quits, is discharged, or retires, the period of time during which the Employee may return and receive credit begins on the Severance From Service Date and ends 1 year after the first day of absence (e.g., first day of layoff).
 - (3) If an Employee returns to service more than 12 months after the date that service first ended, then such date will be a Reemployment Commencement Date and the Plan will resume counting the Employee's service from that date and will not count the intervening Period of Severance.

As a result of the operation of these rules, a severance from service (e.g., a quit), or an absence (e.g., layoff) followed by a severance from service, never results in a period of time of more than 1 year being required to be taken into account after an Employee severs from service or is absent from service.

(c) Additional Rules regarding Vesting.

- (1) An Employee is credited with a number of Years of Vesting Service equal to the number of whole years of the Employee's Period of Service, whether or not the Employee completed such Periods of Service consecutively.
- (2) The whole years of an Employee's Period of Service are determined by aggregating any non-successive Periods of Service and by aggregating any less than whole year Periods of Service (whether or not consecutive) on the basis that 12 Months of Service or 365 days of service equal a whole Year of Vesting Service. Fractional months are aggregated into Months of Service.
- (3) After calculating an Employee's Period of Service in the manner prescribed in this Subsection, any remaining service that is less than a whole year, 12-month, or 365-day Period of Service is disregarded for purposes of determining an Employee's nonforfeitable percentage of accrued benefits derived from Employer Contributions.

- (d) Contribution Allocations. For purposes of contribution allocations, an Employee's service is taken into account from their Participation Commencement Date until their Severance From Service Date. Periods of Severance under any circumstances are not taken into account.
- (e) Break in Service Exception. If an Employee is absent from work for maternity or paternity reasons, the 12-consecutive month period ending on the first anniversary of the first day of such absence will not constitute a Break in Service. For purposes of this Subsection, an absence of work for maternity or paternity reasons means an absence: (1) due to the Employee's pregnancy, (2) due to the birth of the Employee's child, (3) due to a child's placement with the Employee as part of the Employee's adoption of such child, or (4) to care for a child for a period beginning immediately following such birth or placement.

(f) Definitions.

- (1) Severance From Service Date: A "severance from service" occurs on the earlier of:
 - (i) The date on which an Employee quits, retires, is discharged, or dies; or
 - (ii) The first anniversary of the first date of a period in which an Employee remains absent from service (with or without pay) with the Employer or Employers maintaining the Plan for any reason other than a quit, retirement, discharge or death, such as vacation, holiday, sickness, Disability, leave of absence, or layoff.
- (2) Reemployment Commencement Date: The first date, following a Period of Severance from service that is not required to be taken into account under the service spanning rules of Subsection (b), on which the Employee performs an Hour of Service within the meaning of Labor Regulations section 2530.200b-2(a)(1) for the Employer or Employers maintaining this Plan.
- (3) Participation Commencement Date: The date a Participant first commences participation under the Plan.
- (4) Period Of Severance: The period of time commencing on the Severance From Service Date and ending on the date on which the Employee again performs an Hour of Service within the meaning of Labor Regulations section 2530.200b-2(a)(1) for an Employer or Employers maintaining this Plan.

(5) Period of Service:

- (i) General Rule: The period commencing on the Employee's Employment Commencement Date or Reemployment Commencement Date, whichever is applicable, and ending on the Severance from Service Date.
- (ii) Aggregation Rule: Periods of Service will be aggregated unless such periods may be disregarded under Code sections 410(a)(5) or 411(a)(4).
- (iii) Month of Service: 30 days are deemed to be a Month of Service.

"Elective Account": An Account established and maintained for a Participant for accounting purposes with respect to their Elective Deferrals.

"Elective Deferrals": Any contribution made by the Employer to the Plan at the election of the Participant or operation of the Plan, in lieu of cash compensation, including a contribution made pursuant to a salary reduction agreement or other deferral mechanism.

With respect to any taxable year, a Participant's Elective Deferrals are the sum of all Employer contributions made on behalf of the Participant pursuant to an election to defer under any qualified CODA

as described in Code section 401(k), any simplified employee pension cash or deferred arrangement as described in Code section 408(k)(6), any SIMPLE IRA Plan described in Code section 408(p), any plan as described under Code section 501(c)(18), and any employer contributions made on the behalf of a Participant for the purchase of an annuity contract under Code section 403(b) pursuant to a salary reduction agreement. Elective Deferrals will not include any deferrals properly distributed as Excess Annual Additions.

For years beginning after 2005, the term "Elective Deferrals" includes Pre-tax Elective Deferrals and Roth Deferrals.

"Eligibility Computation Period": For purposes of determining Years of Service and Breaks in Service for purposes of eligibility, the initial Eligibility Computation Period is the 12 consecutive month period beginning on the date the Employee first performs an Hour of Service for the Employer (Employment Commencement Date).

If in accordance with the election in the Adoption Agreement, the subsequent periods may be:

- (a) The succeeding 12-consecutive month periods that commence with the first anniversary of the Employee's Employment Commencement Date, or
- (b) The first Plan Year that commences prior to the first anniversary of the Employee's Employment Commencement Date. An Employee who is credited with 1,000 Hours of Service in both the initial Eligibility Computation Period and the first Plan Year that commences prior to the first anniversary of the Employee's initial Eligibility Computation Period will be credited with 2 Years of Service for purposes of eligibility to participate.

"Eligible Automatic Contribution Arrangement" (EACA): An arrangement in a cash or deferred plan as defined in Section 2.2.3(b) of this Plan under which an Employee that has not affirmatively elected to defer Compensation into the Plan will have an amount deferred on their behalf by the Plan Administrator.

"Eligible Employee": An Employee who has met the eligibility requirements of the Plan.

"Eligible Retirement Plan": An eligible plan under Code section 457(b) that is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state and that agrees to separately account for amounts transferred into such plan from this Plan, an individual retirement account described in Code section 408(a), an individual retirement annuity described in Code section 408(b), an annuity plan described in Code section 403(a), an annuity contract described in Code section 403(b), or a qualified plan described in Code section 401(a), that accepts the Distributee's Eligible Rollover Distribution. The definition of Eligible Retirement Plan will also apply in the case of a distribution to a surviving spouse, or to a spouse or former spouse who is an Alternate Payee under a Qualified Domestic Relation Order. If any portion of an Eligible Rollover Distribution is attributable to payments or distributions from a designated Roth Account, an Eligible Retirement Plan with respect to such portion will include only another designated Roth Account of the individual from whose account the payments or distributions were made, or a Roth IRA of such individual.

"Eligible Rollover Distribution": Any distribution of all or any portion of the balance to the credit of the Distributee, except that an Eligible Rollover Distribution does not include: any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the Distributee or the joint lives (or joint life expectancies) of the Distributee and the Distributee's Designated Beneficiary, or for a specified period of ten years or more; any distribution to the extent such distribution is required under Code section 401(a)(9); any Hardship distribution; the portion of any other distribution that is not includible in gross income (determined without regard to the exclusion for net unrealized appreciation with respect to employer securities); and any other distribution that is reasonably expected to total less \$200 during a year. For purposes of the \$200 rule, a distribution from a designated Roth Account and a distribution from other Accounts under the Plan are treated as made under separate plans.

A portion of a distribution will not fail to be an Eligible Rollover Distribution merely because the portion consists of after-tax Employee contributions that are not includible in gross income. However, such portion may be transferred only to an individual retirement account or annuity described in Code sections 408(a) or (b), or to a qualified defined contribution plan described in Code sections 401(a) or 403(a) that agrees to separately account for amounts so transferred, including separately accounting for the portion of such distribution that is includible in gross income and the portion of such distribution that is not so includible.

"Employee": A person who is currently or hereafter employed by the Employer, or by any other employer aggregated under Code sections 414(b), (c), (m), (n), or (o) and the regulations thereunder, including a Leased Employee subject to Code section 414(n) and a self-employed owner of an unincorporated employer, but, unless otherwise provided in the Adoption Agreement, excluding (a) an Employee who is a nonresident alien (within the meaning of Code section 7701(b)(1)(B) deriving no earned income (within the meaning of Code section 911(d)(2)) from the Employer that constitutes income from sources within the United States (within the meaning of Code section 861(a)(3)); and (b) employees who are included in the unit of Employees covered by a collective bargaining agreement between the Employer and employee representatives, provided benefits were the subject of good faith bargaining and 2% or less of the employees of the Employer who are covered pursuant to that agreement are "professional employees" as defined in Treasury Regulations section 1.410(b)-9. For this purpose, the term "employee representatives" does not include any organization more than half of whose members are Employees who are owners, officers, or executives of the Employer.

Except as otherwise provided in this paragraph, if the Employer elects in the Adoption Agreement to exclude Employees who became Employees as the result of a "Code section 410(b)(6)(C) transaction," then such Employees will not be "Eligible Employees" prior to the expiration of the transition period beginning on the date of the transaction and ending on the last day of the first Plan Year beginning after the date of the transaction. A "Code section 410(b)(6)(C) transaction" is an asset or stock acquisition, merger, or similar transaction involving a change in the Employer of the Employees of a trade or business. However, if a separate trade or business becomes a Related Employer as the result of a "Code section 410(b)(6)(C) transaction," then Employees of such Related Employer will not be treated as "Eligible Employees" prior to the date the entity adopts the Plan as a participating Employer. For purposes of this Section, "Related Employer" means any trade or business related to the Employer by the application of Code sections 414(b), (c), (m), (n), or (o).

In the case of an individuals who are not contemporaneously classified as Employees of the Employer for purposes of the Employer's payroll system (including, without limitation, individuals employed by temporary help firms, technical help firms, staffing firms, employee leasing firms, professional employer organizations, or other staffing firms whether or not deemed to be "common law" Employees or Leased Employees) are not considered to be Eligible Employees of the Employer and will not be eligible to participate in the Plan. The preceding sentence will not apply in the case of the Employer out-sourcing its human resource functions for Employees that the Employer would otherwise treat as "common law' Employees. In the event any such individuals are reclassified as Employees for any purpose (including without limitation, common law or statutory employees, by any action of any third party, including, without limitation, any government agency, or as a result of any private lawsuit, action, or administrative proceeding) such individuals shall, notwithstanding such reclassification, remain ineligible for participation hereunder. Notwithstanding foregoing, the exclusive means for individuals who are not contemporaneously classified as an Employee of the Employer on the Employer's payroll system to become eligible to participate in this Plan is through an amendment to this Plan, duly executed by the Employer, that specifically renders such individuals eligible for participation hereunder.

This Plan is to be construed to exclude all individuals who are not considered Employees for purposes of the Employer's payroll system, and the Plan Administrator is authorized to do so, despite the fact that its decision may result in the inadvertent loss of the Plan's tax qualification requiring an amendment of the Plan's eligibility provisions.

"Employer": The Employer maintaining the Plan and those Employers required to be aggregated with the Employer under Code sections 414(b), (c), (m) or (o). An individual who owns the entire interest of an unincorporated trade or business is treated as the Employer. Also, a partnership is treated as the Employer of each partner and each Employee of the partnership as well as any of its affiliates, successors or assigns

that adopt the Plan. However, no mere change in the identity, form, or organization of the Employer will affect its status under the Plan in any manner, and, if the name of the Employer is hereafter changed, a corresponding change will be deemed to have been made in the name of the Plan and references herein to the Employer will be deemed to refer to the Employer as it is then known.

"Employer Account": An Account established and maintained for a Participant for accounting purposes with respect to their share of Employer Contributions, Non-Elective Contributions, and Forfeitures.

"Employer Contribution": A contribution by the Employer to a money purchase pension plan, profit sharing plan, or a Non-Elective Contribution to a cash or deferred profit sharing plan.

"Employment Commencement Date": The date on which an Employee first performs an Hour of Service within the meaning of Labor Regulations section 2530.200b-2(a)(1) for the Employer or Employers maintaining this Plan.

"Entry Date": The date or dates specified as the Entry Date in the Adoption Agreement.

"Excess Aggregate Contributions": With respect to any Plan Year, the excess of:

- (a) The aggregate contribution percentage amounts taken into account in computing the numerator of the contribution percentage actually made on behalf of Highly Compensated Employees for such Plan Year, over
- (b) The maximum contribution percentage amounts permitted by the ACP test (determined by hypothetically reducing contributions made on behalf of Highly Compensated Employees in order of their contribution percentages beginning with the highest of such percentages). Such determination will be made after first determining Excess Elective Deferrals and then determining Excess Contributions, as required in Part II, Article VII.

"Excess Annual Addition": The portion of the allocation of contributions and Forfeitures that cannot be added to a Participant's Accounts due to the limitations on Annual Additions contained in Part III, Article II.

"Excess Contributions": With respect to any Plan Year, the excess of:

- (a) The aggregate amount of Employer contributions actually taken into account in computing the ADP of Highly Compensated Employees for such Plan Year, over
- **(b)** The maximum amount of such contributions permitted by the ADP test (determined hypothetically by reducing contributions made on behalf of Highly Compensated Employees in order of their contribution percentages, beginning with the highest of such percentages).

Excess Contributions will be reduced by Excess Deferrals previously distributed.

"Excess Elective Deferrals": Those Elective Deferrals that are made either:

- (a) During the Participant's taxable year and exceed the dollar limitation under such Code section 402(g) (including, if applicable, the dollar limitation on Catch-up Contributions defined in Code section 414(v)) for such year) or a lesser limitation set by the Adoption Agreement; or
- (b) During a calendar year and exceed the dollar limitation under Code section 402(g) (including, if applicable, the dollar limitation on Catch-up Contributions defined in Code section 414(v)) for the Participant's taxable year beginning in such calendar year, or a lesser limitation set by the Adoption Agreement, counting only Elective Deferrals made under this Plan and any other plan, contract, or arrangement maintained by the Employer.

Excess Elective Deferrals will be treated as Annual Additions under the Plan, unless such amounts are distributed no later than the first April 15 following the close of the Participant's taxable year.

"Fiduciary": The Plan Administrator, the Trustee, and any other person who has discretionary authority or control in the management of the Plan or the disposition of Trust assets.

"5-Percent Owner": A Participant who satisfies the definition of a 5-percent owner within the meaning of Code section 416(i) at any time during the Plan Year ending with or within the calendar year.

"Forfeitures": The non-vested portion of a Participant's Account balance that is forfeited on or after their termination of employment in accordance with Section 2.4.4 and may be allocated to other Participants, used to defray the administrative expenses of the Plan, or considered as part of the Employer's Contribution if permitted by Part II, Article III.

"Highly Compensated Employee" (HCE): Any Employee of an Employer required to be aggregated under Code sections 414(b), (c), (m) or (o) during a Plan Year who:

- (a) Was a 5-Percent Owner of the Employer at any time during the year or the preceding year, or
- **(b)** For the preceding year
 - (1) Had "compensation" from the Employer in excess of \$80,000 (as adjusted by the Secretary of the Treasury pursuant to Code section 415(d)), and
 - (2) If the Employer elects in the Adoption Agreement the application of this clause for such preceding year, was in the top-paid group of Employees for such preceding year.

For this purpose, an Employee is in the top-paid group of Employees for any year if such Employee is in the group consisting of the top 20% of the Employees when ranked on the basis of compensation paid during such year.

For these purposes, the applicable year of the Plan for which a determination is being made is called a "determination year" and the preceding 12-month period is called a "look-back year."

The determination of whether an Employee had compensation in excess of \$80,000, as adjusted, will be made based on compensation paid during the preceding Plan Year, unless the Employer has made a calendar year data election in the Adoption Agreement. The effect of the calendar year data election is that the look-back year becomes the calendar year beginning with or within the look-back year.

The top-paid group election and the calendar year data election must apply consistently to the determination years of all plans of the Employer.

Generally, a former Employee will be treated as a Highly Compensated Employee if:

- (1) Such Employee was a Highly Compensated Employee when such Employee separated from service or
- (2) Such Employee was an active Highly Compensated Employee for any Plan Year that ended on or after the Employee's 55th birthday.

The determination of whether a former Employee is a Highly Compensated Employee will be based on the rules applicable to determining Highly Compensated Employee status as in effect for that determination year, in accordance with temporary Treasury Regulations sections 1.414(q)-1T, A-7 and A-4 and Notice 97-45

The Plan Administrator will apply rules to determine who is a Highly Compensated Employee as set forth in an administrative policy, so long as such rules are reasonable, nondiscriminatory, and uniformly and consistently applied.

"Hour of Service": An hour for which:

- (a) The Employee is paid, or entitled to payment, by the Employer for the performance of duties. These hours will be credited to the Employee for the computation period in which the duties are performed.
- (b) The Employee is paid or entitled to payment by the Employer during which no duties are performed (irrespective of whether the employment relationship has terminated) due to vacation, holiday, illness, incapacity (including Disability), layoff, jury duty, military duty, or leave of absence. These hours will be credited to the Employee for the computation period during which no duties are performed, beginning with the first Hour of Service to which the payment relates. No more than 501 Hours of Service will be credited under this Subsection for any single continuous period (whether or not such period occurs in a single computation period).
- (c) Back pay, irrespective of mitigation of damages, has been either awarded or agreed to by the Employer. These hours will be credited to the Employee for the period to which the award or agreement pertains rather than the period in which the award, agreement, or payment is made.

Hours of Service will not be credited under both (a) and (b), above, as the case may be, and under (c) above. Notwithstanding the preceding, no Hours of Service will be credited to the Employee by reason of a payment made or due under a plan maintained solely for the purpose of complying with applicable worker's compensation, or unemployment compensation or disability insurance laws, and no Hours of Service will be credited by reason of a payment that solely reimburses an Employee for medical or medically related expenses incurred by the Employee. The determination of Hours of Service for reasons other than the performance of duties and the crediting of Hours of Service to computation periods will be made in accord with the provisions of Labor Regulations sections 2530.200b-2(b) and (c) that are incorporated herein by reference.

Solely for the purposes of determining whether an Employee has incurred a Break in Service, an Employee will be credited with the number of Hours of Service that would otherwise have been credited to such individual but for the absence, and in any case in which such hours cannot be determined, with 8 Hours of Service for any day that the Employee is absent from work by reason of the Employee's pregnancy, the birth of a child of the Employee, the placement of a child with the Employee in connection with the adoption of such child by the Employee or for purposes of caring for such child for a period beginning immediately following such birth or placement. Such Hours of Service will be credited in the computation period in which the absence from work begins if such crediting is necessary to prevent the Employee from incurring a Break in Service in such period, and in any other case, in the immediately following computation period. Notwithstanding the foregoing, no credit will be given for such service unless the Employee furnishes to the Plan Administrator information to establish that the absence from work is for the reasons indicated and the number of days for which there was such an absence.

In the case of a Multiple Employer Plan, an Employee will be credited with Hours of Service with all the adopting Employers of the Multiple Employer Plan.

In the event the Employer does not maintain records of the actual hours for which an Employee is paid or entitled to payment, credit for service will be given in accordance with the method selected in the Adoption Agreement.

Service with another business entity that is, along with the Employer, a member of a controlled group of corporations under Code section 414(b), an affiliated service group under Code section 414(m) or trades or businesses under common control under Code section 414(c), or that is otherwise required to be aggregated with the Employer pursuant to Code section 414(o) and the regulations issued thereunder will be treated as service for the Employer. Hours of Service will be credited for any individual considered an Employee for purposes of this Plan under Code sections 414(n) or 414(o) and the regulations issued thereunder.

If elected in the Adoption Agreement, service for a predecessor to the Employer, whether as an Employee or a Self-Employed Individual, will be treated as service for the Employer. If the Employer maintains the plan of a predecessor employer, service with such predecessor will be treated as service for the Employer.

If elected in the Adoption Agreement, service with a prior employer whether as an Employee or a Self-Employer Individual will be treated as service for the Employer for legitimate business purposes so long as the grant of such service does not discriminate in favor of Highly Compensated Employees. Service with a prior employer will be limited to 5 years.

"In-Plan Roth Conversion Account": An Account which maintains pre-tax Elective Deferrals or other Employer Contributions that have been converted to after-tax Roth contributions. If elected in the Adoption Agreement, if the Plan includes a designated Roth Deferral after September 27, 2010, it may permit a Participant to take an Eligible Rollover Distribution pursuant to Section 2.5.12 and transfer such amounts into an In-Plan Roth Conversion Account within the Plan. The Plan will maintain such records as are necessary for the proper reporting of in-plan Roth rollovers.

"Insurer": Any insurance company that has issued a Life Insurance Policy.

"Integration Level": The Integration Level will be equal to the Taxable Wage Base or such lesser amount specified in the Adoption Agreement. The Integration Level will be deemed to be the full amount specified in the Adoption Agreement, even though a Participant's Compensation may include less than a full year's compensation because either they commenced participation after the first day of the Compensation Computation Period.

"Joint and Survivor Annuity": An annuity for the life of the Participant with a survivor annuity for the life of the spouse (or other Beneficiary) that is not less than 50% and not more than 100% of the amount of the annuity that is payable during the joint lives of the Participant and the spouse or other Beneficiary and that is the amount of benefit that can be purchased with the Participant's vested Account balances. The percentage of the survivor annuity will be 50% unless a different percentage is elected by the Employer in the Adoption Agreement.

"Key Employee": For Plan Years beginning after December 31, 2001, an Employee or former Employee (including any deceased Employee) who, at any time during the Plan Year that includes the Determination Date (as defined in Subsection 2.6.2(g)) is either:

- (a) An officer of the Employer having an Annual Compensation (as defined in Subsection 2.6.2(b)) greater \$130,000 (as adjusted by Code section 416(i) for Plan Years beginning after December 31, 2002);
- (b) A 5-Percent Owner; or
- (c) Any person who owns directly or indirectly more than 1% of the outstanding stock of the Employer or stock possessing more 1% of the total combined voting power of all stock of the Employer or, in the case of an unincorporated Employer, the capital or profits interest in the Employer and has Annual Compensation (as defined in Subsection 2.6.2(b)) from the Employer of more than \$150,000.

The determination of who is a Key Employee will be made in accordance with Code section 416(i)(1) and the regulations thereunder.

"Lead Sponsor": The Employer or its successor, agent or representative who is designated as the Plan Sponsor of the Multiple Employer Plan.

"Leased Employee": Any person (other than an Employee of the recipient) who, pursuant to an agreement between the recipient and any other person (the "leasing organization"), has performed services for the recipient (or for the recipient and related persons determined in accordance with Code section 414(n)(6)) on a substantially full time basis for a period of at least 1 year and such services are performed under the primary direction or control of the recipient; provided, that any such person will not be taken into account if:

(a) Such person is covered by a money purchase pension plan providing

- (1) A nonintegrated Employer Contribution rate of at least 10% of compensation, as defined in Code section 415(c)(3) and Subsection 3.2.5(a)(iii) of the Plan, but including in compensation amounts contributed by the employer pursuant to a salary reduction agreement that are excludable from the person's gross income under Code sections 125, 132(f)(4), 402(e)(3), 402(h), or 403(b);
- (2) Immediate participation; and
- (3) Full and immediate vesting; and
- **(b)** Leased Employees do not constitute more than 20% of the workforce of the recipient who are not Highly Compensated Employees.

Contributions or benefits provided a Leased Employee by the leasing organization that are attributable to services performed for the recipient employer will be treated as provided by the recipient employer.

"Life Insurance Policy": A life insurance, group or individual annuity, deposit administration, or endowment policy or contract owned by the Trust, including any Life Insurance Policy on the life of a Participant, the life of someone in whom the Participant has an insurable interest, or the joint lives of a Participant and someone in whom the Participant has an insurable interest.

"Limitation Year": Unless otherwise specified in the Adoption Agreement, the Limitation Year will be the Plan Year, provided that all qualified plans maintained by the Employer must use the same Limitation Year. If the Limitation Year is amended to a different 12 consecutive month period, the new Limitation Year must begin on a date within the Limitation Year during which the amendment is made.

"Mass Submitter": DATAIR Employee Benefits Systems, Inc.

"Matching Account": An Account established and maintained for a Participant for accounting purposes with respect to their share of Matching Contributions and Forfeitures.

"Matching Contribution": A contribution to the Plan by the Employer that matches in whole or in part an Elective Deferral on behalf of an electing Employee.

"Minimum Top-Heavy Allocation": The minimum allocation of a Participant for which provision is made in Sections 2.3.4 and 2.6.1.

"Multiple Employer Plan" (MEP): A plan maintained by two or more employer who are not related pursuant to Code sections 414(d), 414(c), 414(m) or 414(o) and that operates as a single plan as described in Code section 413(c).

"Non-Elective Contribution": A contribution to a cash or deferred profit sharing plan by the Employer that is not a Qualified Non-Elective Contribution, Matching Contribution, Elective Deferral, or Qualified Matching Contribution.

"Non-Highly Compensated Employee" (NHCE): Any Employee who does not meet the definition of a Highly Compensated Employee in effect for a particular Plan Year.

"Non-Key Employee": Any Employee who does not meet the definition of a Key Employee in effect for a particular Plan Year.

"Normal Retirement Age" and "NRA": The age specified as the Normal Retirement Age in the Adoption Agreement, but in no event the later of age 65 or the 5th anniversary of the participation in the Plan. For this purpose only, participation is assumed to commence as of the first day of the first Plan Year in which the Employee became a Participant.

In the case of a money purchase pension plan:

- (1) the age may not be earlier than the earliest retirement age that is reasonably representative of the typical retirement age for the industry in which the Plan Participants work;
- (2) an age of 62 or older automatically meets this requirement;
- (3) the age selected must not be earlier than 55.

NRA is the age selected in the Adoption Agreement. If the Employer enforces a mandatory retirement age, the NRA is the lesser of that mandatory age or the age specified in the Adoption Agreement.

"Normal Retirement Date" and "NRD": The date specified in the Adoption Agreement as the Normal Retirement Date but in no event more than 6 months after the Participant attains their Normal Retirement Age.

"Owner-Employee": An individual who is a sole proprietor or a partner owning more than 10% of either the capital or profits interest of the partnership.

"Participant": Any Eligible Employee who enters the Plan after meeting the requirements of Section 2.1.1.

"Participating Employer": A Related Employer or Affiliated Employer, other than the Lead Sponsor, that has adopted the Plan.

"Plan": The defined contribution plan for Employees as set forth in this Agreement, the Trust, and the Adoption Agreement, together with any amendments or supplements thereto.

"Plan Administrator": The person, persons, or entity appointed by the Employer to administer the Plan, or, if the Employer fails to make such appointment, the Employer.

"Plan Sponsor": The Plan Sponsor specified in the Adoption Agreement. If elected in the Adoption Agreement the Plan Sponsor may serve as the Lead Sponsor for a Multiple Employer Plan.

"Plan Year" or "Year": The 12 consecutive month period designated by the Employer in the Adoption Agreement.

"Post Severance Compensation": Compensation paid a Participant after a severance of employment as defined in the definition of Compensation in Section 3.2.5 of the Plan.

"Preretirement Survivor Annuity": A survivor annuity for the life of the surviving spouse (or other Beneficiary) of the Participant purchasable with an amount equal to at least 50% of the vested Account balance of the Participant as of the date of the Participant's death, reduced if necessary, to reflect any security interest held by the Plan by reason of a loan outstanding to the Participant for which a valid spousal consent has been obtained.

"Pre-tax Elective Account": An Account established and maintained for a Participant for accounting purposes with respect to their Pre-tax Elective Deferrals.

"Pre-tax Elective Deferral": Any Elective Deferral that is not includible in the Participant's gross income at the time deferred.

"Pre-tax IRA Account": An Account established and maintained for a Participant for accounting purposes with respect to their Pre-tax IRA Deferrals.

"Pre-tax IRA Contribution": Any Deemed IRA Contribution that is not includible in the Participant's gross income at the time it is made.

"Prevailing Wage Contribution": An Employer Contribution made pursuant to the Davis Bacon Act or any other state or federal prevailing wage law.

- "Primary Beneficiary under the Plan": An individual named as a Beneficiary under the Plan who has an unconditional right to all or a portion of the Employee's Account balance under the Plan upon the Employee's death.
- "Qualified Automatic Contribution Arrangement" and "QACA": An arrangement in a cash or deferred plan as defined in Section 2.2.3(c) of this Plan under which an Employee that has not affirmatively elected to defer Compensation into the Plan will have an amount deferred on their behalf by the Plan Administrator as well as certain other required Employer Contributions.
- "Qualified Domestic Relations Order" and "QDRO": A domestic relations order that creates or recognizes the existence of an Alternate Payee's right to receive all, or a portion of, the benefits payable with respect to a Participant under this or any other plan of the Employer and that the Plan Administrator has determined to satisfy the conditions of Code section 414(p).
- "Qualified Joint and Survivor Annuity": A Joint and Survivor Annuity that is immediately payable and for which the Beneficiary is the Participant's spouse. The percentage of the survivor annuity will be 50% unless a different percentage is elected by the Employer in the Adoption Agreement.
- "Qualified Matching Account": An Account established and maintained for a Participant for accounting purposes with respect to their share of Qualified Matching Contributions.
- "Qualified Matching Contribution": A Matching Contribution that is 100% vested and nonforfeitable when made and that is distributable only in accordance with the distribution restrictions applicable to Elective Deferrals under the Plan, except that a Qualified Matching Contribution may not be distributed on account of Hardship prior to the Participant's attainment of age 59 1/2.
- "Qualified Military Service": Any service in the uniformed services, whether on a voluntary or involuntary basis, including active duty, active duty for training, initial active duty for training, inactive training, full-time National Guard duty, and the period of time a person is absent to determine their fitness to perform any such duty. Uniformed services refers to the Armed Forces, the Army National Guard, the Air National Guard, the commissioned corps of the Public Health Service, and any other category of persons designated by the President in time of war or national emergency. Notwithstanding any provision of this Plan to the contrary, contributions, benefits, and service credit with respect to Qualified Military Service will be provided in accordance with section 414(u) of the Code and applicable DOL and Treasury rules and regulations.

In addition, the survivors of any Participant who dies on or after January 1, 2007, while performing Qualified Military Service, are entitled to any additional benefits (other than contributions relating to the period of Qualified Military Service, but including vesting service credit for such period and any ancillary life insurance or other survivor benefits) that would have been provided under the Plan had the Participant resumed employment on the day preceding the Participant's death and then terminated employment on account of death.

- "Qualified Non-Elective Account": An Account established and maintained for a Participant for accounting purposes with respect to their share of Qualified Non-Elective Contributions.
- "Qualified Non-Elective Contribution": An Employer Contribution to a cash or deferred profit sharing plan that is neither a Non-Elective Contribution, Matching Contribution, Qualified Matching Contribution, nor an Elective Deferral, that is 100% vested and nonforfeitable when made, that a Participant may not elect to receive in cash, and that is distributable only in accordance with the distribution restrictions applicable to Elective Deferrals under the Plan, except that Qualified Non-Elective Contributions may not be distributed on account of Hardship prior to the Participant's attainment of age 59 1/2.
- "Qualified Optional Survivor Annuity": An immediate annuity for the life of the Participant with a survivor annuity for the life of the spouse which is equal to the applicable percentage of the amount of the annuity which is payable during the joint lives of the Participant and the spouse and which is the amount of benefit which can be purchased with the Participant's vested Account Balance. If the percentage of the survivor

annuity is less than 75%, the applicable percentage is 75%. If the percentage of the survivor annuity is greater than or equal to 75%, the applicable percentage is 50%.

"Qualified Preretirement Survivor Annuity": A Preretirement Survivor Annuity for which the Beneficiary is the Participant's spouse, and the benefit the Participant would have received is a Qualified Joint and Survivor Annuity (or the actuarial equivalent thereof).

"Qualifying Employer Securities or Real Property": Securities or real property of the Employer that the Trustee may acquire and hold pursuant to the applicable provisions of the Code and the Act.

"Related Employer": Any trade or business related to the Employer or affiliated employer by the application of Code sections 414(b), (c), (m), (n), or (o) (a required aggregation group). A Related Employer will not be considered a Participating Employer until they affirmatively elect to adopt the Plan.

"Rollover Account": An Account established and maintained for a Participant for accounting purposes with respect to amounts the Participant has rolled over to this Plan.

"Roth Deferral" (an after-tax contribution): Any contribution made by the Employer to the Plan at the election of the Participant that is: (1) designated irrevocably by the Participant at the time of the cash or deferred election as a Roth Deferral that is being made in lieu of all or a portion of the Pre-tax Elective Deferrals the Participant is otherwise eligible to make under the Plan; (2) treated by the Employer as includible in the Participant's income at the time the Participant would have received the contribution amount in cash if the Participant had not made the deferral election; and (3) maintained by the Plan in a separate account. Roth Deferrals must satisfy all Plan restrictions applicable to Elective Deferrals.

Under the separate accounting requirement, contributions and withdrawals of designated Roth Deferrals must be credited and debited to a designated Roth Deferral Account maintained for the Participant who made the designation, and the Plan must maintain a record of the Participant's investment in the contract (i.e., designated Roth Deferrals that have not been distributed) with respect to the Participant's designated Roth Deferral Account. In addition, gains, losses, and other credits or charges must be separately allocated on a reasonable and consistent basis to the designated Roth Deferral Account and other accounts under the Plan. However, Forfeitures may not be allocated to the designated Roth Deferral Account. The separate accounting requirement applies at the time the designated Roth Deferral is contributed to the Plan and must continue to apply until the designated Roth Deferral Account is completely distributed.

A designated Roth Deferral must satisfy all requirements applicable to Elective Deferrals made under the Plan. Distributions from Roth Deferral Accounts (other than corrective distributions) are not includible in the Participant's gross income if made after the Participant's death, Disability, or age 59-1/2 provided that such distribution occurs after 5 years from the establishment of the Roth Deferral Account for the Participant. Earnings on corrective distributions of Roth Deferrals are includible in gross income in the same manner as earnings on corrective distributions of Pre-tax Elective Deferrals.

"Roth Deferral Account": An Account established and maintained for a Participant for accounting purposes with respect to their Roth Deferrals.

"Roth IRA Account" (Roth IRA): An Account established and maintained for a Participant for accounting purposes with respect to their Roth IRA Contributions.

"Roth IRA Contribution" (an after-tax contribution): Any Deemed IRA Contribution that is (1) designated irrevocably by the Participant at the time of the cash or deferred election as a Roth IRA Contribution; (2) treated by the Employer as includible in the Participant's income at the time the Participant would have received the contribution amount in cash if the Participant had not made the deferral election; and (3) maintained by the Plan in a separate account.

"Segregated Account": An Account established and maintained for a Participant for accounting purposes with respect to their interest in a Segregated Fund.

- "Segregated Fund": Assets held in the name of the Trustee that have been segregated from the other assets of the Trust Fund in accordance with any of the provisions of the Plan.
- "Self-Employed Individual": An individual who has Earned Income for the taxable year from the trade or business for which the Plan is established or who would have had Earned Income but for the fact that the trade or business had no net profits for the taxable year.
- "Straight Life Annuity": An annuity payable in equal installments over the life of the Participant that terminates upon the Participant's death.
- "Taxable Wage Base": The contribution and benefit base in effect under section 230 of the Social Security Act at the beginning of the Plan Year.
- "True-up": The process of increasing an annual contribution that has been limited on a Compensation Computation Period basis to the amount and period determined when applying the limits elected in the Adoption Agreement.
- "Trustee": The persons, corporations, associations, or combination of them who acts as such from time to time hereunder. The term "Trustee" as used herein will also include a person holding the assets of a custodial account, an annuity contract, or other contract that is treated as a qualified trust pursuant to Code section 401(f), and references to the Trust Fund will be construed to apply to such custodial account, annuity contract, or other contract.
- "Trust Fund": All money and property of every kind and character held by the Trustee pursuant to the Plan. Such Trust will be established by a Trust Agreement separate from the Adoption Agreement and Base Plan Document.
- "Valuation Date": The date or dates specified as the Valuation Date(s) in the Adoption Agreement and any additional dates the Plan Administrator may deem necessary in their sole discretion.
- "Voluntary Account": An Account established and maintained for a Participant for accounting purposes with respect to their Voluntary Employee Contributions that were not deductible by the Participant at the time they were made.
- "Voluntary Employee Contributions": Pre-tax contributions made by a Participant at their discretion, prior to Plan Years beginning in 1987, and after-tax contributions made by a Participant at their discretion, but not including Roth Deferrals or Roth IRA Contributions.
- "Year of Credited Service": A 12 consecutive month period used in determining a Participant's benefit allocation. Unless otherwise specified in the Adoption Agreement, an Employee is granted a Year of Credited Service for each Plan Year in which they earn 1,000 Hours of Service. The Plan's benefit formula may limit or exclude certain Years of Credited Service in the calculation of a Participant's annual benefit allocation.
- "Year of Service": A Year of Service is completed on the last day of a 12 consecutive month period (computation period) specified in the Adoption Agreement during which an Employee completes at least 1,000 Hours of Service. If elected in the Adoption Agreement, the hours requirement for the completion of a Year of Service for a particular purpose can be less than 1,000 hours and can be satisfied before the end of the Eligibility Computation Period. For purposes of the Elapsed Time Method, a Year of Service means 12 Months of Service, as defined in the definition of Elapsed Time Method in this Article, or such lesser number of months specified in the Adoption Agreement. All Years of Service will be taken into account for purposes of eligibility.
- "Year of Vesting Service": A Year of Vesting Service is the 12 consecutive month period specified in the Adoption Agreement during which an Employee completes at least 1,000 Hours of Service or such lesser number of hours specified in the Adoption Agreement. For purposes of the Elapsed Time Method, a Year of Vesting Service means either 12 Months of Service, as defined in the definition of Elapsed Time Method in

July 27, 2022 Agenda Item: 5A ATTACHMENT 1

this Article, 365 days of service, or such other period of service specified in the Adoption Agreement. The Plan Administrator will take into account all Years of Vesting Service unless the Employer specifies exclusions in the Adoption Agreement.

PART II

ARTICLE I

PARTICIPATION

2.1.1 Eligibility Requirements. Each Eligible Employee will become a participant in this Plan and receive an appropriate allocation of contributions upon reaching an Entry Date while still an Employee following the satisfaction of the eligibility requirements set forth in the Adoption Agreement.

Under a plan with a cash or deferred arrangement ("CODA"), an Employee's eligibility to make Elective Deferrals may not be conditioned upon the completion of more than 1 Year of Service or the attainment of more than age 21.

An Employee that is a member of an employee classification that indirectly imposes an Hours of Service requirement (e.g., part-time, seasonal, or temporary) will be eligible to participate in the Plan in any Eligibility Computation Period in which the Employee attains age 21 and is credited with at least 1,000 Hours of Service, or such lesser age or Hours of Service required by the Plan.

2.1.2 Commencement of Participation. An Eligible Employee will become a Participant in the Plan on the applicable Entry Date selected in the Adoption Agreement.

2.1.3 Participation upon Reemployment.

(a) Reemployment after Satisfying Eligibility Requirements

Unless otherwise specified in the Adoption Agreement, a Participant whose employment terminates and who is subsequently reemployed will re-enter the Plan immediately on the date of their reemployment notwithstanding the fact that they received a distribution of all of their vested Benefit. In the event that an Employee completes the eligibility requirements set forth in the Adoption Agreement terminates their employment prior to the date they would have become a Participant, and they are subsequently reemployed after the date that they would have become a Participant, the Employee will be deemed to have met the eligibility requirements as of the date of their reemployment and will become a Participant on the date of their reemployment. If such Employee is reemployed prior to the date they would have become a Participant, had their employment had not terminated, they will become a Participant as of the date they would have become a Participant if their employment had not terminated.

(b) Reemployment before Satisfying Eligibility Requirements

Except as provided in the next paragraph, an Employee who has not met the eligibility requirements and whose employment terminates and who is subsequently reemployed will become a Participant in accordance with the provisions of Sections 2.1.1 and 2.1.2.

If the Plan has an eligibility requirement longer than one year, an Employee who terminates after completing a Year of Service and is subsequently reemployed prior to incurring a Break in Service will receive credit for that Year of Service for purposes of eligibility under Sections 2.1.1 and 2.1.2. However, an Employee who incurs a Break in Service will not receive credit for that Year of Service for the year prior to the break.

- (c) An individual returning from a period of Qualified Military Service will recommence participation in the Plan immediately upon reemployment.
- **2.1.4 Termination of Participation.** An Employee who has become a Participant will remain a Participant until the entire amount of their Distributable Benefit is distributed to them or in the event of their death, to their Beneficiary.

2.1.5 Plan Administrator's Determination. In the event any question arises regarding the eligibility of any person to become a Participant or the commencement of participation, the Plan Administrator will determine such question and the Plan Administrator's decision will be conclusive and binding.

2.1.6 One-time Election Not to Participate.

- (a) Employees who made an election not to participate in this Plan or any other plan of the Employer pursuant to (b) and (c) below will continue to be excluded from this Plan on and after the date the Plan was restated for post-EGTRRA.
- **(b)** Where an Employer may have elected to permit an Employee to make an election not to participate in the Plan. The election should:
 - (1) Include plans not yet established;
 - (2) Be for the duration of the Employee's employment with the Employer;
 - (3) Be a one-time irrevocable election;
 - (4) Be made no later than the first date an Employee becomes eligible under the Plan or any plan or arrangement of the Employer that is described in Code section 219(g)(5)(A) (whether or not such other plan or arrangement has terminated).

Such a one-time irrevocable election described in this Section will not be treated as having been made pursuant to a cash or deferred election.

- (c) In the case of an irrevocable election made on or before December 23, 1994:
 - (1) The election does not fail to be treated as a one-time irrevocable election under this Section merely because an Employee was previously eligible under another plan of the Employer (whether or not such other plan has terminated); and
 - (2) In the case of a plan in which partners may participate, the election does not fail to be treated as a one-time irrevocable election under this Section merely because the election was made after the partner's commencement of employment or after the Employee first became eligible under any plan of the Employer, provided that the election was made before the first day of the first Plan Year beginning after December 31, 1988, or, if later, March 31, 1989.
- **2.1.7 Change in Status.** If any Participant continues to be employed by the Employer or an affiliate for which service is required to be taken into account, but ceases to be a member of an eligible class of Employees for any reason (such as becoming covered by a collective bargaining agreement unless the collective bargaining agreement otherwise provides), the Participant will continue to be a Participant until the entire amount of their benefit is distributed. The individual will not be eligible for any Employer Contribution for purposes of Article III ("Allocations") during the period that the Participant is not an Eligible Employee for such reason. This means while a member of an ineligible class the Employee will not receive an allocation of Employer Contributions. Such Participant will continue to receive credit for Years of Vesting Service completed during the period for purposes of determining their vested and nonforfeitable interest in their Accounts. In the event that the individual subsequently again becomes a member of an eligible class of Employees, the individual will participate immediately upon the date of such change in status.

If a Participant incurs a Break in Service and is subsequently reemployed, eligibility to participate will be determined in accordance with Section 2.1.3. In the event that an individual who is not a member of an eligible class of Employees becomes a member of an eligible class, the individual will participate immediately if such individual has satisfied the eligibility requirements and would have otherwise previously become a Participant and will receive credit for Years of Vesting Service completed while in the ineligible class for purposes of determining their vested and nonforfeitable interest in their Accounts.

Any Participant who ceases to be a member of an eligible class of Employees and would otherwise be entitled to a Top-Heavy Minimum Contribution will have any such contribution based on Compensation while a member of an eligible class of Employees.

- **2.1.8 Existing Participants.** Unless otherwise specified in the Adoption Agreement, if an Employee was a Participant on the Effective Date of this Plan, under the provisions of the Plan as in effect immediately prior to the Effective Date, they will be a Participant on the Effective Date and the provisions of Sections 2.1.1 and 2.1.2, pertaining to participation, will not be applicable unless the Employee is a member of an excluded class, as provided in Section 2.1.7. The rights of a Participant who is a member of an excluded class and whose employment terminated prior to the Effective Date of this Plan will be determined under the provisions of the Plan as in effect at the time of such termination.
- **2.1.9 Service Crediting Method.** All Years of Service with the Employer will be used in to determine eligibility and break in service. The Employer may elect in the Adoption Agreement to use the Hours of Service method, Elapsed Time Method or any of the optional service counting methods permitted within the Plan for eligibility to participate, vesting, and application of the benefit formula. Thus, for example, service for eligibility may be determined by counting actual hours worked, while service for vesting could be determined either by the equivalencies as provided in the Adoption Agreement or by the Elapsed Time Method.

ARTICLE II

CONTRIBUTIONS

2.2.1 Employer Contributions.

- (a) Amount of Contribution.
 - (1) Money Purchase Pension Plan. The Employer will contribute to the Trust Fund each Plan Year such amount, including any Forfeitures to be applied, as set forth in the Adoption Agreement.
 - (2) **Profit Sharing Plan.** The Employer may contribute to the Trust Fund each Plan Year such amount as elected in the Adoption Agreement.
 - (3) Cash or Deferred Profit Sharing Plan.
 - (i) Amount of Non-Elective Contribution. The Employer will contribute to the Trust Fund each Plan Year such amount as a Non-Elective Contribution as the Employer elected in the Adoption Agreement.
 - (ii) Amount of Matching Contribution. Subject to applicable limitations provided by the Plan, the Employer will contribute to the Trust Fund each Plan Year with respect to the amount of Elective Deferrals on behalf of each electing Employee, a Matching Contribution determined in the manner set forth in the Adoption Agreement, to the extent permitted under Code section 401(m). If elected in the Adoption Agreement, Catch-up Contributions will not be treated as Elective Deferrals in the calculation of Matching Contributions. If the Participants' Elective Deferral elections are made on an annual basis, but Elective Deferrals are taken from Compensation on some other basis, the Employer must True-Up Matching Contributions at the end of the Plan Year to an annual basis, unless elected otherwise in the Adoption Agreement.
 - (iii) Amount of Qualified Non-Elective Contribution and Qualified Matching Contribution. The Employer may, if elected in the Adoption Agreement, contribute to the Trust Fund each Plan Year such amount as a Qualified Non-Elective Contribution or Qualified Matching Contribution as the Employer may determine. In addition, if the Employer has elected in the Adoption Agreement to use the Current Year Testing Method, in lieu of distributing Excess Contributions or Excess Aggregate Contributions as provided in Article VII below, and to the extent elected by the Employer in the Adoption Agreement, the Employer may make Qualified Non-Elective Contributions or Qualified Matching Contributions on behalf of Participants who are not Highly Compensated Employees that are sufficient to satisfy either the ADP test or the ACP test, or both, pursuant to regulations under the Code.
 - (4) Prevailing Wage Contribution. Notwithstanding any other provision of this subsection 2.2.1, the Employer may, if elected in the Adoption Agreement, contribute a required prevailing wage contribution. If the Employer elects in the Adoption Agreement such amounts may be used to reduce or offset other Employer Contributions. The Plan Administrator will determine the contribution to be reduced or offset.
 - (5) In-kind Contributions The contribution of property, other than Qualifying Employer Securities, to a profit sharing plan or cash or deferred profit sharing plan, will be permitted, so long as the contribution is discretionary and the property is unencumbered. The contribution of Qualifying Employer Securities will be permitted to a pension plan subject to the requirements of ERISA section 408(e).
- **(b) Limitation.** The Employer's Contribution for any Plan Year will not exceed the maximum amount deductible from the Employer's income for such year for federal income tax purposes under the

applicable Code sections.

- (c) Time of Contribution. All Employer Contributions will be delivered to the Trustee not later than the date fixed by law for the filing of the Employer's federal income tax return for the Year for which such contribution is made (including any extensions of time granted by the Internal Revenue Service for filing such return).
- (d) Determination of Amount to be Final. The Employer's determination of the amount of its contribution hereunder will be in all respects final, binding, and conclusive on all persons or parties having or claiming any rights under this Agreement or under the Plan and the Trust created hereby. Under no circumstances and in no event will any Participant, Beneficiary, or other person or party have any right to examine the Employer's books or records. This subparagraph will not operate to limit any right to review granted by ERISA.
- **(e) True-Up.** The Employer may make an additional contribution ("true up") on behalf of each Participant in the amount of the positive difference, if any, between the contributions that would have been allocated to their Account had such contributions been determined on the basis of Compensation for the entire Plan Year and the contributions previously allocated to such Participant's Account.

2.2.2 Elective Deferrals Contributed by the Employer on Behalf of Electing Employees.

(a) Amount of Deferrals. If the Plan is designated in the Adoption Agreement as a Cash or Deferred Profit Sharing Plan, each Participant may elect to have the Employer contribute to the Trust on their behalf for any Plan Year during which they are a Participant such amounts expressed either in dollars or percentages of their Compensation as they may elect, that would otherwise be payable by the Employer as Compensation. If elected in the Adoption Agreement, all or a portion of such Deferral amounts may be designated Roth Deferrals by the Participant. Such amount cannot exceed the dollar limitation on Elective Deferrals provided by Code section 402(g) in effect at the beginning of the taxable year (as adjusted for cost-of-living under Code section 402(g)(4)). In the case of a Participant age 50 or over by the end of the taxable year, Elective Deferrals that do not exceed the dollar limitation described in the preceding sentence may also include any Elective Deferrals that are treated as Catch-up Contributions (described in Subsection (c) below) because of another limitation.

Subsequent to reemployment after a period of Qualified Military Service, a Participant is entitled to make a contribution to make up Elective Deferrals they otherwise could have made if not for the period of Qualified Military Service. No such payment may exceed the amount the Participant would have been permitted or required to contribute had the Participant remained continuously employed by the Employer throughout the period of Qualified Military Service. Any payment to the Plan will be made during the period beginning with the date of reemployment and whose duration is three times the period of the Qualified Military Service, but not greater than five years.

Nevertheless, the Employer may impose reasonable limitations in a uniform, nondiscriminatory manner on the amounts contributed in order to satisfy applicable legal requirements and to ensure the deductibility of amounts contributed by the Employer to the Plan and any other qualified plan of deferred compensation. The Employer may elect in the Adoption Agreement to limit Elective Deferrals made on behalf of electing Employees to a percentage of Compensation or dollar amount that is less than the amount otherwise allowable by Code section 402(g). Where the Plan permits Elective Deferral and Catch-Up Contributions, the limit established by the Plan may not be a percentage that is less than 75 percent.

(b) Catch-up Contributions. If elected in the Adoption Agreement, a Participant may make Catch-up Contributions, which are Elective Deferrals made to the Plan that exceed an otherwise applicable Plan limit and that are made by Participants who are age 50 or over by the end of the applicable taxable year. An otherwise applicable Plan limit is a limit in the Plan that applies to Elective Deferrals without regard to Catch-up Contributions, such as the limits on Annual Additions (see Section 3.2.1), the dollar limitation on Elective Deferrals under Code section 402(g) (not counting Catch-up Contributions), and the limit imposed by the ADP test under Code section 401(k)(3) (see Section 2.7.1), or any Employer-provided limit on the amount of Elective Deferrals. Catch-up Contributions for a Participant for a

taxable year may not exceed the lesser of:

- (1) The dollar limit on Catch-up Contributions under Code section 414(v))(2)(B)(i) for the taxable year (as adjusted for cost-of-living under Code section 414(v)(2)(c)); or
- (2) When added to the Participant's other Elective Deferrals, 100% of the Participant's Compensation for the taxable year.

All Catch-up eligible Participants will be provided with an effective opportunity to make the same dollar amount of Catch-up Contributions.

Catch-up Contributions are not subject to the limits on Annual Additions; are not counted in the ADP test; are not counted in determining the Minimum Top-Heavy Allocation (see Subsection 2.6.1(a)); and are not subject to any Employer-provided limit on the amount of Elective Deferrals. However, Catch-up Contributions made in prior years are counted in determining whether the Plan is Top-Heavy. (See Sections 2.6.1 and 2.6.2.)

- (c) Election. The Plan Administrator will determine the manner in which a Participant may elect to have Elective Deferrals made to the Plan on their behalf. The Plan Administrator will establish reasonable periods during which the election may be made or modified and must allow at least one opportunity per Plan Year for a Participant to make or modify an election. Unless the Plan Administrator establishes another period during which the election may be made or modified, any such election may be made or modified during the 30-day period preceding the first day of the Plan Year. A Participant may not make a retroactive election, and once an election is made, it will remain in effect until modified or terminated. A Participant may revoke an election at any time. Except in the case of an in-plan Roth conversion pursuant to Section 2.5.12(f) (a rollover to a Participant's Roth Elective Deferral account from another Account of the Participant in this Plan), Elective Deferrals contributed to the Plan as one type, either Roth Deferrals or Pre-tax Elective Deferrals, may not later be reclassified as the other type.
- (d) Payment of Contribution. Elective Deferrals will be remitted by the Employer to the Trustee or custodian on the earliest date that they can reasonably be segregated from the Employer's assets, but in no event later than the 15th business day of the month following the month in which the Participant contributions are withheld or received by the Employer, unless under the regulations an extension of up to 10 business days is granted by the Secretary of Labor with respect to Elective Deferrals received or withheld in a single month.

In the case of a small plan (less than 100) participants the Department of Labor has provided a safe harbor option for depositing Participant contributions within 7 business days after the money is received or withheld from payroll. This seven day rule was finalized as of January 14, 2010.

(e) Deferral Limitation After Hardship Distributions. For Hardship distributions occurring before January 1, 2002, a Participant may not have Elective Deferrals made on their behalf for the taxable year following the taxable year of a Hardship distribution in excess of the applicable limit under Code section 402(g) for such taxable year less the amount of the Employee's Elective Deferrals for the taxable year of the Hardship distribution.

2.2.3 Automatic Compensation Reduction (ACA), Eligible Automatic Contribution Arrangements (EACA) and Qualified Automatic Contribution Arrangements QACA).

(a) ACA. If elected in the Adoption Agreement, an Employee who becomes eligible to participate in the Plan and does not affirmatively elect to receive cash or have a specified amount contributed to the Plan will have their Compensation automatically reduced by the percentage elected in the Adoption Agreement. This percentage will be contributed to the Plan as a Pre-tax Elective Deferral. A Participant can make an affirmative election to make no Pre-Tax Elective Deferrals or to defer a different percentage of Compensation at any time. The ACA is effective for any period in which an Employee has not made an affirmative election or until superseded by a subsequent election that is made in writing and filed with the Employer. Under the ACA the Employer will automatically reduce the Employee's Compensation within a reasonable time after giving notice of the arrangement and providing the Employee with an effective opportunity to make an affirmative election.

The Employee must receive a notice that explains the ACA and the Employee's right to elect to have no such ACA made or to alter the amount of those reductions, including the procedure for exercising that right and the timing for implementation of any such election. The Employee must be notified annually of their ACA and the Employee's right to change that percentage. Each notice period must provide the Employee with a reasonable period to exercise their right to make an affirmative election.

If elected in the Adoption Agreement, the Employer may elect to apply the ACA to:

- a current Participant who has an Elective Deferral election in effect for less than the ACA percentage elected in the Adoption Agreement,
- (ii) a current Participant who has no Elective Deferral election in effect,
- (iii) to all Participants hired after a specified date, or
- (iv) to Participants who do not make an Elective Deferral election within the reasonable period specified in the Plan.,

If that Participant thereafter makes an affirmative election to reduce their compensation by another percentage (or by zero percentage), then that affirmative election will be effective for pay periods beginning in the month following the date the Participant files the election with the Plan Administrator.

A Participant that is subject to an Automatic Compensation Reduction (ACA) will have the amount of the ACA increased by the rate elected in the Adoption Agreement, unless elected otherwise by the Participant, beginning the first day of the calendar year following the calendar year in which an ACA first became effective. The total increase of the ACA will be limited as elected in the Adoption Agreement.

(b) EACA. If elected in the Adoption Agreement, the Plan may satisfy the requirement for an Eligible Automatic Contribution Arrangement using an Automatic Compensation Reduction (ACA) arrangement under section 1.401(k)-(1(a) (2) and the uniform default percentage under section 1.414(w)-1 of the Treasury Regulations.

If that Participant thereafter makes an affirmative election to reduce their compensation by another percentage (or by zero percentage), then that affirmative election will be effective for pay periods beginning in the month following the date the Participant files the election with the Plan Administrator.

- (1) **Definition.** For purposes of this Section 2.2.3(b), an "Eligible Automatic Contribution Arrangement" is an ACA arrangement under a cash or deferred profit sharing plan in which:
 - (i) a Covered Employee may elect to have the Employer defer a portion of their Compensation as a Pre-tax Elective Deferral under the Plan or receive such amounts directly in cash;
 - (ii) a Covered Employee who fails to make an affirmative election is treated as having elected to have the Employer make such contributions in an amount equal to a uniform percentage of Compensation;
 - (iii) the Employer has identified the group of Employees who will be covered under the EACA and to whom the default percentage will apply; and
 - (iv) the Employer meets the Annual Notice Requirements of Section 2.2.3(d).

(2) Automatic Default Elective Contribution. Default elective contributions will automatically be made for a Covered Employee who does not have an affirmative election in effect on or before the date the default election applies. The default election will be effective no earlier than a reasonable period after receipt of the notice described in Section 2.2.3(d).

For the purpose of this Section, a default elective contribution will be a Pre-tax Elective Deferral. The amount of the default elective contribution made for a Covered Employee in each pay period will be based on a uniform default percentage elected in the Adoption Agreement, multiplied by the Covered Employee's Compensation for that pay period.

If elected in the Adoption Agreement, the default percentage applied to a Covered Employee's Compensation will increase by a designated percentage each Plan Year, beginning with the second Plan Year that begins after the Plan Year in which the initial default percentage first applies to that Covered Employee. Unless otherwise elected in the Adoption Agreement, the increase will be effective on the first day of the Plan Year.

- (3) Affirmative Election. To avoid having default elective contributions deducted from their Compensation, a Covered Employee must make a written affirmative election. Unless otherwise elected in the Adoption Agreement, each year a Covered Employee will be required to make an affirmative election to have a higher or lower percentage (or zero percentage) applied to their Compensation, otherwise the default election will automatically apply at the default percentage elected in the Adoption Agreement.
- (4) Limitations. To meet the limitations under Code sections 401(a)(17), 402(g) and 415(c) and to satisfy any suspension periods required after a Hardship distribution, the Employer may stop or reduce an Employee's contributions to the Plan.
- (5) Covered Employees. A Covered Employee is an Employee who meets the eligibility requirements of the Plan and who is identified in the Adoption Agreement as a covered under the EACA.
 - (A) All Employees who are eligible to make Pre-tax Elective Deferrals to the Plan will be covered by the EACA and designated as Covered Employees, unless in the Adoption Agreement, the Employer identifies a specific group of Employees or classification of Employees to whom the default percentage will apply.
 - (B) A Covered Employee will have a reasonable opportunity after receipt of the notice described in Section 2.2.3(d) of the Plan to make an affirmative election regarding Pre-tax Elective Deferrals (either to have no Pre-tax Elective Deferrals made to the Plan or to have a different amount of Elective Deferrals made on their behalf) before the default percentage in Section 2.2.3(b)(2) above is applied.
- (6) Unless otherwise elected by the Employer in the Adoption Agreement, when an Employee makes an affirmative election to make Elective Deferrals to the Plan at a percentage that is higher or lower than the default percent, that Employee will cease to be a Covered Employee for the Plan Year and for the purpose of the default percentage in Section 2.2.3(b)(2) and the Annual Notice provisions in Section 2.2.3(d).
- (7) Permissible Withdrawal. The Covered Employee may request a Permissible Withdrawal from the Plan, no later than 90 days after the date on which the default elective contributions were first withheld from their Compensation. A "Permissible Withdrawal" is a distribution from the Plan of the default elective contribution made under an EACA that:
 - (A) is made pursuant to a written election by a Covered Employee;
 - **(B)** consists of the entire amount of contributions described in Section 2.2.3(b)(2) plus earnings attributable thereto through the date of the distribution; and

(C) is includible in the Covered Employee's gross income for the taxable year in which the distribution is made.

The Plan may charge the Covered Employee a withdrawal fee in an amount and manner similar to the fee charged for processing a cash distribution from their account, according to a written expense policy, adopted by the Plan.

Notwithstanding the distribution provisions described in the Plan, spousal consent is not required for this distribution.

- (8) Time for Making a Permissible Withdrawal Election. The provision of Subsection 2.2.3(b)(7) will not apply to a Covered Employee's withdrawal election, unless the election is made no later than 90 days after the date of their first default elective contribution. In the Adoption Agreement, the Employer may specify an earlier date for making this election, provided the election period is at least 30 days.
- (9) Distribution Amount. The provision of Subsection 2.2.3(b)(7) will not apply to any election by a Covered Employee, unless the amount of the Permissible Withdrawal is limited to the amount of the default elective contributions made through the earlier of the second payroll period to which the default percentage first applied or the first payroll period that occurs 30 days before the effective date of the election. The amount of the distribution will be adjusted for earnings, investment gain or losses.
- (10) Special Rules. The amount distributed as a Permissible Withdrawal will not be counted in the annual limitation on Elective Deferrals under Code section 402(g) for the calendar year in which the automatic contributions were contributed.
- (11) Special Rules for Timing of Excess Contribution and Excess Aggregate Contribution. For Plan Years beginning on or after January 1, 2010, where all eligible Non-Highly Compensated Employees and all eligible Highly Compensated Employees are designated as Covered Employees under EACA for the entire Plan Year, the Plan Administrator must distribute Excess Contributions and Excess Aggregate Contributions from the Plan no later than 6 months after the last day of the Plan Year in which the excess amounts were determined.
- (12) Tax Treatment of Excess Contributions. For Plan Years beginning after December 31, 2008, a corrective distribution of Excess Contributions (adjusted for any income or losses) will be included in an Employee's gross income in the tax year in which the distribution occurs.
- (13) Forfeit of Employer Match under an Eligible Automatic Contribution Arrangement. For Plan Years beginning after January 1, 2010, Employer Matching Contributions that are made on amounts distributed from the Plan as Permissible Withdrawals will not be forfeited unless the matching contribution has been allocated to the Employee's account.
- (c) QACA If elected in the Adoption Agreement, the Plan will satisfy the nondiscrimination requirements under Code section 401(k), by including the Safe Harbor contributions provision for a Qualified Automatic Contribution Arrangement and the Automatic Compensation Reduction (ACA) arrangement under sections 1.401(k)-3(j) and 1.401(k)-1(a)(2) of the Treasury Regulations.
 - (1) **Definition.** For purposes of this Section, a "Qualified Automatic Contribution Arrangement" is an ACA arrangement under which:
 - (i) an Employee may elect to have the Employer defer a portion of their Compensation as a Pre-tax Elective Deferral under the Plan or to receive such amounts directly in cash;
 - (ii) an Employee who fails to make an affirmative election is treated as having elected to have the Employer make such contributions as Default Elective Contributions based on a predefined percentage of Compensation (default percentage);

- (iii) the Employer will make either a Safe Harbor Matching Contribution or a Safe Harbor Non-Elective Contribution as described in Section 2.2.3(c)(7); and
- (iv) the Employer meets the Annual Notice Requirements of Section 2.2.3(d).
- (2) **Default Elective Contribution.** Default Elective Contributions will automatically be made for an Employee who does not have an affirmative election in effect, on or before the date a default election applies. The default election will be effective no earlier than a reasonable period after receipt of the notice described in Section 2.2.3(d).

For the purpose of this Section 2.2.3, Default Elective Contributions will be Pre-tax Elective Deferrals. The amount of the Default Elective Contribution made for an Eligible Employee, in each pay period, will be based on the Minimum Default Percentage specified in Subsection 2.2.3(c)(6) and elected in the Adoption Agreement, multiplied by the Employee's Compensation for that pay period.

- (3) Affirmative Election. To avoid having Default Elective Contributions deducted from their Compensation, an Eligible Employee must make a written affirmative election. Each year the Eligible Employee will be required to make an affirmative election to have a higher or lower percentage (or zero percentage) applied to their Compensation otherwise the default election will automatically apply at the default percentage described in Subsection 2.2.3(c)(6) and elected in the Adoption Agreement.
- (4) Limitations. To meet the limitations under Code sections 401(a)(17), 402(g) and 415(c) and to satisfy any suspension periods required after a hardship distribution, the Employer may stop or reduce an Employee's contributions to the Plan.
- (5) Eligible Employee: All Employees eligible to make Elective Deferrals to the Plan, will be covered by the QACA, unless otherwise specified in the Adoption Agreement.
 - (i) **Permissible Exclusion.** The Employer may elect, in the Adoption Agreement, to exclude current Participants who were eligible immediately before the effective date of these QACA provisions and had an affirmative election on file regarding Elective Deferrals.
 - (ii) Statutory Exclusion. The Employer may elect in the Adoption Agreement to exclude Employees who have not met the statutory eligibility requirements, as defined in Code section 410(b). This election will permit the Plan to exclude employees who have not yet reached age 21 and completed a Year of Service since their original Employment Commencement Date.
- (6) Provisions for the Default Elective Contribution. The Employer will specify in the Adoption Agreement how the Default Elective Contribution will be determined for each Eligible Employee based on a minimum default percentage that will automatically increase after the initial period. Unless otherwise elected in the Adoption Agreement, the default percentage will be uniformly applied to an Employee's Compensation under the ACA arrangement for the entire Plan Year and will automatically increase for Plan Years after the initial period.
 - (i) Minimum Default Percentage. The Plan must have a Minimum Default Percentage that is at least 3% of an Employee's Compensation beginning with the initial period. The Minimum Default Percentage must be applied uniformly to all Eligible Employees, under a default election.

The Minimum Default Percentage will automatically increase for Plan Years beginning after the initial default election ("initial period"). Under the Automatic Escalator Method, the Minimum Default Percentage will be at least:

(A) 3% during the initial period, which begins on the date the first Default Elective Contributions are made with respect to such Employee;

- (B) 4% during the period ending on the last day of the first Plan Year which begins after the date on which the first Default Elective Contribution described in Subparagraph (i) is made with respect to such Employee;
- (C) 5% during the second Plan Year following the Plan Year described in Subparagraph (A); and
- (D) 6% during any subsequent Plan Year.

Notwithstanding the preceding, an alternative to the Automatic Escalator Method may be elected in the Adoption Agreement:

- **(E)** the Plan may elect to impose a higher percentage for the initial period and each Plan Year thereafter; or
- **(F)** the Employer may elect in the Adoption Agreement to impose a uniform, flat percentage without annual increases, provided the Minimum Default Percentage is at least 6% or not more than the QACA rate maximum without annual increase.
- (ii) Maximum Default Percentage. The Maximum Default Percentage cannot exceed 10%. If elected in the Adoption Agreement, the Employer may specify a lower maximum for the Plan.
- (iii) Initial Period. For the purpose of applying a Minimum Default Percentage, an initial period begins when the default election for this QACA provision is first applied to an Eligible Employee and ends on the last day of the following Plan Year. The timing for a default election will be determined under the Annual Notice provisions described in Section 2.2.3(d) of the Plan.

If elected in the Adoption Agreement, the Plan may elect to:

- (A) have a new initial period begin for an Employee who did not have any contributions made to the Plan during the entire Plan Year. and
- **(B)** have the default percentage increase after the initial period on a date other than the first day of a Plan Year.
- (7) Safe Harbor Contributions. The Employer is required to make Matching Contributions or Non-Elective Contributions. In the Adoption Agreement, the Employer will select and contribute to the Plan one of the following types of Safe Harbor Contributions for each Plan Year for which the QACA is effective:
 - (i) Matching Contribution. A Matching Contribution for each Non-Highly Compensated Employee in an amount equal to the sum of 100% of their Elective Deferrals, to the extent that such Elective Deferrals do not exceed 1% of their Compensation plus an additional Matching Contribution equal to 50% of the Eligible Employee's Elective Deferral that exceed 1% but does not exceed 6% of their Compensation; or
 - (ii) Non-Elective Contribution. A Non-Elective Contribution for each Eligible Employee in an amount equal to at least 3% of the Employee's Compensation; regardless as to whether or not the Employee elects to make Elective Deferral or Employee Voluntary Contributions to the Plan.

Notwithstanding the preceding, if elected in the Adoption Agreement, the Employer may elect to contribute a Matching Contribution or Non-Elective Contribution in an amount equal to or greater than the percentage allowed herein, provided it satisfies the requirement of sections 1.401(k)-3(b) or 1.401(k)-3(c) of the Treasury Regulations.

- (iii) **Special Requirements.** The Safe Harbor Matching Contributions or the Safe Harbor Non-Elective Contributions made under this Section 2.2.3(c) will be subject to the following distribution restrictions and vesting requirements:
 - (A) Safe Harbor Contributions may not be withdrawn from the Plan before age 59-1/2, under the withdrawal restrictions described in section 1.401(k)-1(d) of the Treasury Regulations; and
 - (B) Safe Harbor Contributions must be 100% vested after an Employee has completed 2 Years of Service.

(d) Annual Notice Requirement. QACA and EACA

- (1) The Plan Administrator will give each Employee to whom the Qualified Automatic Contribution Arrangement in Section 2.2.3(c) or Eligible Automatic Contribution Arrangement in Section 2.2.3(b) applies, a notice explaining the Employee's rights and obligations under the arrangement. The notice must be provided within a reasonable period before the beginning of each Plan Year and it must be:
 - sufficiently accurate and comprehensive to apprise the Employee of such rights and obligations and
 - (ii) written in a manner easily understood by the average Employee to whom the arrangement applies.
 - (iii) In the case of an EACA, the Employer will also include within the notice the Covered Employee's right to make a Permissible Withdrawal pursuant to Subsection 2.2.3(a)(7) and the administrative procedures for making such an election.
- (2) Timing and Form of Notice. A notice will not be treated as meeting the requirements of this Section with respect to an Eligible Employee unless:
 - the notice includes the level of the Default Elective Contribution that will be made for the affected Employee if they do not make an affirmative election;
 - (ii) the notice includes an explanation of the Employee's right to elect not to have Elective Deferral Contributions made on the Employee's behalf or to elect to have such contributions made at a different percentage;
 - (iii) the notice is given before the beginning of each Plan Year and gives the Employee a reasonable period after receipt of the notice to make an affirmative election.
 - (iv) the notice explains how Elective Deferrals and Safe Harbor or Employer Contributions made under the arrangement will be invested in the absence of any investment election by the Employee or where the Plan provides for two or more investment options.
- (3) Reasonable Period. A notice satisfies the timing requirements of this Section only if it is provided sufficiently early so that the Employee has a reasonable period of time, after receipt of the notice, to make the elections described in Sections 2.2.3(b)(3) and 2.2.3(c)(3). The timing of the notice is considered reasonable if it is provided no later than the first pay date in the second payroll period that begins after the date the notice was provided or the first pay date that occurs at least 30 days after the notice is given to a newly hired employee. Notwithstanding, the Plan cannot make the default election effective any later than the earlier of:
 - (i) the pay date for the second payroll period that begins after the date the notice is provided; and

- (ii) the first pay date that occurs at least 30 days after the notice is provided.
- (iii) in the case of an EACA the Plan Administrator will include within the notice the Covered Employee's right to make a Permissible Withdrawal pursuant to Subsections 2.2.3(b)(3) and (7) and the administrative procedures for making such an election.

(e) Expiration of Affirmative Elections

Under a 401(k) plan, the plan may provide that an affirmative election expires annually. If a Participant fails to complete a new affirmative election subsequent to their prior election expiring, the Participant becomes subject to the default deferral percentage as outlined in the plan pursuant to the automatic contribution provision. Each year, the Participant can always complete a new affirmative election and designate a new deferral percentage.

2.2.4 Employee Contributions.

(a) Amount of Contribution. Subject to the Employer's election in the Adoption Agreement, an Employee may make Voluntary Employee Contributions to the Plan on an after-tax basis. Voluntary Employee Contributions for Plan Years beginning after 1986 must meet the nondiscriminatory test of Code section 401(m). Such contributions will be maintained in a Voluntary Account and will be nonforfeitable at all times. The Voluntary Account will share in the gains and losses of the Trust according to the policy adopted pursuant to Section 3.1.3.

The Plan Administrator will not accept Deductible Voluntary Employee Contributions that are made for a taxable year beginning after December 31, 1986. Deductible Voluntary Employee contributions made prior to that date will be maintained in a Deductible Voluntary Account that will be nonforfeitable at all times. This Account will share in the gains and losses of the Trust according to the policy adopted pursuant to Section 3.1.3. No part of the Deductible Voluntary Account will be used to purchase life insurance.

- (b) Withdrawal of Contributions. Subject to the requirements of Section 2.5.5 regarding Qualified Joint and Survivor Annuities (if applicable), the Participant may withdraw any part of the Voluntary Account or Deductible Voluntary Account by making a written application to the Plan Administrator. The Plan Administrator may adopt such procedures with respect to such withdrawals as may be necessary or appropriate. At the Plan Administrator's direction, the Trustee will distribute any such withdrawal to the Participant in accordance with those procedures. Except in the case of the Deductible Voluntary Account, such withdrawals will not include any interest or other increment earned on such contributions if the Participant is still an Employee. No Forfeitures will occur as a result of an Employee withdrawing such contributions. Notwithstanding the foregoing, the Employee's spouse must consent in writing to a withdrawal of Voluntary Employee Contributions subject to the spousal consent requirements of Subsection 2.5.5(c).
- **2.2.5 Return of Contributions.** The Trustee will return contributions by the Employer, including Employer, Qualified Non-Elective, Non-Elective, Matching, and Qualified Matching Contributions, to the Employer in the following instances:
- (a) If a contribution is made by the Employer because of mistake of fact, then the Trustee will return such contribution within 1 year after its payment upon the Employer's written request.
- (b) Each contribution by the Employer is conditioned on initial qualification of the Plan under the applicable sections of the Code. If the Commissioner of Internal Revenue determines that the Plan does not initially qualify, then any contribution made incident to the initial qualification by the Employer will be returned within 1 year after the date of denial of initial qualification of the Plan, but only if the application for initial qualification is made by the time prescribed by law for filing the Employer's tax return for the taxable year in which the Plan is adopted, or such later date as the Secretary of the Treasury may prescribe.

- (c) Each contribution by the Employer is conditioned upon the deductibility of the contribution under the applicable sections of the Code. If the deduction for part or all of a contribution is disallowed, the Trustee will return such contribution, to the extent of the disallowance, within 1 year after such disallowance.
- **2.2.6 Required Diversification For Plans That Hold Publicly Traded Securities.** This Plan will provide Applicable Individuals with the right to divest Employer Securities in their Accounts and reinvest those amounts in certain diversified investments.
- (a) In the case of the portion of an Applicable Individual's Account attributable to Voluntary Employee Contributions and Elective Deferrals invested in Employer Securities, the Plan will provide the Applicable Individual the right to direct the Plan to divest any such securities and to reinvest an equivalent amount in other investment options meeting the requirements of Paragraph (c).
 - (1) An Applicable Individual with respect to Elective Deferrals and Voluntary Employee Contributions is:
 - (A) a Participant in the Plan;
 - (B) an Alternate Payee who has an account under the Plan; or
 - (C) a Beneficiary of a deceased Participant
- (b) In the case of the portion of the Account attributable to Employer Contributions other than Elective Deferrals invested in Employer Securities, an Applicable Individual may elect to direct the plan to divest any such securities and to reinvest an equivalent amount in other investment options meeting the requirements of Paragraph (c).
 - (2) An Applicable Individual with respect to the Employer Contributions is:
 - (A) a Participant who has completed at least 3 years of service;
 - (B) an Alternate Payee who has an Account under the Plan with respect to a Participant who has completed at least 3 Years of Service; or
 - (C) a Beneficiary of a deceased Participant
- (c) Investment options.
 - (1) The Plan will offer not less than 3 investment options, other than Employer Securities, to which an Applicable Individual may direct the proceeds from the divestment of Employer Securities. Each investment option must be diversified and have materially different risk and return characteristics.
 - (2) The Plan will permit an election to divest or reinvest at least quarterly.
 - (3) The Plan will not meet the requirements of this Section if the Plan imposes restrictions or conditions with respect to the investment of Employer Securities that are not imposed on the investment of other Plan assets. Notwithstanding, any restrictions or conditions imposed by under applicable securities laws will apply.
- (d) Applicable Defined Contribution Plan. For purposes of this Section, the term "Applicable Defined Contribution Plan" means:
 - (1) any defined contribution plan which holds any Publicly Traded Employer Securities and
 - (2) does not include a one-participant retirement plan.

- (A) The term "one-participant retirement plan" means a retirement plan that on the first day of the Plan Year:
 - (1) covers only one individual (or the individual and the individual's spouse) and the individual (or the individual and the individual's spouse) owned 100% of the plan sponsor (whether or not incorporated), or
 - (2) covers only one or more partners (or partners and their spouses) in the plan sponsor.
- **(B)** The term "partner" includes a 2% shareholder (as defined in Code section 1372(b)) of an S Corporation.
- (e) Certain plans treated as holding Publicly Traded Employer Securities.
 - (1) In the case of Employer Securities that are not Publicly Traded Employer Securities, the Plan will be treated as holding Publicly Traded Employer Securities if any Employer corporation, or any member of a Controlled Group of Corporations, which includes such Employer corporation, has issued a class of stock that is a Publicly Traded Employer Security.
 - (2) Exception for certain controlled groups with publicly traded securities, Clause (1) will not apply to the Plan if:
 - (A) no Employer Corporation or Parent Corporation of an Employer Corporation has issued any publicly traded Employer Security, and
 - (B) no Employer Corporation or Parent Corporation of an Employer Corporation has issued any special class of stock that grants particular rights to, or bears particular risks for the holder or issuer with respect to any corporation described in Subparagraph (i) that has issued any Publicly Traded Employer Security.
 - (3) **Definitions.** For purposes of this Subparagraph:
 - (A) "Controlled Group of Corporations" has the meaning given such term by Code section 1563(a), except that "50 percent" will be substituted for "80 percent" each place it appears;
 - **(B)** "Employer Corporation" means a corporation which is an Employer maintaining the Plan; and
 - (C) "Parent Corporation" has the meaning given such term by Code section 424(e).
- (f) Other definitions. For purposes of this Section:
 - (1) "Elective Deferrals" means an Employer Contribution as described in Code section 402(g)(3)(A).
 - (2) "Employer Security" has the meaning given by Code section 407(d)(1) of the Employee Retirement Income Security Act of 1974.
 - (3) "Employee Stock Ownership Plan" has the meaning given by Code section 4975(e)(7).
 - (4) "Publicly Traded Employer Securities" means Employer Securities that are readily tradable on an established securities market.
 - (5) "Year of Service" has the meaning given by Code section 411(a)(5).
- (g) Transition rule for securities attributable to Employer Contributions

- (1) Rules phased in over 3 years.
 - (A) In the case of the portion of an Employer Account other than Elective Deferral Account consisting of Employer Securities acquired in a Plan Year beginning before January 1, 2007, Section (b) will only apply to the Applicable Percentage of such securities. This subparagraph applies separately with respect to each class of securities.
 - (B) Exception for certain Participants aged 55 or over. The Transition Rules of Subparagraph (1) will not apply to an Applicable Individual who is a Participant who has attained age 55 and completed at least 3 Years of Service before the first Plan Year beginning after December 31, 2005.
- (2) Applicable Percentage. For purposes of Subparagraph (1), the Applicable Percentage will be determined as follows:

Plan year to which Section (b) applies	The Applicable Percentage is
First Year	33
Second Year	66
Third and following Years	100

(h) Effective Date. The Diversification requirements of Section 2.2.6 will apply to the Plan and are effective with respect to Plan Years beginning after December 31, 2006, subject to certain special effective date rules, including a special rule with respect to plans maintained pursuant to a collective bargaining agreement.

ARTICLE III

ALLOCATIONS

- **2.3.1 Basic Allocation.** Unless otherwise elected by the Employer in the Adoption Agreement, as of the last day of the Plan Year, the contribution made by the Employer, including any Forfeitures to be allocated, with respect to the Plan Year will be allocated among the Employer Accounts of Participants during the Plan Year, in the manner set forth in the Adoption Agreement; provided, that the Employer Contribution must satisfy the requirements of Code section 416 regardless of how the Adoption Agreement is completed.
- **2.3.2 Profit Sharing, Money Purchase Pension Plans.** Unless otherwise elected by the Employer in the Adoption Agreement, as of the last day of the Plan Year, the Employer Contributions made by the Employer with respect to the Plan Year and Forfeitures will be allocated among the Employer Accounts of Participants during the Plan Year in the manner set forth in the Adoption Agreement. A Participant is treated as benefiting under the Plan for any Plan Year during which the Participant received or is deemed to receive an allocation in accordance with Treasury Regulations section 1.410(b)-3(a). If a Plan is integrated with Social Security, Section 2.3.5 or 2.3.6, as the case may be, will also apply.
- (a) As elected by the Employer in the Adoption Agreement, the Employer will determine the total amount of contributions for each Plan Year and either (1) allocate such total amount to Participant groups (the "Participant Group Allocation method"), or (2) allocate such total amount using age weighted allocation rates (the "Age Weighted Allocation method"). Employer Contributions will be allocated to each Eligible Employee.
 - (1) Participant Group Allocation Method. The Employer may elect in the Adoption Agreement to use one of two Participant Group Allocation methods for the purpose of allocating the Employer Contribution: (a) the "employee classification," method that permits the Employer to divide the Eligible Employees into groups under pre-defined classifications or (b) the "separate allocation group" method where each Eligible Employee comprises a separate allocation group for the purpose of allocating the Employer Contribution.

The allocation will be made as follows: First, the total amount of contributions is allocated among the deemed aggregated allocation groups in portions determined by the Employer. A deemed aggregated allocation group consists of all of the separate allocation groups that have the same allocation rate. Second, within each deemed aggregated allocation group, the allocated portion is allocated to each Employee in the ratio that such Employee's Compensation, as defined in Part 1 Article II of the Plan, bears to the total Compensation of all Employees in the group. An allocation rate is the amount of contributions allocated to an Employee for a year, expressed as a percentage of compensation or dollar amount, as defined in Part 1 Article II of the Plan. If the Employer has elected in the Adoption Agreement, the "separate allocation groups" method, each Eligible Employee of the Employer will constitute a "separate allocation group" for purposes of allocating contributions. In the event that an Eligible Employee is included in more than one Participant allocation group, the Participant's share of the Employer Contribution allocated to each such group will be based on the Participant's Compensation for the part of the year the Participant was in the group.

The Employer will specify in written instructions to the Plan Administrator or Trustee, by no later than the due date of the Employer's tax return for the year to which the Employer Contribution relates, including extensions, the portion of such contribution to be allocated to each participant allocation group.

(2) Age Weighted Allocation Method. If the Age Weighted Allocation method is elected in the Adoption Agreement, the total Employer Contribution will be allocated to each Eligible Employee such that the equivalent benefit accrual rate for each Participant is identical. The equivalent benefit accrual rate is the annual annuity commencing at the Participant's testing age, expressed

as a percentage of the Participant's Compensation as defined in Part 1 Article II of the Plan which is provided from the allocation of Employer Contributions and Forfeitures for the Plan Year, using standardized actuarial assumptions that satisfy section 1.401(a)(4)-12 of the Income Tax Regulations. The Employee's testing age is the later of Normal Retirement Age, or the Employee's current age.

- (b) Cross Testing. The Employer may elect in the Adoption Agreement one of the following methods to meet the requirements of Treasury Regulation section 1.401(a)(4)-8.
 - (1) Minimum Allocation Gateway Method. For any Plan Year in which the Employer intends to rely on Treasury Regulation section 1.401(a)(4)-8 (cross testing) in satisfying Code section 401(a)(4). each Non-Highly Compensated Employee entitled to receive a Non-Elective Contribution under this Section 2.3.1 or under Section 2.9.3 will receive a minimum allocation equal to the lesser of (a) 1/3 of the highest allocation rate (as a percentage of Compensation) of any Highly Compensated Employee applied to their Compensation, as defined in Part I, Article II, or (b) 5% of their compensation as defined in Subsection 3.2.5(a)(iii). Compensation for a Participant's initial year of participation will be measured either over the period starting from their Entry Date or over the 12 month period ending in the initial year of participation, as elected in the Adoption Agreement under the Compensation Computation Period. If a defined benefit plan sponsored by the Employer is aggregated with this Plan for purposes of satisfying Code section 410(b) and the Employer intends to rely on Treasury Regulations section 1.401(a)(4)-9, then each Non-Highly Compensated Employee entitled to an allocation will receive an allocation such that their aggregate normal allocation rate equals the lesser of (a) 1/3 of the highest aggregate normal allocation rate (as a percentage of Compensation) of any Highly Compensated Employee or (b) 5% of their compensation as defined in Subsection 3.2.5(a)(iii); provided, if the highest aggregate normal allocation rate of any Highly Compensated Employee exceeds 25%, each Non-Highly Compensated Employee will receive 6% of their Compensation, if the highest aggregate normal allocation rate of any Highly Compensated Employee exceeds 30%, each Non-Highly Compensated Employee will receive 7% of their Compensation, and if the highest aggregate normal allocation rate exceeds 35%, each Non-Highly Compensated Employee will receive 7.5% of their Compensation. Notwithstanding anything in the Plan to the contrary, on or after June 5, 2009, in the event that the Plan is Top-Heavy for a Plan Year, any Non-Highly Compensated Employee eligible for a Minimum Top-Heavy Allocation in Subsection 2.6.1(a) of the Plan will receive an allocation equal to the greater of the Minimum Top-Heavy Allocation or the amount determined above. Any allocation in excess to the Minimum Top-Heavy Allocation determined without regard to this provision must be made to this Plan and may not be made to another Plan in a Top-Heavy Allocation group.
 - (2) Gradual Age or Service Schedule. The Plan has a gradual age or service schedule for the Plan Year if the allocation formula for all Employees under the Plan provides for a single schedule of allocation rates under which; (1) The schedule defines a series of bands based solely on age, Years of Service, or the number of points representing the sum of age and Years of Service (age and service points), under which the same allocation rate applies to all Employees whose age, Years of Service, or age and service points are within each band; and (2) The allocation rates under the schedule increase smoothly at regular intervals, within the meaning of this paragraph and the following two paragraphs of this section.
 - (3) Smoothly Increasing Schedule Of Allocation Rates. A schedule of allocation rates increases smoothly if the allocation rate for each band within the schedule is greater than the allocation rate for the immediately preceding band (i.e., the band with the next lower number of years of age, Years of Service, or age and service points) but by no more than 5 percentage points. However, a schedule of allocation rates will not be treated as increasing smoothly if the ratio of the allocation rate for any band to the rate for the immediately preceding band is more than 2 or if it exceeds the ratio of allocation rates between the two immediately preceding bands.
 - (4) Regular Intervals. A schedule of allocation rates has regular intervals of age, Years of Service or age and service points, if each band, other than the band associated with the highest age, years of service, or age and service points, is the same length. For this purpose, if the schedule is

based on age, the first band is deemed to be of the same length as the other bands if it ends at or before age 25. If the first age band ends after age 25, then, in determining whether the length of the first band is the same as the length of other bands, the starting age for the first age band is permitted to be treated as age 25 or any age earlier than 25. For a schedule of allocation rates based on age and service points, the rules of the preceding two sentences are applied by substituting 25 age and service points for age 25. For a schedule of allocation rates based on service, the starting service for the first service band is permitted to be treated as one year of service or any lesser amount of service.

(5) Minimum Allocation Rates Permitted. A schedule of allocation rates under the Plan does not fail to increase smoothly at regular intervals, within the meaning of the preceding paragraphs of this Section, merely because a minimum uniform allocation rate is provided for all Employees or the minimum benefit described in section 416(c)(2) of the Code is provided for all Non-Key Employees (either because the Plan is Top-Heavy or without regard to whether the Plan is Top-Heavy) if the schedule satisfies one of the following conditions; (1) The allocation rates under the Plan that are greater than the minimum allocation rate can be included in a hypothetical schedule of allocation rates that increases smoothly at regular intervals, within the meaning of the preceding paragraphs of this section, where the hypothetical schedule has a lowest allocation rate no lower than 1% of Plan Year Compensation; or (2) For a Plan using a schedule of allocation rates based on age, for each age band in the schedule that provides an allocation rate greater than the minimum allocation rate, there could be an Employee in that age band with an equivalent accrual rate that is less than or equal to the equivalent accrual rate that would apply to an Employee whose age is the highest age for which the allocation rate equals the minimum allocation rate.

2.3.3 Cash or Deferred Profit Sharing Plans.

(a) Matching Contributions. Unless otherwise specified in the Adoption Agreement, as of the last day of the Plan Year, the Matching Contribution made by the Employer with respect to the Plan Year, and Forfeitures, will be allocated to the Matching Accounts of Participants for whom Elective Deferrals were made in the manner specified in the Adoption Agreement. If elected in the Adoption Agreement, Catch-up Contributions will not be treated as Elective Deferrals in the calculation of Matching Contributions allocated to Participant's Matching Accounts.

If a discretionary Matching Contribution formula applies (i.e., a formula that provides an Employer with discretion regarding how to allocate a Matching Contribution to Participants) and the Employer makes a discretionary Matching Contribution to the Plan, the Employer must provide the Plan Administrator (or Trustee, if applicable), written instructions describing (1) how the discretionary Matching Contribution formula will be allocated to Participants (e.g., a uniform percentage of Elective Deferrals or a flat dollar amount); (2) the computation period(s) to which the discretionary Matching Contribution formula applies; and (3) if applicable, a description of each business location or business classification subject to separate discretionary Matching Contribution allocation formulas. Such instructions must be provided no later than the date on which the discretionary Matching Contribution is made to the Plan. A summary of these instructions must be communicated to Participants who receive discretionary Matching Contributions. The summary must be communicated to Participants no later than 60 days following the date on which the discretionary Matching Contribution is made to the Plan. This written instruction requirement will become applicable the first day of the Plan Year following the Plan Year in which the Plan first executes a Cycle 3 (Post-PPA) pre-approve plan document.

- **(b) Pre-Tax Elective Deferrals.** The Pre-Tax Elective Deferrals by the Employer on behalf of an electing Employee will be allocated to the Pre-Tax Elective Account of such electing Employee.
- **(c) Roth Deferrals.** The Roth Deferrals contributed by the Employer on behalf of an electing Employee will be allocated to the Roth Deferral Account of such electing Employee.
- (d) Qualified Non-Elective Contributions and Qualified Matching Contributions. As of the last day of the Plan Year, the Qualified Non-Elective Contributions and Qualified Matching Contributions made by the Employer with respect to the Plan Year will be allocated to the Qualified Non-Elective Account or

Qualified Matching Account of Participants during the Plan Year in the manner specified in the Adoption Agreement. Contributions are treated as Qualified Non-Elective or Qualified Matching Contributions only if they meet the requirements of Treasury Regulations sections 1.401(k)-2(a)(6)(iv) or 1.401(m)-2(a)(6)(v), respectively

2.3.4 Minimum Top-Heavy Allocation. In the event the Plan becomes a Top-Heavy Plan during any Plan Year, the provisions of Subsection 2.6.1(a) will apply. The Employer may elect to make an additional allocation to meet the Minimum Top-Heavy requirement or otherwise follow the Plan provisions. The allocation of Employer Contributions must satisfy the requirements of Code section 416 regardless of how the Adoption Agreement is completed. Elective Deferrals and Matching Contributions allocated to Key Employees, but not Catch-up Contributions, are taken into account for the purpose of determining the minimum contribution under Code section 416. However, Elective Deferrals (and, for Plan Years beginning before 2002, Matching Contributions) made on behalf of Non-Key Employees may not be taken into account for the purpose of satisfying the minimum contribution requirement under Code section 416.

2.3.5 Integration with Social Security - Profit Sharing Plans.

(a) If the Employer has elected in the Adoption Agreement that the Plan will be integrated with Social Security, then the applicable contributions plus Forfeitures will be allocated to Participants' Accounts as follows (provided that Steps One and Two, below, need only be applied in years in which the Plan is Top-Heavy):

<u>STEP ONE:</u> Contributions and Forfeitures will be allocated to each Participant's Account in the ratio that each Participant's Compensation bears to all Participant's Compensation, but not in excess of 3% of each Participant's Compensation.

<u>STEP TWO:</u> Any contributions and Forfeitures remaining after the allocation in Step One will be allocated to each Participant's Account in the ratio that each Participant's Compensation for the Plan Year in excess of the Integration Level bears to the excess compensation of all Participants, but not in excess of 3%.

STEP THREE: Any contributions and Forfeitures remaining after the allocation in Step Two (or starting with this Step Three if so elected in the Adoption Agreement, all contributions and Forfeitures) will be allocated to each Participant's Account in the ratio that the sum of each Participant's Compensation and Compensation in excess of the Integration Level bears to the sum of all Participants' Compensation and Compensation in excess of the Integration Level, but not in excess of the Maximum Profit Sharing Disparity Rate.

<u>STEP FOUR:</u> Any remaining contributions and Forfeitures will be allocated to each Participant's account in the ratio that each Participant's Compensation for the Plan Year bears to all Participants' Compensation for that year.

The Maximum Profit Sharing Disparity Rate is equal to the lesser of:

- (1) 5.7% (minus the percentage of Compensation allocated in Step One, if any); or,
- (2) 5.4% (minus the percentage of Compensation allocated in Step One, if any) if the Integration Level (IL) is more than 80% but less than 100% of the Taxable Wage Base (TWB) under section 230 of the Social Security Act at the beginning of the Plan Year; or
- (3) 4.3% (minus the percentage of Compensation allocated in Step One, if any) if the IL is greater than 20% of the TWB, but not more than 80% of the TWB, and greater than \$10,000.
- (b) In the event the Plan is Top-Heavy and the Employer elects to use Steps Three and Four of Subsection (a) above, then allocations made due to Compensation greater than the Integration Level will be reduced proportionately to the extent necessary to provide all Employees entitled to a Top-Heavy Minimum Allocation with the required allocation.

- 2.3.6 Integration with Social Security Money Purchase Plans.
- (a) Annual overall permitted disparity limit. Notwithstanding the formula elected in the Adoption Agreement, for any Plan Year this Plan benefits any Participant who benefits under another qualified plan or simplified employee pension, as defined in Code section 408(k), maintained by the Employer that provides for permitted disparity (or imputes disparity), the Employer will contribute for each eligible Participant an amount equal to the Excess Contribution Percentage multiplied by the Participant's total Compensation. For purposes of this Subsection, the "Excess Contribution Percentage" is the percentage of Compensation at which Employer-derived contributions are made with respect to Compensation at or above the Integration Level in a defined contribution plan.
- (b) Cumulative permitted disparity limit. The cumulative permitted disparity limit for a Participant is 35 total cumulative permitted disparity years. Total cumulative permitted years means the number of years credited to the Participant for allocation or accrual purposes under this Plan or any other qualified plan or simplified employee pension plan (whether or not terminated) ever maintained by the Employer. For purposes of determining the Participant's cumulative permitted disparity limit, all years ending in the same calendar year are treated as the same year. If the Participant has not benefited under a defined benefit or target benefit plan for any year beginning on or after January 1, 1994, the Participant has no cumulative disparity limit.
- (c) The Maximum Money Purchase Disparity Rate is equal to the lesser of the Base Contribution Percentage or the applicable percentage determined in accordance with the table below.

If the Integration Level:

is more than	but not more than	the applicable percentage is:
\$0	X*	5.7%
X* of TWB	80% of TWB	4.3%
80% of TWB If TWB	Y**	5.4% 5.7%

^{*}X = the greater of \$10,000 or 20% of the TWB

If the Integration Level is equal to the Taxable Wage Base ("TWB"), the applicable percentage is 5.7%.

For purposes of this Subsection, the "Base Contribution Percentage" will be the percentage of Compensation at which Employer-derived contributions are made with respect to Compensation at or below the Integration Level in a defined contribution plan.

- (d) **Top-Heavy Years.** In the event the Plan becomes Top-Heavy, the Employer will make an additional contribution to the Plan to provide the Minimum Top-Heavy Allocation pursuant to Section 2.3.4.
- **2.3.7 Fail-Safe Allocation.** Unless elected otherwise in the Adoption Agreement, failure of both the coverage tests under section 410(b) of the Code requires correction using the principles set forth in this Section if such use is elected in the Adoption Agreement. Testing is done as of the last day of the Plan Year. Notwithstanding any provision of the Plan or Adoption Agreement to the contrary, if the Plan would otherwise fail to satisfy the requirements of Code section 410(b) and the regulations thereunder because

^{**}Y = any amount more than 80% of the TWB but less than 100% of the TWB.

Employer Contributions have not been allocated to a sufficient number or percentage of Participants for the Plan Year, and the right to receive a contribution is conditioned upon participation on the last day of the Plan Year or the completion of more than 500 Hours of Service, or conditioned upon membership in a certain Employee classification, an additional contribution will be made by the Employer and will be allocated to the Employer Accounts of affected Participants, considering all the applicable exclusions of Code sections 410(b)(3) and (4) subject to the following provisions:

(a) The NHCE Participants eligible to share in the allocation of the Employer's Contribution will be expanded to include the minimum number of NHCE Participants who are not otherwise eligible to the extent necessary to satisfy the ratio percentage test of Code section 410(b)(1)(B). The specific NHCE Participants who will become eligible are those NHCE Participants who are members of an eligible classification of Employees, who are actively employed on the last day of the Plan Year, but have not met the Plan's Hours of Service requirement.

These Participants will be given an allocation one Participant at a time beginning with the Participant with the greatest number of Hours of Service and continuing in descending order until the ratio percentage test is passed. If two or more Participants have the same number of Hours of Service then the Participant with the highest Compensation will receive an allocation first. Fail-safe allocations will be given to the smallest number of Participants that allows the Plan to pass the ratio percentage test.

(b) If the ratio percentage test is still not satisfied, the NHCE Participants eligible to share in the allocation will be further expanded to include the minimum number of NHCE Participants who are members of an eligible classification of Employees who are not employed on the last day of the Plan Year as is necessary to satisfy the ratio percentage test. The specific NHCE Participants who will become eligible are those NHCE Participants who have completed the greatest number of Hours of Service (in excess of 500 hours) during the Plan Year.

These Participants will be given an allocation one Participant at a time in descending order until the ratio percentage test is passed. If two or more Participants have been credited with the same number of Hours of Service then the Participant with the highest Compensation will receive an allocation first.

(c) If the ratio percentage test is still not satisfied, the NHCE Participants eligible to share in the allocation of the Employer Contribution will be expanded to include the minimum number of NHCE Participants who are not otherwise eligible to the extent necessary to satisfy the ratio percentage test of Code section 410(b)(1)(B). The specific NHCE Participants who will become eligible are those NHCE Participants who are members of an ineligible classification of Employees, who are actively employed on the last day of the Plan Year, and have met the Plan's Hours of Service requirement.

These Participants will be given an allocation one Participant at a time beginning with the Participant with the greatest number of Hours of Service and continuing in descending order until the ratio percentage test is passed. If two or more Participants have the same number of Hours of Service then the Participant with the highest Compensation will receive an allocation first. Fail-Safe allocations will be given to the smallest number of Participants that allows the Plan to pass the ratio percentage test.

- (d) If the ratio percentage test is still not satisfied, the NHCE Participants eligible to share in the allocation will be further expanded to include the minimum number of NHCE Participants who are members of an ineligible classification of NHCE Employees who are employed on the last day of the Plan Year but have not met the hours requirement of the Plan as is necessary to satisfy the ratio percentage test. The specific NHCE Participants who will become eligible are those NHCE Participants who have completed the greatest number of Hours of Service during the Plan Year. These Participants will be given an allocation one Participant at a time in descending order until the ratio percentage test is passed. If two or more Participants have been credited with the same number of Hours of Service then the Participant with the highest Compensation will receive an allocation first.
- (e) If the ratio percentage test is still not satisfied, the NHCE Participants eligible to share in the allocation will be further expanded to include the minimum number of NHCE Participants who are members of an

ineligible classification of NHCE Employees who are not employed on the last day of the Plan Year as is necessary to satisfy the ratio percentage test. The specific NHCE Participants who will become eligible are those NHCE Participants who have completed the greatest number of Hours of Service (in excess of 500 hours) during the Plan Year. These Participants will be given an allocation one Participant at a time in descending order until the ratio percentage test is passed. If two or more Participants have been credited with the same number of Hours of Service then the Participant with the highest Compensation will receive an allocation first.

- (f) The Employer may elect in the Adoption Agreement whether to apply this Section 2.3.7. If the Employer elects to apply this section 2.3.7 then for each Plan Year it is used, a list of each Participant benefiting on account of this Section and the associated percentage or dollar amount of Employer Contribution will be prepared for each Plan Year and provided to the Plan Administrator or Trustee not later than the time prescribed by law for filing the return for such applicable taxable year (including any extensions), and will be maintained as part of the administrative records of the Plan.
- **2.3.8 Contributions on Behalf of Disabled Participants.** If elected in the Adoption Agreement by the Employer, Employer Contributions will be made to the Plan on behalf of each Participant who has a Disability.

2.3.9 Treatment in the Case of Death or Disability Resulting from Active Military Service.

- (a) In the Adoption Agreement, the Employer may elect for benefit accrual purposes to treat an individual who dies or becomes disabled (as defined under the terms of the Plan) while performing Qualified Military Service, with respect to the Employer maintaining the Plan, as if the individual had resumed employment in accordance with the individual's reemployment rights under Code section 414(u) on the day preceding death or disability (as the case may be) and terminated employment on the actual date of death or disability. In the case of any such treatment, and subject to subparagraphs (b) and (c) of this Subparagraph, any full or partial compliance by the Plan with respect to the benefit accrual requirements of Code section 414(u)(8) with respect to such individual, will be treated for purposes of Code section 414(u)(1) as if such compliance were required under such section.
- **(b) Nondiscrimination Requirements.** The provisions of this Section 2.3.9 will apply only if all individuals performing Qualified Military Service with respect to the Employer maintaining the Plan (as determined under Code sections 401(b), (c), (m), and (o)) who die or became disabled as a result of performing Qualified Military Service prior to reemployment by the Employer are credited with service and benefits on reasonably equivalent terms.
- (c) Determination of Benefits. The amount of Voluntary Employee Contributions and the amount of Elective Deferrals of an individual treated as reemployed under this Section 2.3.9 for purposes of applying Code section 414(u)(8)(C) will be determined on the basis of the individual's average actual Voluntary Employee Contributions or Elective Deferrals for the lesser of:
 - (1) the 12-month period of service with the Employer immediately prior to Qualified Military Service; or
 - (2) the actual length of continuous service with the Employer, if service with the Employer is less than such 12-month period.

ARTICLE IV

VESTED BENEFITS

2.4.1 Vesting. A Participant will at all times be 100% vested and have a nonforfeitable interest in their Elective Deferral, Catch-up, Qualified Non-Elective, Qualified Matching, Prevailing Wage, Deductible Voluntary and Voluntary Accounts. In addition, ADP Test Safe Harbor Contributions made pursuant to Section 2.9.3 of the Plan will be 100% vested at all times. The vested and nonforfeitable interest of the Participant in their Controlled Account will be determined by reference to the Account from which the funds were originally transferred. The vested and nonforfeitable interest in a Participant's Employer and Matching Accounts will be determined subject to the vesting schedule selected in the Adoption Agreement and will not increase after the date the Participant terminates employment.

If the "6-year tired vesting schedule" is specified in the Adoption Agreement, the vested and nonforfeitable interest in a Participant's Account associated with the source will vest according to the following schedule:

Years of Vesting Service	Percentage
Less than 2	0%
2	20%
3	40%
4	60%
5	80%
6	or more 100%

For the purpose in determining the vested and nonforfeitable interest of a Participant, all of an Employee's Years of Vesting Service with the Employer or Related Employers maintaining the Plan will be applied to the vesting schedule selected for the Plan, unless otherwise elected in the Adoption Agreement.

Each period of Qualified Military Service served by an individual is, upon reemployment deemed to constitute service with the Employer for the purpose of determining the non-forfeitability of the Participant's Account balances under the Plan.

Changes in Vesting Schedule. The Employer may elect to amend the vesting schedule in the Adoption Agreement, provided the nonforfeitable interest of the Participant in their Employer and Matching Accounts is not less than the nonforfeitable interest in such Accounts under the Plan prior to the adoption the amendment.

Election of Former Schedule. Furthermore, each Participant with at least 3 Years of Service with the Employer may elect within a reasonable period after the adoption of an amendment changing the vesting schedule, to have their nonforfeitable percentage computed under the Plan without regard to such amendment. An election will be permitted by such a Participant, if the Participant has at least one Hour of Service in any Plan Year beginning on or after the effective date of the amendment. The period during which the election may be made will commence with the date the amendment is adopted or deemed to be made and will end on the latest of: (1) 60 days after the amendment is adopted; (2) 60 days after the amendment becomes effective; or (iii) 60 days after the Participant is issued written notice of the amendment by the Employer. With respect to benefits accrued as of the later of the adoption or effective date of such amendment, the vested percentage of each Participant will be the greater of the vested percentage under the old vesting schedule or the vested percentage under the new vesting schedule.

(a) Normal Retirement Age or Postponed Retirement. A Participant, on attaining their Normal Retirement Age will be 100% vested and have a nonforfeitable interest in their Employer and Matching Accounts. If a Participant continues in active employment following their Normal Retirement Age, they

will continue to participate under the Plan.

- **(b) Disability.** If a Participant terminates employment prior to their Normal Retirement Age due to a Disability, they will be vested and have a nonforfeitable interest in their Employer and Matching Accounts as elected in the Adoption Agreement.
- (c) Death. In the event of the death of a Participant before termination of employment with the Employer and before Normal Retirement Age, the Participant will be vested and have a nonforfeitable interest in their Employer and Matching Accounts as elected in the Adoption Agreement.
- (d) Termination of Plan. In the event of Plan termination (including termination resulting from a complete discontinuance of contributions by the Employer), each Participant employed by the Employer as of the date of such termination will be 100% vested and have a nonforfeitable interest in all of their Accounts. In the event of a partial Plan termination, each Participant employed by the Employer with respect to whom such partial termination has occurred will be 100% vested and have a nonforfeitable interest in all of their Accounts.
- (e) Early Retirement, Resignation, or Discharge. If the employment of a Participant terminates by reason of early retirement, resignation, or discharge prior to their Normal Retirement Age, they will be vested and have a nonforfeitable interest in a percentage of their Employer and Matching Accounts taking into account all of their Years of Vesting Service as of such termination date in accordance with the schedule set forth in the Adoption Agreement.
- **(f) All Other Times.** A Participant will be vested and have a nonforfeitable interest in a percentage of their Employer and Matching Accounts determined by taking into account all of their Years of Vesting Service in accordance with the schedule set forth in the Adoption Agreement.
- (g) Elapsed Time Method. A Participant will be vested and have a nonforfeitable interest in a percentage of their Employer and Matching Accounts determined by taking into account all of their Years of Vesting Service in accordance with the schedule set forth in the Adoption Agreement.
- **2.4.2 Leave of Absence.** The Plan Administrator will not treat a temporary cessation from active employment with the Employer pursuant to an authorized leave of absence in accordance with the nondiscriminatory policy of the Employer, whether occasioned by illness, military service, or any other reason, as either a termination of employment or a Break in Service so long as the Employee returns to employment in a manner consistent with the authorized leave of absence policy.
- **2.4.3 Reemployment.** Unless otherwise elected by the Employer in the Adoption Agreement, in the case of a Participant who has 5 or more consecutive Breaks in Service, all Years of Vesting Service after such Breaks in Service will be disregarded for vesting the Employer Account balance that accrued before such breaks, however, both pre-break and post-break service will count for vesting the Employer Account balance that accrues after such breaks.

Account balances that accrued before and after such Break in Service will share in the earnings and losses of the Trust Fund.

If a Participant receives a distribution from their Account when they have less than a 100% vested and nonforfeitable interest in the Account and the Participant resumes employment covered under the Plan, the Plan Administrator will restore the Participant's Employer-derived Account balance to the amount on the date of distribution if the Participant repays to the Plan the full amount of the distribution attributable to Employer Contributions before the earlier of 5 years after the first date on which the Participant is subsequently reemployed by the Employer, or the date the Participant incurs 5 consecutive Breaks in Service following the date of the distribution. If a Participant is deemed to receive a distribution pursuant to this Section, and the Participant resumes employment covered under the Plan before the date the Participant incurs 5 consecutive Breaks in Service, upon the reemployment of such Participant, the Plan Administrator will restore the Employer Account balance of the Participant to the amount on the date of such deemed distribution as of the date of the reemployment of such Participant.

2.4.4 Forfeitures. Unless otherwise elected in the Adoption Agreement, if a Participant terminates service and elects, in accordance with the provisions of the Plan, to receive the value of the their vested Account balance, the non-vested portion will be treated as a Forfeiture as of the earlier of the last day of the Plan Year in which the Plan Administrator distributes the Participant's entire vested interest, or the last day of the Plan Year of the 5th consecutive Break in Service. If the Participant elects to receive less than the entire vested portion of the Account balance derived from Employer contributions, the portion forfeited will equal the total non-vested portion of such Account multiplied by a fraction, the numerator of which is the distributed amount attributable to Employer contributions and the denominator of which is the entire vested portion of such Account.

If a Participant terminates service, and the value of the Participant's vested Account balance derived from Employer and Employee contributions is not greater than \$5,000 (or a dollar amount that is less, if elected in the Adoption Agreement), and the Participant receives a distribution of the value of the entire vested portion of such Account balance, the Plan Administrator will treat the non-vested portion as a Forfeiture, unless elected otherwise in the Adoption Agreement, as of the earlier of the last day of the Plan Year in which the Plan Administrator distributes the Participant's entire vested interest, or the last day of the Plan Year of the 5th consecutive Break in Service. If elected in the Adoption Agreement, the portion of the Account balance that is attributable to rollover contributions (and earnings allocable thereto) within the meaning of Code sections 402(c), 403(a)(4), 403(b)(8), 408(d)(3)(A)(ii), and 457(e)(16), will be excluded from the Participant's Account balance for purposes of this Section. If a Participant would have received a distribution under the preceding but for the fact that the Participant's vested Account balance exceeded \$5,000, or such lesser amount as elected above, when the Participant terminated service and if at a later time such Account balance is reduced such that it is not greater than \$5,000, or such lesser amount as elected above, the Participant will receive a distribution of such Account balance and the Plan Administrator will treat the non-vested portion as a Forfeiture. If the value of a Participant's vested Account balance is zero, the Participant will be deemed to have received a distribution of such vested Account balance. A Participant's vested Account balance will not include accumulated deductible Employee contributions within the meaning of Code section 72(o)(5)(B) for Plan Years beginning prior to January 1, 1989.

Unless otherwise elected in the Adoption Agreement, the Plan Administrator will allocate such Forfeiture in the same manner as a contribution by the Employer for the year in which said Forfeiture occurred. For each Plan Year that Forfeitures are allocated, the Plan Administrator will designate the specific Employer Contribution or Contributions that the Forfeitures reduce or supplement. The Employer may elect in the Adoption Agreement to use Forfeitures to offset administrative expenses of the Plan to the extent the expenses are Plan expenses and not settlor expenses and any remaining Forfeitures will be allocated in the same manner as a contribution by the Employer for the year in which said Forfeiture occurred. Forfeitures that are used to supplement a contribution will increase that contribution that was otherwise specified for the Plan Year. Forfeitures that are used to reduce a fixed required contribution reduce the amount that the Employer would otherwise have to contribute to the Plan for the Plan Year. A Participant's forfeitable amount must remain in the Participant's account until used as a Forfeiture. The forfeitable amount must be exhausted by the end of the next plan year, within the limitations of the Code and Treasury Regulations, but in any event, as soon as administratively feasible.

If a Participant is reemployed following a Break in Service and is entitled to restoration of any amount of their Account that was forfeited because of such Break in Service, the Plan Administrator will restore such amount in the manner specified in the Adoption Agreement.

ARTICLE V

DISTRIBUTIONS

2.5.1 Distributable Benefit.

- (a) When a Participant incurs a severance of employment (for any reason) and they are no longer employed by the Employer or another Related Employer, they or their Beneficiary will be entitled to a benefit equal to the vested and nonforfeitable interest in their Accounts as of the Distribution Determination Date. This will include a Disability benefit if so elected in the Adoption Agreement. Such Accounts will include the allocable share of contributions and Forfeitures, if any, that may be allocated to said Accounts as of such Distribution Determination Date and will be determined after making the adjustments for which provision is made in the Plan.
- (b) If the value of the Participant's vested and nonforfeitable interest in the Plan at the time of their termination of employment is zero, the Participant will be deemed to have received a distribution of such interest as their Distributable Benefit as of the date their employment terminated.
- **(c) Minimum Distributable Benefit.** If a Participant terminates service, and the value of the Participant's vested Account balance derived from Employer Contributions is zero, the Employer may elect in the Adoption Agreement to provide such Participant with a minimum Distributable Benefit.
- (d) **Distribution Determination Date.** For purposes of determining the amount to be distributed, the Distribution Determination Date will be determined in the manner specified in the Adoption Agreement.
- (e) Distributions following Distribution Determination Date. Subject to the necessity, if any, of obtaining the consent of a Participant and spouse, distribution of a Participant's Distributable Benefit will commence within a reasonable period after the Distribution Determination Date, unless otherwise elected by the Participant in accordance with the provisions of the Plan or as required by the provisions of the Plan. A distribution will not be made to a Participant if they are reemployed prior to the date their benefit would have been distributed.
- **2.5.2 Forms of Distribution.** Subject to the Employer's elections in the Adoption Agreement distribution will be made by one of the following methods, as determined in accordance with the election of the Participant (or in the case of death, their Beneficiary) with such spousal consents as may be required by law.

Notwithstanding any election in the Adoption Agreement, Beneficiaries may elect to receive Required Minimum Distributions in the form of installment payments.

(a) Lump Sums. Payments made in a single taxable year representing the Participant's entire Distributable Benefit.

If the Participant fails to elect a distribution option and the Participant's vested Account balance is at or below the Mandatory Cash-out threshold of \$5,000, (or lesser threshold amount, elected in the Adoption Agreement), then in lieu of a lump sum distribution, payment will be made pursuant to Subsection 2.5.3.(c) Mandatory Cash-out. The Employer may set a minimum Account balance to which this will apply. This Mandatory Cash-out provision applies only if the terminating Participant fails to request affirmatively a cash payment, or a Direct Rollover to another qualified plan, (or an IRA designated by the terminating Participant). If the value of a Participant's vested Account balance exceeds the threshold amount, an immediate cash-out will not be made unless the Participant, and if applicable, their spouse, or their Beneficiary consent to such distribution.

(b) Installments.

- (1) Payments made in substantially equal annual, quarterly, or monthly installments over a period of more than 1 taxable year but not in excess of the period designated in the Adoption Agreement, or if permitted, as selected by the Participant (provided that such period is not greater than the Participant's life expectancy or the joint life expectancy of Participant and the Participant's Designated Beneficiaries), plus accrued net income. If distribution is to be made in installments, the Plan Administrator may transfer the undistributed portion of the Distributable Benefit to a Segregated Account, from which installment payments will thereafter be withdrawn from time to time.
- (2) Partial Nonperiodic Distribution. If elected in the Adoption Agreement, ad hoc distributions at the times and in the amounts requested by the Participant or Beneficiary may be made subject to the limits, if any, in the Adoption Agreement.
- (c) Annuities. Payments made by the purchase and delivery of a single premium, nontransferable, fully refundable, annuity contract issued by a legal reserve life insurance company. Payments must provide for equal periodic payments for the life of the Participant, or upon the Participant's death for payments to continue to the Participant's Designated Beneficiary over a specified period as may be designated in the Adoption Agreement and as selected by the Participant. Notwithstanding the Employer may elect a period certain annuity distribution option in the Adoption Agreement, the annuity under this election will be for a fixed duration that is less than the Participant's life expectancy as of the Annuity Starting Date. Such annuity will conform to all provisions set forth herein. Any refund feature under such annuity contract following the death of the Participant will inure to the benefit of the Participant's Beneficiary.

Notwithstanding, If the Distributable Benefit is subject to the funding standards of Code Section 412, the distribution to a married Participant will be made in the form of a Qualified Joint and Survivor Annuity or in the form of a life annuity to an unmarried Participant, subject to the terms and provisions under Subsection 2.5.6.

- (d) Alternative Distribution Methods. Any alternative distribution method of equivalent value that was contained in the Plan at any time on or after the first day of the first Plan Year beginning after 1988 to which the Participant and spousal consents.
- **2.5.3 Methods of Distribution.** Subject to the provisions of Section 2.5.7 and 2.5.8 below, any security interest in a loan from the Plan for which any necessary spousal consent has been obtained (to the extent such security interest is used as repayment of the loan), distribution will be made by one of the following methods, as determined in accordance with the election of the Participant (or in the case of death, their Beneficiary) with such spousal consents as may be required by law. Notwithstanding any restrictions on optional forms of distribution in the Adoption Agreement, active Participants may elect to receive Required Minimum Distributions in the form of installment payments in the amount of the Required Minimum Distribution.
- (a) Lump Sum Payment to Participant with 60-Day Rollover Option. The Plan may pay a Distributable Benefit in a single sum distribution directly to a Participant, subject to the Spousal Consent requirements under Subsection 2.5.5 and the Automatic Rollover requirements under Subsection 2.5.3(d) below. Upon receipt and within 60 days, a Participant may elect to transfer, rollover, or deposit to an individual retirement account or another qualified plan all or a portion of the Distributable Benefit that is an Eligible Rollover Distribution.
- (b) Direct Rollovers. Notwithstanding any provision of the Plan to the contrary that would otherwise limit a Participant's election under the Plan, a Participant may elect, at the time and in the manner prescribed by the Plan Administrator, to have any portion of an Eligible Rollover Distribution that is equal to at least \$500 paid directly to an Eligible Retirement Plan specified by the Participant in a Direct Rollover. If an Eligible Rollover Distribution is less than \$500, a Participant may not make the election described in the preceding sentence to roll over a portion of the Eligible Rollover Distribution.

(c) Mandatory Cash-out. The Employer may elect in the Adoption Agreement to require the mandatory distribution of a Participant's vested Account balance if its value, determined as of a date following the Participant's date of termination of employment with the Employer, is equal to no more than \$5,000 (or some lesser threshold amount as elected in the Adoption Agreement). If a distribution is required under this provision, the Participant will receive an immediate cash-out of their vested Account balance and no further benefits will be payable from the Plan. If the value of a Participant's vested Account balance exceeds the threshold amount, an immediate cash-out will not be made unless the Participant, and if applicable, their spouse, or their Beneficiary consent to such distribution.

The Employer may elect in the Adoption Agreement to exclude that portion of a Participant's Account balance that is attributable to Rollover contributions (and earnings allocable thereto) within the meaning of Code sections 402(c), 403(a)(4), 403(b)(8), 408(d)(3)(A)(ii), and 457(e)(16) in the determination of the Participant's nonforfeitable Account balance.

- (d) Automatic Rollovers. Effective for distributions on and after March 28, 2005, in the event a Mandatory Cash-out distribution is greater than \$1,000, or some lesser amount, in accordance with the provisions of Subsection 2.5.3(c), if the Participant does not elect to have such distribution paid directly to an Eligible Retirement Plan specified by the Participant in a Direct Rollover or to receive the distribution directly in accordance with Subsection 2.5.3(a), then the Plan Administrator will pay the distribution in a Direct Rollover to an individual retirement plan designated by the Plan Administrator. The Plan Administrator will select an IRA trustee, custodian, or issuer (the "trustee") that is unrelated to the Employer, will establish the IRA with that trustee on behalf of the terminating Participant who fails affirmatively to elect a Direct Rollover of a cash distribution, and make the initial investment choices for the account. For purposes of determining whether a mandatory distribution is greater than \$1,000, the portion of the Participant's distribution attributable to any rollover contribution is included.
- **2.5.4 Commencement of Distributions.** For purposes of determining when a Distributable Benefit can be distributed under this Article V, unless otherwise elected in the Adoption Agreement, by the Employer:
- (a) Immediate Distribution. A distribution to a Participant whose vested Account balance is Immediately Distributable. A Participant's Account Balance is Immediately Distributable if any part of the Account balance could be distributed to the Participant, (or surviving spouse), before the Participant attains, (or would have attained if not deceased), the later of age 62 or their Normal Retirement Age.
 - (1) If the account is distributable, due to the Participant's death, Disability, or attainment of Early or Normal Retirement Age, the Participant may elect (or their Beneficiary may elect) to begin distribution of their Distributable Benefit within a reasonable period after the Distribution Determination Date. A Participant (or their Beneficiary), when permitted under the terms of the Plan, may elect a deferred Distribution under Subsection (b) below.
 - (2) If the account is immediately distributable, the Participant will have the distribution of their Distributable Benefit governed by Subsection (b) below.
- (b) Deferred Distribution. Except in the case of amounts subject to Subsection 2.5.6(I), or mandatory distributions for which a Participant's consent is not required, a Participant with a Distributable Benefit in excess of \$5,000 or the specified Mandatory Cash-out threshold selected by the Employer in the Adoption Agreement, a Participant may elect to delay the commencement of distributions. Upon such an election, the distribution of a Participant's Distributable Benefit will be deferred and must commence no later than the sixtieth (60th) day after the last day of the Plan Year in which the following occurs:
 - (1) The Participant attains the earlier of age sixty-five (65) or the Normal Retirement Age;
 - (2) The Participant reaches the tenth (10th) anniversary of their participation in the Plan; or
 - (3) The Participant terminates their employment with the Employer.

A Participant who terminates employment before satisfying the age requirement for Normal Retirement, or Early Retirement Age, but has satisfied any service requirement will be entitled to a distribution of their Distributable Benefit in accordance with Subsection (a) above upon attaining such age.

The Plan Administrator will notify the Participant and the Participant's spouse of the right to defer any distribution while the Participant's Account balance in the Plan is Immediately Distributable, as defined in Subsection 2.5.7(a). Such notice will be given to the Participant at least 30, but no more than 180 days (90 days for Plan Years beginning before January 1, 2007) before the distribution. A Participant and the Participant's spouse may elect a distribution commencing less than 30 days after the notice is provided to the Participant, if the notice clearly indicates that the Participant has at least 30 days to decide whether to consent to the distribution. The failure of a Participant or the Participant's spouse to consent to a distribution while a benefit is Immediately Distributable will be deemed to be an election to defer commencement of benefits under this Section. If the Plan meets the requirements of Subsection 2.5.5(c), only the Participant must consent to the distribution of an Account balance that is Immediately Distributable.

Such distribution may commence less than 30 days after the notice required under Treasury Regulations section 1.411(a)-11(c) is given, provided that:

- (1) The Plan Administrator clearly informs the Participant that the Participant has a right to a period of at least 30 days after receiving the notice to consider the decision of whether or not to elect a distribution (and if applicable, a particular distribution option),
- (2) The Participant, after receiving the notice, affirmatively elects a distribution,
- (3) The distribution commences more than 7 days after such explanation is provided,
- (4) The Participant is permitted to revoke any affirmative distribution election at least until the Annuity Starting Date or, if later, at any time prior to the expiration of the 7 day period that begins the day after the explanation of the Qualified Joint and Survivor Annuity is provided to the Participant, and
- (5) The Annuity Starting Date is a date after the date that the written explanation was provided to the Participant.
- (c) Advance Distributions. If the Employer elects in the Adoption Agreement to permit advance distribution to a Participant or their Beneficiary after their employment has terminated or after they reach Normal Retirement Age but continues their employment, and before they are otherwise entitled to distribution of their Distributable Benefit but in no event earlier than a reasonable period following the Distribution Determination Date, the Trustee upon the request of the Participant or Beneficiary will make advance distributions to them or to their Beneficiary. The aggregate of such advance distributions will not exceed the sum of the vested and nonforfeitable interest in the Participant's Accounts.

If the Employer elects in the Adoption Agreement to forfeit non-vested amounts immediately upon distribution of the Employee's entire vested Account balance on termination of service, an Employee who terminates service and elects to receive the value of the Employee's vested Account balance will forfeit the non-vested portion. If the Employee elects to receive a distribution of less than the entire vested portion of the Account balance derived from Employer Contributions, the part of the non-vested portion that is treated as a Forfeiture is the total non-vested portion multiplied by a fraction, the numerator of which is the amount of the distribution attributable to Employer Contributions and the denominator of which is the total value of the vested Employer derived Account balance.

Except as provided in the preceding paragraph, if a Participant receives a distribution that reduces the balance in their Employer Account when they have less than a 100% vested and nonforfeitable interest in the Account, the amount, if any, of the Participant's vested and nonforfeitable interest in the undistributed balance of said Account at any relevant time will not be less than an amount ("X") determined by the formula: $X = P(AB + (R \times D)) - (R \times D)$. For purposes of applying the formula: P is

the vested percentage at the relevant time; AB is the Account balance at the relevant time; D is the amount of the distribution; and R is the ratio of the Account balance at the relevant time to the Account balance after distribution.

- (d) Distribution upon Plan Termination. In the event of Plan termination (including termination resulting from a complete discontinuance of contributions by the Employer), the Distribution Determination Date will be the date of such termination. In the event of a partial Plan termination, the Distribution Determination Date for each Participant with respect to whom such partial termination has occurred will be the last day of the Plan Year coinciding with or immediately following the date of such partial termination.
- (e) Distribution upon Death. In the event of the Participant's death, the Distributable Benefits will be paid to the Participant's Designated Beneficiary according to the Required Minimum Distribution provision under Section 2.5.7. If the Participant dies before the distribution has begun under the Required Beginning Date of Subsection 2.5.7(b), unless the Plan or a Participant's Account is subject to Qualified Joint and Survivor Annuity requirement, then a Qualified Preretirement Survivor Annuity will be payable to the Participant's Spouse, unless the Participant and Spouse elects to waive subject to Subsection 2.5.6(k).

Notwithstanding, the survivors of any Participant who dies on or after January 1, 2007, while performing Qualified Military Service, are entitled to any additional benefits (other than contributions relating to the period of Qualified Military Service, but including vesting service credit for such period and any ancillary life insurance or other survivor benefits) that would have been provided under the Plan had the Participant resumed employment on the day preceding the Participant's death and then terminated employment on account of death.

(f) Distribution upon Disability. In the event the Participant is determined to be Disabled, the Participant will be entitled to receive their Distributable Benefit within a reasonable period as elected by the Employer in the Adoption Agreement.

If the Participant becomes Disabled after termination of employment, they will be entitled to commence distribution of their vested Account balance as if they became Disabled prior to termination of employment, however their vested Account balance will not increase if the Participant becomes Disabled after the termination of their employment.

- **2.5.5 General Distributions Requirements.** Where a Plan permits a distribution under the terms of this Article, the Plan Administrator must notify the Participant (or in the event of the Participant's death, the Participant's spouse or survivor) of their right to elect a distribution (or the right to elect an optional forms of payment) under the terms of the Plan.
- (a) Notice and Consent. The Plan Administrator will notify the Participant of their right to a distribution by distributing the 402(f) Safe Harbor Tax Notice provided by the Internal Revenue Service and request an affirmative election to one of the distribution options under the Plan. The consent notice must be provided at least 30 days but in no case more than 180 days (90 days for Plan Years beginning before January 1, 2007) before the distribution commence.

Generally, no payment can be made under the Plan until at least 30 days after the Participant has received the consent notice. The consent notice must clearly inform the Participant that they have:

- (1) a right to a period of at least 30 days after receiving the notice to consider the decision of whether or not to elect a distribution (and if applicable, a particular distribution option),
- (2) a right to defer their distribution under Subsection 2.5.4, until they reach the Plan's NRA and/or commencement date as elected by the employer in the Adoption Agreement, and, for Plan Years beginning after December 31, 2006, the consequences of failing to defer any distribution, and
- (3) a right to elect a Direct Rollover under Subsection 2.5.3.(b).

After receipt of the consent notice, the Participant has the right to waive all or a portion of the unexpired 30-day notice period and affirmative elects a distribution under the terms of the Plan.

Notwithstanding, where the Participant's vested Account balance derived from Employer and Employee Contribution is less than \$5,000 or some lesser amount elected by the Employer in the Adoption Agreement, the Participant will not be required to consent to a certain form or method of distribution or the waiver of the 30 day notice period. Absent an affirmative election the Participant's distribution will be made as an Automatic Rollover under Subsection 2.5.3(d).

(b) Spousal Consent. If payment in the form of a Qualified Joint and Survivor Annuity is not required with respect to a Participant and the value of a Participant's vested Account balance derived from Employer and Employee contributions exceeds \$5,000, and the Account balance is Immediately Distributable, then the Participant and the Participant's spouse must consent to any distribution of such Account Balance prior to reaching the Normal Retirement Age under the Plan. If payment in the form of a Qualified Joint and Survivor Annuity is required with respect to a Participant and the value of a Participant's vested Account balance derived from Employer and Employee contributions exceeds \$5,000, and the Account balance is Immediately Distributable, then the Participant and the Participant's spouse must consent to any distribution of such Account Balance prior to reaching the Normal Retirement Age under the Plan. The notice and consent requirements under Subsection 2.5.5 will apply.

Neither the consent of the Participant nor the Participant's spouse will be required to the extent that a distribution is required to satisfy Code sections 401(a)(9) or 415.

- (c) Exception to Spousal Consent Requirements. If a plan designated as either a Cash or Deferred Arrangement or Profit Sharing Plan, unless otherwise elected by the Employer in the Adoption Agreement, a Participant will not be required to obtain the consent of their spouse for any distribution or Participant loan if the following conditions are satisfied:
 - (1) The Participant does not or cannot elect payments in the form of a life annuity; and
 - (2) On the death of a Participant, the Participant's vested Account balance will be paid to the Participant's surviving spouse. If there is no surviving spouse or if the surviving spouse has consented in a manner conforming to a qualified election, then the Participant's vested Account balance will be paid to the Participant's Designated Beneficiary.

Notwithstanding, if a Participant's vested Account Balance in a plan designated as a Profit Sharing Plan or Cash or Deferred Arrangement contains any assets that were a direct or indirect transferee of a defined benefit plan, money purchase plan or target benefit plan, then spousal consent will be required for that portion of the Plan that is subject to the joint and survivor annuity requirements of Code sections 401(a)(11) and 417.

2.5.6 Annuity and Consent Requirements. If the only forms of distribution selected by the Employer in the Adoption Agreement are annuities or if the Plan or Distributable Benefit is subject to the minimum funding requirement of Code section 412, the Participant's distribution will be subject to joint and survivor provisions of Code sections 401(a)(11) and 417.

Notwithstanding the other provisions of this Section, if the Plan is designated in the Adoption Agreement as a Cash or Deferred Profit Sharing Plan or a Profit Sharing Plan and the Participant does not select a life annuity distribution option, the Qualified Joint and Survivor Annuity and Qualified Preretirement Survivor Annuity forms of distribution will not be available, except in the case of transfers under Subsection 3.9.3(b). However, a Participant's surviving spouse will be entitled to elect distribution of the Participant's vested Account balance in the manner provided by Section 3.6.3.

(a) Automatic Qualified Joint and Survivor Annuity. If benefits in the form of a life annuity become payable to a married Participant who:

- (i) Begins to receive payments under the Plan on or after Normal Retirement Age; or
- (ii) Dies on or after Normal Retirement Age while still working for the Employer; or
- (iii) Begins to receive payments on or after the qualified Early Retirement Age; or
- (iv) Separates from service on or after attaining Normal Retirement Age (or the qualified Early Retirement Age) and after satisfying the eligibility requirements for the payment of benefits under the Plan and thereafter dies before beginning to receive such benefits;

then such benefits will be received under this Plan in the form of a Qualified Joint and Survivor Annuity, unless the Participant has elected otherwise during the election period. The election period must begin at least 6 months before the Participant attains qualified Early Retirement Age and end not more than 90 days before the commencement of benefits. Any election hereunder will be in writing and may be changed by the Participant at any time.

- **(b) Terms of Annuity Contracts.** Any annuity contract distributed from the Plan must be nontransferable. The terms of any annuity contract purchased and distributed by the Plan to a Participant or a Participant's spouse will comply with the requirements of the Plan as elected by the Employer in the Adoption Agreement.
- **(c) Survivor Annuity.** If elected by the Employer in the Adoption Agreement the Plan will provide an immediate annuity for the life of the Participant and if the Participant is married, a survivor annuity for the life of the spouse which is not less than 50% (or a higher percentage as selected by the employer in the Adoption Agreement).

Unless a survivor annuity form of benefit is waived, pursuant to a qualified election within the applicable election period, a married Participant's vested Account balance will be applied toward the purchase of a Qualified Joint and Survivor Annuity (QJSA) and an unmarried Participant's vested account balance will be applied toward the purchase of a Straight Life Annuity (SLA). The Participant may elect to receive another form of payment or to receive a Qualified Optional Survivor Annuity within the applicable election period and waive the survivor annuity payments under the Plan.

A Qualified Optional Survivor Annuity (QOSA) is an optional form of benefit that provides a survivor annuity based on the survivor annuity provided under the Plan's Qualified Joint and Survivor Annuity as described in Subsection 2.5.6(j).

If the married Participant dies before the annuity start date, then the Participant's vested Account balance will be applied toward the purchase of a Qualified Pre-retirement Survivor Annuity (QPSA). The value of the Qualified Pre-retirement Survivor Annuity will not be less than the amount that would have been payable under the Qualified Joint and Survivor Annuity form of benefit. The surviving spouse may receive a payment under the Qualified Pre-retirement Survivor Annuity not later than the month in which the Participant would have attained the earliest retirement date under the Plan. Notwithstanding, the surviving spouse may elect to have such annuity distributed within a reasonable period after death.

(d) Exception if Participant is not Married for One Year. If the "One Year Marriage Rule" is elected in the Adoption Agreement, notwithstanding the foregoing, the benefits under the Plan will not be provided in the form of a Qualified Joint and Survivor Annuity or a Qualified Preretirement Survivor Annuity unless the Participant and their spouse have been married throughout the 1 year period ending on the earlier of the Participant's Annuity Starting Date or the date of the Participant's death.

If a Participant marries within 1 year before the Annuity Starting Date and the Participant and their spouse in such marriage have been married for at least a 1 year period ending on or before the date of the Participant's death, the Participant and such spouse will be treated as having been married throughout the required period. A former spouse will be treated as the spouse or surviving spouse and a current spouse will not be treated as the spouse or surviving spouse, to the extent provided under a

Qualified Domestic Relations Order as described in Code section 414(p).

- (e) Notice and Consent to Distributions. The Plan Administrator will notify the Participant and the Participant's spouse of the right to elect this form of distribution or the right to elect an optional form under the terms of the Plan. Such notification will include a general description of the material features and an explanation of the relative values of the optional forms of benefit available under the Plan in a manner that would satisfy the notice requirements of Code section 417(a)(3) and within a reasonable time before the Annuity Start Date.
- **(f) Written Explanations of Qualified Joint and Survivor Annuity.** In general, the Plan Administrator will provide each Participant, with a written explanation of:
 - (1) The terms and conditions of a Qualified Joint and Survivor Annuity and of the Qualified Optional Survivor Annuity;
 - (2) The Participant's right to make and the effect of an election to waive the Qualified Joint and Survivor Annuity form of benefit;
 - (3) The rights of the Participant's spouse to consent to a Participant's election; and
 - (4) The right to make and the effect of a revocation of an election.

The written explanation will comply with the requirements of section 1.417(a)(3)-1 of the Income Tax Regulations.

- (g) Applicable Election Period (QJSA). For the purpose of a Qualified Joint and Survivor Annuity the applicable election period will be no less than 30 days and no more than 180 days (90 days for Plan Years beginning before January 1, 2007) before the Annuity Starting Date with a written explanation of the survivor annuity option. During this period the Participant may elect a Qualified Optional Survivor Annuity.
- (h) Written Explanation of Qualified Preretirement Survivor Annuity. The Plan Administrator will provide to each Participant within the applicable election period, a written explanation of a Qualified Preretirement Survivor Annuity comparable to that provided with respect to a Qualified Joint and Survivor Annuity percentage elected in the Plan. The written explanation will comply with the requirements of section 1.417(a)(3)-1 of the Income Tax Regulations.
- **(i) Applicable Election Period (QPSA).** The Applicable election period for a Qualified Preretirement Survivor Annuity means with respect to a Participant, the later of the following periods:
 - (1) The period beginning with the first day of the Plan Year in which the Participant attains age 32 and ending with the close of the Plan Year preceding the Plan Year in which the Participant attains age 35.
 - (2) A reasonable period ending after the individual becomes a Participant.
 - (3) A reasonable period ending after the Plan ceases to fully subsidize costs.
 - (4) A reasonable period ending after Code section 401(a)(11) first applies to the Participant.
 - (5) A reasonable period ending after separation from service in case of a Participant who separates before attaining age 35. For purposes of applying the foregoing, a reasonable period ending after the enumerated events described in (2), (3) and (4) is the end of the two-year period beginning 1 year prior to the date the applicable event occurs and ending 1 year after that date. In the case of a Participant who separates from service before the Plan Year in which age 35 is attained, notice will be provided within the two-year period beginning prior to separation and ending 1 year after separation. If such a Participant thereafter returns to employment with the Employer, the Applicable Period for such Participant will be redetermined.

(j) Qualified Optional Survivor Annuity. Plans to which the Annuity and Consent requirements of Code sections 401(a)(11) and 417 apply must provide a Qualified Optional Survivor Annuity.

For purposes of a Qualified Optional Survivor Annuity, the term "Survivor Annuity Percentage" means the percentage by which the survivor annuity under the plan's Qualified Joint and Survivor Annuity bears to an annuity payable during the joint lives of the Participant and their spouse. If the Survivor Annuity Percentage:

- (1) is less than 75%, then the applicable percentage is 75%; or
- (2) is greater than or equal to 75%, then the applicable percentage is 50%.

The Qualified Optional Survivor Annuity provisions will apply to the Plan and are effective for Plan Years beginning after December 31, 2007, subject to certain special effective date rules for collectively bargained plans.

- (k) Election to Waive Annuity Payment. At any time during the applicable election period, a Participant and their spouse may elect to waive the Qualified Joint and Survivor Annuity form of benefit or the Qualified Preretirement Survivor Annuity form of benefit (or both) and may revoke any such election. There is no limit to the number of times that an election or a revocation may be made by a Participant.
- (I) Spousal Consent Required. An election to waive the Qualified Joint and Survivor Annuity form of benefit or the Qualified Preretirement Survivor Annuity form of benefit (or both) will not take effect unless:
 - (1) the spouse of the Participant consents in writing to the election, and such election designates a specific Beneficiary, including any class of Beneficiaries or contingent Beneficiaries, or, solely in the case of a waiver of a Qualified Joint and Survivor Annuity, designates an alternative form of benefit, that may not be changed without spousal consent (unless the consent of the spouse expressly permits designations by the Participant without any requirement of further consent by the spouse),
 - (2) the spouse's consent acknowledges the effect of such election and is witnessed by a Plan representative or a notary public, or
 - (3) the Participant establishes to the satisfaction of the Plan Administrator that such consent cannot be obtained because there is no spouse, because the spouse cannot be located, or because of other circumstances permitted by applicable regulations.

Any consent by a spouse obtained under this provision (or establishment that the consent of a spouse may not be obtained) will be effective only with respect to such spouse. A consent that permits designations by the Participant without any requirement of further consent by such spouse must acknowledge that the spouse has the right to limit consent to a specific Beneficiary, and a specific form of benefit where applicable, and that the spouse voluntarily elects to relinquish either or both of such rights.

A Participant may revoke a prior waiver without the consent of the spouse at any time before the commencement of benefits. The number of waivers and revocations will not be limited. No consent obtained under this provision will be valid unless the Participant has received notice as provided in Notice and Consent to Distribution provision of this section.

If payment in the form of a Qualified Joint and Survivor Annuity is required with respect to a Participant and either the value of a Participant's vested Account balance derived from Employer and Employee contributions exceeds \$5,000 (or a lesser Mandatory Cash-out amount selected in the Adoption Agreement), or there are remaining payments to be made with respect to a particular distribution option that previously commenced, prior to the Participant reaching the earliest retirement age under the Plan, then the Participant and the Participant's spouse must consent to any distribution from the

Plan.

If payment in the form of a Qualified Joint and Survivor Annuity is required with respect to a Participant and the value of a Participant's vested Account balance derived from Employer and Employee contributions does not exceeds \$5,000(or a lesser Mandatory Cash-out amount selected in the Adoption Agreement), and the Account balance is Immediately Distributable, then the consent requirement will not apply to the distribution from the Plan.

If payment in the form of a Qualified Joint and Survivor Annuity is required with respect to a Participant or the Plan, and the vested Account balance is used as security for a loan from the Plan, the Participant's spouse must consent in writing and acknowledge the effect of using the vested Account balance on survivor benefit due under the Plan during the 180 day period ending on the date the loan is secured.

The consent of a Participant or Participant's spouse must in writing and obtained within the 180-day period (90 days for Plan Years beginning before January 1, 2007) ending on the Annuity Starting Date.

2.5.7 Required Minimum Distributions. All distributions required under this Article V will be determined and made in accordance with Code section 401(a)(9) and the applicable regulations, including the minimum distribution incidental benefit requirements of Code section 401(a)(9)(G). In the event that a provision of this Plan conflicts with such requirements, such requirements will govern.

As of the first Distribution Calendar Year (defined in Subsection 2.5.7(i)(1)), distributions, if not made in a single sum, may only be made over one of the following periods (or a combination thereof): (1) the life of the Participant; (2) the life of the Participant and a Designated Beneficiary; (3) a period certain not extending beyond the Life Expectancy of the Participant; or (4) a period certain not extending beyond the joint and last survivor expectancy of the Participant and a Designated Beneficiary.

The Plan was amended pursuant to section 401(a)(9)(H) of the Code to provide for the waiver of the Required Minimum Distributions for the 2009 Distribution Calendar Year. The Plan suspended all Required Minimum Distributions attributable to the 2009 Distribution Calendar Year. The otherwise Required Minimum Distributions were not required to be distributed from the Plan unless a Participant submitted a written request electing to receive a distribution for 2009. The Required Beginning Date for 2009 for any Participant who attained age 70 1/2 in 2009 was postponed from April 1, 2010 until December 31, 2010. The Plan was not required to make a distribution under this provision until December 31, 2010 unless a Participant submitted a written request electing to receive a distribution for 2009. The Required Minimum Distribution payable during the lifetime of a Participant or after the death of a Participant to their Designated Beneficiary that was attributable to the 2009 Distribution Calendar Year was not distributed unless a Participant or the Designated Beneficiary submitted a written request electing to receive a distribution for 2009. Under this 2009 suspension of the Required Minimum Distribution a Participant or a Designated Beneficiary could have requested that the Plan make a Required Minimum Distribution for the 2009 Distribution Calendar Year by completing an individual election form. When made this distribution would be treated as an Eligible Rollover Distribution from the Plan. Where a Required Minimum Distribution was payable and the five year rule applies with respect to a deceased Participant's account, then the Required Minimum Distribution for the 2009 Distribution Calendar Year will be disregarded in the calculation of the 5 year period; resulting in a one year extension of this rule.

- (a) Required Beginning Date. Notwithstanding anything herein to the contrary, unless the Participant has made an appropriate TEFRA 242(b)(2) election, as described in Subsection 2.5.7(j), that has not been revoked or modified, the Participant's entire benefit will be distributed, or begin to be distributed, to the Participant no later than the Required Beginning Date.
- (b) Timing of Distributions if Participant Dies Before Distributions Begin.
 - (1) If the Participant dies before required distributions begin, the Participant's entire interest will be distributed, or begin to be distributed, no later than as follows:

- (A) If the Participant's surviving spouse is the Participant's sole Designated Beneficiary, then, except as provided in the Adoption Agreement, distributions to the surviving spouse will begin by December 31 of the calendar year immediately following the calendar year in which the Participant died, or by December 31 of the calendar year in which the Participant would have attained age seventy and one-half (701/2), if later.
- **(B)** If the Participant's surviving spouse is not the Participant's sole Designated Beneficiary, then, except as provided in the Adoption Agreement, distributions to the Designated Beneficiary will begin by December 31 of the calendar year immediately following the calendar year in which the Participant died.
- **(C)** If there is no Designated Beneficiary as of the date of the Participant's death who remains a Beneficiary as of September 30 of the year immediately following the year the Participant died, the Participant's entire interest will be distributed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
- (D) If the Participant's surviving spouse is the Participant's sole Designated Beneficiary and the surviving spouse dies after the Participant but before distributions to the surviving spouse begin, this Subsection 2.5.7(b), other than Subsection 2.5.7(b)(1)(A), will apply as if the surviving spouse were the Participant.
- (2) For purposes of this Subsection 2.5.7(b) and Subsections 2.5.7(f) and (g), distributions are considered to begin on the Participant's Required Beginning Date unless Subsection 2.5.7(b)(1)(D) applies. If Subsection 2.5.7(b)(1)(D) applies, distributions are considered to begin on the date distributions are required to begin to the surviving spouse under Subsection 2.5.7(b)(1)(A). If distributions under an annuity contract purchased from an insurance company irrevocably commence to the Participant before the Participant's Required Beginning Date (or to the Participant's surviving spouse before the date distributions are required to begin to the surviving spouse under Subsection 2.5.7(b)(1)(A)), the date distributions are considered to begin is the date distributions actually commence.
- (3) If the Adoption Agreement permits Participants and Beneficiaries to elect the 5-year rule, the election must be made no later than the earlier of:
 - (A) September 30 of the calendar year in which the distribution would be required under Subsections 2.5.7(b)(1)(A) or (B), or
 - **(B)** By September 30 of the calendar year that contains the 5th anniversary of the Participant's (or if applicable, the surviving spouse's) death.

If this election is made, then the Participant's entire interest must be distributed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.

- (c) Forms of Distribution. Unless the Participant's interest is distributed in the form of an annuity contract purchased from an insurance company or in a single sum on or before the Required Beginning Date, distributions will be made in accordance with Subsections 2.5.7(d) and (f) as of the first Distribution Calendar Year. If the Participant's interest is distributed in the form of an annuity contract purchased from an insurance company, distributions thereunder will be made in accordance with the requirements of Code section 401(a)(9) and the applicable regulations.
- (d) Required Minimum Distributions During Participant's Lifetime. During the Participant's lifetime, the minimum amount that will be distributed for each Distribution Calendar Year is the lesser of:
 - (1) The quotient obtained by dividing the Participant's Account Balance by the distribution period set forth in the Uniform Lifetime Table found in Treasury Regulations section 1.401(a)(9)-9, Q&A-2, using the Participant's age as of the Participant's birthday in the Distribution Calendar Year; or

- (2) If the Participant's sole Designated Beneficiary for the Distribution Calendar Year is the Participant's spouse, the quotient obtained by dividing the Participant's Account Balance by the number in the Joint and Last Survivor Table set forth in Treasury Regulations section 1.401(a)(9)-9, Q&A-3, using the Participant's and spouse's attained ages as of the Participant's and spouse's birthdays in the Distribution Calendar Year.
- (e) Lifetime Required Minimum Distributions Continue Through Year of Participant's Death. Required Minimum Distributions will be determined under Subsection 2.5.7(d) beginning with the first Distribution Calendar Year and continuing up to, and including, the Distribution Calendar Year that includes the Participant's date of death.
- (f) Required Minimum Distributions On or After Date Required Distributions Begin.
 - (1) Participant Survived by Designated Beneficiary. If the Participant dies on or after the date required distributions begin and there is a Designated Beneficiary, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the longer of the remaining Life Expectancy of the Participant or the remaining Life Expectancy of the Participant's Designated Beneficiary, determined as follows:
 - (A) The Participant's remaining Life Expectancy is calculated in accordance with the Single Life Table found in Treasury Regulations section 1.401(a)(9)-9, Q&A-1, using the age of the Participant in the year of death, reduced by one for each subsequent year.
 - (B) If the Participant's surviving spouse is the Participant's sole Designated Beneficiary, the remaining Life Expectancy of the surviving spouse is calculated using the Single Life Table found in Treasury Regulations section 1.401(a)(9)-9, Q&A-1, for each Distribution Calendar Year after the year of the Participant's death using the surviving spouse's age as of the spouse's birthday in that year. For Distribution Calendar Years after the year of the surviving spouse's death, the remaining Life Expectancy of the surviving spouse is calculated using the age of the surviving spouse as of the spouse's birthday in the calendar year of the spouse's death, reduced by one for each subsequent calendar year.
 - (C) If the Participant's surviving spouse is not the Participant's sole Designated Beneficiary, the Designated Beneficiary's remaining Life Expectancy is calculated under the Single Life Table using the age of the Beneficiary in the year following the year of the Participant's death, reduced by one for each subsequent year.
 - (2) No Designated Beneficiary Survives. If the Participant dies on or after the date required distributions begin and there is no Designated Beneficiary as of the Participant's date of death who remains a beneficiary as of September 30 of the year after the year of the Participant's death, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account balance by the Participant's remaining Life Expectancy under the Single Life Table calculated using the age of the Participant in the year of death, reduced by one for each subsequent year.
- (g) Distribution Amount If Participant Dies Before Date Required Distributions Begin.
 - (1) Participant Survived by Designated Beneficiary. If the Participant dies before the date required distributions begin and there is a Designated Beneficiary, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the remaining Life Expectancy of the Participant's Designated Beneficiary, determined as provided in Subsection 2.5.7(f). If permitted in the Adoption Agreement, a Participant or Designated Beneficiary may elect to receive distributions over the 5-year period described in Subsection 2.5.7(b)(3).
 - (2) No Designated Beneficiary Survives. If the Participant dies before the date distributions begin and all Designated Beneficiaries, if any, as of the Participant's date of death are no longer

Beneficiaries as of September 30 of the year following the year of the Participant's death, distribution of the Participant's entire interest must be completed by December 31 of the calendar year containing the 5th anniversary of the Participant's death.

- (3) Death of Surviving Spouse Before Distributions to Surviving Spouse Are Required to Begin. If all of the following are true, this Subsection 2.5.7(g) will apply as if the surviving spouse were the Participant:
 - (A) The Participant dies before the date distributions begin,
 - (B) The Participant's surviving spouse is the Participant's sole Designated Beneficiary, and
 - **(C)** The surviving spouse dies before distributions are required to begin to the surviving spouse under Subsection 2.5.4(b)(1).

(h) Asset Rollovers and Transfers.

If a Participant requests a rollover of their benefits to a qualified plan not maintained by the Employer during their first or subsequent Distribution Calendar Year, or a Participant's benefits are transferred to another qualified plan, such rollover or transfer will be reduced by the Required Minimum Distribution for the Distribution Calendar Year under Code section 401(a)(9) and the reduction will be distributed to the Participant by the appropriate date.

If the Plan receives a rollover or transfer with respect to a Participant who has reached their Required Beginning Date, the Plan will distribute the Required Minimum Distribution under Code section 401(a)(9) beginning with the following calendar year. In the event that the beneficiary or method of distribution with respect to transferred assets differs from the Beneficiary or method of distribution of this Plan, the Plan will segregate the transferred assets and continue to distribute the benefits from the originating plan in accordance with the method of distribution and Life Expectancy of the originating plan.

(i) Definitions.

- (1) Distribution Calendar Year. A calendar year for which a minimum distribution is required. For distributions beginning before the Participant's death, the first Distribution Calendar Year is the calendar year immediately preceding the calendar year that contains the Participant's Required Beginning Date. For distributions beginning after the Participant's death, the first Distribution Calendar Year is the calendar year in which distributions are required to begin under Subsection 2.5.7(b)(1). The Required Minimum Distribution for the Participant's first Distribution Calendar Year will be made on or before the Participant's Required Beginning Date. The Required Minimum Distribution for the Distribution Calendar Years, including the Required Beginning Date occurs, will be made on or before December 31 of that Distribution Calendar Year.
- (2) Life Expectancy. Life Expectancy as computed by use of one of the following tables, found in Treasury Regulations section 1.401(a)(9)-9, as appropriate:
 - (A) Single Life Table;
 - (B) Uniform Life Table; or
 - (C) Joint and Last Survivor Table
- (3) Participant's Account Balance. (For purposes of Required Minimum Distributions only.)
 - (A) The calendar year immediately preceding the Distribution Calendar Year is the "valuation calendar year." The Participant's Account Balance as of the last Valuation Date in the valuation calendar year is increased by the amount of any contributions made and allocated

or Forfeitures allocated to the Participant's Account Balance as of dates in the valuation calendar year after the Valuation Date and decreased by distributions made in the valuation calendar year after the Valuation Date. If an amount is distributed or transferred from another source during the valuation calendar year, that amount is included in the Participant's Account Balance for the valuation calendar year regardless if such amount is transferred or rolled over to the Plan in the valuation calendar year or in the Distribution Calendar Year.

(B) Exception for second Distribution Calendar Year. For purposes of Subsection (i) above, if any portion of the minimum distribution for the first Distribution Calendar Year is made in the second Distribution Calendar Year on or before the Required Beginning Date, the amount of the minimum distribution made in the second Distribution Calendar Year will be treated as if it had been made in the immediately preceding Distribution Calendar Year.

(4) Required Beginning Date.

- (A) General rule. The Required Beginning Date of a Participant is the first day of April of the calendar year following the calendar year in which the Participant attains age 70 1/2.
- **(B) Non-5-Percent Owner Exception.** If so elected in the Adoption Agreement, the Required Beginning Date of a Participant who is not a 5-Percent Owner will be the first day of April of the calendar year following the later of the calendar year in which the Participant retires or the calendar year in which the Participant attains age 70 1/2.

(C) Transition Rule:

- (A) If previously elected by the Employer, any Participant who is not a 5-Percent Owner and attains age 70 1/2 in years after 1995 may elect by April 1 of the calendar year following the year in which the Participant attained age 70 1/2, (or by December 31, 1997 in the case of a Participant attaining age 70 1/2 in 1996) to defer distributions until the calendar year following the calendar year in which the Participant retires. If no such election is made the Participant who is not a 5-Percent Owner will begin receiving distributions by April 1 of the calendar year following the year in which the Participant attained age 70 1/2 (or by December 31, 1997 in the case of a Participant attaining age 70 1/2 in 1996).
- **(B)** If previously elected by the Employer, any Participant who attained age 70 1/2 in years prior to 1997 but not earlier than January 1, 1996, was permitted to elect to stop distributions and recommence by the April 1 of the calendar year following the year in which the Participant retires. An Employer who adopted this option selected either:
 - (1) A new Annuity Starting Date upon recommencement, or
 - (2) No new Annuity Starting Date upon recommencement.
- (C) If previously elected by the Employer, the requirement that distributions begin no later than age 70 1/2 only eliminated with respect to Employees who reach age 70 1/2 in or after a calendar year that begins after the later of December 31, 1998, or the date that the Employer adopted the GUST Restatement of this Plan.
- (D) 5-Percent Owner. A Participant is treated as a 5-Percent Owner for purposes of this Section if such Participant is a 5-Percent Owner as defined in Part I, Article II at any time during the Plan Year ending with or within the calendar year in which such owner attains age 70 1/2. Once distributions have begun to a 5-Percent Owner under this Section, they must continue to be distributed, even if the Participant ceases to be a 5-Percent Owner in a subsequent year.

(j) TEFRA Section 242(b)(2) Election.

- (1) Notwithstanding the other requirements of this Article and subject to the requirements of Sections 2.5.4 and 2.5.6, distribution on behalf of any Employee, including a 5-Percent Owner, may be made in accordance with all of the following requirements (regardless of when such distribution commences):
 - (A) The distribution by the Plan is one that would not have disqualified the Plan under Code section 401(a)(9) as in effect prior to amendment by the Deficit Reduction Act of 1984.
 - **(B)** The distribution is in accordance with a method of distribution designated by the Employee whose interest in the Plan is being distributed or, if the Employee is deceased, by a Beneficiary of such Employee.
 - **(C)** Such designation was in writing, was signed by the Employee or the Beneficiary, and was made before January 1, 1984.
 - (D) The Employee had an Account balance under the Plan as of December 31, 1983.
 - **(E)** The method of distribution designated by the Employee or the Beneficiary specifies the time at which distribution will commence, the period over which distributions will be made, and in the case of any distribution upon the Employee's death, the Beneficiaries of the Employee listed in order of priority.
- (2) A distribution upon death will not be covered by this transitional rule unless the information in the designation contains the required information described above with respect to the distributions to be made upon the death of the Employee.
- (3) For any distribution that commences before January 1, 1984, but continues after December 31, 1983, the Employee or the Beneficiary to whom such distribution is being made will be presumed to have designated the method of distribution under which the distribution is being made if the method of distribution was specified in writing and the distribution satisfies the requirements in Subsections 2.5.7(j)(1)(A) and (E).
- (4) If a designation is revoked, any subsequent distribution must satisfy the requirements of Code section 401(a)(9) and the applicable regulations. If a designation is revoked subsequent to the date distributions to a Participant were required to begin, the Plan must distribute, by the end of the calendar year following the calendar year in which the revocation occurs, the total amount not yet distributed that would have been required to have been distributed to the Participant to satisfy Code section 401(a)(9) and the applicable regulations, but for the TEFRA section 242(b)(2) election. For calendar years beginning after December 31, 1988, such distributions must meet the minimum distribution incidental benefit requirements. Any changes in the designation will be considered to be a revocation of the designation. However, the mere substitution or addition of another Beneficiary (one not named in the designation) under the designation will not be considered to be a revocation of the designation, so long as such substitution or addition does not alter the period over which distributions are to be made under the designation, directly or indirectly (for example, by altering the relevant measuring life).
- (5) If an amount is transferred or rolled over from one plan to another plan, the rules in Treasury Regulations section 1.401(a)(9)-8, Q&A 14 and Q&A 15 will apply.

2.5.8 Nature of Distributions.

(a) Trust Fund and Segregated Funds. Except as provided in Subsection (b) with regard to Life Insurance Policies and the Qualified Joint and Survivor Annuity requirements in Section 2.5.6, distribution of a Participant's Distributable Benefit will consist of cash, property, or an annuity contract, or a combination thereof. If a distribution includes property, the property will be valued at its fair market value as of the date of distribution.

- **(b) Insurance Policies.** In the event that the Trustee has purchased one or more Life Insurance Policies on the life of the Participant, or someone in whom the Participant has an insurable interest, the values and benefits available with respect to each such Policy will be distributed as follows:
 - (1) If the Participant's employment terminates for any reason other than death, the Trustee, at the election of the Participant, will either (i) surrender the Life Insurance Policy for its available cash value and distribute the proceeds as provided in Subsection (a) above or (ii) distribute the Life Insurance Policy to the Participant, provided the Participant's vested Account balance is at least equal to the policy's cash value.
 - (2) If the Participant dies, the Participant's Beneficiary will be entitled to the full amount of the policy proceeds.

The Trustee will apply for and be the owner of any policy purchased under the terms of the Plan. The Policy must provide that the proceeds are payable to the Trustee subject to the Trustee's obligation to pay over the proceeds to the Participant's Beneficiary. A Participant's spouse will be the Designated Beneficiary of the proceeds of such Policy unless a qualified election has been made in accordance with Section 2.5.6 of the Plan, if applicable. Under no circumstances will the Trust retain any part of the proceeds. In the event of any conflict between the terms of the Plan and the terms of any policy purchased hereunder, the Plan provisions will control.

- (3) In the event the Trustee purchases a Life Insurance Policy on the life of someone other than the Participant pursuant to this Plan, and such other person will die, the Participant may withdraw the life insurance proceeds to the extent they exceed the policy's cash value immediately prior to such death.
- **2.5.9 Qualified Military Service.** Distributions to Individuals Called to Active Duty. Any individual who receives a Qualified Reservist Distribution may, at any time during the 2 year period beginning on the day after the end of the active duty period, make one or more repayment contributions to an Individual Retirement Plan of such individual in an aggregate amount not to exceed the amount of such distribution. The dollar limitations otherwise applicable to contributions to Individual Retirement Plans will not apply to any contribution made pursuant to the preceding sentence. No deduction will be allowed for any contribution pursuant to this Section. For purposes of this Subparagraph, the term "Qualified Reservist Distribution" means any distribution to an individual if:
- (a) such distribution is from an Individual Retirement Plan, or from amounts attributable to elective deferrals under a 401(k) plan, 403(b) annuity contract, or certain similar arrangements;
- (b) such individual was (by reason of being a member of a reserve component (as defined in section 101 of title 37, United States Code) was ordered or called to active duty for a period in excess of 179 days or for an indefinite period; and
- (c) such distribution is made during the period beginning on the date of such order or call and ending at the close of the active duty period.
- (d) This Paragraph 2.5.9 is effective August 17, 2001, and applies to individuals ordered or called to active duty after September 11, 2001, and before December 31, 2007. Effective June 17, 2008, it also applies to individuals ordered or recalled to active duty after December 31, 2007.
- **2.5.10** Hardship Distributions of Elective Deferrals. If the Plan is designated in the Adoption Agreement as a Cash or Deferred Profit Sharing Plan, the Employer may elect in the Adoption Agreement to permit Hardship Distributions of Elective Deferrals. A Hardship Distribution is a distribution that meets the requirements of a distribution on account of immediate and heavy financial need enumerated in this Section 2.5.10. A Participant may request a distribution from the Plan as a result of immediate and heavy financial needs of the Participant to the extent that the distribution is necessary to satisfy such financial needs. The availability of Hardship Distributions to Participants is limited to Participants who have not terminated employment. Effective August 17, 2006, the Plan may permit a Participant to request a Hardship

Distribution where the Hardship or unforeseeable financial emergency occurs to the Participant's spouse, dependents, or Primary Beneficiary.

Hardship distributions are subject to the spousal consent requirements contained in Code sections 401(a)(11) and 417 unless the Plan meets the exception of Subsection 2.5.5(c). The Plan Administrator will determine whether a Participant has an immediate and heavy financial need. A distribution will be deemed to be made on account of an immediate and heavy financial need if the distribution is due to:

- (a) Deductible medical expenses described in Code section 213(d) incurred or necessary for medical care of the Participant, their spouse, dependents, or Primary Beneficiary under the Plan;
- (b) Purchase (excluding mortgage payments) of a principal residence for the Participant;
- (c) Cost of tuition and related educational fees and room and board expenses for the next 12 months of post-secondary education for the Participant, their spouse, children, Primary Beneficiary or dependents; or
- (d) The need to prevent the eviction of the Participant from their principal residence or foreclosure on the mortgage of the Participant's principal residence.
- (e) Payments for funeral or burial expenses for the Participant's deceased parent, spouse, child, dependent or Primary Beneficiary under the Plan. (For Plan Years beginning after 2005.)
- (f) Expenses to repair damage to the Participant's principal residence that would qualify for a casualty loss deduction under Code section 165 (determined without regard to whether the loss exceeds 10% of adjusted gross income). (For Plan Years beginning after 2005.)
 - A distribution will be considered as necessary to satisfy an immediate and heavy financial need of the Participant only if:
- **(g)** The Participant has obtained all distributions, other than Hardship Distributions, and all nontaxable loans under all plans maintained by the Employer;
- (h) All plans maintained by the Employer provide that the Participant's Elective Deferrals and Voluntary Employee Contributions will be suspended for 6 months after the receipt of the Hardship Distribution; and
- (i) The distribution is not in excess of the amount of an immediate and heavy financial need (including amounts necessary to pay any federal, state, or local income taxes or penalties reasonably anticipated to result from the distribution).
- **2.5.11 Distributions in the Event of Incapacity.** In the sole judgment of the Plan Administrator, if any Beneficiary entitled to benefits under the Plan is not capable of managing the benefit about to be distributed, such distribution may be made to legal representative of that Beneficiary, or in the case of a minor Beneficiary the guardian, under a valid power of attorney, a court appointed guardian or any other person authorized under state law to receive the benefit. The Plan Administrator may require an order of a court of competent jurisdiction before making such distribution.

2.5.12 In Service Distributions.

(a) Cash or Deferred Profit Sharing Plans. If the Plan is designated in the Adoption Agreement as a Cash or Deferred Profit Sharing Plan, the Employer may elect to permit distributions of Elective Deferrals, Qualified Non-Elective Contributions, Qualified Matching Contributions, and ADP Test Safe Harbor Contributions to a Participant after attaining age 59 1/2 but only prior to their termination of employment. A Participant will be entitled to receive a distribution of all or a part of these Accounts upon filing a written request with the Plan Administrator; provided, that no distribution will be made unless the balance in the Account to be distributed has accumulated for at least 2 years or the individual has been a Participant for 5 or more Plan Years or on account of Hardship unless otherwise

elected in the Adoption Agreement; and, the distribution of Elective Deferrals, Qualified Non-Elective Contributions, Matching Contributions (to the extent used in the ADP Test), Qualified Matching Contributions and ADP Test Safe Harbor Contributions must satisfy the limitations imposed by Section 2.7.8. In-service distributions due to Hardship must meet the Hardship requirements of Section 2.5.10.

The Adoption Agreement may provide that In-service Distributions are available before age 59 1/2 for Accounts other than Elective Deferrals, Qualified Non-Elective Contributions, Qualified Matching Contributions, Matching Contributions that are used to pass the ADP test, and ADP Test Safe Harbor Contributions but only prior to a Participant's termination of employment. In-Service Distributions from the Profit Sharing portion of a Cash or Deferred Profit Sharing Plan will be governed by Subsection (b) below.

- (b) Profit Sharing Plans. If the Plan is designated in the Adoption Agreement as a Profit Sharing Plan, (including the Profit Sharing portion of a Cash or Deferred Profit Sharing Plan), the Employer may elect in the Adoption Agreement to permit distributions to a Participant prior to their termination of employment. A Participant will be entitled to receive a distribution of all or part of their interest in the Plan upon filing a written request with the Plan Administrator; provided, that no distribution will be made unless the amount to be distributed to the Participant is fully vested and nonforfeitable (unless otherwise elected in the Adoption Agreement) and has accumulated for at least 2 years or the individual has been a Participant for 5 or more Plan Years, or on account of Hardship; provided, further that In-service distributions on account of Hardship will be subject to the restrictions of Section 2.5.10. In-service Distributions are subject to the spousal consent requirements contained in Code sections 401(a)(11) and 417 unless the Plan meets the exception of Section 2.5.5(c).
- **(c) Money Purchase Plans.** The Employer may elect in the Adoption Agreement to permit In-service distributions upon a Participant reaching their Normal Retirement Age or an age that may not be earlier than age 62.
- (d) All Plans. The Employer may elect in the Adoption Agreement to permit In-service Distributions at the election of the Participant for amounts held in a Rollover Account, in a Deductible Voluntary Account, or in a Voluntary Account regardless of age or periods of participation.
- (e) **Spousal Consent.** In-service distributions are subject to the spousal consent requirements contained in Code sections 401(a)(11) and 417 unless the Plan meets the exception of Subsection 2.5.5(c).
- (f) If the Plan is a Cash or Deferred Profit Sharing Plan permitting a Participant to contribute all or a portion of their Elective Deferrals to the Trust as designated Roth Deferrals (an after-tax deferral); and also contains distribution options that permit a Participant (active or former Participants with an Account Balance in the Plan) to request an Eligible Rollover Distribution from the Plan, the Employer may elect in the Adoption Agreement to include in the Plan an in-plan Roth conversion feature.

Notwithstanding, the Employer will not amend the Plan to restrict or eliminate any distribution rights that a Participant is entitled to under the Plan (such as a right to an Immediate Distribution of the amount rolled over or right to an In-service Distribution existing under the terms of the Plan).

(1) Conversion. The in-plan Roth conversion will be made through a Direct Rollover election. Under the in-plan Roth conversion feature, all or a portion of an Eligible Rollover Distribution from a nondesignated Roth Account in this Plan is directly rolled into an In-plan Roth Rollover Conversion Account. An In-Plan Roth Conversion Account may be established for Eligible Rollover Distributions made on or after September 27, 2010.

Unless otherwise elected in the Adoption Agreement, any Eligible Rollover Distribution made under the Plan's In-service Distribution options that is otherwise eligible for a Direct Rollover election may be converted under the in-plan Roth conversion feature.

(2) Applicable Definitions.

- (i) In-plan Roth conversion means the Direct Rollover of amounts from a non-designated Roth Account without removing the funds from the Plan into an In-plan Roth Conversion Account.
- (ii) In-plan Roth Conversion Account means a separate Participant subaccount within the Plan that is treated as a designated Roth account under section 402A of the Code.
- (iii) Non-designated Roth Account means any subaccount within the Plan that contains pre-tax contributions or earnings.

ARTICLE VI

CONTINGENT TOP HEAVY PROVISIONS

- **2.6.1 Top-Heavy Requirements.** If the Plan becomes a Top-Heavy Plan during any Plan Year, the following provisions will supersede any conflicting provisions in the Plan or Adoption Agreement and apply for such Plan Year:
- (a) Except as otherwise provided below, the Employer Contributions and Forfeitures allocated on behalf of any Participant who is not a Key Employee will not be less than the lesser of 3% of such Participant's Compensation, as defined in Section 2.6.2(b), or in the case where the Employer has no defined benefit plan that designates this Plan to satisfy Code section 401, the largest percentage of Employer Contributions and Forfeitures (not including Catch-up Contributions), as a percentage of the Key Employee's Compensation as defined in Section 2.6.2(b), and limited by Code section 401(a)(17), allocated on behalf of any Key Employee for that year. The Minimum Top-Heavy Allocation is determined without regard to any Social Security contribution. This Minimum Top-Heavy Allocation will be made even though, under other Plan provisions, the Participant would not otherwise be entitled to receive an allocation, or would have received a lesser allocation for the year because of (i) the Participant's failure to complete 1,000 Hours of Service (or any equivalent provided in the Plan), or (ii) the Participant's failure to make mandatory Employee contributions to the Plan, or (iii) compensation less than a stated amount. The Employer may elect in the Adoption Agreement to have this provision provide Key Employees with the Top-Heavy minimum benefit.

Elective Deferrals (and, for Plan Years beginning before 2002, Matching Contributions) may not be used to satisfy the Minimum Top-Heavy Allocation.

For purposes of computing the Minimum Top-Heavy Allocation, Compensation will mean a Participant's Compensation as defined in the Adoption Agreement.

The Minimum Top-Heavy Allocation provided above will not apply to any Participant who was not employed by the Employer on the last day of the Plan Year.

If so elected in the Adoption Agreement, the Minimum Top-Heavy Allocation provided above will not apply to any Participant who is in a unit of Employees covered by a collective bargaining agreement between the Employer and Employee representatives, provided benefits were the subject of good faith bargaining and 2% or less of the Employees of the Employer who are covered pursuant to that agreement are professionals as defined in Treasury Regulations section 1.410(b)-9(g).

The Minimum Top-Heavy Allocation provided above will not apply to any Participant to the extent the Participant is covered under any other plan or plans of the Employer and the Employer has provided in the Adoption Agreement that the Minimum Top-Heavy Allocation or benefit requirement applicable to Top-Heavy plans will be met in the other plan or plans and the Participant receives the minimum allocation or benefit under such plan or plans.

Effective for Plan Years beginning after December 31, 2007 if contributions consist solely of an "Automatic Contribution Arrangement" or section 401(k)(12) of the Code plan with matching contributions complying with section 401(m)(11) of the Code the Plan will not be considered Top-Heavy.

(b) The vested and nonforfeitable interest of each Participant will be equal to the percentage determined under the vesting schedule specified in the Adoption Agreement if the Plan becomes a Top-Heavy Plan, or if no vesting schedule is specified, the percentage determined under the following schedule:

Years of Vesting Service	Percentage
Less than 2	0%
2	20%
3	40%
4	60%
5	80%
6	or more 100%

The Top-Heavy minimum vesting schedule applies to all benefits within the meaning of Code section 411(a)(7), except those attributable to Employee Contributions, including benefits accrued before the effective date of Code section 416 and benefits accrued before the Plan becomes Top-Heavy.

No decrease in a Participant's nonforfeitable percentage may occur in the event the Plan's status as Top-Heavy changes for any Plan Year. Any Minimum Top-Heavy Allocation required (to the extent required to be nonforfeitable under Code section 416(b)) may not be forfeited under Code sections 411(a)(3)(B) or (D).

- 2.6.2 Top-Heavy Definitions. The following terms, as used in this Plan, will have the following meaning:
- (a) "Aggregation Group": Each qualified retirement plan of the Employer in which a Key Employee is a participant and each other qualified retirement plan of the Employer that enables any plan in which a Key Employee is a participant to meet the requirements of Code sections 401(a)(4) or 410.
- **(b)** "Annual Compensation": One of the following 3 definitions of Compensation as elected in the Adoption Agreement:
 - (1) W-2 Wages. Information required to be reported under Code sections 6041, 6051, and 6052 (wages, tips and other compensation as reported on Form W-2). Compensation is defined as wages within the meaning of Code section 3401(a) and all other payments of compensation to an Employee by the Employer (in the course of the Employer's trade or business) for which the employer is required to furnish the Employee a written statement under Code sections 6041(d), 6051(a)(3), and 6052. Compensation must be determined without regard to any rules under section 3401(a) that limit the remuneration included in wages based on the nature or location of the employment or the services performed (such as the exception for agricultural labor in Code section 3401(a)(2)).
 - (2) Section 3401(a) wages. Compensation is defined as wages within the meaning of Code section 3401(a) for the purposes of income tax withholding at the source but determined without regard to any rules that limit the remuneration included in wages based on the nature or location of the employment or the services performed (such as the exception for agricultural labor in section Code 3401(a)(2)).
 - (3) 415 Simplified Compensation. Compensation is defined as wages, salaries, and fees for professional services and other amounts received (without regard to whether or not an amount is paid in cash) for personal services actually rendered in the course of employment with the Employer maintaining the Plan to the extent that the amounts are includable in gross income (including, but not limited to, commissions paid salesmen, compensation for services on the basis of a percentage of profits, commissions on insurance premiums, tips, bonuses, fringe benefits, and reimbursements or other expense allowances under a non-accountable plan (as described in Treasury Regulation 1.62-2(c)), and excluding the following:
 - (A) Employer contributions to a plan of deferred compensation which are not includible in the Employee's gross income for the taxable year in which contributed, or Employer contributions under a simplified employee pension plan, or any distributions from a plan of deferred compensation;

- **(B)** Amounts realized from the exercise of a nonqualified stock option, or when restricted stock (or property) held by the employee either becomes freely transferable or is no longer subject to a substantial risk of forfeiture;
- **(C)** Amounts realized from the sale, exchange or other disposition of stock acquired under a qualified stock option; and
- (D) Other amounts which received special tax benefits, or contributions made by the Employer (whether or not under a salary reduction agreement) towards the purchase of an annuity contract described in Code section 403(b) (whether or not the contributions are actually excludable from the gross income of the Employee).

For any Self-Employed Individual, Compensation will mean Earned Income.

For purposes of applying the limitations of this Article, Compensation paid or made available during such limitation year will include any Elective Deferral (as defined in Code section 402(e)(3)), and any amount which is contributed or deferred by the Employer at the election of the Employee and which is not includible in the gross income of the Employee by reason of Code sections 125, 132(f)(4), 402(k), 402(h)(1)(B) or 457.

If elected by the Employer in the Adoption Agreement, amounts under Code section 125 include any amounts not available to a Participant in cash in lieu of group health coverage because the Participant is unable to certify that they have other health coverage (deemed Code section 125 Compensation). An amount will be treated as an amount under Code section 125 only if the Employer does not request or collect information regarding the Participant's other health coverage as part of the enrollment process for the health plan.

- (c) "Top-Heavy Plan": For any Plan Year beginning after December 31, 1983, the Plan is Top-Heavy if any of the following conditions exists:
 - (1) If the Top-Heavy Ratio for the Plan exceeds 60% and the Plan is not part of any Required Aggregation Group or Permissive Aggregation Group of plans.
 - (2) If the Plan is a part of a Required Aggregation Group of plans but not part of a Permissive Aggregation Group and the Top-Heavy Ratio for the group of plans exceeds 60%.
 - (3) If the Plan is a part of a Required Aggregation Group and part of a Permissive Aggregation Group of plans and the Top-Heavy Ratio for the Permissive Aggregation Group exceeds 60%.

SIMPLE 401(k) plans and certain Safe Harbor CODA plans and Qualified Automatic Contribution Arrangements are not subject to the Top-heavy requirements of this Article VI.

(d) "Top-Heavy Ratio":

(1) If the Employer maintains one or more defined contribution plans (including any simplified Employee pension plan) and the Employer has not maintained any defined benefit plan that during the 1 year period ending on the Determination Date(s) has or has had accrued benefits, the Top-Heavy Ratio for this Plan alone or for the Required or Permissive Aggregation Group as appropriate is a fraction, the numerator of which is the sum of the Account balances of all Key Employees as of the Determination Date(s) (including any part of any Account balance distributed in the 1 year period ending on the Determination Date(s) (five (5) year period ending on the Determination Date in the case of a distribution made for a reason other than severance from employment, death, or Disability)), and the denominator of which is the sum of all Account balances (including any part of any Account balance distributed in the 1 year period ending on the Determination Date(s) (five (5) year period ending on the Determination Date in the case of a distribution made for a reason other than severance from employment, death, or Disability)), both computed in accordance with Code section 416 and the regulations thereunder.

Both the numerator and denominator of the Top-Heavy Ratio are increased to reflect any contribution not actually made as of the Determination Date, but that is required to be taken into account on that date under Code section 416 and the regulations thereunder.

- (2) If the Employer maintains one or more defined contribution plans (including any simplified Employee pension plan) and the Employer maintains or has maintained one or more defined benefit plans that during the 1 year period ending on the Determination Date(s) has or has had any accrued benefits, the Top-Heavy Ratio for any Required or Permissive Aggregation Group as appropriate is a fraction, the numerator of which is the sum of Account balances under the aggregated defined contribution plan or plans for all Key Employees, determined in accordance with (i) above, and the present value of accrued benefits under the aggregated defined benefit plan or plans for all Key Employees as of the Determination Date(s), and the denominator of which is the sum of the Account balances under the aggregated defined contribution plan or plans for all Participants, determined in accordance with (i) above, and the present value of accrued benefits under the defined benefit plan or plans for all Participants as of the Determination Date(s), all determined in accordance with Code section 416 and the regulations thereunder. The accrued benefits under a defined benefit plan in both the numerator and denominator of the Top-Heavy Ratio are increased for any distribution of an accrued benefit made in the 1 year period ending on the Determination Date (five (5) year period ending on the Determination Date in the case of a distribution made for a reason other than severance from employment, death or Disability beginning before January 1, 2002.)
- (3) For purposes of (1) and (2) above, the value of Account balances and the present value of accrued benefits will be determined as of the most recent that falls within or ends with the 12 month period ending on the Determination Date, except as provided in Code section 416 and the regulations thereunder for the first and second Plan Years of a defined benefit plan. The Account balances and accrued benefits of a Participant: (1) who is not a Key Employee but was a Key Employee in a prior year, or (2) who has not been credited with at least 1 Hour of Service with any Employer maintaining the Plan at any time during the 1 year period ending on the Determination Date will be disregarded. The calculation of the Top-Heavy Ratio and the extent to which distributions, rollovers, and transfers are taken into account will be made in accordance with Code section 416 and the regulations thereunder. Deductible Employee contributions will not be taken into account for purposes of computing the Top-Heavy Ratio, however, Catch-up Contributions will be taken into account. When aggregating plans, the value of Account balances and accrued benefits will be calculated with reference to the Determination Dates that fall within the same calendar year.

The accrued benefit of a Participant other than a Key Employee will be determined under: (a) the method, if any, that uniformly applies for accrual purposes under all defined benefit plans maintained by the Employer, or (b) if there is no such method, as if such benefit accrued not more rapidly than the slowest accrual rate permitted under the fractional rule of Code section 411(b)(1)(C).

(e) "Permissive Aggregation Group": The Required Aggregation Group of plans plus any other plan or plans of the Employer that, when considered as a group with the Required Aggregation Group, would continue to satisfy the requirements of Code sections 401(a)(4) and 410.

(f) "Required Aggregation Group":

- (1) Each qualified plan of the Employer in which at least one Key Employee participates or participated at any time during the Plan Year containing the Determination Date (regardless of whether the plan has terminated).
- (2) Any other qualified plan of the Employer that enables a plan described in (i) to meet the requirements of Code sections 401(a)(4) or 410.
- **(g)** "Determination Date": For any Plan Year subsequent to the first Plan Year, the last day of the preceding Plan Year. For the first Plan Year of the Plan, the last day of that year.

- (h) "Valuation Date": The date elected by the Employer in the Adoption Agreement as of which account balances or accrued benefits are valued for purposes of calculating the Top-Heavy Ratio.
- (i) "Present Value": Present value will be based only on the interest and mortality rates specified in the Adoption Agreement.

ARTICLE VII

SPECIAL CODA LIMITATIONS

2.7.1 Limitation on Deferral Percentage for Highly Compensated Employees.

- (a) Prior Year Testing. The Employer may elect in the Adoption Agreement to use the current year ADP for all eligible Highly Compensated Employees must not exceed the greater of the prior year's ADP for all Participants who were Non-Highly Compensated Employees in such year multiplied by:
 - (1) 1.25; or
 - (2) 2, but provided the sum is not more than 2 percentage points greater than the prior Plan Year ADP of such Non-Highly Compensated Employees.

The Employer may switch from Prior Year Testing to Current Year testing for any Plan Year.

For the first Plan Year that the Plan permits any Participant to make Elective Deferrals and this is not a successor plan, the ADP of Non-Highly Compensated Employees for the prior Plan Year will be deemed to be 3% or, at the Employer's election, the test will use actual data for the first Plan Year to determine the ADP of eligible Non-Highly Compensated Employees.

- **(b) Current Year Testing.** The Employer may elect in the Adoption Agreement to use the current Plan Year ADP for all eligible Non-Highly Compensated Employees rather than the ADP for the prior Plan Year. Once the Employer has made this election, it may change to Prior Year Testing only if the Plan meets the requirements of Treasury Regulations section 1.401(k)-2(c)(1)(ii), including that:
 - (1) The Plan has used Current Year Testing for each of the preceding 5 Plan Years or, if lesser, the number of Plan Years that the Plan has been in existence; or
 - (2) As a result of a merger or acquisition described in Code section 410(b)(6)(C)(i), the Employer maintains both a plan using Prior Year Testing and a plan using Current Year Testing, and the change is made within the transition period described in Code section 410(b)(6)(C)(ii).
- (c) Other Rules. An Employee who would be a Participant but for the failure to have Elective Deferrals made on their behalf will be treated as a Participant on whose behalf no Elective Deferrals are made. For purposes of the ADP test compensation may be compensation for the entire Plan Year whether or not the Participant was a Participant for the entire Plan Year or compensation from a Participant's Plan Entry Date determined each year by the Plan Administrator and applied to all eligible Participants in a uniform and non-discriminatory manner.

For purposes of determining the ADP test, Elective Deferrals (other than Catch-up Contributions), Qualified Non-Elective Contributions, and Qualified Matching Contributions must be made before the last day of the 12 month period immediately following the Plan Year to which the contributions relate. For any Plan Year in which Qualified Non-Elective Contributions, and Qualified Matching Contributions are made the Plan must use the Current Year testing method.

Pursuant to Code section 410(b)(4)(B), the Employer may elect to treat that portion of the Plan that benefits only Participants who satisfy age and service conditions under the Plan that are lower than the greatest minimum age and service conditions permitted under Code section 410(a) ("Otherwise Excludable Employees") and that portion of the Plan that benefits Participants that meet the greatest minimum age and service conditions permitted under Code section 410(a) ("Not Otherwise Excludable Employees") as 2 separate plans for purposes of Code section 401(k), and the ADP test need not be satisfied with respect to both plans in order for one of the plans to pass the ADP test. An Employee's membership in the class of Otherwise Excludible Employees will be determined using either the Plan's

Entry Date provision or the earlier of the first day of the next Plan Year of 6 months after completing the greatest minimum age and service conditions of Code section 410(a) as determined each year by the Plan Administrator.

Elective Deferrals that are made as Catch-up Contributions after a period of Qualified Military Service pursuant to Section 2.2.2(a) will not be used to determine a Participant's deferral percentage when the ADP of Participants is calculated for a particular Plan Year.

The determination and treatment of the ADP amounts of any Participant will satisfy such other requirements as may be prescribed by the Secretary of the Treasury.

Use of Qualified Non-Elective Contributions in the ADP or ACP test requires compliance with Reg. 1.401(k)-2(a)(6)(iv) or Reg. 1.401(m)-2(a)(6)(v), respectively, that require the Qualified Non-Elective Contribution be no greater than 5% of a Participant's compensation or no more than twice a "representative contribution rate" determined by analyzing the Qualified Non-Elective Contributions provided to all Non-Highly Compensated Employees.

2.7.2 Limitation on Matching Contributions.

- (a) Prior Year Testing. The Employer may elect in the Adoption Agreement to use the current year ACP for eligible Highly Compensated Employees which must not exceed the greater of the prior year's ACP for all Participants who were Non-Highly Compensated Employees in such year multiplied by:
 - (1) 1.25; or
 - (2) 2, provided the sum is not more than 2 percentage points greater than the prior Plan Year ACP of such Non-Highly Compensated Employees.

The Employer may switch from Prior Year Testing to Current Year Testing for any Plan Year.

For the first Plan Year that the Plan permits any Participant to make Elective Deferrals and this is not a successor plan, the ACP of Non-Highly Compensated Employees for the prior Plan Year will be deemed to be 3% or, at the Employer's election, the test will use actual data of the first Plan Year to determine the ACP for eligible Non-Highly Compensated Employees.

- (b) Current Year Testing. The Employer may elect in the Adoption Agreement to use the current Plan Year ACP for all eligible Non-Highly Compensated Employees rather than the ADP of the prior Plan Year. Once the Employer has made this election, it may change to Prior Year Testing only if the Plan meets the requirements of Treasury Regulations sections 1.401(m)-2(c)(1) and 1.401(k)-2(c)(1)(ii), including that:
 - (1) The Plan has used Current Year Testing for each of the preceding 5 Plan Years or, if lesser, the number of Plan Years that the Plan has been in existence; or
 - (2) As a result of a merger or acquisition described in Code section 410(b)(6)(C)(i), the Employer maintains both a plan using Prior Year Testing and a plan using Current Year Testing, and the change is made within the transition period described in Code section 410(b)(6)(C)(ii).
- (c) Other Rules. If so elected in the Adoption Agreement, the Employer may include Qualified Non-Elective Contributions in the contribution percentage amounts. In order to be taken into account in the calculation of the ACP for a year under Current Year Testing, a Qualified Non-Elective Contribution or Qualified Matching Contribution must be allocated as of a date within the year using current data and must actually be paid to the Trust no later than the end of the 12-month period following the end of the Plan Year to which the contribution relates. The Employer may also elect to use Elective Deferrals in the contribution percentage amounts so long as the ADP test is met before the Elective Deferrals are used in the ACP test and continues to be met following the exclusion of those Elective Deferrals that are used to meet the ACP test. An ACP will be rounded to the nearest hundredth of a percentage point. If an Elective Deferral or other contribution by an Employee is required as a condition of

participation in the Plan, any Employee who would be a Participant if such Employee made such a contribution will be treated as an eligible Participant on behalf of whom no such contributions are made. For purposes of the ACP test compensation may be compensation for the entire Plan Year whether or not the Participant was a Participant for the entire Plan Year or compensation from a Participant's Plan Entry Date determined each year by the Plan Administrator and applied to all eligible Participants in a uniform and non-discriminatory manner.

The amount of Excess Aggregate Contributions for a Plan Year will be determined only after first determining the Excess Contributions that are treated as Voluntary Employee Contributions due to recharacterization.

Pursuant to section 410(b)(4)(B), the Employer may elect to treat that portion of the Plan that benefits only Participants who satisfy age and service conditions under the Plan that are lower than the greatest minimum age and service conditions permitted under Code section 410(a) ("Otherwise Excludable Employees") and that portion of the Plan that benefits Participants that meet the greatest minimum age and service conditions permitted under Code section 410(a) ("Not Otherwise Excludable Employees") as 2 separate plans for purposes of Code section 401(m), and the ACP test need not be satisfied with respect to both plans in order for one of the plans to pass the ACP test. An Employee's membership in the class of Otherwise Excludible Employees will be determined using either the Plan's Entry Date provision or the earlier of the first day of the next Plan Year of 6 months after completing the greatest minimum age and service conditions of Code section 410(a) as determined each year by the Plan Administrator.

Employee contributions are considered to have been made in the Plan Year in which contributed to the Trust. Matching Contributions, Voluntary Employee Contributions, Qualified Non-Elective Contributions, and Qualified Matching Contributions will be considered made for a Plan Year if made no later than the end of the twelve-month period beginning on the day after the close of the Plan Year.

Matching Contributions made on account of Elective Deferrals that are made as Catch-up Contributions after a period of Qualified Military Service pursuant to Section 2.2.2(a) will not be used to determine a Participant's contribution percentage when the ACP of Participants is calculated for a particular Plan Year.

The Employer will maintain records sufficient to demonstrate satisfaction of the ACP test and the amount of Qualified Non-Elective Contributions and Qualified Matching Contributions used in such test.

The determination and treatment of the contribution percentage of any Participant will satisfy such other requirements as may be prescribed by the Secretary of the Treasury.

Qualified Matching Contributions and Qualified Non-Elective Contributions may be used with Current Year testing method only if the contribution is made within the 12-month period following the last day of the Prior Year. Use of Qualified Matching Contributions in the ADP or ACP test requires compliance with Reg. 1.401(k)-2(a)(6)(iv) or Reg. 1.401(m)-2(a)(6)(v), respectively, that require the Qualified Matching Contribution be no greater than 5% of a Participant's compensation or no more than twice a "representative contribution rate" determined by analyzing the Qualified Matching Contributions provided to all Non-Highly Compensated Employees.

2.7.3 Multiple Plan Limitations For ADP and ACP Testing.

- (a) No Participant will be permitted to have Elective Deferrals made under this Plan, or any other qualified plan maintained by the Employer, during any taxable year, in excess of the dollar limitation contained in Code section 402(g)(1) in effect at the beginning of such taxable year.
- (b) The ADP for any Participant who is a Highly Compensated Employee for the Plan Year and who is eligible to have Elective Deferrals (and Qualified Matching Contributions, and Qualified Non-Elective Contributions if treated as Elective Deferrals for purposes of the ADP test) allocated to their Accounts under two or more arrangements described in Code section 401(k) that are maintained by the

Employer, will be determined as if such Elective Deferrals and, if applicable, such Qualified Matching Contributions and Qualified Non-Elective Contributions) were made under a single arrangement. If a Highly Compensated Employee participates in two or more cash or deferred arrangements of the Employer that have different plan years, all Elective Deferrals made during the Plan Year under all such arrangements will be aggregated. Notwithstanding the foregoing, certain plans will be treated as separate if mandatorily disaggregated under regulations under Code section 401(k).

- (c) In the event that this Plan satisfies the requirements of Code sections 401(k), 401(a)(4), or 410(b) only if aggregated with one or more other plans, or if one or more other plans satisfy the requirements of such sections of the Code only if aggregated with this Plan, then this Section will be applied by determining the ADP of Employees as if all such plans were a single plan. If more than 10% of the Employer's Non-Highly Compensated Employees are involved in a plan coverage change as defined in Treasury Regulations section 1.401(k)-2(c)(4), then any adjustments to the Non-Highly Compensated Employee ADP for the prior year will be made in accordance with such Regulations, unless the Employer has elected in the Adoption Agreement to use the Current Year Testing method. Plans may be aggregated in order to satisfy Code section 401(k) only if they have the same Plan Year and use the same ADP testing method.
- (d) The ACP for any Participant who is a Highly Compensated Employee and who is eligible to have contribution percentage amounts allocated to their Accounts under two or more plans described in Code section 401(a), or arrangements described in Code section 401(k), that are maintained by the Employer, will be determined as if the total of such contribution percentage amounts was made under each plan. If a Highly Compensated Employee participates in two or more cash or deferred arrangements that have different plan years, all types of contributions includible in the ACP test from all such arrangements that were made during the Plan Year will be aggregated. Notwithstanding the foregoing, certain plans will be treated as separate if mandatorily disaggregated under regulations under Code section 401(m).
- (e) In the event that this Plan satisfies the requirements of Code sections 401(m), 401(a)(4), or 410(b) only if aggregated with one or more other plans, or if one or more other plans satisfy the requirements of such Code sections only if aggregated with this Plan, then Section 2.7.2 will be applied by determining the ACP of Employees as if all such plans were a single plan. If more than 10% of the Employer's Non-Highly Compensated Employees are involved in a plan coverage change as defined in Treasury Regulations section 1.401(m)-2(c)(4), then any adjustments to the Non-Highly Compensated Employee ACP for the prior year will be made in accordance with such regulations, unless the Employer has elected in the Adoption Agreement to use the Current Year Testing method. Plans may be aggregated in order to satisfy Code section 401(m) only if they have the same Plan Year and use the same ACP testing method.
- **2.7.4 Different Testing Methods for Elective Deferrals and Matching Contributions.** The Plan must use the same testing method for the ADP test under Section 2.7.1 and the ACP test under Section 2.7.2 unless the Employer prohibits the use of Elective Deferrals in the ACP test and Recharacterization of Excess Contributions and the use of Qualified Matching Contributions in the ADP test.
- **2.7.5 Distribution of Excess Elective Deferrals.** A Participant may assign to the Plan any Excess Elective Deferrals made during a taxable year of the Participant by notifying the Plan Administrator on or before March 15th of each calendar year of the amount of the Excess Elective Deferrals to be assigned to the Plan. A Participant is deemed to notify the Plan Administrator of any Excess Elective Deferrals that arise by taking into account only those Elective Deferrals made to this Plan and any other plans of the Employer.

Notwithstanding any other provision of the Plan, Excess Elective Deferrals, plus any income and minus any loss allocable thereto, will be distributed no later than April 15th to any Participant to whose Account Excess Elective Deferrals were assigned for the preceding year and who claims Excess Elective Deferrals for such taxable year. For years beginning after 2005, distribution of Excess Elective Deferrals for a year will be made first from the Participant's Pre-tax Elective Account to the extent Pre-tax Elective Deferrals were made for the year, unless the Participant specifies otherwise. For taxable years beginning before January 1, 2006, income or loss allocable to the period between the end of the taxable year and the date of

distribution (the "gap-period") could be disregarded in determining income or loss on Excess Elective Deferrals for such years. Gap-period income loss must be included in any distribution of Excess Elective Deferrals occurring in taxable years beginning in 2007, but must be excluded from any distribution of Excess Elective Deferrals occurring in taxable years beginning after 2007.

Excess Elective Deferrals distributed under this Section will be adjusted for any income or loss based on a reasonable method of computing the allocable income or loss. The method selected must be applied consistently to all Participants and used for all corrective distributions under the Plan for the Plan Year, and must be the same method that is used by the Plan for allocating income or loss to Participants' Accounts. The income allocable to Excess Deferrals is equal to the allocable gain or loss through the end of the Plan Year for which the Deferrals were made.

One reasonable method treats the income and loss allocable to Excess Elective Deferrals as the sum of:

- (a) Income or loss allocable to the Participant's Elective Account for the taxable year, multiplied by a fraction, the numerator of which is such Participant's Excess Elective Deferrals for the year and the denominator is the Participant's Account balance attributable to Elective Deferrals without regard to any income or loss occurring during such taxable year; and
- (b) 10% of the amount determined under (a) multiplied by the number of whole calendar months between the end of the Participant's taxable year and the date of the distribution, counting the month of distribution if distribution occurs after the 15th of such month. The Gap-period calculation is not applicable after the end of the 2007 Plan Year.
- **2.7.6 Distribution of Excess Contributions.** If elected by the Employer in the Adoption Agreement, if the Plan permits all Participants to make Voluntary Employee Contributions, Elective Deferrals allocated to a Highly Compensated Employee as Excess Contributions will be re-characterized as Voluntary Employee Contributions (after-tax). A Participant must or may, as elected by the Employer in the Adoption Agreement, treat Excess Contributions allocated to them as an amount distributed to the Participant and then contributed by the Participant to the Plan. Re-characterized amounts will remain nonforfeitable. A Highly Compensated Employee may not recharacterize an amount to the extent that such amount in combination with other Voluntary Employee Contributions made by that Employee would exceed any stated limit under the Plan on Voluntary Employee Contributions. Recharacterized Excess Contributions will be reduced by Excess Deferrals previously distributed.

Re-characterization must occur no later than 2 1/2 months after the last day of the Plan Year in which such Excess Contributions arose and is deemed to occur no earlier than the date the last Highly Compensated Employee is informed in writing of the amount re-characterized and the consequences thereof.

If elected in the Adoption Agreement, Excess Contributions that are not recharacterized, plus any income and minus any loss allocable thereto, will be distributed no later than 12 months after the last day of each Plan Year to Highly Compensated Employees to whose Accounts such Excess Contributions were allocated for such Plan Year, except to the extent such Excess Contributions are classified as Catch-up Contributions. For years beginning after 2005, distribution of Excess Contributions for a year will be made first from the Participant's Pre-tax Elective Account to the extent Pre-tax Elective Deferrals were made for the year, unless the Participant specifies otherwise. For Plan years beginning before 2006, income or loss allocable to the period between the end of the Plan Year and the date of distribution (the "gap-period") could be disregarded in determining income or loss on Excess Aggregate Contributions for such years. Gap-period income or loss must be included in any distribution of Excess Aggregate Contributions occurring in Plan Years beginning in 2006 and 2007, but must be excluded from any distribution of Excess Aggregate Contributions occurring in Plan Years beginning after 2007.

The amount of Excess Contributions (and allocable income) to be distributed under this Section of the Plan or the amount of Excess Contributions recharacterized under this Section of the Plan with respect to a Participant for a Plan Year, is reduced by any amounts previously distributed to the Participant from the Plan to correct Excess Deferrals for the Participant's taxable year ending with or within the Plan Year in accordance with Code section 402(g)(2).

The amount required to be distributed to correct an Excess Deferral to an Employee for a taxable year is reduced by any Excess Contributions (and allocable income) previously distributed or excess contributions recharacterized with respect to the Participant for the Plan Year beginning with or within the taxable year. The amount of Excess Contributions includible in the gross income of the Employee, and the amount of Excess Contributions reported by the payer or Plan Administrator as includible in the gross income of the Participant, does not include the amount of an reduction under section 1.402(g)-1(e)(6) of the Treasury regulations.

Excess Contributions are applied to Highly Compensated Employees with the largest amounts of Elective Deferrals taken into account when calculating the ADP test for the Plan Year in which the excess arose, beginning with the Highly Compensated Employee with the largest contribution and continuing in descending order until all the Excess Contributions have been allocated. To the extent that a Highly Compensated Employee has not reached their Catch-up Contribution limit under the Plan, Excess Contributions allocated to such Highly Compensated Employee are Catch-up Contributions and will not be treated as Excess Contributions. If such excess amounts (other than Catch-up Contributions) are distributed more than 2 1/2 months after the last day of the Plan Year in which such excess amounts arose, a 10% excise tax will be imposed on the Employer maintaining the Plan with respect to such amounts.

Excess Contributions will be treated as Annual Additions under Part III, Article II even if distributed to the Participant.

Excess Contributions distributed under this Section will be adjusted for any income or loss based on a reasonable method of computing the allocable income or loss. The method selected must be applied consistently to all Participants and used for all corrective distributions under the Plan for the Plan Year, and must be the same method that is used by the Plan for allocating income or loss to Participants' Accounts. For Plan Years beginning after 2005, income or loss allocable to the period between the end of the Plan Year and the date of distribution may not be disregarded in determining income or loss.

One reasonable method treats the income and loss allocable to Excess Contributions as the sum of:

- (a) Income or loss allocable to the Participant's Elective Account (and, if applicable, the Qualified Non-Elective Account or the Qualified Matching Account, or both) for the Plan Year, multiplied by a fraction, the numerator of which is such Participant's Excess Contributions for the year and the denominator is the Participant's Account balance attributable to Elective Deferrals (and Qualified Non-Elective Contributions or Qualified Matching Contributions, or both, if any such contributions are included in the ADP test) without regard to any income or loss occurring during such Plan Year; and
- **(b)** 10% of the amount determined under (a) multiplied by the number of whole calendar months between the end of the Plan Year and the date of the distribution, counting the month of distribution if distribution occurs after the 15th of such month.

Matching Contributions attributable to Excess Contributions that have been returned will be forfeited according to the requirements of Section 2.7.9.

Excess Contributions applied to a Participant will be distributed from the Participant's Elective Account and Qualified Matching Account (if applicable) in proportion to the Participant's Elective Deferrals and Qualified Matching Contributions (to the extent used in the ADP test) for the Plan Year. Excess Contributions will be distributed from the Participant's Qualified Non-Elective Account only to the extent that such Excess Contributions exceed the amount of Excess Contributions in the Participant's Elective Account and Qualified Matching Account.

2.7.7 Distribution of Excess Aggregate Contributions. Notwithstanding any other provision of this Plan, Excess Aggregate Contributions, plus any income and minus any loss allocable thereto, will be forfeited, if forfeitable, or if not forfeitable, distributed no later than 12 months after a Plan Year to Participants to whose accounts such Excess Aggregate Contributions were allocated for such Plan Year. For Plan Years beginning after December 31, 2008, a corrective distribution of Excess Aggregate Contributions (adjusted for any income or loss) will be included in an Employee's gross income in the tax year in which the distribution occurs.

Excess Aggregate Contributions are determined only after first determining the amount of Excess Contributions treated as Employee Contributions due to recharacterization under a CODA in this or any other plan of the Employer.

Excess Aggregate Contributions are applied to Highly Compensated Employees with the largest contribution percentage amounts taken into account in calculating the ACP test for the year in which the excess arose, beginning with the Highly Compensated Employee with the largest contribution percentage amount and continuing in descending order until all the Excess Aggregate Contributions have been allocated. If such Excess Aggregate Contributions are distributed more than 2 1/2 months after the last day of the Plan Year in which such excess amounts arose, a 10% excise tax will be imposed on the Employer maintaining the Plan with respect to those amounts. However, for Plan Years beginning on or after January 1, 2010 where all eligible Non-Highly Compensated Employees and all eligible Highly Compensated Employees are Covered Employees under the Eligible Automatic Contribution Arrangement for the entire Plan Year, the Plan Administrator must distribute Excess Aggregate Contributions from the Plan no later than 6 months after the last day of the Plan Year in which the excess amounts were determined.

Excess Aggregate Contributions distributed under this Section will be adjusted for any income or loss based on a reasonable method of computing the allocable income or loss. The method selected must be applied consistently to all Participants and used for all corrective distributions under the Plan for the Plan Year, and must be the same method that is used by the Plan for allocating income or loss to Participants' Accounts. For Plan Years beginning after 2005, income or loss allocable to the period between the end of the Plan Year and the date of distribution may not be disregarded in determining income or loss. For Plan Years beginning on or after January 1, 2008, the income allocable to Excess Aggregate Contributions is equal to the allocable gain or loss through the end of the Plan Year for which the ACP Test is being corrected.

One reasonable method treats the income and loss allocable to the Participant's Excess Aggregate Contributions as the sum of:

- (a) Income or loss allocable to the Participant's Voluntary Account, Matching Account, Qualified Matching Account, (if any, and if all amounts therein are not used in the ADP test) and, if applicable, Qualified Non-Elective Contribution and Elective Deferral Accounts for the Plan Year, multiplied by a fraction, the numerator of which is such Participant's Excess Aggregate Contributions for the year and the denominator is the balance(s) of the Participant's Account previously described, if any such contributions are included in the ACP test without regard to any income or loss occurring during such Plan Year; and
- (b) 10% of the amount determined under (a) multiplied by the number of whole calendar months between the end of the Plan Year and the date of the distribution, counting the month of distribution if distribution occurs after the 15th of such month.

Forfeitures of Excess Aggregate Contributions may either be reallocated to the accounts of Participants who are not Highly Compensated Employees or applied to reduce Employer Contributions, as elected by the Employer in the Adoption Agreement.

Excess Aggregate Contributions allocated to a Participant will be forfeited, if forfeitable, or distributed on a pro-rata basis from the Participant's Voluntary, Matching, and Qualified Matching Accounts (and, if applicable, the Participant's Qualified Non-Elective Account or Elective Account, or both).

- **2.7.8 Limitation on Distributions.** Except as otherwise provided in this Article, Elective Deferrals, Qualified Non-Elective Contributions, Qualified Matching Contributions, and income allocable thereto are not distributable to a Participant or their Beneficiary in accordance with such Participant's or Beneficiary's election prior to severance from employment, death, or Disability. Such amounts may, however, be distributed upon:
- (a) Termination of the Plan without the establishment of another defined contribution plan, other than an employee stock ownership plan (as defined in Code sections 4975(e)(7) or 409(a)) or a simplified

employee pension plan as defined in Code section 408(k), a SIMPLE IRA plan as defined in Code section 408(p), a plan or contract described in Code section 403(b) or a plan described in Code section 457(b) or (f)) at any time during the period beginning on the date of plan termination and ending 12 months after all assets have been distributed from the Plan. Such a distribution must be made in a lump sum.

- (b) The attainment of age 59 1/2.
- (c) The Hardship of a Participant in accordance with Section 2.5.10.
- (d) The Participant's call to active duty after September 11, 2001, (because of the Participant's status as a member of a reserve component) for a period of at least 180 days (90 days for Plan Years beginning before January 1, 2007) or for an indefinite period. (A "qualified reservist distribution.")
- (e) The Participant's service in the uniformed services while on active duty for a period of at least 30 days. If a Participant receives a distribution under this provision, the Participant's Elective Deferrals (and Employee Contributions) will be suspended for 6 months after receipt of the distribution.
- (f) A federally declared disaster, where resulting legislation authorizes such a distribution.

All such distributions are subject to the spousal and Participant consent requirements, if applicable, contained in Code sections 401(a)(11) and 417 unless the Plan meets the exception of Subsection 2.5.5(c). In addition, distributions that are triggered by any of the first three events enumerated above must be made in a lump sum.

2.7.9 Forfeiture of Matching Contributions. Matching Contributions will be forfeited when:

- (a) The Elective Deferral to which the Matching Contribution or Qualified Matching Contribution relates is returned because it was determined to be Excess Deferrals (unless the Excess Deferrals are for Non-Highly Compensated Employees), Excess Contributions, or Excess Annual Additions; or
- **(b)** The Employee contribution to which the Matching Contribution relates is returned because it was determined to be either Excess Aggregate Contributions or Excess Annual Additions.

The requirements of this Section will be met in whole or in part if the Matching Contribution in question is returned to the Participant as an Excess Aggregate Contribution.

Following the forfeiture of Matching Contributions pursuant to this Section, the highest rate of Matching Contribution allocated to any Highly Compensated Employee may not exceed the lowest rate of Matching Contribution allocated to any Non-Highly Compensated Employee eligible to receive an allocation of Matching Contributions under this Plan or under all plans in a mandatory or permissive aggregation group. This provision will not apply in the case of a Non-Highly Compensated Employee whose allocation of Matching Contributions is limited by the application of Code section 415. Allocation of Forfeitures under this Section will be governed by elections regarding the allocation of the forfeiture of Matching Contributions in the Adoption Agreement.

ARTICLE VIII

SIMPLE 401(k) LIMITATIONS

Notwithstanding any provision in the Plan or Adoption Agreement to the contrary, during any year for which the Employer has elected to treat this Plan as a SIMPLE 401(k) plan under Code sections 401(k)(11) and 401(m)(10), this Article VIII and the attendant election of the Adoption Agreement or any amendment thereof will prevail.

2.8.1 Establishing a SIMPLE 401(k) Plan. If the Employer has elected in the Adoption Agreement to have the 401(k) SIMPLE Provisions apply, then the provisions of this Article will apply for a Year only if (a) the Employer is an Eligible Employer and (b) no contributions are made, or benefits accrued for services during the Year, on behalf of any Eligible Employee under any other plan, contract, pension, or trust described in Code section 219(g)(5)(A) or (B) maintained by the Employer. To the extent that any other provision of the Plan is inconsistent with the provisions of this Article, the provisions of this Article govern.

2.8.2 Definitions.

- (a) "Compensation" means, for purposes of Sections 2.8.2(b), 2.8.3, and 2.8.4 of this Article, the sum of the wages, tips, and other compensation from the Employer subject to federal income tax withholding (as described in Code section 6051(a)(3)) and the Employee's salary reduction contributions made under this or any other section 401(k) plan, and, if applicable, Elective Deferrals under a section 408(p) SIMPLE IRA Plan, a SARSEP, or a section 403(b) annuity contract and compensation deferred under a section 457 plan, required to be reported by the Employer on Form W-2 (as described in Code section 6051(a)(8)). For Self-Employed Individuals, Compensation means net earnings from self-employment determined under Code section 1402(a) prior to subtracting any contributions made under this Plan on behalf of the individual. The provisions of the Plan implementing the limit on compensation under Code section 401(a)(17) apply to the Compensation under Sections 2.8.3 and 2.8.4 of this Article.
- (b) "Eligible Employer" means, with respect to any Year (as defined in Subsection 2.8.2(d) below), an employer that had no more than 100 employees who received at least \$5,000 of Compensation from the Employer for the preceding Year. In applying the preceding sentence, all employees of controlled groups of corporations under Code section 414(b), all employees of trades or businesses (whether incorporated or not) under common control under Code section 414(c), all employees of affiliated service groups under Code section 414(m), and leased employees required to be treated as the Employer's employees under Code section 414(n), are taken into account.

An Eligible Employer that elects to have the 401(k) SIMPLE Provisions apply to the Plan and that fails to be an Eligible Employer for any subsequent Year is treated as an Eligible Employer for the 2 Years following the last Year the Employer was an Eligible Employer. If the failure is due to any acquisition, disposition, or similar transaction involving an Eligible Employer, the preceding sentence applies only if the provisions of Code section 410(b)(6)(C)(i) are satisfied.

- (c) "Eligible Employee" means, for purposes of the 401(k) SIMPLE Provisions, any Employee who is entitled to make Elective Deferrals under the terms of the Plan.
- (d) "Year" means the calendar year.

2.8.3 Salary Reduction Contributions.

(a) Each Eligible Employee may make a salary reduction election to have their Compensation reduced for the Year in any amount selected by the Employee subject to the limitation in Subsection 2.8.3(b). The Employer will make a salary reduction contribution to the Plan, as an Elective Deferral, in the amount by which the Employee's Compensation has been reduced.

- (b) The total salary reduction contribution for the Year cannot exceed \$10,000 for any Employee. To the extent permitted by law, this amount will be adjusted to reflect any annual cost-of-living increases announced by the IRS. After 2005, the \$10,000 limit will be adjusted by the Secretary of the Treasury for cost-of-living increases under Code section 408(p)(2)(E). Any such adjustments will be in multiples of \$500.
- (c) The amount of a Participant's salary reduction contributions permitted for a Year is increased for Participants aged 50 or over by the end of the Year by the amount of allowable Catch-up Contributions. Allowable Catch-up Contributions are \$2,500 for 2006. After 2006, the \$2,500 limit will be adjusted by the Secretary of the Treasury for cost-of-living increases under Code section 414(v)(2)(C). Any such adjustments will be in multiples of \$500. Catch-up Contributions are otherwise treated the same as other salary reduction contributions.

2.8.4 Other Contributions.

- (a) Matching Contributions. Each Year, the Employer will contribute a Matching Contribution to the Plan on behalf of each Employee who makes a salary reduction election under Section 2.8.3. The amount of the Matching Contribution will be equal to the Employee's salary reduction contribution up to a limit of 3% of the Employee's Compensation for the full Year.
- **(b) Non-Elective Contribution.** For any Year, instead of a Matching Contribution, the Employer may elect to contribute a Non-Elective Contribution of 2% of Compensation for the full Year for each Eligible Employee who received at least \$5,000 of Compensation (or such lesser amount as elected by the Employer in the Adoption Agreement) for the Year.
- **2.8.5** Limitation on Other Contributions. No Employer or Employee contributions may be made to this Plan for the Year other than a salary reduction contributions described in Section 2.8.3, Matching or Non-Elective Contributions described in Section 2.8.4, and rollover contributions described in Treasury Regulations section 1.402(c)-2, Q&A-1(a).
- **2.8.6 Section 415 Limitations.** The provisions of the Plan implementing the limitations of Code section 415 apply to contributions made pursuant to Sections 2.8.3 (other than Catch-up Contributions) and 2.8.4.

2.8.7 Election and Notice Requirements.

(a) Election Period

- (1) In addition to any other election periods provided under the Plan, each Eligible Employee may make or modify a salary reduction election during the 60-day period immediately preceding each January 1.
- (2) For the Year an Employee becomes eligible to make salary reduction contributions under the 401(k) SIMPLE Provisions, the 60-day election period requirement of Subsection 2.8.7(a)(1) is deemed satisfied if the Employee may make or modify a salary reduction election during a 60-day period that includes either the date the Employee becomes eligible or the day before.
- (3) Each Employee may terminate a salary reduction election at any time during the Year.

(b) Notice Requirements

- (1) The Employer will notify each Eligible Employee prior to the 60-day election period described in Subsection 2.8.7(a) that they can make a salary reduction election or modify a prior election during that period.
- (2) The notification described in Subsection 2.8.7(b)(1) will indicate whether the Employer will provide a 3% Matching Contribution described in Subsection 2.8.4(a) or a 2% Non-Elective Contribution described in Subsection 2.8.4(b).

- **2.8.8 Vesting Requirements.** All benefits attributable to contributions described in Sections 2.8.3 and 2.8.4 are nonforfeitable at all times, and all previous contributions made under the Plan are nonforfeitable as of the beginning of the Year the 401(k) SIMPLE Provisions apply.
- **2.8.9 Top-Heavy Rules.** The Plan is not treated as a Top-Heavy plan under Code section 416 for any Year for which this Article applies.
- **2.8.10 Nondiscrimination Tests.** The ADP and ACP tests described in Sections 2.7.1 and 2.7.2 of the Plan are treated as satisfied for any Year for which this Article applies.
- **2.8.11 Revocation.** An amendment to have the 401(k) SIMPLE Provisions no longer apply is effective the next January 1 following the adoption of the amendment.

ARTICLE IX

SAFE HARBOR CODA PROVISIONS

2.9.1 Safe Harbor CODA Rules.

(a) If the Employer has elected in the Adoption Agreement to make a Safe Harbor CODA contribution pursuant to Code section 401(k)(12) or 401(k)(13), this Article will apply for the Plan Year and any provisions relating to the ADP test described in Code section 401(k)(3) or the ACP test described in Code section 401(m)(2) will not apply. In accordance with Treasury Regulations 1.401(k)-1(e)(7) and 1.401(m)-1(c)(2), it is impermissible for the Plan to use ADP and ACP testing for a Plan Year in which it is intended for the Plan to be a safe harbor plan pursuant to sections 401(k) and 401(m) of the Code and the Plan fails to satisfy the requirements of such safe harbors for the Plan Year.

If the Employer has not elected a Safe Harbor CODA contribution option, the Employer may amend the Plan not later than 30 days before the last day of the Plan Year to specify that the Safe Harbor Non-Elective Contribution method will be used for the Plan Year, provided the Notice Requirement of Subsection 2.9.1(c) below, is met.

- **(b)** If the Employer has not elected a Safe Harbor contribution option, the Plan must satisfy the ADP (and, if applicable, ACP) test for the Plan Year.
- (c) Notice Requirement. For the purpose of a Safe Harbor contribution pursuant to Code section 401(k)(12), at least 30 days, or any other reasonable period but not more than 90 days, before the beginning of the Plan Year, the Employer must provide each Eligible Employee with a comprehensive notice of the Employee's rights and obligations under the Plan, written in a manner calculated to be understood by the average Eligible Employee, as described in Code section 401(k)(12)(D) and the applicable regulations. If an Employee becomes eligible after the Employer distributes the notice for a particular Plan Year, the Employer must provide the notice no more than 90 days before such Employee becomes eligible but not later than the date the Employee becomes eligible.

For the purpose of a Safe Harbor contribution pursuant to Code section 401(k)13), the Employer will give each Employee to whom the Qualified Automatic Contribution Arrangement in Section 2.2.3(c) applies, a notice that meets the requirements of Section 2.2.3(d).

The Employer reserves the right to amend the Plan to make a Safe Harbor Non-Elective Contribution under Subsection 2.9.1(b) above. If the Employer wishes to exercise that right, the notice given to Eligible Employees before the beginning of the Plan Year must provide that: (1) the Plan may be amended during the Plan Year to provide that the Employer will make a Safe Harbor Non-Elective Contribution of at least 3% to the Plan for the Plan Year, and (2) if the Plan is so amended, the Employer will give a supplemental notice to Eligible Employees 30 days prior to the last day of the Plan Year informing them of such an amendment.

- (d) Election Periods. In addition to any other election periods provided under the Plan, each Eligible Employee may make or modify a deferral election during the 30-day period immediately following receipt of the notice described in Subsection 2.9.1(c) above.
- (e) Timing of Matching Contributions. The Employer may also elect in the Adoption Agreement to meet the matching contribution requirements of this Article IX either (1) with respect to the Plan Year as a whole, or (2) separately with respect to each payroll period (or with respect to all payroll periods ending with or within each month or plan-year quarter) taken into account under the Plan for the Plan Year (the "payroll period method"). The payroll period method applies only for purposes of satisfying the ADP Safe Harbor Matching Contribution requirements of Section 2.9.3 of the Plan and the ACP Safe Harbor Matching Contribution requirements of Section 2.9.4 of the Plan, if applicable.

- (f) Interaction With Other Plan Provisions. For any Plan Year in which the Plan relies on this Article instead of Subsection 2.7.2 to satisfy Code section 401(k)(3), the Plan is treated as having used Current Year Testing for purposes of Sections 2.7.1 and 2.7.2. The Plan cannot switch to Prior Year Testing under those Sections unless the Plan meets the requirements of Treasury Regulations section 1.401(k)-2(c)(1)(ii).
- (g) Suspension, Reduction, or Discontinuance of a Safe Harbor Matching Contribution option.
 - (1) Pursuant to the Treasury Regulations sections 1.401(k)-3(g) and 1.401(m)-3(h) the Employer may elect to suspend, reduce or discontinue the Safe Harbor Matching Contributions on Elective Deferrals provided:
 - (A) all Eligible Employees are provided a notice of the suspension;
 - (B) all Eligible Employees are given a reasonable opportunity (including a reasonable period after receipt of the notice), prior to the discontinuance of the Safe Harbor Matching Contributions, to change their cash or deferred elections and, if applicable, their Employee Contribution elections;
 - **(C)** The Plan satisfies the Safe Harbor Matching Contribution requirement with respect to Safe Harbor Compensation paid through the effective date of the Amendment;
 - (D) The ADP and/or the ACP test will be satisfied for the entire Plan Year in which the Safe Harbor Matching Contribution was reduced or suspended; using the current year testing method described in Treasury Regulations 1.401(k)-2(a)(2)(ii) and 1.401(m)-2(a)(1)(ii); and
 - **(E)** The Plan must continue to provide Safe Harbor Matching Contributions on Elective Deferrals (or if applicable Voluntary Employee Contributions) until the effective date of the Amendment.
 - (2) The Plan is no longer a safe harbor plan described in Code sections 401(k)(12), 401(k)(13), 401(m)(11), or 401(m)(12) for the entire Plan Year. Accordingly, the Plan is not described in section 416(g)(4)(H) of the Code and, thus, will be subject to the Top-Heavy rules under section 416.
- (h) Suspension, Reduction or Discontinuance of a Safe Harbor Non-Elective Contribution option.

 The Employer may elect to suspend, reduce, or discontinue the Safe Harbor Non-Elective Contribution option provided the Employer has incurred a substantial business hardship (comparable to a substantial business hardship described in Code section 412(c)).
 - (1) For purposes of this subsection the factors taken into account in determining a substantial business hardship will include (but will not be limited to) whether or not:
 - (A) the Employer is operating at an economic loss,
 - **(B)** there is substantial unemployment or underemployment in the trade or business and in the industry concerned,
 - (C) the sales and profits of the industry concerned are depressed or declining, and
 - (D) it is reasonable to expect that the Plan will be continued only if the Safe Harbor Non-Elective Contributions are suspended.
 - (2) The Employer may elect to suspend, reduce, or discontinue the Safe Harbor Non-Elective Contribution with an effective date on the later of the date that is 30 days after:

- (A) all Eligible Employees are provided a notice of the suspension;
- (B) Eligible Employees are given a reasonable opportunity (including a reasonable period after receipt of this notice) prior to the reduction or suspension of the Safe Harbor Non-Elective Contributions to change their cash or deferred elections and, if applicable, their Voluntary Employee Contribution elections;
- (C) The Employer makes the Safe Harbor Non-Elective Contribution with respect to Safe Harbor Compensation paid through the effective date of the amendment. The Plan must pro-rate the otherwise applicable compensation limit under section 401(a)(17) of the Code in accordance with the requirements of section 1.401(a)(17)-1(b)(3)(iii)(A) of the Treasury Regulations.
- (D) The ADP and/or the ACP test will be satisfied for the entire Plan Year in which the Safe Harbor Non-Elective Contribution was reduced or suspended; using the current year testing method described in Treasury Regulations 1.401(k)-2(a)(2)(ii) and 1.401(m)-2(a)(1)(ii).
- (3) The Plan is no longer a safe harbor plan described in section 401(k)(12), 401(k)(13), 401(m)(11), or 401(m)(12) of the Code for the entire Plan Year. Accordingly, the Plan is not described in section 416(g)(4)(H) of the Code and, thus, will be subject to the top-heavy rules under section 416.

Safe Harbor Matching Contributions and Safe Harbor Non-Elective Contributions are immediately nonforfeitable regardless of the age and service of the Participant, or whether the Participant is employed on a specific date.

(i) Interaction with the Top-Heavy Rules.

- (1) If the Plan is Top-Heavy, within the meaning of Subsection 2.6.2(c), an ADP Test Safe Harbor Non-Elective Contribution provided pursuant to Subsection 2.9.3(a) can be counted toward the Plan's Top-Heavy Minimum Contribution requirement. To the extent that the ADP Test Safe Harbor Non-Elective Contribution and other eligible Employer Contributions are less than the Plan's Top-Heavy Minimum Contribution requirement for a particular Participant, then that Participant must receive the an additional contribution sufficient to satisfy the applicable Top-Heavy Minimum Contribution requirement. Although for purposes of the ADP Test Safe Harbor Non-Elective Contribution, the Plan can limit a Participant's compensation to a portion of the Plan Year, the Plan must calculate the Top-Heavy Minimum Contribution using the Participant's compensation for the entire Plan Year.
- (2) Effective for Plan Years beginning after December 31, 2001, an ADP Test Safe Harbor Matching Contribution provided pursuant to Subsection 2.9.3(a) can be counted toward the Plan's Top-Heavy Minimum Contribution requirement. To the extent that the ADP Test Safe Harbor Matching Contribution and other eligible Employer Contributions are less than the Plan's Top-Heavy Minimum Contribution requirement for a particular Participant, then that Participant must receive an additional contribution sufficient to satisfy the applicable Top-Heavy Minimum Contribution requirement.
- (3) If the Plan receives only Elective Deferrals (including Catch-up Contributions) and contributions pursuant to this Article in a Plan Year, Code section 416(g)(4)(H) provides that the Plan will be deemed not to be Top-Heavy even if it would otherwise be considered a Top-Heavy Plan pursuant to Subsection 2.6.2(c). This rule is effective for Plan Years beginning after December 31, 2001. To meet this requirement for a particular Plan Year, no other Employer Contributions may be allocated to Participants for that Plan Year, including the allocation of any Forfeitures. In addition, all matching contributions must satisfy the requirements of Code section 401(m)(11).
- (j) This Article Governs. To the extent that any other provision of the Plan is inconsistent with the provisions of this Article, the provisions of this Article will govern.

2.9.2 Definitions.

- (a) "Compensation" is defined in Part I, Article II, except, for purposes of this Article, no dollar limit, other than the limit imposed by Code section 401(a)(17), applies to the Compensation of a Non-Highly Compensated Employee. However, solely for purposes of determining the Compensation subject to a Participant's deferral election, the Employer may use an alternative definition to the one described in the preceding sentence, provided such alternative definition is a reasonable definition within the meaning of Treasury Regulations section 1.414(s)-1(d)(2) and permits each Participant to elect sufficient Elective Deferrals to receive the maximum amount of matching contributions (determined using the definition of Compensation described in the preceding sentence) available to the Participant under the Plan.
- (b) "Eligible Employee" means an Employee eligible to make Elective Deferrals under the Plan for any part of the Plan Year or who would be eligible to make Elective Deferrals but for a suspension due to a Hardship distribution described in Section 2.5.10 or to statutory limitations, such as Code sections 402(g) and 415. The Employer may elect in the Adoption Agreement to exclude Participants from the ADP Safe Harbor or ACP Safe Harbor portion of the Plan who have not completed a Year of Service since their original Employment Commencement Date, or have not yet reached age 21, and are not employed on the earlier of the first day of the next Plan Year after meeting the preceding requirements or 6 months after meeting the preceding requirements.

2.9.3 ADP Test Safe Harbor Contributions.

- (a) ADP Test Safe Harbor Contributions Options. The Employer may elect in the Adoption Agreement to make a Qualified Automatic Contribution Arrangement, Basic or Enhanced Safe Harbor Matching Contributions, or Safe Harbor Non-Elective Contributions, for the Plan Year to this Plan in lieu of satisfying the ADP testing requirements under Sections 2.7.1 et Seq.
 - (1) In the case where the Employer elects to make ADP Test Safe Harbor Contributions for the Plan Year, unless the Employer elects in the Adoption Agreement to make Enhanced Matching Contributions or Safe Harbor Non-Elective Contributions, the Employer will contribute for the Plan Year a Safe Harbor Matching Contribution to the Plan on behalf of each Eligible Employee equal to (i) 100% of the amount of the Employee's Elective Deferrals that do not exceed 3% of the Employee's Compensation for the Plan Year, plus (ii) 50% of the amount of the Employee's Elective Deferrals that exceed 3% of the Employee's Compensation but that do not exceed 5% of the Employee's Compensation ("Basic Matching Contributions").
 - (2) Notwithstanding the requirement in (a) above that the Employer make ADP Test Safe Harbor Contributions to this Plan, a plan may provide in the Adoption Agreement that ADP Test Safe Harbor Contributions will be made to the defined contribution plan indicated in the Adoption Agreement. However, such contributions will be made to this Plan unless (i) each Employee eligible under this Plan is also eligible under the other plan and (ii) the other plan has the same plan year as this Plan.
 - (3) The Participant's Account balance derived from ADP Test Safe Harbor Contributions is nonforfeitable and may not be distributed earlier than severance from employment, death, Disability, an event described in Code section 401(k)(10), or, in the case of a profit sharing plan, the attainment of age 59 1/2. No distributions on account of Hardship are permitted. In addition, such contributions must satisfy the ADP Test Safe Harbor without regard to permitted disparity under Code section 401(l).
- **(b)** Suspension of ADP Safe Harbor Contributions. If the Plan is making Matching Contributions under Subsection (a), the Plan may be amended to reduce or eliminate such Contributions by complying with the requirements of Section 2.9.1.

2.9.4 ACP Test Safe Harbor Matching Contribution Options.

- (a) In addition to the ADP Test Safe Harbor Contribution Options described in Subsection 2.9.3(a) of this Article, the Employer will make the ACP Test Safe Harbor Matching Contributions, if any, indicated in the Adoption Agreement for the Plan Year.
- (b) ACP Test Safe Harbor Matching Contributions will be vested as indicated in the Adoption Agreement, but, in any event, such contributions will be fully vested at Normal Retirement Age, upon the complete or partial termination of the Plan, or upon the complete discontinuance of Employer Contributions. Forfeitures of non-vested ACP Test Safe Harbor Matching Contributions will be used to reduce the Employer's Contribution.

ARTICLE X

PARTICIPATING EMPLOYERS

If elected by the Employer in the Adoption Agreement Related Employers and Affiliated Employers may adopt the Plan. If the Plan is adopted by one or more Affiliated Employers the Plan will be a Multiple Employer Plan (MEP) established and maintained pursuant to Code section 413(c) as a tax qualified defined contribution plan under Code sections 401 and 501. The Plan will be operated for the exclusive benefit of the Eliqible Employees of the Participating Employers.

2.10.1 Adoption. If permitted by the Lead Sponsor, Related Employers and Affiliated Employers may adopt the plan by signing the same Adoption Agreement as the Lead Sponsor, executing a Participating Employer Addendum, or executing an Adoption Agreement that specifies the Employer is a Participating Employer in the Plan. No employer will become a Participating Employer of the Plan without the written consent of the Lead Sponsor.

The Participating Employer will be bound by the terms of the Plan, including all amendments, as adopted by the Lead Sponsor. If permitted by the Lead Sponsor, the Participating Employer may elect separate provisions to apply to only the Employees employed by the Participating Employer. These elections can be made using either a Participating Employer Addendum or a separate Adoption Agreement. The Lead Sponsor may limit the options available to Participating Employers if a Participating Employer Addendum or a separate Adoption Agreement is used. A Participating Employer Addendum or separate Adoption Agreement will not be considered valid unless it is executed by the Lead Sponsor.

2.10.2 Plan Maintenance. The Lead Sponsor may at any time or times amend the Plan, in whole or in part, without the consent of the Participating Employer(s). A Participating Employer may not modify the Plan, in whole or in part, without the consent of the Lead Sponsor. Any changes made by a Participating Employer without the consent of the Lead Sponsor will cause such Participating Employer to cease being a Participating Employer in the Plan.

In the event the Lead Sponsor terminates their participation in the Plan, at the discretion of the Lead Sponsor any Participating Employer may become the Lead Sponsor by execution of a new Adoption Agreement.

- **2.10.3 Operational Rules.** The rules contained in this Plan Section govern the application of certain operational provisions and qualification requirements contained in the Plan with respect to the Lead Sponsor and each Participating Employer. With respect to the Lead Sponsor and each Participating Employer, the following rules apply the operation of the Plan.
- (a) Service. An Employee's service, Hours of Service, Years of Service (or any fractional period thereof) and periods of employment will be determined by treating the Lead Sponsor, all Participating Employers, and Related Employers of either the Lead Sponsor or a Participating Employer as a single Employer.

An Employee's employment status will be determined by treating the Lead Sponsor, all Participating Employers, and Related Employers of either the Lead Sponsor or a Participating Employer as a single Employer. An Employee who terminates employment from the Lead Sponsor or a Participating Employer but starts or continues employment with the Lead Sponsor or a Participating Employer will not have a severance of employment.

(b) Testing.

Participating Employers will be treated as a single Employers for the purposes of meeting the requirements under:

- (1) Code section 415,
- (2) Code section 401(a)(2) relating to the "exclusive benefit rule",
- (3) Code section 402(g), and
- (4) Code section 414(v).

Except as required under Code section 414(b), (c), (m), or (o), the requirements of the following Code sections will be applied separately to the Lead Sponsor and each Participating Employers:

- (1) Code section 410(b),
- (2) Code section 401(a)(4),
- (3) Code section 401(k),
- (4) Code section 401(m),
- (5) Code section 414(q), and
- (6) Code section 416.
- **2.10.4 Duties.** The Lead Sponsor and Participating Employer(s) agree to provide to the Plan Administrator, in a timely fashion, all information the Plan Administrator, it its sole discretion, deems necessary for the operation of the Plan and to keep the Plan in compliance with all Code and regulatory requirements. The Lead Sponsor and Participating Employer(s) agree to provide all required contributions related to Employees that they employee in compliance with the provisions of the Plan and all Code and regulatory requirements. The Lead Sponsor and each Participating Employer(s) will be jointly and severally liable for all Plan expenses.

The Lead Sponsor and each Participating Employer will indemnify and hold harmless the Plan Administrator (and their delegates), any other Participating Employer, and any person serving as the Trustee from all claims, liabilities, losses, damages and expenses, including reasonable attorneys' fees and expenses for its failure to operate in accordance with the Plan or any intentional or negligent act or omission with respect to the Plan including but not limited to failure of oversight and or appointment.

The Plan Administrator may, in its sole discretion, utilize any IRS or DOL correction program to correct any Plan failures. Any fees or costs associated with such program will be the responsibility of the Lead Sponsor or Participating Employer(s) whose actions lead to the Plan failures.

- **2.10.5 Withdrawal of an Employer.** The Lead Sponsor or any Participating Employer may elect to discontinue or revoke their participation in the Plan by providing written notice to the Plan Administrator within a reasonable time prior to the effective date of such discontinuance. Such discontinuance means there are no further benefits accruing after the effective date of such discontinuance. The Lead Sponsor may discontinue the participation in the Plan of any Participating Employer by providing written notice to the Participating Employer within a reasonable time prior to the effective date of such discontinuance. At the time of discontinuance the Plan Administrator has the right, but not the duty, to take any of the following actions:
- (a) Spin-Off. The Lead Sponsor or the former Participating Employer can set up a successor plan. This successor plan will receive the assets of the Plan related to the former Participating Employer thru a non-elective transfer of assets. The Lead Sponsor may name either itself or the former Participating Employer as the Plan Administrator and/or adopting Employer.
- **(b) Existing Plan.** If the former Participating Employer sponsors or participates in an existing qualified defined contribution plan, the Lead Sponsor may direct that the assets of the Plan related to the former Participating Employer be transferred to such plan thru a non-elective transfer of assets.
- (c) Retain Assets. The Lead Sponsor may chose to retain the assets related to the former Participating Employer in the Plan.

At the time of such discontinuance the former Participating Employer much promptly contribute amounts necessary to cover all accrued benefits related to its Participants under the Plan that are not covered by contributions already made. The discontinuance of participation in the Plan by the Lead Sponsor or any Participating Employer does not, in and of itself, result in a termination from employment for the Participants of the former Participating Employer.

PART III

ARTICLE I

ACCOUNTING

- **3.1.1 Accounts.** All income, profits, recoveries, contributions, and any and all moneys, securities, and properties of any kind at any time received or held by the Trustee will be held as a commingled Trust Fund, except to the extent such assets are transferred to a Segregated Fund or Controlled Account. For accounting purposes, the Plan Administrator shall, as necessary, establish and maintain the following Accounts for each Participant:
- (a) Employer Account
- (b) Controlled Account
- (c) Elective Account
- (d) Matching Account
- (e) Qualified Matching Account
- (f) Qualified Non-Elective Account
- (g) Voluntary Account
- (h) Deductible Voluntary Account
- (i) Pre-tax Elective Account
- (i) Roth Deferral Account
- (k) Deemed IRA Account
- (I) Pre-tax IRA Account
- (m) Roth IRA Account
- (n) Rollover Account
- (o) ADP Test Safe Harbor Account
- (p) ACP Test Safe Harbor Account
- (q) Segregated Account
- (r) In-Plan Roth Conversion Account
- (s) Paid Time-Off Contribution Account

If, to fulfill any purposes of the Plan, assets are either deposited initially or transferred to a Segregated Fund for the benefit of a Participant, the Plan Administrator will establish and maintain a Segregated Account for the Participant. If a Participant elects to exercise investment control over all or a portion of their Accounts, the Plan Administrator will establish and maintain a Controlled Account for the Participant.

3.1.2 Valuation Adjustments.

- (a) As of each Valuation Date, the following adjustments will be made to each Participant's Accounts in a manner consistent with the Plan's recordkeeping system:
 - (1) Distributions. Any distribution made to or on behalf of a Participant since the last preceding Valuation Date will be deducted from the Participant's Account from which the distribution was made.
 - (2) Expenses. Expenses earmarked for an individual Participant, if any, will reduce their Account balance.
 - (3) Contributions. Each Participant's Account will be increased by their portion of any Employer Contributions, Non-Elective Contributions, other contributions to the Plan by the Employer, Elective Deferrals, any allocated Forfeitures, and any other contributions made on behalf of the Participant.
 - (4) Forfeitures. If a Participant terminates service with the Employer, is less than 100% vested, and forfeits the non-vested portion of their Account, then their Account and the amount available for distribution will be reduced by the amount of such Forfeiture. To the extent that the Plan provides additional allocations from Forfeitures, the Accounts of those eligible for such an allocation will increase.
 - (5) Adjustment to Fair Market Value. The Trustee will appraise all moneys, securities, and other property in the Trust Fund, including Segregated Funds and Controlled Accounts but excluding Life Insurance Policies, at the then fair market value for each asset. In determining such value, all income and contributions, if any, received by the Trustee from the Employer or Participants will be included and all expenses will be deducted, such amounts being determined under the accounting method of the Trust. If the total net value so determined by the Trustee exceeds (or is less than) the total amount in the respective Accounts of all Participants, the excess (or deficiency) will be added to (or deducted from) the respective Accounts of all Participants in the ratio that each such Participant's Account bears to the total amount in all such Accounts.
 - (6) Insurance. If this Plan is funded by individual contracts that provide a Participant's benefit under the Plan, such individual contracts will constitute the Participant's Account balance. If this Plan is funded by group annuity contracts or group insurance contracts, premiums or other consideration received by the insurance company must be allocated to Participants' Accounts under the Plan. Payments made since the last preceding Valuation Date for Life Insurance Policies on the life of a Participant or someone in whom the Participant has an insurable interest (including without limitation payments of premiums and interest on policy loans) will be deducted from the Account of the Participant from which the payment was made. Dividends or credits received since the last preceding Valuation Date on any Life Insurance Policy will be added to the Account of the Participant from which the premiums for such Life Insurance Policy have been paid.
 - (7) Loans. The issuance of a loan to a Participant reduces the Account from which it is issued and increases the Participant's outstanding loan obligation. Loan repayments, including principal and interest, will increase the Account to which the repayment is made and decrease the outstanding loan obligation.
 - (8) Transfers To or From Segregated Funds. To the extent that funds are transferred to a Segregated Fund from the general assets of the Trust or from a Segregated Fund to the general assets of the Trust pursuant to any of the provisions of the Plan, the Account from which the funds were transferred will be decreased and the Account to which the funds were transferred will be increased for each affected Participant.
- **(b)** Every adjustment made pursuant to this Section will be considered as having been made no later than the applicable Valuation Date pursuant to policies established by the Trustee in a nondiscriminatory

manner. The Trustee's determination regarding the valuation of assets and charges or credits to the individual Accounts of the respective Participants will be conclusive and binding on all persons. If funds are transferred from the Trust Fund to a Segregated Fund as of any date other than a Valuation Date pursuant to the terms of the Plan, the adjustment made pursuant to this Section will be made as of the date such transfer was made, as if such date is a Valuation Date. If any Participant receives a distribution pursuant to the terms of the Plan as of any date other than a Valuation Date, then the adjustments made pursuant to this Section will be made in the manner specified in the Adoption Agreement.

- **3.1.3 Allocation of Earnings, Gains, and Losses.** As of each Valuation Date, the net income of the Trust Fund recognized since the prior Valuation Date will be allocated among Participant Accounts in a nondiscriminatory manner according to policies established by the Trustee. Net income is the interest, dividends, net gains or losses from the sale of investments, and unrealized appreciation (depreciation) in Trust Fund assets, less investment expenses of the Trust Fund. The portion of a Participant's Account held in a Controlled Account will not participate in the allocation of earnings of this Section, but instead will be credited with the actual earnings of such Controlled Account. In the event that the Participant's Accounts (or portion thereof) are invested in assets that are valued on a daily basis, their Account will reflect the daily activity of such assets and the method of allocating earnings described above will not apply.
- **3.1.4 Interim Valuations.** It is contemplated that the Trust Fund will be valued by the Trustee and allocations made only on a Valuation Date. At any time that the Plan's valuations are not performed on a daily basis, should it be necessary to make distributions under the provisions hereof and the Plan Administrator in good faith determines that, because of (a) an extraordinary change in general economic conditions, (b) the occurrence of some casualty materially affecting the value of the Trust Fund or a substantial part thereof, or (c) a significant fluctuation in the value of the Trust Fund has occurred since the immediately preceding Valuation Date, the Plan Administrator may, in their sole discretion, prevent the payee from receiving a substantially greater or lesser amount than what they would be entitled to, based on current values, and cause a re-valuation of the Trust Fund to be made and a reallocation of the interests therein as of the date the payee's right of distribution becomes fixed. The Plan Administrator's determination to make such special valuation and the valuation of the Trust Fund as determined by the Trustee will be conclusive and binding on all persons ever interested hereunder. Such interim valuation will not discriminate in favor of Highly Compensated Employees.
- **3.1.5 Earnings on Forfeitures.** In the event that a Participant forfeits part of their Account balance, such Forfeiture will not be credited with earnings after the date the Forfeiture is recognized pursuant to Section 2.4.4.
- **3.1.6 Plan Expenses.** Reasonable expenses for administration, investments, and processing will be charged against the Trust Fund earnings, paid by the Employer, or allocated among terminated and active Participants according to an expense policy adopted by the Employer and applied in a uniform manner consistent with DOL and Treasury regulations.

ARTICLE II

LIMITATIONS

3.2.1 Limitations on Annual Additions. The Annual Additions contributed or allocated to a Participant's Account for a Limitation Year will not exceed the Maximum Permissible Amount, as determined in Section 3.2.2. If the total amounts contributed or allocated would otherwise produce Annual Additions in excess of the Maximum Permissible Amount, the allocations or contributions will be reduced pursuant to Section 3.2.3 so that the Annual Additions for the Limitation Year equal the Maximum Permissible Amount.

If a Participant participates in, or has ever participated in, any of the following types of plans or funds maintained by the Employer, then the Annual Additions derived from the allocations of all such plans and this Plan credited to a Participant will not exceed the lesser of the Maximum Permissible Amount for the Limitation Year or any other limitation contained in the Plan:

- (a) Another qualified defined contribution plan;
- (b) A welfare benefit fund, as defined in Code section 419(e), under which amounts attributable to postretirement medical benefits are held in separate accounts for Key Employees;
- (c) An individual medical account, as defined in Code section 415(I)(2); or
- (d) A simplified employee pension, as defined in Code section 408(k),

If such Annual Additions exceed the Maximum Permissible Amount for the Limitation Year, the amounts contributed or allocated will be reduced in accordance with Section 3.2.4 so that the Annual Additions under all such plans and funds for the Limitation Year will equal the Maximum Permissible Amount.

3.2.2 Determination of Annual Additions and Maximum Permissible Amount.

- (a) Annual Additions. The term "Annual Additions" will mean the sum of the following amounts credited to a Participant's Accounts for the Limitation Year:
 - (1) Employer Contributions:
 - (2) Employee contributions;
 - (3) Forfeitures;
 - (4) Excess Elective Deferrals, Excess Contributions and Excess Aggregate Contributions;
 - (5) Amounts allocated after March 31, 1984, to an individual medical account, as defined in Code section 415(I)(2), that is part of a pension or annuity plan maintained by the Employer;
 - (6) Amounts derived from contributions paid or accrued after December 31, 1985, in taxable years ending after such date, that are attributable to post-retirement medical benefits held in separate accounts for Key Employees under a welfare benefit fund as defined in Code section 419(e), maintained by the Employer; and
 - (7) Allocations under a simplified employee pension as defined in Code section 408(k).

For this purpose, Excess Elective Deferrals are disregarded to the extent distributed to the Participant. Excess Contributions and Excess Aggregate Contributions are included even if distributed to the Participant. Any other amounts are disregarded if they are held in a suspense account or reallocated to other Participants in accordance with Section 3.2.3. Any excess amounts applied under Section 3.2.3 to

reduce Employer Contributions in a Limitation Year are considered Annual Additions for such Limitation Year.

Exception for Restorative Payments. Restorative payments allocated to a Participant's Account, which include payments made to restore losses to the Plan resulting from actions (or a failure to act) by a Fiduciary for which there is a reasonable risk of liability under Title I of ERISA or under other applicable federal or state law, where similarly situated Participants are similarly treated, do not give rise to an Annual Addition for any Limitation Year.

If, in a particular Limitation Year, the Employer contributes an amount to an Employee's account with respect to a prior Limitation Year and such contribution is required by reason of a Period of Qualified Military Service, then such contribution is not considered an Annual Addition with respect to the Employee for that particular Limitation Year in which the contribution is made, but, in accordance with section 414(u)(1)(B) of the Code, is considered an Annual Addition for the Limitation Year to which the contribution relates.

- **(b) Maximum Permissible Amount.** Except for Catch-up Contributions, the term "Maximum Permissible Amount" will mean, for any Limitation Year beginning on or after January 1, 2002, the lesser of:
 - (1) \$40,000, as adjusted for increases in the cost-of-living under Code section 415(d), or
 - (2) 100% of a Participant's compensation, as defined in Subsection 3.2.5(a), for the Limitation Year.

The compensation limit referred to in (2) above will not apply to any contribution for medical benefits after separation from service (within the meaning of Code sections 401(h) or 419A(f)(2)) that is otherwise treated as an Annual Addition.

Prior to determining a Participant's actual Compensation for a Limitation Year, the Employer may determine the Maximum Permissible Amount for the Participant based on a reasonable estimate of the Participant's Compensation for the Limitation Year, uniformly determined for all Participants similarly situated. As soon as is administratively feasible after the end of the Limitation Year, the Employer will determine the Maximum Permissible Amount for the Limitation Year based on the Participant's actual Compensation for the Limitation Year.

- (c) Short Limitation Year. If a short Limitation Year is created because of an amendment changing the Limitation Year to a different 12 consecutive month period, the limitation determined in subsection (b)(1) above will be multiplied by a fraction, the numerator of which is the number of months in the short Limitation Year and the denominator of which is 12. Any new Limitation Year must begin within the Limitation Year in which the amendment is made.
- **3.2.3 Excess Annual Additions.** If, due to the allocation of Forfeitures, a reasonable estimation of a Participant's Compensation that exceeds their actual Compensation, or a miscalculation in a Participant's Elective Deferrals, a Participant's Annual Additions exceed their Maximum Permissible Amount for the Limitation Year, the Employer may correct such amount using the methods permitted in the Employer Compliance Resolution System defined in Revenue Procedure 2008-50 as amended from time to time.
- **3.2.4 Participation in Certain Other Plans.** If the Annual Additions with respect to the Participant under such other plans and funds maintained by the Employer are less than the Maximum Permissible Amount and the Employer Contribution that would otherwise be contributed or allocated to the Participant's Account under this Plan would cause the Annual Additions for the Limitation Year to exceed this limitation, the amount contributed or allocated will be reduced so that the Annual Additions under all such plans and funds for the Limitation Year will equal the Maximum Permissible Amount. If the Annual Additions with respect to the Participant under such other plans and funds in the aggregate are equal to or greater than the Maximum Permissible Amount, no amount will be contributed or allocated to the Participant's Account under this Plan for the Limitation Year.

If a Participant's Annual Additions under this Plan and such other plans and funds maintained by the Employer would result in an excess amount for a Limitation Year, such excess will be distributed or

reallocated according to the provisions of Section 3.2.3, and the excess amount will be deemed to consist of the Annual Additions last allocated. In addition, allocations to qualified defined contribution plans will be adjusted first, followed by adjustments to allocations to a simplified employee pension, followed by adjustments to allocations to an individual medical account, followed by adjustments to allocations to a welfare benefit fund, as those terms are referenced in Section 3.2.1.

If the excess amount was allocated to a Participant on an allocation date of this Plan that coincides with an allocation date of another plan, the excess amount attributed to this Plan will be the product of:

- (i) The total excess amount allocated as of such date, times
- (ii) The ratio of (I) the Annual Additions allocated to the Participant for the Limitation Year as of such date under this Plan to (II) the total Annual Additions allocated to the Participant for the Limitation Year as of such date under this and all the other qualified defined contribution plans of the Employer. Any excess amount attributed to this Plan will be disposed in the manner described in Section 3.2.3.

If the Participant is covered under another qualified defined contribution plan maintained by the Employer that is not a pre-approved plan, Annual Additions that may be credited to the Participant's Account under this Plan for any Limitation Year will be limited as provided above, unless the Employer specifies other limitations in the Adoption Agreement.

For purposes hereof, the excess amount is the excess of the Participant's Annual Additions for the Limitation Year over the Maximum Permissible Amount.

- **3.2.5 Definitions.** For purposes of this Article, the terms below have the following meanings:
- (a) Compensation. As elected by the Employer in the Adoption Agreement, Compensation will mean all of a Participant's:
 - (i) Wages, Tips, and Other Compensation Box on Form W-2. Compensation includes Information required to be reported under sections 6041, 6051, and 6052 of the Code. Compensation is defined as wages, within the meaning of section 3401(a) of the Code, and all other payments of compensation to an Employee by the Employer (in the course of the Employer's trade or business) for which the Employer is required to furnish the Employee written statement under sections 6041(d), 6051(a)(3), and 6052 of the Code. Compensation will be determined without regard to any rules under section 3401(a) of the Code that limit the remuneration included in wages based on the nature or location of the employment or the services performed (such as the exception for agricultural labor in section 3401(a)(2) of the Code).
 - (ii) Section 3401(a) Wages. Wages as defined in Code section 3401(a) for the purposes of income tax withholding at the source but determined without regard to any rules that limit the remuneration included in wages based on the nature or location of the employment or the services performed (such as the exception for agricultural labor in section 3401(a)(2) of the Code).
 - (iii) Section 415 Simplified Compensation. Wages, salaries, differential wage payments under section 3401(h) of the Code and fees for professional services and other amounts received (without regard to whether or not an amount is paid in cash) for personal services actually rendered in the course of employment for the Employer maintaining the Plan to the extent that the amounts are includible in gross income (including but not limited to commissions paid salesmen, compensation for services on the basis of a percentage of profits, commissions on insurance premiums, tips, bonuses, fringe benefits, and reimbursements or other expense allowances under a non-accountable plan (as described in Treasury Regulations section 1.62-2(c) of the Code), but excluding items described in Section 3.2.5(a)(iv)(II)(A) through (E):
 - (iv) Section 415 Compensation. Except as otherwise provided in this Subparagraph (iv) Section 415 Compensation will mean, compensation from the Employer within the meaning of section

415(c)(3) of the Code, which is used for purposes of section 415 and regulations promulgated under section 415, and includes all items of remuneration described in Subparagraph (iv)(I) of this section, but excludes the items of remuneration described in Subparagraph (iv)(II) of this section. Paragraphs (i) (ii) and (iii) of this section provides safe harbor definitions of Compensation that are permitted to be provided in the Plan in lieu of the generally applicable definition of compensation contained in this Paragraph (iv).

- (I) Items Includible As Compensation. For purposes of applying the limitations of section 415, except as otherwise provided in this section, the term compensation means remuneration for services of the following types:
 - (A) The Employee's wages, salaries, fees for professional services, and other amounts received (without regard to whether or not an amount is paid in cash) for personal services actually rendered in the course of employment with the Employer maintaining the Plan, to the extent that the amounts are includible in gross income (or to the extent amounts would have been received and includible in gross income but for an election under sections 125(a), 132(f)(4), 402(e)(3), 402(h)(1)(B), 402(k), or 457(b) of the Code). These amounts include, but are not limited to, commissions paid to salespersons, compensation for services on the basis of a percentage of profits, commissions on insurance premiums, tips, bonuses, fringe benefits, and reimbursements or other expense allowances under a non-accountable plan as described in Section 1.62-2(c) of the Code.
 - (B) In the case of an Employee who is an Employee within the meaning of section 401(c)(1) of the Code and regulations promulgated thereunder, the Employee's Earned Income (as described in section 401(c)(2) of the Code and regulations promulgated thereunder), plus amounts deferred at the election of the Employee that would be includible in gross income but for the rules of sections 402(e)(3), 402(h)(1)(B), 402(k), or 457(b) of the Code.
 - (C) Amounts described in sections 104(a)(3), 105(a), or 105(h) of the Code, but only to the extent that these amounts are includible in the gross income of the Employee.
 - **(D)** Amounts paid or reimbursed by the Employer for moving expenses incurred by an Employee, but only to the extent that at the time of the payment it is reasonable to believe that these amounts are not deductible by the Employee under section 217 of the Code.
 - **(E)** The value of a nonstatutory option (which is an option other than a statutory option as defined in section 1.421-1(b) of the Treasury Regulations) granted to an Employee by the Employer, but only to the extent that the value of the option is includible in the gross income of the Employee for the taxable year in which granted.
 - **(F)** The amount includible in the gross income of an Employee upon making the election described in section 83(b) of the Code.
 - (G) Amounts that are includible in the gross income of an Employee under the rules of section 409A or section 457(f)(1)(A) of the Code or because the amounts are constructively received by the Employee.

(II) Items Not Includible As Compensation.

(A) Contributions (other than elective contributions described in sections 402(e)(3), 408(k)(6), 408(p)(2)(A)(i), or 457(b) of the Code) made by the Employer to a plan of deferred compensation (including a simplified employee pension described in section 408(k) of the Code or a simple retirement account described in section 408(p) of the Code, and whether or not qualified) to the extent that the contributions are not includible in the gross income of the Employee for the taxable year in which

contributed. In addition, any distributions from a plan of deferred compensation (whether or not qualified) are not considered as compensation for section 415 purposes, regardless of whether such amounts are includible in the gross income of the Employee when distributed. However, if the Plan so provides, any amounts received by an Employee pursuant to a nonqualified unfunded deferred compensation plan are permitted to be considered as compensation for section 415 purposes in the year the amounts are actually received, but only to the extent such amounts are includible in the Employee's gross income.

- (B) Amounts realized from the exercise of a non-statutory option (which is an option other than a statutory option as defined in section 1.421-1(b) of the Treasury Regulations), or when restricted stock or other property held by an Employee either becomes freely transferable or is no longer subject to a substantial risk of forfeiture (see section 83 of the Code and regulations promulgated thereunder).
- **(C)** Amounts realized from the sale, exchange, or other disposition of stock acquired under a statutory stock option (as defined in section 1.421-1(b) of the Code).
- **(D)** Other amounts that receive special tax benefits, such as premiums for group-term life insurance (but only to the extent that the premiums are not includible in the gross income of the Employee and are not salary reduction amounts that are described in section 125 of the Code).
- (E) Other items of remuneration that are similar to any of the items listed in Subparagraphs (I)(A) through (I)(E) of this section.

Notwithstanding the preceding sentence, if elected in the Adoption Agreement, Compensation for a Participant who is permanently and totally disabled (as defined in Code section 22(e)(3)) is the Compensation such Participant would have received for the Limitation Year if the Participant had been paid at the rate of Compensation paid immediately before becoming permanently and totally disabled.

- (b) For Limitation Years beginning on or after July 1, 2007, or such earlier date as specified in the Adoption Agreement, Compensation for a Limitation Year will also include Compensation paid by the later of 2-1/2 months after an Employee's severance from employment with the Employer or the end of the Limitation Year that includes the date of the Employee's severance from employment with the Employer, if:
 - (i) the payment is regular Compensation for services during the Employee's regular working hours, or Compensation for services outside the Employee's regular working hours (such as overtime or shift differential), commissions, bonuses, or other similar payments, and, absent a severance from employment, the payments would have been paid to the Employee while the Employee continued in employment with the Employer; or, if the Employer so elects in the Adoption Agreement;

Or, if the Employer so elects in the Adoption Agreement,

- (ii) the payment is for unused accrued bona fide sick, vacation or other leave that the Employee would have been able to use if employment had continued; or
- (iii) the payment is received by the Employee pursuant to a nonqualified unfunded deferred compensation plan and would have been paid at the same time if employment had continued, but only to the extent includible in gross income.

Any payments not described above will not be considered Compensation if paid after severance from employment, even if they are paid by the later of 2-1/2 months after the date of severance from employment or the end of the Limitation Year that includes the date of severance from employment, except, if elected by the Employer in the Adoption Agreement, Compensation paid to a Participant who is permanently and totally disabled, as defined in section 22(e)(3) of the Code, provided, as

elected by the Employer in the Adoption Agreement, salary continuation applies to all Participants who are permanently and totally disabled for a fixed or determinable period, or the Participant was not a Highly Compensated Employee, as defined in Part 1 Article 2 of the Plan immediately before becoming disabled.

Back pay, within the meaning of section 1.415(c)-2(g)(8) of the Treasury Regulations, will be treated as Compensation for the Limitation Year to which the back pay relates to the extent the back pay represents wages and Compensation that would otherwise be included under this definition.

Compensation paid or made available during a Limitation Year will include amounts that would otherwise be included in Compensation but for an election under sections 125(a), 132(f)(4), 402(e)(3), 402(h)(1)(B), 402(k), or 457(b) of the Code.

Unless the Employer elects otherwise in the Adoption Agreement, Compensation will not include Deemed 125 Compensation. Deemed section 125 Compensation is an amount that is excludable under section 106 of the Code that is not available to Participant in cash in lieu of group health coverage under a section 125 arrangement solely because the Participant is unable to certify that they or she has other health coverage. Amounts are Deemed section 125 Compensation only if the Employer does not request or otherwise collect information regarding the Participant's other health coverage as part of the enrollment process for the health plan.

If elected by the Employer in the Adoption Agreement, Compensation will not include amounts paid as Compensation to a nonresident alien, as defined in section 7701(b)(1)(B) of the Code, who is not a Participant in the Plan to the extent the Compensation is excludable from gross income and is not effectively connected with the conduct of a trade or business within the United States.

- (c) Certain de minimis timing differences. Notwithstanding the provisions of this Section 3.2.5, the Employer may elect in the Adoption Agreement whether to include in Compensation for the Limitation Year amounts earned during the Limitation Year but not paid during the Limitation Year solely because of the timing of pay periods and pay dates if:
 - (i) These amounts are paid during the first few weeks of the next Limitation Year;
 - (ii) The amounts are included on a uniform and consistent basis with respect to all similarly situated Employees; and
 - (iii) No Compensation is included in more than one Limitation Year.
- (d) Employer. Employer will mean the employer that adopts this Plan and all members of a group of employers that constitutes a controlled group of corporations or trades or businesses under common control (as defined in Code sections 414(b) and (c), as modified by Code section 415(h)), or an affiliated service group (as defined in Code section 414(m)) of which the adopting employer is part and any other entity required to be aggregated with the Employer under Code section 414(o) and the regulations issued thereunder. In addition, if the Plan is a Multiple Employer Plan, then the Employer will include any Participating Employer, along with their Related Employer who adopts this Plan.
- **3.2.6 Controlled Businesses.** If this Plan provides contributions or benefits for one or more Owner-Employees who control both the business for which this Plan is established and one or more other trades or businesses, this Plan must, when looked at as a single plan, satisfy Code section 401(a), and contributions on behalf of any Owner-Employee may be made only with respect to the Earned Income of such Owner-Employee that is derived from the trade or business with respect to which this Plan is established.

ARTICLE III

FIDUCIARIES

- **3.3.1 Standard of Conduct.** The duties and responsibilities of the Plan Administrator with respect to the Plan will be carried out (a) in a nondiscriminatory manner; (b) for the exclusive benefit of Participants and their Beneficiaries; (c) by defraying the reasonable expenses of administering the Plan; (d) with the care, skill, prudence, and diligence under the circumstances then prevailing that a prudent person acting in a like capacity and familiar with such matters would use in the conduct of an enterprise of a like character and with like aims; (e) by diversifying the investments of the Plan to minimize the risk of large losses, unless under the circumstances it is clearly prudent not to do so; and (f) in accordance with the documents and instruments governing the Plan insofar as such documents and instruments are consistent with the provisions of the Act.
- **3.3.2 Individual Fiduciaries.** At any time that a group of individuals is acting in the same Fiduciary capacity, the Employer will determine the number of such persons who will act in such capacity from time to time. Such persons will be appointed by the Employer and may or may not be Participants or Employees of the Employer. Unless otherwise provided for by agreement or by direction of the Employer, any action taken by a group of individuals acting as Plan Administrator will be taken at the direction of a majority of such persons, or, if the number of such persons is 2, by unanimous consent.
- **3.3.3 Disqualification from Service.** No person will be permitted to serve as a Fiduciary, custodian, counsel, agent, or employee of the Plan or as a consultant to the Plan who has been convicted of any of the criminal offenses specified in the Act.
- **3.3.4 Bonding.** Except as otherwise permitted by law, every Fiduciary or other person who handles funds or other property or assets of the Plan will be bonded in accordance with the requirements of the Act.
- **3.3.5 Prior Acts.** No Fiduciary will be liable for any acts occurring prior to the period of time during which the Fiduciary was actually serving in such capacity with respect to the Plan.
- **3.3.6 Insurance and Indemnity.** The Employer may purchase or cause the Trustee to purchase and keep current as an authorized expense liability insurance for the Plan, its Fiduciaries, and any other person to whom any financial or other administrative responsibility with respect to the Plan and the Trust is allocated or delegated, from and against any and all liabilities, costs, and expenses incurred by such persons as a result of any act or omission to act in connection with the performance of the duties, responsibilities, and obligations under the Plan and under the Act; provided that any such insurance policy purchased with Plan assets permits subrogation by the insurer against the Fiduciary in the case of breach by such Fiduciary.

Unless otherwise determined and communicated to affected parties by the Employer, the Employer will indemnify and hold harmless each such person, other than a corporate trustee, for and from any such liabilities, costs, and expenses that are not covered by any such insurance, except to the extent that any such liabilities, costs, or expenses are judicially determined to be due to the gross negligence or willful misconduct of such person. No Plan assets may be used for any such indemnification.

3.3.7 Expenses. Expenses incurred by the Plan Administrator or the Trustee in the administration of the Plan and the Trust, including appropriate fees for legal services regarding the Plan, such compensation to the Trustee as may be agreed upon in writing from time to time between the Employer and the Trustee, and all other proper charges and expenses of the Plan Administrator or the Trustee and of their agents and counsel will be paid by the Employer, or at its election at any time or from time to time, may be charged against the assets of the Trust, but until so paid will constitute a charge upon the assets of the Trust.

The Trustee will have the authority to charge the Trust Fund for its compensation and reasonable expenses unless paid or contested by written notice by the Employer within 60 days after mailing of the written billing

by the Trustee. All taxes of any and all kinds whatsoever that may be levied or assessed under existing or future laws upon the assets of the Trust or the income thereof will be paid from such assets. Notwithstanding the foregoing, no compensation will be paid to any Employee for services rendered under the Plan and the Trust as a Trustee.

- **3.3.8 Agents, Accountants, and Legal Counsel.** The Plan Administrator will have authority to employ suitable agents, custodians, investment counsel, accountants and legal counsel who may, but need not be, legal counsel for the Employer. The Plan Administrator and the Trustee will be fully protected in acting upon the advice of such persons. The Trustee will at no time be obliged to institute any legal action or to become a party to any legal action unless the Trustee has been indemnified to the Trustee's satisfaction for any fees, costs, and expenses to be incurred in connection therewith.
- **3.3.9 Investment Manager.** The Employer may employ as an investment manager or managers to manage all or any part of the Trust Fund any (a) investment advisor registered under the Investment Advisors Act of 1940; (b) bank as defined in said Investment Advisors Act; or (c) insurance company qualified to perform investment management services in more than one state. Any investment manager will have all powers of the Trustee in the management of such part of the Trust Fund, including the power to acquire or dispose of assets.

In the event an investment manager is so appointed, the Trustee will not be liable for the acts or omissions of such investment manager or be under any obligation to invest or otherwise manage that part of the Trust Fund that is subject to the management of the investment manager. The Employer will notify the Trustee in writing of any appointment of an investment manager, and will provide the Trustee with the investment manager's written acknowledgment that it is a Fiduciary with respect to the Plan.

- **3.3.10 Finality of Decisions or Acts.** Except for the right of a Participant or Beneficiary to appeal the denial of a claim, any decision or action of the Plan Administrator or the Trustee made or done in good faith upon any matter within the scope of authority and discretion of the Plan Administrator or the Trustee will be final and binding upon all persons. In the event of judicial review of actions taken by any Fiduciary within the scope of their duties in accordance with the terms of the Plan and the Trust, such actions will be upheld unless determined to have been arbitrary and capricious.
- **3.3.11 404(c) Election.** The Employer may designate in the Adoption Agreement that this Plan is intended to be administered pursuant to Labor Regulations section 2550.404c-1 as an ERISA section 404(c) Plan.

ARTICLE IV

PLAN ADMINISTRATOR

- **3.4.1 Administration of Plan.** The Employer will designate the Plan Administrator from time to time. The primary responsibility of the Plan Administrator is to administer the Plan for the exclusive benefit of the Participants and their Beneficiaries, subject to the specific terms of the Plan. The Plan Administrator will administer the Plan and will construe and determine all questions of interpretation or policy in their sole discretion. The Plan Administrator may correct any defect, supply any omission, or reconcile any inconsistency in such manner and to such extent as they will deem necessary or advisable to carry out the purpose of the Plan; provided, however, that any interpretation or construction will be done in a nondiscriminatory manner and will be consistent with the intent that the Plan will continue to be a qualified Plan pursuant to the Code, and will comply with the terms of the Act. The Plan Administrator will have all powers necessary or appropriate to accomplish their duties under the Plan.
- (a) The Plan Administrator will be responsible for the general administration of the Plan, including but not limited to the following duties:
 - (1) To determine all questions relating to the eligibility of an Employee to participate in the Plan or to remain a Participant in the Plan.
 - (2) To compute, certify, and direct the Trustee with respect to the amount and kind of benefits to which any Participant is entitled.
 - (3) To collect all contributions or other payments, as applicable, and transfer them to the Trustee.
 - (4) To authorize and direct the Trustee with respect to all disbursements from the Trust Fund.
 - (5) To maintain all the necessary records for the administration of the Plan.
 - (6) To interpret the provisions of the Plan and to make and publish rules and regulations for the Plan as the Plan Administrator may deem reasonably necessary for the proper and efficient administration of the Plan and consistent with its terms.
 - (7) To select the Insurer to provide any Life Insurance Policy to be purchased for any eligible Participant.
 - (8) To advise the Fiduciary responsible for Plan investments regarding the short and long-term liquidity needs of the Plan in order that the Fiduciary might direct its investment accordingly.
 - (9) To advise, counsel, and assist any Participant regarding any rights, benefits, or elections available under the Plan.
 - (10) To instruct the Trustee regarding the management, investment, and reinvestment of the Trust Fund unless the investment authority has been delegated to the Trustee or an investment manager.
- **(b)** The Plan Administrator will also be responsible for preparing and filing such annual disclosure reports and tax forms as may be required from time to time by the Secretary of Labor, the Secretary of the Treasury, or other governmental authorities.
- (c) Whenever the Plan Administrator determines it is in the best interest of the Plan and its Participants or Beneficiaries, the Plan Administrator may request such variances, deferrals, extensions, or exemptions or make such elections for the Plan as may be available under the law.

- (d) The Plan Administrator will procure bonding for all persons dealing with the Plan or its assets as may be required by law.
- **3.4.2 Disclosure Requirements.** Every Participant covered under the Plan and every Beneficiary receiving benefits under the Plan will receive from the Plan Administrator a summary plan description and such other information as may be required by law or by the terms of the Plan.
- **3.4.3 Information Generally Available.** The Plan Administrator will make copies of this Plan and the Trust, the Adoption Agreement, the summary plan description, latest annual report, Life Insurance Policies, or other instruments under which the Plan was established or is operated available for examination by any Participant or Beneficiary in the principal office of the Plan Administrator and such other locations as may be necessary to make such information reasonably accessible to all interested parties. Subject to a reasonable charge to defray the cost of furnishing such copies, the Plan Administrator shall, upon written request of any Participant or Beneficiary, furnish a copy of any of the above documents to the respective party.
- **3.4.4 Statement of Account.** Upon written request to the Plan Administrator once during any 12 month period, a Participant or Beneficiary will be furnished with a written statement, based on the latest available information, of their then vested Account balance and the earliest date upon which the same will become fully vested and nonforfeitable. The statement will also include a notice to the Participant of any benefits that are forfeitable if the Participant dies before a certain date.
- **3.4.5 Explanation of Rollover Treatment.** The Plan Administrator shall, when making a distribution eligible for rollover treatment, provide a written explanation to the recipient of the provisions under which such distribution will not be subject to tax if transferred to an Eligible Retirement Plan within 60 days after the date on which the recipient received the distribution and, if applicable, the provisions of law pertaining to the tax treatment of lump sum distributions.
- **3.4.6 Electromechanical Communications.** Electromechanical communications are permitted, but not required, among all parties to a communication regarding the Plan, including the Employer, Trustee, Plan Administrator, and Participant. Each Plan Fiduciary may adopt policies to govern its use of electromechanical communications, including policies developed by third parties working with the Fiduciary on Plan matters.

The Trustee (or other agent appointed for this purpose) may, but is not required to, act upon receipt of directions (including, without limitation, directions pursuant to voice response systems, facsimile, or other electromechanical means) with regard to all Plan transactions. The Trustee may comply with such direction and will incur no liability for doing so. If the Trustee permits the use of electromechanical directions, the giving of such directions will have the same meaning as if the directions were in writing and delivered as a document.

In the sole judgment of the Plan Administrator, if any Participant or Beneficiary entitled to benefits is not capable of managing their benefits under the Plan, any election that may be made by a Participant or Beneficiary may be made by the legal representative of such individual.

If the Plan Administrator, in their sole discretion is not satisfied with the power of a legal representative to act on behalf of a Participant or Beneficiary, the Plan Administrator may require an order of a court of competent jurisdiction.

3.4.7 Elections on Behalf of an Incapacitated Person. In the sole judgment of the Plan Administrator, if any person entitled to benefits under the Plan is incapacitated, any election that such person is permitted to make under the Plan may be made by the legal representative of such person.

If the Plan Administrator, in its sole discretion, is not satisfied with the power of a legal representative to act on behalf of a person, the Plan Administrator may require an order from a court of competent jurisdiction.

- **3.4.8 Operational Rules for Related Employer Groups.** If the Plan Sponsor and/or an Participating Employer has one or more Related Employers, that Employer and such Related Employer(s) constitute a Related Employer group in such case, the following rules apply to the operation of the Plan:
- (a) If the term "Employer" is used in the context of administrative functions necessary in the operation, maintenance, or termination of the Plan, only the Employer executing the Signature Page of this Plan is treated as the Employer.
- **(b)** Hours of Service are determined by treating all members of the Related Employer group as the Employer.
- (c) The term Excluded Employee is determined by treating all members of the Related Employer group as the Employer, except as specifically provided in the Plan.
- (d) Compensation is determined by treating all members of the Related Employer group as the Employer, except as specifically provided in the Plan.
- (e) An Employee is not treated as separated from service or terminated from employment if the Employee is employed by any member of the Related Employer group.
- (f) The Annual Additions Limitation described in Part 3 Article 2 and the Top-Heavy Plan rules described in Part 2, Article 6 are applied by treating all members of the Related Employer group as the Employer.

In all other contexts, the term "Employer" generally means a reference to all members of the Related Employer group, unless the context requires otherwise. If the terms of the Plan are ambiguous with respect to the treatment of the Related Employer group as the Employer, the Plan Administrator has the authority to make a final determination or the proper interpretation of the Plan.

3.4.9 Use of Electronic Medium. The use of an electronic medium to provide applicable notices and to make Participant elections must comply with Treasury Regulation 1.401(a)-21 that reflects the requirements for electronic signatures contained in the Electronic Signatures in Global and National Commerce Act, Public Law 106-229 (114 Stat. 464 (2000) (ESIGN).

ARTICLE V

PARTICIPANT LOANS

- **3.5.1 Authorization.** If the Employer elects in the Adoption Agreement to permit loans to Participants or Beneficiaries, the Plan Administrator will establish a Participant loan program in compliance with Treasury Regulations and Labor Regulations section 2550.408b. The terms of such Participant loan program will be in writing and will constitute part of the Plan. Such terms will include:
- (a) The identity of the person or positions authorized to administer the Participant loan program;
- **(b)** A procedure for applying for loans;
- (c) The basis on which loans will be approved or denied;
- (d) Limitations (if any) on the loan amount available, the number of loans, and the permitted purposes for a loan;
- (e) The procedure under the program for determining a reasonable rate of interest;
- (f) The types of collateral that may secure a Participant loan; and
- (g) The events constituting default and the steps that will be taken to preserve Plan assets in the event of default.
- **3.5.2 Spousal Consent.** Subject to Subsection 2.5.5(c), a Participant must obtain the written consent of their spouse, if any, to the use of the Participant's interest in the Plan as security for a Participant loan within 90 days before the date on which the loan is to be so secured, unless the Participant establishes the consent of the spouse cannot be obtained or the total value of the Participant's Accounts is less than \$5,000. A new consent must be obtained whenever the amount of the loan is increased or if the loan is renegotiated, extended, renewed, or otherwise revised. The consent must acknowledge the effect of such consent and be witnessed by a Plan representative or a notary public. Such consent will thereafter be binding with respect to the consenting spouse or any subsequent spouse with respect to that loan.

If a valid spousal consent has been obtained, then notwithstanding any other provision of the Plan, the portion of the Participant's vested Account balance used as a security interest held by the Plan by reason of a loan outstanding to the Participant will be taken into account for purposes of determining the amount of the Account balance payable at the time of death or distribution, but only if the reduction is used as repayment of the loan. If less than the entire amount of the Participant's vested Account balance (determined without regard to the preceding sentence) is payable to the surviving spouse, the Account balance will first be reduced by the amount of the security used as repayment of the loan before determining the benefit payable to each Beneficiary, including the surviving spouse.

3.5.3 Limitations. Except as provided in the Participant loan program, in no event will the amount loaned to any Participant or Beneficiary, when added to the outstanding balance of all other loans to the Participant or Beneficiary, exceed the lesser of (a) \$50,000.00 (reduced by the excess, if any, of the highest outstanding balance of loans from the Plan during the one-year period ending on the day before the date on which the loan was made over the outstanding balance of loans from the Plan on the date on which such loan was made) or (b) one-half of the vested and nonforfeitable interest in their Accounts, determined as of the Valuation Date coinciding with or immediately preceding such loan or \$10,000, if greater.

For purposes hereof, all loans from all plans of the Employer and other members of a group of employers described in Code sections 414(b), (c), (m), and (o) will be aggregated. All loans must be adequately secured and bear a reasonable interest rate. No Participant loan will exceed the present value of the Participant's vested Account balance. In the event of a default, foreclosure on the note evidencing the loan

and attachment of the security will not occur until a distributable event occurs.

- **3.5.4 Availability.** If loans are authorized in the Adoption Agreement, they must be available to all Participants and Beneficiaries on a reasonably equivalent basis without regard to any individual's race, color, religion, sex, age, or national origin. Loans will not be made available to Highly Compensated Employees in an amount greater than the amount made available to other Employees.
- **3.5.5 Default.** In the event of default, foreclosure on the note and attachment of security will not occur until a distributable event occurs for the Participant.
- **3.5.6 Qualified Military Service.** If elected in the Participant loan program, the obligation to repay any loan made to a Participant who is performing service in the uniformed services will be suspended, as permitted under Code section 414(u)(4). Such a suspension of loan repayments will not trigger a distribution under Code section 72(p) or constitute a loan default.

ARTICLE VI

BENEFICIARIES

3.6.1 Designation of Beneficiaries. Subject to the provisions of Sections 2.5.5 and 2.5.6, each Participant will have the right to designate a Beneficiary or Beneficiaries and contingent or successive Beneficiaries to receive any benefits provided by this Plan that become payable upon the Participant's death. The Participant may change the Beneficiaries at any time or times by the filing of a new designation with the Plan Administrator, and the most recent designation will govern.

In the absence of a designation, the Designated Beneficiary will be the surviving spouse of the Participant, unless such surviving spouse consents in writing to an alternate designation and the terms of such consent acknowledge the effect of such alternate designation and the consent is witnessed by a representative of the Plan or by a notary public. The designation of a Beneficiary other than the spouse of the Participant or the designation of an optional form of benefit with the consent of such spouse may not be changed without the consent of such spouse, and any consent must acknowledge the specific nonspouse Beneficiary, including any class of Beneficiaries or any contingent Beneficiaries, unless the spouse expressly permits designations by the Participant without any further spousal consent.

Regardless of any prior designation, a Participant, at any time, may designate their spouse as Beneficiary, and with respect to Accounts providing annuities, may also designate the form of benefit as a Qualified Joint and Survivor Annuity.

In the event the Participant is divorced, the ex-spouse will be treated as having predeceased the Participant and benefits will go to the secondary Designated Beneficiaries or if none survive the Participant, to their surviving children, equally, or if none, such other heirs, or the executor or administrator of their estate, as the Plan Administrator will select. In the event the Participant intends that their ex-spouse remain their Beneficiary, a post-divorce Beneficiary designation must be completed naming the ex-spouse as a Designated Beneficiary.

Any consent by a spouse obtained under this provision (or establishment that the consent of a spouse may not be obtained) will be effective only with respect to such spouse. A consent that permits designations by the Participant without any requirement of further consent by such spouse must acknowledge that the spouse has the right to limit consent to a specific Beneficiary, and a specific form of benefit where applicable, and that the spouse voluntarily elects to relinquish either or both of such rights. A Participant may revoke a prior waiver without the consent of the spouse at any time prior to the commencement of benefits. The number of revocations will not be limited. No consent obtained under this provision will be valid unless the Participant has received notice as provided in Sections 2.5.5 and 2.5.6.

3.6.2 Absence or Death of Beneficiaries. Except with respect to the process of any life insurance payable upon the death of the Participant, if a Participant dies without having a Beneficiary designation then in force, or if all of the Participant's Designated Beneficiaries predecease them, their Beneficiary will be their surviving spouse, or if none, their surviving children, equally, or if none, such other heirs, or the executor or administrator of their estate, as the Plan Administrator will select.

Unless otherwise designated by the Participant, if a Participant dies and is survived by some, but not all, of their Designated Beneficiaries, such surviving Beneficiaries will be deemed their sole Beneficiaries. In the event that a surviving Beneficiary dies before the complete distribution of the deceased Participant's interest, the estate of such Beneficiary will be deemed to be the Beneficiary of the undistributed portion of such interest.

3.6.3 Surviving Spouse Election. If the Plan is designated in the Adoption Agreement as a Cash or Deferred Profit Sharing Plan or a Profit Sharing Plan and the Employer does not elect a life annuity form of distribution in the Adoption Agreement, a surviving spouse, who has not consented to an alternate designation under Section 3.6.1, above, may elect to have distribution of the Participant's vested Account

balance commence within the 90-day period following the date of the Participant's death. The Account balance will be adjusted for gains or losses occurring after the Participant's death in accordance with the provisions of the Plan governing the adjustment of Account balances for other types of distributions, as well as for the outstanding balance of any Plan loan to the Participant.

3.6.4 Right to Direct on Behalf of a Participant or Beneficiary. In the sole judgment of the Plan Administrator, if any Participant or Beneficiary entitled to benefits is not capable of managing their benefits under the Plan, any election that may be made by a Participant or Beneficiary may be made by the legal representative of such individual, or in the case of a minor Beneficiary, the guardian, under valid power of attorney, a court appointed guardian or any other person authorized under state law to receive the benefit.

If the Plan Administrator, in their sole discretion is not satisfied with the power of a legal representative to act on behalf of a Participant or Beneficiary the Plan Administrator may require an order of a court of competent jurisdiction.

ARTICLE VII

CLAIMS

3.7.1 Claim Procedure (Non-Disability). Any Participant or Beneficiary who is entitled to a payment of a benefit for which provision is made in this Plan will file a written claim with the Plan Administrator on such forms as furnished by the Plan Administrator and will furnish such evidence of entitlement to benefits as the Plan Administrator may require. The Plan Administrator will notify the Participant or Beneficiary in writing of the amount of benefit to which they are entitled, the duration of such benefit, the time the benefit is to commence, and other pertinent information concerning their benefit.

If the Plan Administrator denies a claim for benefits, in whole or in part, the Plan Administrator will provide adequate notice in writing to the Participant or Beneficiary whose claim for benefits has been denied within 90 days after receipt of the claim unless special circumstances require an extension of time for processing the claim. If such an extension of time for processing is required, the Plan Administrator will furnish written notice to the Participant or Beneficiary indicating the special circumstances and the date by which a final decision is expected to be rendered. In no event will the period of extension exceed 180 days after receipt of the claim. The notice of denial of the claim will include (a) the specific reason or reasons for the denial: (b) specific reference to pertinent Plan provisions on which the denial is based; (c) a description of any additional material or information necessary for the claimant to perfect the claim and an explanation of why such material or information is necessary; and (d) a statement that any appeal of the denial must be made by giving to the Plan Administrator, within 60 days after receipt of the notice of the denial, written notice of such appeal, such notice to include a full description of the pertinent issues and basis of the claim. The Participant or Beneficiary (or their duly authorized representative) may review pertinent documents and submit issues and comments in writing to the Plan Administrator. If the Participant or Beneficiary fails to appeal such action to the Plan Administrator in writing within the prescribed period of time, the Plan Administrator's adverse determination will be final, binding, and conclusive. Benefits under this Plan will be paid only if the Plan Administrator decides in their discretion that the applicant is entitled to them.

3.7.2 Appeal (Non-Disability). If the Plan Administrator receives from a Participant or a Beneficiary, within the prescribed period of time, a notice of an appeal of the denial of a claim for benefit, such notice and all relevant materials will immediately be submitted to the Employer. The Employer may hold a hearing or otherwise ascertain such facts as it deems necessary and will render a decision that will be binding upon both parties.

The Employer's decision will be made within 60 days after the Plan Administrator receives the notice of appeal, unless special circumstances require an extension of time for processing, in which case the Employer will render a decision as soon as possible but not later than 120 days after receipt of the request for review. If such an extension of time is required, written notice of the extension will be furnished to the claimant prior to the commencement of the extension. The Employer's decision will be in writing, will include specific reasons for the decision, written in a manner calculated to be understood by the claimant, as well as specific references to the pertinent Plan provisions on which the decision is based, and will be promptly furnished to the claimant. Benefits under this Plan granted pursuant to such an appeal will be paid only if the Employer decides in its discretion that the applicant is entitled to them.

3.7.3 Claims Involving Disability. If the Plan Administrator receives a Disability claim from a Participant or a Beneficiary, and the final determination of Disability is being made by the Plan Administrator, the claim will be reviewed in accordance with this subsection. If a claim for a Disability benefit is denied by the Plan Administrator, in whole or in part, the Plan Administrator will provide adequate notice, in writing, to the Participant or Beneficiary whose claim for benefit has been denied. The notice must be provided within 45 days after receipt of the claim that has been filed in accordance with the reasonable procedures of the Plan unless the Plan Administrator determines that an extension of time of up to 30 days is required due to matters beyond control of the Plan. If such an extension is required, the Plan Administrator will furnish written notice of the extension prior to the end of the initial 45 day period. The notice of extension must include information to indicate the special circumstances that required the extension and the date by which

the Plan Administrator will issue their determination (no later than 30 days beyond the end of the first deadline). If prior to the end of the first extension period the Plan Administrator determines that another extension is required due to matters beyond the control of the Plan, the deadline can be extended for a further 30 days. If such further extension is required, the Plan Administrator will furnish written notice of the extension prior to the end of the above notice period. The notice of extension must include information to indicate the special circumstances giving rise to the extension and the date by which the Plan Administrator will issue their determination. In no event will the determination extend more than 30 days beyond the end of the first deadline. Both notices of extension must contain (1) the standards on which the determination is being made, (2) the unresolved issues that prevent the Plan Administrator from making the decision, and (3) the additional information that would be needed to allow the Plan Administer to make the decision. After receipt of an extension notice the Participant or Beneficiary will have no less than 45 days to provide the information specific as need for the decision.

Following receipt of notice of the Plan Administrator decision the Participant or Beneficiary will have 180 days to appeal such denial. Upon receipt of such appeal, the Plan Administrator must act within 45 days.

In addition, with respect to a claim for Disability benefits as described above, the Participant or Beneficiary must be provided, free of charge, any new or additional evidence or rationale obtained by the Plan during pendency of appeal. The Participant or Beneficiary will have the right to review and respond to the additional information. This information must include any new or additional evidence considered, relied upon, or generated by the Plan, insurer, or other person making the benefit determination in connection with the claim. This information must be provided as soon as possible and sufficiently in advance of the Plan's decision on appeal to give the Participant or Beneficiary a reasonable opportunity to respond.

All claims subject to this Section 3.7.3 must be adjudicated in a manner designed to ensure the independence and impartiality of the persons involved in making the decision. Therefore, decisions regarding hiring, compensation, termination, promotion, or similar matters with respect to any individual must not be made based upon the likelihood that the individual will support the denial of the Disability claim. In addition, with respect to a Disability claim, both the initial denial letter and any appeal denial letter must be provided in a culturally and linguistically appropriate manner and must specifically set forth the following:

- 1. The specific basis for disagreeing with any determination by the Social Security Administration or other third party disability payer, or any views of health care professionals or vocational professionals treating a claimant, regardless of whether the information was relied upon.
- 2. If the adverse determination is based on a medical necessity, experimental treatment, or similar exclusion or limit, either (1) an explanation of the scientific or clinical judgment for the determination, or (2) the denial letter must make a statement that such an explanation will be made available upon request at no charge.
- **3.** All internal rules, guidelines, protocols, standards or other similar criteria that were relied upon in denying the claim or a statement that such criteria do not exist.
- A statement that the Participant or Beneficiary is entitled to receive, upon request and free of charge, relevant documents.
- **5.** In the case of an appeal denial letter, the letter must describe any contractual limitation period for a lawsuit and the expiration date for that limitation period along with a statement that the limitation period may not expire before the conclusion of the Plan's internal appeals process.
- 6. If a claimant's address is in a county where at least 10% of the individuals in that population are literate only in the same non-English language (as determined by the United States Census Bureau), the notices must include a statement in the non-English language about the availability of language services. In addition, the Plan must provide a customer assistance process (such as a telephone hotline) with oral language services in the non-English language and provide written notices in that language upon request.

Failure to adhere to the special requirements applicable to benefit determinations pursuant to this Section 3.7.3 will result in the Participant or Beneficiary being deemed to have exhausted the Plan's administrative remedies, thus giving such Participant or Beneficiary the right to pursue court action, unless the violation was de minimis; non-prejudicial; attributable to good cause or matters beyond the Plan's control; in the context of on-going good-faith exchange of information; and not reflective of a pattern or practice of non-compliance.

The administrative remedies available under the Plan with respect to claims for disability benefits will not be deemed exhausted based on de minimis violations that do not cause, and are not likely to cause, prejudice or harm to the claimant so long as the Plan demonstrates that the violation was for good cause or due to matters beyond the control of the Plan and that the violation occurred in the context of an ongoing, good faith exchange of information between the Plan and the claimant. This exception is not available if the violation is part of a pattern or practice of violations by the Plan. The claimant may request a written explanation of the violation from the Plan, and the Plan must provide such explanation within 10 days, including a specific description of its basis, if any, for asserting that the violation should not cause the administrative remedies available under the Plan to be deemed exhausted. If a court rejects the claimant's request for immediate review under on the basis that the Plan met the standards that the violation was de minimis; non-prejudicial; attributable to good cause or matters beyond the Plan's control, the claim will be considered as re-filed on appeal upon the Plan's receipt of the decision of the court. Within a reasonable time after the receipt of the decision, the Plan will provide the claimant with notice of the resubmission.

3.7.4 Claims Appeal Involving Disability. If the Plan Administrator receives a notice of an appeal of the denial of a claim for benefit from a Participant or a Beneficiary, within the prescribed period of time, such notice and all relevant materials will immediately be submitted to the Employer. The Employer may hold a hearing or otherwise ascertain such facts as it deems necessary and will render a decision, which will be binding upon both parties.

The decision of the Employer will be made within 60 days (45 days for Disability claims) after the receipt by the Plan Administrator of the notice of appeal, unless special circumstances require an extension of time for processing, in which case a decision of the Employer will be rendered as soon as possible but not later than 120 days after receipt of the request for review. If such an extension of time is required, written notice of the extension will be furnished to the claimant prior to the commencement of the extension. The decision of the Employer will be in writing, will include specific reasons for the decision, written in a manner calculated to be understood by the claimant, as well as specific references to the pertinent Plan provisions on which the decision is based and will be promptly furnished to the claimant. Benefits under this Plan granted pursuant to such an appeal will be paid only if the Employer decides in their discretion that the applicant is entitled to them.

ARTICLE VIII

AMENDMENT AND TERMINATION

3.8.1 Right to Amend.

- (a) The Employer may at any time or times amend the Plan, Trust, and the provisions of the Adoption Agreement, in whole or in part. The Employer specifically reserves the right to amend the Plan retroactively. Subject to Subsection (b), an Employer that amends the Plan will no longer participate in this Plan and will be considered to have an individually designed plan.
- (b) The Employer may change the choice of options in the Adoption Agreement, add overriding language in the Adoption Agreement when such language is necessary to satisfy Code sections 415 or 416 because of the required aggregation of multiple plans, and add certain sample or model amendments published by the Internal Revenue Service or other required good faith amendments that specifically provide that their adoption will not cause the Plan to be treated as individually designed. An Employer that amends the Plan for any other reason, including a waiver of the minimum funding requirements under Code section 412(d), will no longer have reliance on the Sponsor's advisory letter.

An Employer that has adopted the Plan may amend the Trust or custodial account document provided such amendment merely involves the specifications of the names of the Plan, Employer, Trustee or custodian, Plan Administrator or other Fiduciaries, the Trust Year, or the name of any pooled trust in which the Plan's Trust will participate.

An Employer that has adopted the Plan will not be considered to have an individually designed plan merely because the Employer amends administrative provisions of the Trust or custodial account document (such as provisions relating to investments and duties of Trustees) so long as the amended provisions are not in conflict with any other provision of the Plan and do not cause the Plan to fail to qualify under Code section 401(a).

An Employer may also add or change provisions permitted under the Plan and/or specify or change the effective date of a provision as permitted under the Plan and correct obvious and unambiguous typographical errors and/or cross-references that merely correct a reference but that do not in any way change the original intended meaning of the provisions.

3.8.2 Manner of Amending. The Employer will make each Plan amendment by delivery to the Plan Administrator of a copy of the Employer resolution that adopts and sets forth such amendment.

3.8.3 Limitations On Amendments.

- (a) No Plan amendment shall:
 - (1) Directly or indirectly operate to give the Employer any interest whatsoever in the assets of the Trust or custodial account or to deprive any Participant or Beneficiary of their vested and nonforfeitable interest in the assets of the Trust as then constituted, or cause any part of the income or corpus of the Trust to be used for, or diverted to, purposes other than the exclusive benefit of Employees or their Beneficiaries;
 - (2) Increase the duties or liabilities of the Trustee without the Trustee's prior written consent;
 - (3) Change the vesting schedule under the Plan if the nonforfeitable percentage of the Account balance derived from Employer Contributions (determined as of the later of the date such amendment is adopted or the date such amendment becomes effective) of any Participant is less than such nonforfeitable percentage computed without regard to such amendment; or

- (4) Reduce or eliminate the accrued benefit of a Participant within the meaning of Code section 411(d)(6), or otherwise places greater restrictions or conditions on a participant's rights, except to the extent permitted under Code section 412(d)(2) or to the extent permitted under sections 1.411(d)-3 and 1.411(d)-4 of the regulations. An amendment that has the effect of decreasing a Participant's Account balance with respect to benefits attributable to service before the amendment will be treated as reducing an accrued benefit.
- (b) If a Plan amendment changes the vesting schedule or the Plan is amended in any way that directly or indirectly affects the computation of the Participant's nonforfeitable percentage or if the Plan is deemed amended by an automatic change to or from a Top-Heavy vesting schedule, each Participant who has completed 3 Years of Service (or in the case of Participants who do not have at least 1 Hour of Service in any Plan Year beginning after 1988, 5 or more Years of Service) may elect within a reasonable period after the adoption of such amendment to have their nonforfeitable percentage computed without regard to such amendment or change. The period during which such Participant may make the election will commence with the date the amendment is adopted or deemed to be made and will end on the latest of 60 days after the latest of the date:
 - (1) The amendment is adopted;
 - (2) The amendment becomes effective; or
 - (3) The Employer or Plan Administrator issues the Participant written notice of the amendment.
- (c) No Plan amendment will eliminate or restrict an optional form of benefit. The preceding sentence will not apply to a Plan amendment that eliminates or restricts the ability of a Participant to receive payment of their Account balance under a particular optional form of benefit if the amendment provides a single-sum distribution form that is otherwise identical to the optional form of benefit eliminated or restricted. For purposes of this condition, a single-sum distribution form is otherwise identical only if it is identical in all respects to the eliminated or restricted optional form of benefit (or would be identical except that it provided greater rights to the Participant) except with respect to the timing of payments after commencement.
- **3.8.4 Plan Termination.** The Employer may elect to terminate the Plan or termination may occur based on certain business transactions. Upon termination, the Plan must provide for the immediate 100% vesting of the nonforfeitable interest in all of an affected Participant's account balance. Notwithstanding the effective date of the termination, the Trust will continue until the entire Distributable Benefit of each Participant has been distributed.
- (a) Voluntary Termination. The Employer may terminate the Plan at any time by delivering to the Trustee an instrument in writing that designates such termination. Following termination of the Plan, the Trust will continue until the entire Distributable Benefit of each Participant has been distributed.
- (b) Involuntary Termination. The Plan will terminate if: (a) the Employer is dissolved or adjudicated bankrupt or insolvent in appropriate proceedings, or if a general assignment is made by the Employer for the benefit of creditors, or (b) the Employer loses its identity by consolidation or merger into one or more corporations or organizations, unless within 90 days after such consolidation or merger, such corporations or organizations elect to continue the Plan.
- (c) Partial Termination. The Plan will be subject to the rules under section 411(d)(3) of the Code, where it has been determined that a partial termination has occurred by reason of: (1) a plan amendment that adversely affect the rights of the Employees, or (2) by the Employer's severance of a group of Employees who were previously covered under the Plan. Such determination will depend on the facts and circumstances. Notwithstanding, if the Employer's turnover rate is at least 20% due to severance of employment, there is a presumption that a partial termination of the Plan has occurred. If a partial termination occurs, all Participants who had a severance from employment on account of the Partial Termination during the period will become immediately 100% vested in their account.

- (d) Liquidation of all Unallocated Accounts. Following the termination of the Plan, the Employer will allocate any unallocated amounts due to excess annual additions or forfeitures that are held in a suspension account subject to limitation in section 415(c) of the Code. Notwithstanding any unallocated amounts remaining that exceed the limitations imposed under section 415(c) of the Code to the extent set forth in rules prescribed by the Commissioner will revert to the Employer.
- (e) Limitation on Establishing a Successor Plan in the Case of a Cash or Deferred Arrangement. The Employer who elects to terminate a Plan that is designated in the Adoption Agreement as a Cash or Deferred Profit Sharing Plan and thereafter liquidates the Trust may not establish a new plan any time during the period beginning on the effective date of the Plan's termination and ending 12 months after all assets have been distributed from the Plan, and further, a distribution may not be made under Article V if the Employer establishes or maintains an alternative defined contribution plan. For purposes of the preceding sentence, the term "Employer" is applied as of the date of plan termination, and a plan is an alternative defined contribution plan only if it is a defined contribution plan that exists at any time during the period beginning on the date of plan termination and ending 12 months after distribution of all assets from the terminated plan. However, if at all times during the 24-month period beginning 12 months before the date of plan termination, fewer than 2% of the employees who were eligible under the defined contribution plan that includes the cash or deferred arrangement as of the date of plan termination are eligible under the other defined contribution plan, the other plan is not an alternative defined contribution plan. In addition, a defined contribution plan is not treated as an alternative defined contribution plan if it is an employee stock ownership plan as defined in Code sections 4975(e)(7) or 409(a), a simplified employee pension as defined in Code section 408(k), a SIMPLE IRA plan as defined in section 408(p), a plan or contract that satisfies the requirements of Code section 403(b), or a plan that is described in Code sections 457(b) or (f).
- **3.8.5 Withdrawal By Employer.** The Employer may withdraw from participation under the Plan without terminating the Trust upon making a transfer of the Trust assets to another Plan, which will be deemed to constitute an amendment in its entirety of the Trust.
- **3.8.6 Powers Pending Final Distribution.** Until final distribution of the assets of the Trust, the Plan Administrator and Trustee will continue to have all the powers provided under this Plan as are necessary for the orderly administration, liquidation, and distribution of the assets of the Trust.
- **3.8.7 Delegation to Sponsor.** The Employer expressly delegates authority to the Document Sponsor the right to amend any part of the Plan on its behalf to the extent necessary to preserve the qualified status of the Plan. For purposes of amendments by the Document Sponsor, the Mass Submitter will be recognized as the agent of the Document Sponsor. If the Document Sponsor does not adopt the amendments made by the Mass Submitter, the Plan will no longer be identical to or a minor modifier of the Mass Submitter plan. The Document Sponsor will submit a copy of the amendment to each Employer who has adopted the Plan after first having received a ruling or favorable determination from the Internal Revenue Service that the Plan as amended satisfies the applicable requirements of the Code. The Employer may revoke the authority of the Document Sponsor to amend the Plan on its behalf by written notice to the Document Sponsor of such revocation.

However, for purposes of reliance on an advisory or determination letter, the Document Sponsor will no longer have the authority to amend the Plan on behalf of the Employer as of the date (1) the Employer amends the plan to incorporate a type of plan described in section 6.03 of Rev. Proc. 2017-41 that is not permitted under the M&P program, or (2) the Internal Revenue Service notifies the Employer, in accordance with section 24.03 of Rev. Proc. 2017-41, that the Plan is an individually designed plan due to the nature and extent of Employer amendments to the Plan.

ARTICLE IX

PORTABILITY

- **3.9.1 Continuance by Successor.** In the event of the dissolution, consolidation, or merger of the Employer, or the sale by the Employer of its assets, the resulting successor person or persons, firm, or corporations may continue this Plan by (a) adopting the Plan by appropriate resolution; (b) appointing a new Trustee as though the Trustee (including all members of a group of individuals acting as Trustee) had resigned; and (c) executing a proper agreement with the new Trustee. In such event, each Participant in this Plan will have an interest in the Plan after the dissolution, consolidation, merger, or sale of assets at least equal to the interest that they had in the Plan immediately before the dissolution, consolidation, merger, or sale of assets. Any Participants who do not accept a position with such successor within a reasonable time will be deemed to be terminated. If, within 90 days from the effective date of such dissolution, consolidation, merger, or sale of assets, such successor does not adopt this Plan, as provided herein, the Plan will automatically be terminated and deemed an involuntary termination.
- **3.9.2 Merger With Other Plan.** In the event of the merger or consolidation with, or transfer of assets or liabilities to, any other deferred compensation plan and trust, each Participant will have an interest in such other plan that is equal to or greater than the interest that they had in this Plan immediately before such merger, consolidation, or transfer, and if such other plan thereafter terminates, each Participant will be entitled to a Distributable Benefit that is equal to or greater than the Distributable Benefit to which they would have been entitled immediately before such merger, consolidation, or transfer if this Plan had then been terminated, as adjusted for any investment gains or losses since such merger, consolidation, or transfer.

3.9.3 Transfers and Rollovers From Other Plans.

(a) The Employer may cause all or any of the assets held in connection with any other plan or trust that is maintained by the Employer for the benefit of its Employees and satisfies the applicable requirements of the Code relating to qualified plans and trusts to be transferred to the Trustee, whether such transfer is made pursuant to a merger or consolidation of this Plan with such other plan or trust within the meaning of Code section 414(I) or for any other allowable purpose. Any such assets so transferred to the Trustee will be accompanied by written instructions from the Employer, or the trustee, custodian, or individual holding such assets, setting forth the name of each Employee for whose benefit such assets have been transferred and showing separately the respective contributions by the Employer and by the Employee and the current value of the assets attributable thereto.

Upon receipt by the Trustee of such assets, the Trustee will place such assets in appropriate Accounts for each Employee. If the transferor plan maintained a vesting schedule that differs from the schedule maintained under this Plan, the vesting schedule of this Plan will apply, but eligible Participants will have the option to apply the vesting schedule of the transferor plan to the transferred assets in a manner consistent with the provisions of Section 3.8.3(b).

(b) If assets are transferred to the Plan from a defined benefit pension plan, money purchase pension plan, target benefit pension plan, stock bonus or profit sharing plan that are subject to the survivor annuity requirements of Code sections 401(a)(11) and 417, such transferred assets will remain subject to such requirements, and this Plan will provide life annuity distribution options as provided under the transferor plan. Unless this Plan is also subject to such survivor annuity requirements, the annuity distribution option will only apply to those assets that were transferred and any earnings therein.

Any such transferred assets and associated earnings will be distributed to the Participant only if: (1) they terminate employment with the Employer, attains Normal Retirement Age, becomes Disabled, or dies, (2) the Plan should subsequently terminate, or (3) the Employer spins off the business unit to which the Participant is a member. To implement the restrictions on distributions this Subsection imposes, the Plan must maintain separate accounting between the benefits attributable to the transferred account balances

and all other Account balances of each Participant who has Account balances attributable to such transfers.

- (c) At the direction of the sponsoring Employer, the Trustee will accept the transfer of account balances from another plan of Employees that have become Employees of the sponsoring Employer due to the spin-off of the business unit to which they belong.
- (d) If elected in the Adoption Agreement, the Plan will accept rollovers of Eligible Rollover Distributions made after December 31, 2001, from the types of plans specified in the Adoption Agreement, beginning on the effective date specified in the Adoption Agreement. Such rollover amounts will be placed in a Rollover Account for the benefit of the Employee in question.

Unless otherwise provided by the Employer in the Adoption Agreement, the Plan will accept a rollover contribution to a Roth Elective Deferral Account only if it is a Direct Rollover from another Roth elective deferral account under an applicable retirement plan described in Code section 402A(e)(1) and only to the extent the rollover is permitted under the rules of Code section 402(c).

The Plan will not provide for a Direct Rollover (including an automatic rollover) for distributions from a Participant's Roth Elective Deferral Account if the amount of the distributions that are Eligible Rollover Distributions are reasonably expected to total less than \$200 during a year. In addition, any distribution from a Participant's Roth Elective Deferral Account is not taken into account in determining whether distributions from a Participant's other accounts are reasonably expected to total less than \$200 during a year. However, Eligible Rollover Distributions from a participant's Roth Elective Deferral Account are taken into account in determining whether the total amount of the Participant's Account balances under the plan exceeds \$1,000 for purposes of mandatory distributions from the Plan.

The provisions of the Plan that allow a Participant to elect a Direct Rollover of only a portion of an Eligible Rollover Distribution but only if the amount rolled over is at least \$500 is applied by treating any amount distributed from the Participant's Roth Elective Deferral Account as a separate distribution from any amount distributed from the Participant's other accounts in the Plan, even if the amounts are distributed at the same time.

3.9.4 Transfer to Other Plans.

(a) The Trustee, upon written direction by the Employer, will transfer some or all of the assets held under the Trust to another plan or trust of the Employer or any other employer meeting the requirements of the Code relating to qualified plans and trusts, whether such transfer is made pursuant to a merger or consolidation of this Plan with such other plan or trust or for any other allowable purpose.

At the direction of the Employer, the Trustee will spin-off a portion of the Plan assets to a plan of the buyer of a portion of the assets of the Employer. The amount of the spin-off will be the portion of the assets representing the Account balances of Participants who are terminated due to the sale. In order to be a part of the spin-off, a Participant must become employed by the buyer within one year of the date of sale.

In addition, upon the termination of employment of any Participant and receipt by the Plan Administrator of a request in writing, the Participant may request that any distribution from the Trust to which they are entitled will be transferred to an individual retirement account, an individual retirement annuity, or any other plan or trust that is maintained by some other employer for the benefit of its employees and satisfies the applicable Code requirements relating to qualified plans and trusts. Upon receipt of any such written request, the Plan Administrator will cause the Trustee to transfer the assets so directed and, as appropriate, will direct the Insurer to transfer to the new trustee any applicable insurance policies issued for the benefit of the Participant.

(b) Transfer to Nonqualified Foreign Trust. If the interest of a Participant or Beneficiary under the Plan is transferred into a nonqualified foreign trust or the assets and liabilities are transferred into a plan described in Section 1165 of the Puerto Rico Code will be a taxable distribution. For this purpose a "Nonqualified Foreign Trust" is a trust: (i) created or organized outside of the United States or (ii)

attached to a plan created or organized outside of the United States and is not subject to the Internal Revenue Code.

- (c) Exception with respect to Paragraph (b) above.
 - (1) The Plan Administrator of a pension, profit sharing or stock bonus plan and trust created or organized in Puerto Rico may elect under Section 1022(i)(2)(A) of the Employee Retirement Income Security Act of 1974 (ERISA), to be treated as a qualified trust under Code section 501(a). For purposes of ERISA 1022(i): (i) the plan must cover participants who are all "bona fide" residents of Puerto Rico; or (ii) the plan must covers participants who perform labor or services primarily within Puerto Rico, regardless of their place of residence; and (iii) the plan is exempt from income tax under the laws of Puerto Rico.
 - (2) Upon such an election, the plan will be treated as a qualified trust under Code section 401(a).

ARTICLE X

INSURANCE

3.10.1 Participants Insurable at Standard Rates. The Trustee, when directed by the Plan Administrator, and in accordance with the provisions specified in the Adoption Agreement, will provide incidental insurance benefits by purchasing for each Participant insurable at standard rates, either term insurance, universal life insurance, or ordinary life insurance policies. The manner in which insurance benefits are provided under this Part III Article X must not discriminate in favor of Highly Compensated Employees.

Said Life Insurance Policies will provide a 3.10.6, prior to the Normal Retirement Date, equal to the proceeds of such policies, subject to the following provisions:

- (a) The Trustee may invest any portion or all of the assets of the Trust Fund that are attributable to a Participant in the purchase of group or individual Life Insurance Policies issued on the life of the Participant or in the case of a profit sharing plan (including a 401(k) plan), someone in whom the Participant has an insurable interest and for the benefit of the Participant with the consent of the Participant, subject to the following conditions, as they are represented by the Plan Administrator:
 - (1) If "Ordinary Life Insurance Policies" are purchased, the aggregate life insurance premiums must be less than 1/2 of the aggregate Employer contributions and Forfeitures allocated to the Participant's Account at any particular time, without regard to Trust earnings, capital gains, or losses. For purposes of this Plan, the term "Ordinary Life Insurance Policies" will mean policies with both nondecreasing death benefits and nonincreasing premiums.
 - (2) The aggregate Premiums paid for Life Insurance Policies that are either term, universal or any other policies that are not ordinary whole life Policies will not at any time exceed 25% of the aggregate amount of Employer Contributions and Forfeitures that have been allocated to the Accounts of such Participant.
 - (3) The sum of 1/2 of the aggregate premiums for Ordinary Life Insurance Policies and all premiums for other Life Insurance Policies will not at any time exceed 25% of the aggregate amount of Employer Contributions and Forfeitures that have been allocated to the Accounts of such Participant.
 - (4) For purposes of this Section, a Participant's Elective Deferrals are considered Employer Contributions.
 - (5) The Employer may elect in the Plan's policy on insurance to set minimum amounts for the initial amount of insurance purchased as well as for the purchase of additional amounts. The minimum amount will be expressed in terms of the face amount of the insurance and may not be greater than \$1,000. The Plan's policy on insurance may require that insurance must be purchased in multiples not to exceed \$1,000.
- (b) If the case of a profit sharing plan (including a 401(k) plan) that permits In-service distributions to a Participant prior to their Normal Retirement Date in accordance with Subsection 2.5.12(a) or (b), the amount that may be distributed to the Participant may be used to purchase Life Insurance Policies without limitation unless otherwise specified in the Plan's policy on insurance. Voluntary Employee Contributions and rollovers from other plans may also be used to purchase Life Insurance Policies without limitation.
- **3.10.2 Uninsurable Participants.** For each Participant found by an Insurer to be uninsurable, the Trustee, when requested by the Plan Administrator, will purchase a retirement annuity contract on the life of such Participant with a death benefit before Normal Retirement Date in an amount equal to the cash surrender value of such contract. The Participant must consent to the purchase of such a retirement

annuity contract.

- **3.10.3 Participants Insurable at Above Standard Rates.** If an Insurer finds a Participant to be uninsurable, except at above standard rates, the Trustee, when requested by the Plan Administrator, will pay premiums that do not exceed the percentage of standard rates specified in the Plan's policy on insurance. The face value of any insurance policy will be limited by the percentage of the aggregate Employer Contributions and Forfeitures allocated to the Participant's Account that the Employer dedicated to purchase insurance. The Participant must consent to the purchase of any insurance policy above standard rates.
- **3.10.4 Purchase of Policies.** No Life Insurance Policy will be purchased under the Plan unless at the time of purchase it benefits an active Participant and such policy or separate definite written agreement between the Employer and the Insurer provides that no value or credits determined by the Insurer (on account of dividends, earnings, or other experience rating credits, or surrender or cancellation credits) with respect to such policy may be paid or returned to the Employer or diverted to or used for other than the exclusive benefit of the Participants or their Beneficiaries. Any Life Insurance Policy must provide that any proceeds are payable to the Trustee, subject to the Trustee's obligation to transfer such proceeds to the Beneficiary. Under no circumstances will the Trust retain any part of the proceeds.

If, at any time, a policy described herein will not be available from or will be determined to be invalid or unenforceable by an Insurer, the Trustee, when directed by the Plan Administrator, will purchase a policy or make such other alternative arrangements that, in the opinion of the Plan Administrator, conform most closely to the descriptions herein. In the event of a conflict between the terms of the Plan and the terms of any Life Insurance Policy, the Plan provisions will control.

- **3.10.5** Applications for Policies. The Trustee will be the applicant, owner, and beneficiary for all Life Insurance Policies required under the terms of the Plan. Any Life Insurance Policy distributed from the Plan must be nontransferable. The terms of any Life Insurance Policy purchased and distributed by the Plan to a Participant or spouse will comply with the requirements of the Plan. The Trustee will not be responsible for the validity or execution of any Life Insurance Policy, failure of an Insurer to pay proceeds when due, or failure of any policy to meet requirements or conform to the provisions of the Plan.
- **3.10.6 Incidents of Ownership.** The Trustee will have the right to receive all sums payable under the terms of all Life Insurance Policies issued hereunder. The Trustee will be the complete and absolute owner of the Life Insurance Policies held in the Trust and of each and every incident of ownership therein and will have the power to exercise the rights, options, and privileges of an absolute owner with respect to the Life Insurance Policies, subject to the provisions of the Plan. The Plan Administrator will have the power to deal with and settle all claims.
- **3.10.7 Payment of Premiums.** In a Plan Year in which the Employer Contributions are less than the amount required to sustain insurance premiums, the Plan Administrator will direct the Trustee to take one of the following steps with respect to payment of insurance premiums on Life Insurance Policies:
- (a) Pay premiums due on policies from the Participant's Account, within the limits of 3.10.1 and 3.10.3 above.
- **(b)** Pay premiums due on policies from the Participant's Account using the balance described in 3.10.1(b) above,
- (c) Surrender any policies for their cash surrender values and add such amount to the Participant's Account,
- (d) Convert the cash surrender value of any policies to paid up insurance for which there will be no further premiums, or
- (e) Offer to sell any policies to the Participant for their cash surrender value for transfers or distributions made on or before February 13, 2004. Distributions or transfers of policies after February 13, 2004 sale be valued at Fair Market Value.

- **3.10.8 Discontinuance of Policies.** The Trustee or other Fiduciary responsible for making investment decisions may discontinue the investment in Life Insurance Policies at any time. If the Plan provides for Participant-directed investments, life insurance, as an investment option, may be eliminated at any time by the Plan Administrator. If life insurance investment options are discontinued, the Plan Administrator, in its sole discretion, may offer to sell the policies to the Participant, or to another person, provided the prohibited transaction exemption requirements of the Department of Labor are satisfied.
- **3.10.9 Protection of the Insurer.** An Insurer will not be responsible for the validity of the Plan or the Trust and will have no responsibility for action taken or not taken by the Trustee, for determining the propriety of accepting premium payments or other contributions, for making payments in accordance with the direction of the Trustee, or for the application of such payments. The Insurer will be fully protected in dealing with any representative of the Employer or any one of a group of individuals acting as Trustee. Until an Insurer receives written notice of a change of Trustee at its home office, the Insurer will be fully protected in dealing with any party acting as Trustee according to the latest information received by the Insurer at its home office.
- **3.10.10 No Responsibility for Act of Insurer.** Neither the Employer, the Plan Administrator, nor the Trustee will be responsible for any of the following, nor will they be liable for instituting action in connection with:
- (a) The validity of Life Insurance Policies or policy provisions;
- (b) Failure or refusal by the Insurer to provide benefits under a policy;
- (c) An act by a person that may render a policy invalid or unenforceable; or
- (d) Inability to perform or delay in performing an act, which inability or delay is occasioned by a provision of a policy or a restriction imposed by the Insurer.

ARTICLE XI

MISCELLANEOUS

3.11.1 No Reversion to Employer.

- (a) Except as specifically provided in the Plan, no part of the corpus or income of the Trust will revert to the Employer or be used for, or diverted to, purposes other than for the exclusive benefit of Participants and their Beneficiaries.
- **(b) Reversion Upon Termination.** Notwithstanding Paragraph (a), any other Plan provision or section 1.401(a)-2 of the Treasury Regulations, upon termination of the Plan, amounts contributed to the Plan that exceed the limitations imposed under section 415(c) of the Code, to the extent set forth in rules prescribed by the Commissioner will revert to the Employer.
- **3.11.2 Employer Actions.** Any action by the Employer pursuant to the provisions of the Plan will be evidenced by appropriate resolution or by written instrument executed by any person authorized by the Employer to take such action.
- **3.11.3 Execution of Receipts and Releases.** Any payment to any person eligible to receive benefits under this Plan, in accordance with the provision of the Plan or the Trust, shall, to the extent thereof, be in full satisfaction of all claims hereunder. The Plan Administrator may require such person, as a condition precedent to such payment, to execute a receipt and release therefore in such form as it will determine.
- **3.11.4 Rights of Participants Limited.** Neither the creation of this Plan and the Trust nor anything contained in this Plan or the Adoption Agreement will be construed as giving any Participant, Beneficiary, or Employee any equity or other interest in the assets, business, or affairs of the Employer, or the right to complain about any action taken by or about any policy adopted or pursued by the Employer, or as giving any Employee the right to be retained in the service of the Employer; and all Employees will remain subject to discharge to the same extent as if the Plan had never been executed. Prior to the time that distributions are made in conformity with the provisions of the Plan, neither the Participants nor their spouses, Beneficiaries, heirs-at-law, or legal representatives will receive or be entitled to receive cash or any other thing of current exchangeable value from either the Employer or the Trustee as a result of the Plan or the Trust.
- **3.11.5 Inalienability.** The right of any Participant or their Beneficiary in any distribution hereunder or to any Account will not be subject to alienation, assignment, or transfer, voluntarily or involuntarily, by operation of law or otherwise, except as may be expressly permitted herein. No Participant will assign, transfer, or dispose of such right, nor will any such right be subjected to attachment, execution, garnishment, sequestration, or other legal, equitable, or other process. The preceding will also apply to the creation, assignment, or recognition of a right to any benefit payable with respect to a Participant pursuant to a domestic relations order, unless such order is determined to be a Qualified Domestic Relations Order or any domestic relations order entered before January 1, 1985.

The preceding provisions will not apply to judgments, orders, and decrees issued, and settlement agreements entered into, on or after August 5, 1997, that offset a Participant's benefits provided under this Plan if:

- (a) The order or requirement to pay arises:
 - (1) Under a judgment of conviction for a crime involving this Plan,
 - (2) Under a civil judgment (including a consent order or decree) entered by a court in an action brought in connection with a violation (or alleged violation) of Part 4 of Subtitle B of Title I of the Act, or

- (3) Pursuant to a settlement agreement between the Secretary of Labor and the Participant, or a settlement agreement between the Pension Benefit Guaranty Corporation and the Participant, in connection with a violation (or alleged violation) of Part 4 of such subtitle by a Fiduciary or any other person,
- (b) The judgment, order, decree, or settlement agreement expressly provides for the offset of all or part of the amount ordered or required to be paid to the Plan against the Participant's benefits provided under the Plan, and
- (c) In a case in which the survivor annuity requirements of Code section 401(a)(11) apply with respect to distributions from the Plan to the Participant, if the Participant has a spouse at the time at which the offset is to be made:
 - (1) Either such spouse has consented in writing to such offset and such consent is witnessed by a notary public or representative of the Plan (or it is established to the satisfaction of a Plan representative that such consent may not be obtained by reason of circumstances described in Code section 417(a)(2)(B)), or an election to waive the right of the spouse to either a Qualified Joint and Survivor Annuity or a Qualified Preretirement Survivor Annuity is in effect in accordance with the requirements of Code section 417(a),
 - (2) Such spouse is ordered or required in such judgment, order, decree, or settlement to pay an amount to the Plan in connection with a violation of Part 4 of such subtitle, or
 - (3) In such judgment, order, decree, or settlement, such spouse retains the right to receive the survivor annuity under a Qualified Joint and Survivor Annuity or under a Qualified Preretirement Survivor Annuity,) where such survivor annuity equals the survivor annuity determined under the Plan assuming that:
 - (i) The Participant terminated employment on the date of the offset,
 - (ii) There was no offset,
 - (iii) Benefits commence at the Participant's Normal Retirement Date (or current age if later),
 - (iv) The Qualified Joint and Survivor Annuity survivor percentage is 50% (even if another percentage is selected in the definition of Qualified Joint and Survivor Annuity), and
 - (v) The Qualified Preretirement Survivor Annuity is 50% of such Qualified Joint and Survivor Annuity.

In the event a Participant's benefits are attached by order of any court, the Plan Administrator may bring an action for a declaratory judgment in a court of competent jurisdiction to determine the proper recipient of the benefits to be paid by the Plan. During the pendency of the action, the Plan Administrator will cause any benefits payable to be paid to the court for distribution by the court as it considers appropriate.

3.11.6 Qualified Domestic Relations Orders. The Plan Administrator will adhere to the terms of any judgment, decree, or order (including approval of a property settlement agreement) that relates to the provision of child support, alimony payments, or marital property rights to a spouse, former spouse, child, or other dependent of a Participant and is made pursuant to a state domestic relations law (including a community property law) and that creates or recognizes the existence of an Alternate Payee's right to, or assigns to an Alternate Payee the right to, receive all or a portion of the benefits payable with respect to a Participant.

Any such domestic relations order must clearly specify the name and last known mailing address of the Participant and the name and mailing address of each Alternate Payee covered by the order, the amount or percentage of the Participant's benefit to be paid by the Plan to each such Alternate Payee, or the manner in which such amount or percentage is to be determined, the number of payments or period to which such

order applies, and each plan to which such order applies.

Any such domestic relations order will not require the Plan to provide any type or form of benefit or any option not otherwise provided under the Plan, to provide increased benefits, or the payment of benefits to an Alternate Payee that are required to be paid to another Alternate Payee under another order previously determined to be a Qualified Domestic Relations Order. Notwithstanding the foregoing sentence, the Employer may elect to permit distributions to an Alternate Payee before the Participant has separated from service, on or after the date on which the Participant attains or would have attained the earliest retirement age under the Plan, as if the Participant had retired on the date on which such payment is to begin under such order. For this purpose, the "earliest retirement age" under the Plan means the earlier of: (a) the date on which the Participant is entitled to a distribution under the Plan, or (b) the later of the date the Participant attains age 50, or the earliest date on which the Participant could begin receiving benefits under the Plan if the Participant separated from service.

The payment of benefits to an Alternate Payee before the Participant has severed from employment will be determined as if the Participant had retired, or otherwise terminated employment, on the date on which such payment is to begin under the Qualified Domestic Relations Order, (but taking into account only the benefits actually accrued), and may be paid in any form in which such benefits may be paid under the Plan to the Participant (other than the form of a Joint and Survivor Annuity with respect to the Alternate Payee and their subsequent spouse).

If the Employer so elects in the Adoption Agreement, distributions to an Alternate Payee are permitted while the Participant continues to be employed on or after the date a Domestic Relations Order is determined to be a Qualified Domestic Relations Order by the Plan Administrator.

To the extent provided in the Qualified Domestic Relations Order, the former spouse of a Participant will be treated as a surviving spouse of such Participant for purposes of Code sections 401(a)(11) and 417 (and any spouse of the Participant will not be treated as a spouse of the Participant for such purposes) and if married for at least 1 year to the Participant, the surviving former spouse will be treated as meeting the requirements of Code section 417(d).

The Plan Administrator will promptly notify the Participant and each Alternate Payee of the receipt of a domestic relations order by the Plan and the Plan's procedures for determining the qualified status of domestic relations orders. Within a reasonable period after receipt of a domestic relations order, the Plan Administrator will determine whether such order is a Qualified Domestic Relations Order and will notify the Participant and each Alternate Payee of such determination. If the Participant or any affected Alternate Payee disagrees with the determinations of the Plan Administrator, the Plan Administrator will treat the disagreeing party as a claimant and follow the claims procedure of the Plan. The Plan Administrator may bring an action for a declaratory judgment in a court of competent jurisdiction to determine the proper recipient of the benefits to be paid by the Plan.

During any period in which the issue of whether a domestic relations order is a Qualified Domestic Relations Order is being determined (by the Plan Administrator, by a court of competent jurisdiction or otherwise), the Plan Administrator will separately account for the amounts that would have been payable to the Alternate Payee during such period if the order had been determined to be a Qualified Domestic Relations Order. If, within the 18 month period beginning on the date on which the first payment would be required to be made under the domestic relations order, the order (or modification thereof) is determined to be a Qualified Domestic Relations Order, the Plan Administrator will pay the segregated amounts, including any interest thereon, to the person or persons entitled thereto. If within such 18 month period it is determined that the order is not a Qualified Domestic Relations Order or the issue whether such order is a Qualified Domestic Relations Order is not resolved, then the Plan Administrator will pay the segregated amounts, including any interest thereon, to the person or persons who would have been entitled to such amounts if there had been no order. Any determination that an order is a Qualified Domestic Relations Order that is made after the close of the 18-month period will be applied prospectively only.

3.11.7 Missing Persons. If the Trustee mails by registered or certified mail, postage prepaid, to the last known address of a Participant or Beneficiary, a notification that the Participant or Beneficiary is entitled to a distribution and if (a) the notification is returned by the post office because the addressee cannot be

located at such address and if neither the Employer, the Plan Administrator nor the Trustee will have any knowledge of the whereabouts of such Participant or Beneficiary within a reasonable timeframe from the date such notification was mailed, or (b) within a reasonable timeframe after such notification was mailed to such Participant or Beneficiary, they does not respond thereto by informing the Trustee of their whereabouts, the ultimate disposition of the then undistributed vested Account balance of such Participant or Beneficiary will be determined in accordance with the then applicable Federal laws, rules, and regulations.

- **3.11.8 Notices.** Any notice or direction to be given in accordance with the Plan will be deemed to have been effectively given if hand delivered to the recipient or sent by certified mail, return receipt requested, to the recipient at the recipient's last known address. At any time that a group of individuals is acting in a Fiduciary capacity, notice to such Fiduciary may be given by giving notice to any one or more of such individuals.
- **3.11.9 Governing Law.** The provisions of this Plan will be construed, administered, and enforced in accordance with the provisions of the Act and, to the extent applicable, the laws of the state specified in the Adoption Agreement. All contributions to the Trust will be deemed to take place in such state.
- **3.11.10 Severability of Provisions.** In the event that any provision of this Plan will be held to be illegal, invalid, or unenforceable for any reason, said illegality, invalidity, or unenforceability will not affect the remaining provisions, but will be fully severable, and the Plan will be construed and enforced as if said illegal, invalid, or unenforceable provisions had never been inserted herein.
- **3.11.11 Gender and Number.** Whenever appropriate, words used in the singular will include the plural, and the masculine gender will include the feminine gender and vice versa.
- **3.11.12 Binding Effect.** The Plan and Adoption Agreement, and all actions and decisions hereunder, will be binding upon the heirs, executors, administrators, successors, and assigns of any and all parties hereto and Participants, present and future, and their Beneficiaries.

ARTICLE XII

Deemed IRA

- **3.12.1 Deemed Pre-tax IRA.** This Section will apply after December 31, 2005, if the Employer elects to permit Participants to make Pre-tax IRA contributions to the Plan.
- (a) The Pre-tax IRA Account will be established for the exclusive benefit of the Participant or their Beneficiaries.
- (b) The maximum permissible annual contribution will be:
 - (1) Except in the case of a rollover contribution (as permitted by Code sections 402(c), 402(e)(6), 403(a)(4), 403(b)(8), 403(b)(10), 408(d)(3), and 457(e)(16)), no contributions will be accepted unless they are in cash, and the total of such contributions will not exceed:

\$3,000 for any taxable year beginning in 2002 through 2004; \$4,000 for any taxable year beginning in 2005 through 2007; and \$5,000 for any taxable year beginning in 2008 and years thereafter.

After 2008, the limit will be adjusted by the Secretary of the Treasury for cost-of-living increases under Code section 219(b)(5)(C). Such adjustments will be in multiples of \$500.

- (2) In the case of a Participant who is 50 or older, the annual cash contribution limit is increased by:
 - \$500 for any taxable year beginning in 2002 through 2005; and \$1,000 for any taxable year beginning in 2006 and years thereafter.
- (3) No contributions will be accepted under a SIMPLE IRA plan established by any employer pursuant to Code section 408(p). Also, no transfer or rollover of funds attributable to contributions made by a particular employer under its SIMPLE IRA plan will be accepted from a SIMPLE IRA, that is, an IRA used in conjunction with a SIMPLE IRA plan, prior to the expiration of the 2 year period beginning on the date the Participant first participated in that employer's SIMPLE IRA plan.
- (c) If the Trust acquires collectibles within the meaning of Code Section 408(m) after December 31, 1981, for the benefit of any Pre-tax IRA Account, those Trust assets will be treated as a distribution in an amount equal to the cost of such collectibles.
- (d) No part of the Trust funds attributable to a Pre-tax IRA Contribution will be invested in life insurance contracts.
- (e) Distributions before death must commence no later than when the Participant attains age 70 1/2.
 - (1) Notwithstanding any provision of this Section to the contrary, the distribution of the Participant's interest in the Pre-tax IRA Account will be made in accordance with the requirements of Code section 408(a)(6) and the regulations thereunder, the provisions of which are herein incorporated by reference. If distributions are made from an annuity contract purchased from an insurance company, distributions thereunder must satisfy the requirements of Q&A-4 of Treasury Regulations section 1.401(a)(9)-6, rather than Subsections (2), (3), and (4) below and Subsection (f). The Required Minimum Distributions calculated for an IRA created under this Section may be withdrawn from another IRA of the Participant in accordance with Q&A-9 of Treasury Regulations section 1.408-8.

- (2) The entire value of the Account of the Participant for whose benefit the Account is maintained will commence to be distributed no later than the first day of April following the calendar year in which such Participant attains age 70 1/2 (the "required beginning date") over the life of such Participant or the lives of such Participant and their Designated Beneficiary.
- (3) The amount to be distributed each year, beginning with the calendar year in which the Participant attains age 70 1/2 and continuing through the year of death, will not be less than the quotient obtained by dividing the value of the IRA created under this Section (as determined under Subsection (h)) as of the end of the preceding year by the distribution period in the Uniform Lifetime Table in Q&A-2 of Treasury Regulations section 1.401(a)(9)-9, using the Participant's age as of their birthday in the year. However, if the Participant's sole Designated Beneficiary is their surviving spouse and such spouse is more than ten (10) years younger than the Participant, then the distribution period is determined under the Joint and Last Survivor Table in Q&A-3 of Treasury Regulations section 1.401(a)(9)-9, using the ages as of the Participant's and spouse's birthdays in the year.
- (4) The Required Minimum Distribution for the year the Participant attains age 70 1/2 can be made as late as April 1 of the following year. The Required Minimum Distribution for any other year must be made by the end of such year.
- (f) Death On or After Required Beginning Date. If the Participant dies on or after the required beginning date, the remaining portion of their interest will be distributed at least as rapidly as follows:
 - (1) If the Designated Beneficiary is someone other than the Participant's surviving spouse, the remaining interest will be distributed over the remaining life expectancy of the Designated Beneficiary, with such life expectancy determined using the Beneficiary's age as of their birthday in the year following the year of the Participant's death, or over the period described in Subsection (3) below if longer.
 - (2) If the Participant's sole Designated Beneficiary is the Participant's surviving spouse, the remaining interest will be distributed over such spouse's life or over the period described in Subsection (3) below if longer. Any interest remaining after such spouse's death will be distributed over such spouse's remaining life expectancy determined using the spouse's age as of their birthday in the year of the spouse's death, or, if the distributions are being made over the period described in Subsection (3) below, over such period.
 - (3) If there is no Designated Beneficiary, or if applicable by operation of Subsections (f)(1) or (f)(2) above, the remaining interest will be distributed over the individual's remaining life expectancy determined in the year of the individual's death.
 - (4) The amount to be distributed each year under Subsections (1), (2), or (3), beginning with the calendar year following the calendar year of the Participant's death, is the quotient obtained by dividing the value of the IRA created under this Section as of the end of the preceding year by the remaining life expectancy specified in such Subsection. Life expectancy is determined using the Single Life Table in Q&A-1 of Treasury Regulations section 1.401(a)(9)-9. If distributions are being made to a surviving spouse as the sole Designated Beneficiary, such spouse's remaining life expectancy for a year is the number in the Single Life Table corresponding to such spouse's age in the year. In all other cases, remaining life expectancy for a year is the number in the Single Life Table corresponding to the Beneficiary's or Participant's age in the year specified in Subsections (1), (2), or (3) and reduced by 1 for each subsequent year.
- (g) Death Before Required Beginning Date. If the Participant dies before the required beginning date, their entire interest will be distributed at least as rapidly as follows:
 - (1) If the Designated Beneficiary is someone other than the Participant's surviving spouse, the entire interest will be distributed, starting by the end of the calendar year following the calendar year of the Participant's death, over the remaining life expectancy of the Designated Beneficiary, with such life expectancy determined using the age of the Beneficiary as of their birthday in the year

following the year of the Participant's death, or, if elected, in accordance with Subsection (3) below.

- (2) If the Participant's sole Designated Beneficiary is the Participant's surviving spouse, the entire interest will be distributed, starting by the end of the calendar year following the calendar year of the Participant's death (or by the end of the calendar year in which the Participant would have attained age 70 1/2, if later), over such spouse's life, or, if elected, in accordance with Subsection (3) below. If the surviving spouse dies before distributions are required to begin, the remaining interest will be distributed, starting by the end of the calendar year following the calendar year of the spouse's death, over the spouse's Designated Beneficiary's remaining life expectancy determined using such Beneficiary's age as of their birthday in the year following the death of the spouse, or, if elected, will be distributed in accordance with Subsection (3) below. If the surviving spouse dies after distributions are required to begin, any remaining interest will be distributed over the spouse's remaining life expectancy determined using the spouse's age as of their birthday in the year of the spouse's death.
- (3) If there is no Designated Beneficiary, or if applicable by operation of Subsections (1) or (2) above, the entire interest will be distributed by the end of the calendar year containing the fifth anniversary of the Participant's death (or of the spouse's death in the case of the surviving spouse's death before distributions are required to begin under Subsection (2) above).
- (4) The amount to be distributed each year under Subsections (1) or (2) is the quotient obtained by dividing the value of the IRA created under this Section as of the end of the preceding year by the remaining life expectancy specified in such Subsection. Life expectancy is determined using the Single Life Table in Q&A-1 of Treasury Regulations section 1.401(a)(9)-9. If distributions are being made to a surviving spouse as the sole Designated Beneficiary, such spouse's remaining life expectancy for a year is the number in the Single Life Table corresponding to such spouse's age in the year. In all other cases, remaining life expectancy for a year is the number in the Single Life Table corresponding to the Beneficiary's age in the year specified in Subsections (1) or (2) and reduced by 1 for each subsequent year.
- (h) The "value" of the IRA created under this Section includes the amount of any outstanding rollover, transfer, and recharacterization under Q&As-7 and -8 of Treasury Regulations section 1.408-8.
- (i) If the sole Designated Beneficiary is the Participant's surviving spouse, the spouse may elect to treat the IRA created under this Section as their own IRA. This election will be deemed to have been made if such surviving spouse makes a contribution to the IRA or fails to take required distributions as a Beneficiary.
- (j) The interest of a Participant in the balance in their Account is nonforfeitable at all times.

(k) Trust Provisions.

- (1) The trustee of a Deemed IRA Account must be a bank, as required by Code section 408(a)(2), or if the trustee is not a bank as defined in Code section 408(n), the Trustee must have received approval from the Commissioner of the Internal Revenue Service to serve as a nonbank trustee or nonbank custodian pursuant to Treasury Regulations section 1.408-2(e).
- (2) The requirements of Code section 408(a)(5) regarding commingling of assets do not apply to Deemed IRAs. Accordingly, the assets of a Deemed IRA may be commingled for investment purposes with those of the Plan. However, the restrictions on the commingling of Plan and IRA assets with other assets apply to the assets of the Plan and Deemed IRA Accounts established under this Plan.
- (I) Separate records will be maintained for the interest of each Participant.
- (m) The trustee of the Deemed IRA Accounts will furnish annual calendar-year reports concerning the status of the Deemed IRA Accounts and such information concerning Required Minimum Distributions

as is prescribed by the Commissioner of Internal Revenue Service.

- (n) The nonbank trustee or custodian will substitute another trustee or custodian if the nonbank trustee or custodian receives notice from the Commissioner of Internal Revenue Service that such substitution is required because it has failed to comply with the requirements of Treasury Regulations section 1.408-2(e).
- (o) For purposes of this Section, "compensation" means wages, salaries, professional fees, or other amounts derived from or received for personal services actually rendered (including, but not limited to commissions paid salesmen, compensation for services on the basis of a percentage of profits, commissions on insurance premiums, tips, and bonuses) and includes Earned Income, as defined in Code section 401(c)(2) (reduced by the deduction the Self-Employed Individual takes for contributions made to a self-employed retirement plan). For purposes of this definition, Code section 401(c)(2) will be applied as if the term trade or business for purposes of Code section 1402 included service described in Subsection (c)(6). Compensation does not include amounts derived from or received as earnings or profits from property (including but not limited to interest and dividends) or amounts not includible in gross income determined without regard to section 112 of the Code. Compensation also does not include any amount received as a pension or annuity or as deferred compensation. The term "compensation" will include any amount includible in the Participant's gross income under Code section 71 with respect to a divorce or separation instrument described in Code section 71(b)(2)(A) and Differential Wage Payments as defined in section 3401(h)(2) of the Code.
- **3.12.2 Deemed Roth IRA.** This Section will apply after December 31, 2005, if the Employer elects to permit Participants to make Roth IRA contributions to this Plan.
- (a) A Roth IRA Account will be established for the exclusive benefit of the Participant or their Beneficiaries.
- (b) The maximum permissible annual contribution will be:
 - (1) Except in the case of a qualified rollover contribution or a recharacterization (as defined in Subsection (6) below), no contribution will be accepted unless it is in cash and the total of such contributions to all the Participant's Roth IRAs for a taxable year does not exceed the applicable amount (as defined in Subsection (2) below), or the Participant's compensation (as defined in Subsection (8) below), if less, for that taxable year. The contribution described in the previous sentence that may not exceed the lesser of the applicable amount or the Participant's compensation is referred to as a "regular contribution." A "qualified rollover contribution" is a rollover contribution that meets the requirements of Code section 408(d)(3), except the one-rollover-per-year rule of Code section 408(d)(3)(B) does not apply if the rollover contribution is from an IRA other than a Roth IRA (a "non-Roth IRA"). Contributions may be limited under Subsections (3) through (5) below. The applicable amount is will not exceed:

\$3,000 for any taxable year beginning in 2002 through 2004; \$4,000 for any taxable year beginning in 2005 through 2007; and \$5,000 for any taxable year beginning in 2008 and years thereafter.

After 2008, the limit will be adjusted by the Secretary of the Treasury for cost-of-living increases under Code section 219(b)(5)(C). Such adjustments will be in multiples of \$500.

(2) In the case of a Participant who is 50 or older, the annual cash contribution limit is increased by: PI=6.6

\$500 for any taxable year beginning in 2002 through 2005; and \$1,000 for any taxable year beginning in 2006 and years thereafter.

(3) Regular Contribution Limit. If Subsections (i) and/or (ii) below apply, the maximum regular contribution that can be made to all the Participant's Roth IRAs for a taxable year is the smaller amount determined under Subsections (i) or (ii).

(i) The maximum regular contribution is phased out ratably between certain levels of modified adjusted gross income ("modified AGI," defined in Subsection (7) below) in accordance with the following table:

Filing Status	Full	Phase-out	No
	Contribution	Range	Contribution
		Modified AGI	
Single or Head of	\$95,000 or less	Between \$95,000 and	\$110,000 or more
Household		\$110,000	
Joint Return or Qualifying	\$150,000 or less	Between \$150,000 and	\$160,000 or more
Widow(er)		\$160,000	
Married Separate Return	\$0	Between \$0 and \$10,000	\$10,000 or more

If the Participant's modified AGI for a taxable year is in the phase-out range, the maximum regular contribution determined under this table for that taxable year is rounded up to the next multiple of \$10 and is not reduced below \$200 and after 2006, the dollar amount above will be adjusted by the Secretary of the Treasury for cost of living increases under section 408A(c)(3) of the Code. Such adjustments are in multiples of one \$1,000.

- (ii) If the Participant makes regular contributions to both Roth and non-Roth IRAs for a taxable year, the maximum regular contribution that can be made to all the individual's Roth IRAs for that taxable year is reduced by the regular contributions made to the Participant's non-Roth IRAs for the taxable year.
- (4) Qualified Rollover Contribution Limit. A rollover from a non-Roth IRA cannot be made to this IRA if, for the year the amount is distributed from the non-Roth IRA, (i) the Participant is married and files a separate return, (ii) the Participant is not married and has modified AGI in excess of \$100,000, or (iii) the Participant is married and together the Participant and the Participant's spouse have modified AGI in excess of \$100,000. For purposes of the preceding sentence, a husband and wife are not treated as married for a taxable year if they have lived apart at all times during that taxable year and file separate returns for the taxable year.
- (5) SIMPLE IRA Limits. No contributions will be accepted under a SIMPLE IRA plan established by any employer pursuant to Code section 408(p). Also, no transfer or rollover of funds attributable to contributions made by a particular employer under its SIMPLE IRA plan will be accepted from a SIMPLE IRA, that is, an IRA used in conjunction with a SIMPLE IRA plan, prior to the expiration of the 2 year period beginning on the date the individual first participated in that employer's SIMPLE IRA plan.
- **(6) Recharacterization.** A regular contribution to a non-Roth IRA may be recharacterized pursuant to the rules in Treasury Regulations section 1.408A-5 as a regular contribution to this IRA, subject to the limits in Subsection (3) above.
- (7) Modified AGI. For purposes of (3) and (4) above, a Participant's modified AGI for a taxable year is defined in Code section 408A(c)(3)(C)(i) and does not include any amount included in adjusted gross income as a result of a rollover from a non-Roth IRA (a "conversion").
- (8) Compensation. For purposes of Subsection (b) above, compensation is defined as wages, salaries, professional fees, or other amounts derived from or received for personal services actually rendered (including, but not limited to commissions paid salesmen, compensation for services on the basis of a percentage of profits, commissions on insurance premiums, tips, and bonuses) and includes Earned Income, as defined in Code section 401(c)(2) (reduced by the deduction the Self-Employed Individual takes for contributions made to a self-employed retirement plan). For purposes of this definition, Code section 401(c)(2) will be applied as if the term trade or business for purposes of Code section 1402 included service described in Subsection (c)(6). Compensation does not include amounts derived from or received as earnings

or profits from property (including but not limited to interest and dividends) or amounts not includible in gross income. Compensation also does not include any amount received as a pension or annuity or as deferred compensation. The term "compensation" will include any amount includible in the individual's gross income under Code section 71 with respect to a divorce or separation instrument described in Code section 71(b)(2)(A). In the case of a married Participant filing a joint return, the greater compensation of their spouse is treated as their own compensation, but only to the extent that such spouse's compensation is not being used for purposes of the spouse making a contribution to a Roth IRA or a deductible contribution to a non-Roth IRA. Compensation does not include amounts derived from or received as earnings or profits from property (including but not limited to interest and dividends) or amounts not includible in gross income determined without regard to section 112 of the Code. The term "compensation" will include any amount includible in the Participant's gross income under Code section 71 with respect to a divorce or separation instrument described in Code section 71(b)(2)(A) and Differential Wage Payments as defined in section 3401(h)(2) of the Code.

- (c) If the trust acquires collectibles within the meaning of Code section 408(m) after December 31, 1981, for the benefit of any Roth IRA Account, those trust assets will be treated as a distribution in an amount equal to the cost of such collectibles.
- (d) No part of the Trust funds attributable to a Roth IRA Contribution will be invested in life insurance contracts.
- (e) No amount is required to be distributed prior to the death of the Participant for whose benefit the account was originally established.

(f) Distribution upon Death.

- (1) Notwithstanding any provision of this Section to the contrary, the distribution of the Participant's interest in the Roth IRA Account will be made in accordance with the requirements of Code section 408(a)(6), as modified by Code section 408A(c)(5), and the regulations thereunder, the provisions of which are herein incorporated by reference. If distributions are made from an annuity contract purchased from an insurance company, distributions thereunder must satisfy the requirements of Temporary Treasury Regulations section 1.401(a)(9)-6T (taking into account Code section 408A(c)(5)), rather than the distribution rules in Subsections (2), (3), and (4) below.
- (2) Upon the death of the Participant, their entire interest will be distributed at least as rapidly as follows:
 - (A) If the Designated Beneficiary is someone other than the Participant's surviving spouse, the entire interest will be distributed, starting by the end of the calendar year following the calendar year of the Participant's death, over the remaining life expectancy of the Designated Beneficiary, with such life expectancy determined using the age of the Beneficiary as of their birthday in the year following the year of the Participant's death, or, if elected, in accordance with Subsection (C) below.
 - (B) If the Participant's sole Designated Beneficiary is the Participant's surviving spouse, the entire interest will be distributed, starting by the end of the calendar year following the calendar year of the Participant's death (or by the end of the calendar year in which the Participant would have attained age 70 1/2, if later), over such spouse's life expectancy, or, if elected, in accordance with Subsection (C) below. If the surviving spouse dies before distributions are required to begin, the remaining interest will be distributed, starting by the end of the calendar year following the calendar year of the spouse's death, over the spouse's Designated Beneficiary's remaining life expectancy determined using such Beneficiary's age as of their birthday in the year following the death of the spouse, or, if elected, will be distributed in accordance with Subsection (C) below. If the surviving spouse dies after distributions are required to begin, any remaining interest will be distributed over the spouse's remaining life expectancy determined using the spouse's age as of their birthday in the year of the spouse's death.

- (C) If the Participant's sole Designated Beneficiary is the Participant's surviving spouse, the entire interest will be distributed, starting by the end of the calendar year following the calendar year of the Participant's death (or by the end of the calendar year in which the Participant would have attained age 70 1/2, if later), over such spouse's life expectancy, or, if elected, in accordance with Subsection (C) below. If the surviving spouse dies before distributions are required to begin, the remaining interest will be distributed, starting by the end of the calendar year following the calendar year of the spouse's death, over the spouse's Designated Beneficiary's remaining life expectancy determined using such Beneficiary's age as of their birthday in the year following the death of the spouse, or, if elected, will be distributed in accordance with Subsection (C) below. If the surviving spouse dies after distributions are required to begin, any remaining interest will be distributed over the spouse's remaining life expectancy determined using the spouse's age as of their birthday in the year of the spouse's death.
- (D) The amount to be distributed each year under Subsection (A) or (B) is the quotient obtained by dividing the value of the IRA created under this Section as of the end of the preceding year by the remaining life expectancy specified in such Subsection. Life expectancy is determined using the Single Life Table in Q&A-1 of Treasury Regulations section 1.401(a)(9)-9. If distributions are being made to a surviving spouse as the sole Designated Beneficiary, such spouse's remaining life expectancy for a year is the number in the Single Life Table corresponding to such spouse's age in the year. In all other cases, remaining life expectancy for a year is the number in the Single Life Table corresponding to the Beneficiary's age in the year specified in Subsections (A) or (B) and reduced by 1 for each subsequent year.
- (3) The "value" of the IRA created under this Section includes the amount of any outstanding rollover, transfer and recharacterization under Q&As-7 and -8 of Treasury Regulations section 1.408-8.
- (4) If the sole Designated Beneficiary is the Participant's surviving spouse, the spouse may elect to treat the IRA created under this Section as their own IRA. This election will be deemed to have been made if such surviving spouse makes a contribution to the IRA or fails to take required distributions as a Beneficiary.
- (g) The interest of a Participant in the balance in their Account is nonforfeitable at all times.

(h) Trust Provisions.

- (1) The trustee of a Deemed IRA Account must be a bank, as required by Code section 408(a)(2), or if the trustee is not a bank as defined in Code section 408(n), the trustee must have received approval from the Commissioner of the Internal Revenue Service to serve as a nonbank trustee or nonbank custodian pursuant to Treasury Regulations section 1.408-2(e).
- (2) The requirements of Code section 408(a)(5) regarding commingling of assets do not apply to Deemed IRA Accounts. Accordingly, the assets of a Deemed IRA Account may be commingled for investment purposes with those of the Plan. However, the restrictions on the commingling of Plan and IRA assets with other assets apply to the assets of the Plan and the Deemed IRA Accounts established under this Plan.
- (i) Separate records will be maintained for the interest of each Participant.
- (j) The trustee of the Deemed IRA Accounts will furnish annual calendar-year reports concerning the status of the Deemed IRA Accounts and such information concerning Required Minimum Distributions as is prescribed by the Commissioner of Internal Revenue.
- (k) The nonbank trustee or custodian will substitute another trustee or custodian if the nonbank trustee or custodian receives notice from the Commissioner of Internal Revenue that such substitution is

July 27, 2022 Agenda Item: 5A ATTACHMENT 1

required because it has failed to comply with the requirements of Treasury Regulations section 1.408- 2(e).

Appendix A
Age-Weighted Tables
Table I

NRA	UP84 7.50%	- male/female 8.00%	8.50%	7.50%	G71 - male 8.00%	8.50%
50	133.39558	127.81104	122.64462	133.87491	128.27223	123.08767
51	131.68326	126.25593	121.22873	132.15556	126.71284	121.66974
52	129.91096	124.64274	119.75675	130.37691	125.09602	120.19650
53	128.08231	122.97462	118.23141	128.53619	123.41915	118.66534
54	126.19427	121.24857	116.65005	126.63053	121.67908	117.07298
55	124.24537	119.46290	115.01035	124.65642	119.87202	115.41560
56	122.23012	117.61205	113.30687	122.60942	117.99381	113.68869
57	120.15072	115.69789	111.54129	120.48454	116.03902	111.88699
58	118.01011	113.72295	109.71577	118.27602	114.00182	110.00446
59	115.81245	111.69104	107.83357	115.98196	111.88010	108.03873
60	113.55288	109.59709	105.88958	113.61351	109.68388	105.99902
61	111.23508	107.44439	103.88693	111.17770	107.42004	103.89146
62	108.86392	105.23735	101.82937	108.67722	105.09058	101.71801
63	106.44537	102.98163	99.72218	106.11093	102.69407	99.47675
64	103.98697	100.68415	97.57192	103.47950	100.23073	97.16756
65	101.49368	98.34959	95.38290	100.79213	97.70930	94.79850
66	98.97540	95.98734	93.16400	98.06584	95.14576	92.38490
67	96.44411	93.60886	90.92637	95.31689	92.55574	89.94173
68	93.89662	91.21125	88.66695	92.55865	89.95208	87.48110
69	91.32062	88.78211	86.37369	89.79449	87.33800	85.00626
70	88.70352	86.30928	84.03460	87.04327	84.73176	82.53467
71	86.04948	83.79645	81.65305	84.32727	82.15516	80.08784
72	83.36293	81.24763	79.23264	81.65112	79.61276	77.67027
73	80.64861	78.66750	76.77767	78.99038	77.08109	75.25922
74	77.92240	76.07114	74.30287	76.31234	74.52798	72.82319
75 76 77 78 79	75.19300 72.47065 69.76697 67.09586 64.44038 61.81054	73.46702 70.86510 68.27672 65.71559 63.16517	71.81622 69.32745 66.84769 64.39026 61.93914 59.50387	73.59988 70.85419 68.10189 65.39577 62.76892	71.93665 69.30790 66.66751 64.06704 61.53910 59.08950	70.34555 67.82691 65.29201 62.79139 60.35702 57.99492

NRA	G71 - female 7.50%	9.000/	9 F00/	7.50%	G83 - male	9 FO9/
NKA	7.50%	8.00%	8.50%	7.50%	8.00%	8.50%
50	144.90527	138.26324	132.16028	138.63492	132.59926	127.03027
51	143.60635	137.11101	131.13533	137.10255	131.22171	125.78854
52	142.23274	135.88840	130.04425	135.50742	129.78433	124.49022
53	140.78142	134.59258	128.88425	133.84552	128.28325	123.13102
54	139.24881	133.21983	127.65148	132.11209	126.71333	121.70618
55	137.63167	131.76671	126.34254	130.30165	125.06941	120.21023
56	135.92700	130.22987	124.95377	128.40761	123.34502	118.63663
57	134.13313	128.60760	123.48346	126.42384	121.53326	116.97877
58	132.24942	126.89888	121.93011	124.34516	119.62924	115.23140
59	130.27652	125.10399	120.29366	122.16888	117.63004	113.39128
60	128.21500	123.22312	118.57417	119.89545	115.53522	111.45762
61	126.06589	121.25681	116.77169	117.52676	113.34643	109.43158
62	123.82951	119.20499	114.88587	115.06712	111.06731	107.31606
63	121.50521	117.06681	112.91550	112.52366	108.70404	105.11699
64	119.09157	114.84023	110.85811	109.90545	106.26530	102.84183
65	116.58681	112.52340	108.71172	107.22400	103.76178	100.50089
66	113.98596	110.11094	106.47042	104.49343	101.20648	98.10645
67	111.28113	107.59468	104.12611	101.72785	98.61320	95.67153
68	108.46410	104.96593	101.66982	98.93819	95.99223	93.20585
69	105.54134	102.23046	99.10625	96.13028	93.34889	90.71464
70	102.53171	99.40595	96.45211	93.30456	90.68378	88.19812
71	99.46349	96.51942	93.73310	90.45657	87.99231	85.65193
72	96.37133	93.60395	90.98105	87.57864	85.26683	83.06821
73	93.27601	90.67967	88.21532	84.66842	82.50491	80.44450
74	90.18710	87.75586	85.44496	81.73067	79.71081	77.78463
75	87.11538	84.84304	82.68002	78.77675	76.89552	75.09910
76	84.06201	81.94230	79.92174	75.82433	74.07600	72.40433
77	81.02953	79.05620	77.17252	72.89544	71.27376	69.72120
78	78.02448	76.19112	74.43866	70.00987	68.50821	67.06872
79	75.05164	73.35197	71.72505	67.18340	65.79488	64.46223
80	72.12192	70.54948	69.04218	64.42863	63.14638	61.91418

NRA	G83 - female 7.50%	8.00%	8.50%	7.50%	l71 - male 8.00%	8.50%
50	147.91302	140.92288	134.51962	136.59149	130.66277	125.19499
51	146.76982	139.91631	133.63052	135.09132	129.30840	123.96842
52	145.56105	138.84839	132.68423	133.55057	127.91486	122.70451
53	144.28224	137.71524	131.67719	131.96756	126.48077	121.40180
54	142.93045	136.51337	130.60559	130.34036	125.00422	120.05840
55	141.50316	135.24040	129.46710	128.66701	123.48285	118.67197
56	139.99799	133.89372	128.25923	126.94415	121.91390	117.23962
57	138.41386	132.47191	126.98009	125.16914	120.29424	115.75829
58	136.74878	130.97319	125.62766	123.33832	118.62031	114.22449
59	135.00087	129.39540	124.20000	121.44854	116.88880	112.63481
60	133.16838	127.73611	122.69443	119.49602	115.09599	110.98557
61	131.24841	125.99266	121.10789	117.47707	113.23811	109.27267
62	129.23820	124.16174	119.43694	115.38790	111.31123	107.49220
63	127.13458	122.24028	117.67835	113.22541	109.31195	105.64069
64	124.93456	120.22482	115.82841	110.98680	107.23723	103.71486
65	122.63529	118.11200	113.88367	108.67109	105.08606	101.71346
66	120.23364	115.89850	111.84019	106.27953	102.85897	99.63642
67	117.72697	113.58125	109.69447	103.81445	100.55810	97.48575
68	115.11600	111.16051	107.44659	101.27956	98.18649	95.26405
69	112.40616	108.64074	105.10009	98.67918	95.74812	92.97465
70	109.60621	106.03004	102.66224	96.01785	93.24705	90.62141
71	106.72905	103.34048	100.14451	93.30055	90.68764	88.20808
72	103.79031	100.58642	97.56031	90.53223	88.07468	85.73910
73	100.80493	97.78230	94.92322	87.71814	85.41285	83.21877
74	97.78660	94.94115	92.24557	84.86385	82.70737	80.65186
75	94.74739	92.07442	89.53842	81.97568	79.96405	78.04391
76	91.69794	89.19222	86.81143	79.06057	77.18962	75.40114
77	88.64714	86.30317	84.07282	76.12621	74.39142	72.73066
78	85.60244	83.41460	81.32977	73.18116	71.57761	70.04018
79	82.56990	80.53235	78.58787	70.23399	68.75650	67.33794
80	79.55392	77.66068	75.85136	67.29335	65.93657	64.63201

NRA	I71 - female 7.50%	8.00%	8.50%	7.50%	l83 - male 8.00%	8.50%
INIXA	7.5070	0.0070	0.30 /0	7.5070	0.0070	0.50 /0
50	145.27345	138.55022	132.37967	140.91751	134.59421	128.77736
51	144.00299	137.42172	131.37393	139.58740	133.40520	127.71147
52	142.66699	136.23143	130.31013	138.20573	132.16762	126.59975
53	141.26707	134.98100	129.18979	136.76822	130.87721	125.43825
54	139.80559	133.67233	128.01442	135.27022	129.52931	124.22237
55	138.28543	132.30786	126.78642	133.70628	128.11873	122.94695
56	136.70839	130.88954	125.50736	132.07126	126.64031	121.60671
57	135.07635	129.41867	124.17855	130.35991	125.08850	120.19650
58	133.38971	127.89588	122.80050	128.56627	123.45756	118.71015
59	131.64841	126.32069	121.37219	126.68475	121.74173	117.14211
60	129.84888	124.68968	119.89085	124.71191	119.93716	115.48811
61	127.98476	122.99654	118.34991	122.64626	118.04228	113.74637
62	126.04571	121.23108	116.73965	120.48904	116.05761	111.91704
63	124.01839	119.38033	115.04700	118.24356	113.98596	110.00253
64	121.88731	117.42877	113.25709	115.91489	111.83218	108.00687
65	119.63954	115.36368	111.35684	113.50918	109.60117	105.93467
66	117.26546	113.17531	109.33646	111.03240	107.29890	103.79118
67	114.76195	110.85983	107.19194	108.49094	104.93080	101.58132
68	112.13113	108.41859	104.92365	105.89092	102.50282	99.31059
69	109.37867	105.85674	102.53592	103.23881	100.02051	96.98409
70	106.51318	103.18160	100.03551	100.54041	97.48946	94.60693
71	103.54390	100.40190	97.42999	97.80114	94.91457	92.18357
72	100.48090	97.52644	94.72774	95.02578	92.30039	89.71836
73	97.33402	94.56453	91.93686	92.21877	89.65089	87.21460
74	94.11313	91.52514	89.06599	89.38364	86.96923	84.67558
75	90.82800	88.41741	86.12322	86.52577	84.26062	82.10585
76	87.48845	85.25047	83.11730	83.65211	81.53162	79.51167
77	84.10457	82.03371	80.05694	80.77127	78.79035	76.90089
78	80.68719	78.77757	76.95184	77.89309	76.04643	74.28268
79	77.24883	75.49375	73.81338	75.02843	73.31034	71.66727
80	73.80450	72.19685	70.65545	72.18888	70.59340	69.06559

	183 - female			G83 - 50/50) blend	
NRA	7.50%	8.00%	8.50%	7.50%	8.00%	8.50%
50	148.15100	141.09944	134.64568	142.96988	136.50050	130.55119
51	147.05406	140.13394	133.79305	141.61377	135.29164	129.47044
52	145.90085	139.11630	132.89178	140.19287	134.02167	128.33231
53	144.68849	138.04356	131.93953	138.70319	132.68671	127.13249
54	143.41400	136.91241	130.93291	137.14035	131.28218	125.86707
55	142.07372	135.24040	135.71965	135.50061	129.80421	124.53176
56	140.66418	133.89372	134.46191	133.77962	128.24854	123.12223
57	139.18233	132.47191	133.13576	131.97313	126.61064	121.63393
58	137.62485	130.97319	131.73755	130.07793	124.88714	120.06329
59	135.98819	129.39540	130.26413	128.09181	123.07571	118.40765
60	134.27019	128.71306	123.56113	126.01379	121.17482	116.66528
61	132.46928	127.08224	122.08134	123.84377	119.18411	114.83548
62	130.58479	125.37126	120.52419	121.58291	117.10397	112.9183
63	128.61720	123.57960	118.88966	119.23342	114.93645	110.91517
64	126.56706	123.37900	117.17753	116.79882	112.68443	108.82849
04	120.50700	121.70795	117.17733	110.79002	112.00443	100.02049
65	124.43398	119.75546	115.38697	114.28428	110.35238	106.66235
66	122.21593	117.71976	113.51550	111.69566	107.94578	104.42131
67	119.90926	115.59705	111.55884	109.03972	105.47054	102.11105
68	117.50813	113.38139	109.51103	106.32292	102.93255	99.73679
69	115.00637	111.06619	107.36526	103.55064	100.33683	97.30324
70	112.40035	108.64758	105.11721	100.72765	97.68764	94.81405
71	109.69025	106.12501	102.76605	97.85741	94.98814	92.27195
72	106.87985	103.50166	100.31418	94.94323	92.24105	89.6795
73	103.97594	100.78369	97.76710	91.99074	89.45176	87.04153
74	100.98827	97.98005	95.13290	89.00923	86.62891	84.36604
75	97.92718	95.10007	92.42042	86.01041	83.78374	81.66405
76	94.80299	92.15358	89.63850	83.00905	80.93050	78.94916
77	91.62602	89.15000	86.79614	80.02145	78.08511	76.23672
78	88.40592	86.09861	83.90184	77.06220	75.26165	73.54055
79	85.15211	83.00822	80.96399	74.14289	72.47154	70.87187
80	81.87574	79.88931	77.99260	71.27254	69.72387	68.23969

	GAR		
NRA	7.50%	8.00%	8.50%
50	144.66357	138.03784	131.95119
51	143.30405	136.82698	130.86949
52	141.86853	135.54434	129.72020
53	140.35698	134.18947	128.50233
54	138.76709	132.76016	127.21368
55	137.09410	131.25116	125.84917
56	135.33905	129.66351	124.40922
57	133.50423	127.99902	122.89558
58	131.59505	126.26231	121.31211
59	129.61377	124.45566	119.66088
60	127.55850	122.57665	117.93940
61	125.43149	120.62740	116.14915
62	123.24136	118.61538	114.29734
63	120.98935	116.54224	112.38506
64	118.68845	114.41988	110.42379
65	116.33940	112.24870	108.41366
66	113.94389	110.03037	106.35605
67	111.50807	107.77058	104.25637
68	109.02035	105.45798	102.10343
69	106.45750	103.06983	99.87505
70	103.81262	100.59907	97.56403
71	101.07156	98.03152	95.15617
72	98.24187	95.37407	92.65754
73	95.34370	92.64577	90.08628
74	92.37423	89.84346	87.43903
75	89.33565	86.96893	84.71690
76	86.23708	84.03058	81.92786
77	83.08691	81.03609	79.07889
78 70	79.93101	78.03008	76.21336
79	76.78197	75.02472	73.34282
80	73.65159	72.03147	70.47867

Age-Weighted

		TABLĚ II	
years to NRA		discount factor	
•	7.50%	8.00%	8.50%
at or over NRA	1.00000	1.00000	1.00000
1	0.93023	0.92593	0.92166
2	0.86533	0.85734	0.84946
3	0.80496	0.79383	0.78291
4	0.74880	0.73503	0.72157
4	0.74000	0.75505	0.72137
5	0.69656	0.68058	0.66505
6	0.64796	0.63017	0.61295
7	0.60275	0.58349	0.56493
8	0.56070	0.54027	0.52067
9	0.52158	0.50025	0.47988
10	0.48519	0.46319	0.44229
11	0.45134	0.42888	0.40764
12	0.41985	0.39711	0.37570
13	0.39056	0.36770	0.34627
14	0.36331	0.34046	0.31914
15	0.33797	0.31524	0.29414
16	0.31439	0.29189	0.27110
17	0.29245	0.27027	0.24986
18	0.27205	0.25025	0.23028
19	0.25307	0.23171	0.21224
20	0.23541	0.21455	0.19562
21	0.21899	0.19866	0.18029
22	0.20371	0.18394	0.16617
23	0.18950	0.17032	0.15315
24	0.17628	0.15770	0.14115
25	0.16398	0.14602	0.13009
26	0.15254	0.13520	0.11990
27	0.14190	0.12519	0.11051
28	0.13200	0.11591	0.10185
29	0.12279		0.10103
29	0.12279	0.10733	0.09367
30	0.11422	0.09938	0.08652
31	0.10625	0.09202	0.07974
32	0.09884	0.08520	0.07349
33	0.09194	0.07889	0.06774
34	0.08553	0.07305	0.06243
35	0.07956	0.06763	0.05754
36	0.07401	0.06262	0.05303
37	0.06885	0.05799	0.04888
38	0.06404	0.05369	0.04505
39	0.05958	0.04971	0.04152
	0.00900	0.04371	0.04132
40	0.05542	0.04603	0.03827

41	0.05155	0.04262	0.03527
42	0.04796	0.03946	0.03251
43	0.04461	0.03654	0.02996
44	0.04150	0.03383	0.02761
45	0.03860	0.03133	0.02545
46	0.03591	0.02901	0.02345
47	0.03340	0.02686	0.02162
48	0.03107	0.02487	0.01992
49	0.02891	0.02303	0.01836
50	0.02689	0.02132	0.01692

Appendix B INSURANCE ADDENDUM

Notwithstanding anything in the Plan to the contrary, if the Employer elects in the Adoption Agreement to fund the Plan exclusively with Insurance Contracts, the Plan will be deemed a "NonTrusteed Plan". As such, the following provisions will apply, and all references to the Trustee and the Trust Fund will be construed to apply to the annuity contracts as described in this Addendum. Any duties of the Trustees, including but not limited to the establishment of a Participant loan program, that do not expressly apply to the annuity contracts will apply to the Plan Administrator.

- **1. Purchase of Contracts.** The benefits provided under this Plan will be funded exclusively through the purchase of annuity Contracts issued by an Insurance Company. The provisions of this Addendum will apply to any such Contracts which, as determined by the Employer, will not be held by the Trustee. The Employer will pay within a reasonable period of time all contributions that are made to this Plan to the Insurer for the purchase of such Contracts.
- 2. Employer Designated as Owner. Each Contract will designate the Employer as sole owner, with rights reserved to said Employer to exercise those rights or options contained therein that apply to the owner of the Contract. All such Contracts will be held by the Employer who will have the power and right to take such actions with respect to such Contracts as will be in accordance with this Plan for the exclusive purpose of providing benefits to Participants. The Employer will be treated as Trustee to the extent that the Contracts are treated as trusts pursuant to Code Section 401(f).
- **3. Type of Contracts(s).** The Employer will have the right to determine whether to have fixed of a combination of fixed and variable Contracts and whether to have group or individual Contracts. The Employer will base its decision on which Contract(s) would be more beneficial for the Participants and on the administrative tasks imposed by each Contract. Such decision will be in the sole discretion of the Employer.
- **4. Voting Rights.** The Employer will solicit and act in accordance with the instructions of the Participant in regard to any voting rights that pertain to a Contract for a variable accumulation of benefits. During the accumulation period, Participants will have the right to instruct the Employer with respect to the votes attributable to any Vested interest they have in the Contract. All other votes entitled to be cast during the accumulation period may be cast by the Employer in its sole discretion. During the annuity period, every Participant will have the right to instruct the Employer with respect to all votes attributable to the amount of assts established in the appropriate separate account to meet the annuity obligations related to such Participant. The Insurer will provide all notices and proxy materials to the Employer for distribution to the Participants. The Employer may cast all votes for which instruction were not received in accordance with the Employer's sole discretion.
- **5. Certificate of Participant.** The Insurer will issue a certificate of participant and/or a Contract, as applicable, to each Participant. Each such certificate of participation will set forth in substance the benefits or other rights to which such Participant is entitled under the Contract.
- **6. Insurer Indemnification.** The Employer agrees to indemnify and hold harmless the Insurer against any and all claims, losses, damages, expenses and liabilities the Insurer may incur in the exercise and performance of the Insurer's duties hereunder, unless the same are determined to be due to gross negligence or willful misconduct on the part of the Insurer.

Appendix C EXPANDED HARDSHIP DISTRIBUTIONS AND REQUIRED MINIMUM DISTRIBUTION ADDENDUM

Section 1. General

- **1.1 Purpose and Adoption.** The following amendment will amend the Plan in accordance with the Bipartisan Budget Act of 2018, the SECURE Act, and any subsequent guidance promulgated by the Internal Revenue Service.
- **1.2 Precedence.** The requirements of this Amendment will take precedence over any inconsistent provisions of the Plan, including any previous amendments adopted by the Employer. Where appropriate, the term "Plan" shall mean the Plan and Trust.
- 1.3 Requirements of Treasury Regulations Incorporated. This Amendment is intended to update the Plan for changes made to the Code and Treasury Regulations by the Bipartisan Budget Act of 2018 and the SECURE Act. All matters addressed under this Amendment will be determined and made in accordance with both existing and future guidance related to the provisions contained in this Amendment.
- **1.4 Effective Date.** This Amendment amends the Plan effective January 1, 2020, unless an earlier date is elected in an associated Adoption Agreement Amendment.

Section 2. Base Plan Document Provisions Related to Bipartisan Budget Act of 2018.

- **2.1 Qualified Matching Contribution.** The definition of "Qualified Matching Contribution" is replaced in its entirety with the following:
 - "Qualified Matching Contribution" and "QMAC": A Matching Contribution that is one hundred percent (100%) vested and nonforfeitable when allocated to Participants' accounts and that is distributable only in accordance with the distribution restrictions applicable to Elective Deferrals under the Plan.
- **2.2 Qualified Non-Elective Contribution.** The definition of "Qualified Non-Elective Contribution" is replaced in its entirety with the following:
 - "Qualified Non-Elective Contribution" and "QNEC": An Employer Contribution to a cash or deferred profit sharing plan that is neither a Non-Elective Contribution, Matching Contribution, Qualified Matching Contribution, nor an Elective Deferral, that is one hundred percent (100%) vested and nonforfeitable when allocated to Participants' accounts, that a Participant may not elect to receive in cash, and that is distributable only in accordance with the distribution restrictions applicable to Elective Deferrals under the Plan.
- **2.3 EACA Limitations.** Section 2.2.3(b)(4) of the Base Plan Document is replaced in its entirety with the following:
 - (4) Limitations. To meet the limitations under Code sections 401(a)(17), 402(g), 414(u)(12)(B)(ii), and 415(c), the Employer may stop or reduce an Employee's contributions to the Plan.
- **2.4 QACA Limitations.** Section 2.2.3(c)(4) of the Base Plan Document is replaced in its entirety with the following:
 - (4) Limitations. To meet the limitations under Code sections 401(a)(17), 402(g), 414(u)(12)(B)(ii), and 415(c), the Employer may stop or reduce an Employee's contributions to the Plan.

2.5 Hardship Distributions of Elective Deferrals. Section 2.5.10 of the Base Plan Document is replaced in its entirety with the following:

2.5.10 Hardship Distributions of Elective Deferrals, ADP Test Safe Harbor Contribution Accounts, QACA ADP Safe Harbor Contribution Accounts, QNEC Accounts, and QMAC Accounts. The Employer may elect in the Adoption Agreement or an Adoption Agreement Amendment to permit Hardship Distributions of Elective Deferrals (including attributed earnings if elected in the Adoption Agreement or Adoption Agreement Amendment), ADP Test Safe Harbor Contribution Accounts, QACA ADP Safe Harbor Contribution Accounts, QNEC Accounts, and QMAC Accounts. A Hardship Distribution is a distribution that meets the requirements of a distribution on account of immediate and heavy financial need enumerated in this Section 2.5.10. A Participant may request a distribution from the Plan as a result of immediate and heavy financial needs of the Participant to the extent that the distribution is necessary to satisfy such financial needs. The availability of Hardship Distributions to Participants is limited to Participants who have not terminated employment. Effective August 17, 2006, the Plan may permit a Participant to request a Hardship Distribution where the Hardship or unforeseeable financial emergency occurs to the Participant's spouse, dependents, or Primary Beneficiary.

Hardship distributions are subject to the spousal consent requirements contained in Code sections 401(a)(11) and 417 unless the Plan meets the exception of Subsection 2.5.5(c). The Plan Administrator shall determine whether a Participant has an immediate and heavy financial need. A distribution shall be deemed to be made on account of an immediate and heavy financial need if the distribution is due to:

- (a) Deductible medical expenses described in Code section 213(d) incurred or necessary for medical care of the Participant, his spouse, dependents, or Primary Beneficiary under the Plan;
- (b) Purchase (excluding mortgage payments) of a principal residence for the Participant;
- (c) Cost of tuition and related educational fees and room and board expenses for the next twelve (12) months of post-secondary education for the Participant, his spouse, children, Primary Beneficiary or dependents;
- (d) The need to prevent the eviction of the Participant from his principal residence or foreclosure on the mortgage of the Participant's principal residence;
- (e) Payments for funeral or burial expenses for the Participant's deceased parent, spouse, child, dependent or Primary Beneficiary under the Plan (for Plan Years beginning after 2005.);
- (f) Expenses to repair damage to the Participant's principal residence that would qualify for a casualty loss deduction under Code section 165 (determined without regard to Code section 165(h)(5) or whether the loss exceeds ten percent (10%) of adjusted gross income) (for Plan Years beginning after 2005.); or
- (g) Expenses and losses (including loss of income) incurred by the Participant on account of a disaster declared by the Federal Emergency Management Agency (FEMA) under the Robert T. Stafford Disaster Relief and Emergency Assistance Act, provided that Participant's principal residence or principal place of employment at the time of the disaster was located in an area designated by FEMA for individual assistance with respect to the disaster.

A distribution shall be considered as necessary to satisfy an immediate and heavy financial need of the Participant only if:

(a) The Participant has obtained all distributions, other than Hardship Distributions, under all plans maintained by the Employer;

- (b) The distribution is not in excess of the amount of an immediate and heavy financial need (including amounts necessary to pay any federal, state, or local income taxes or penalties reasonably anticipated to result from the distribution); and
- (c) For hardship distributions made on or after January 1, 2020, the Participant must represent, in writing, that they have insufficient cash or other liquid assets to satisfy the need. The Plan Administrator may rely on the Participant's representation unless the Plan Administrator has actual knowledge to the contrary.
- **2.6 Definitions.** Section 2.9.2(b) of the Base Plan Document is replaced in its entirety with the following:
 - (b) "Eligible Employee" means an Employee eligible to make Elective Deferrals under the Plan for any part of the Plan Year or who would be eligible to make Elective Deferrals but for statutory limitations, such as Code sections 402(g) and 415. The Employer may elect in the Adoption Agreement to exclude Participants from the ADP Safe Harbor or ACP Safe Harbor portion of the Plan who have not completed a Year of Service since their original Employment Commencement Date, or have not yet reached age 21, and are not employed on the earlier of the first day of the next Plan Year after meeting the preceding requirements or 6 months after meeting the preceding requirements.
- **2.7 ADP Test Safe Harbor Contributions.** Section 2.9.3(a)(3) of the Base Plan Document is replaced in its entirety with the following:
 - (3) The Participant's Account balance derived from ADP Test Safe Harbor Contributions is nonforfeitable and may not be distributed earlier than severance from employment, death, Disability, an event described in Code section 401(k)(10), or, in the case of a profit sharing plan, the attainment of age fifty-nine and one-half (59-1/2). Unless otherwise elected in an Adoption Agreement amendment, no distributions on account of Hardship are permitted. In addition, such contributions must satisfy the ADP Test Safe Harbor without regard to permitted disparity under Code section 401(l).

Section 3. Base Plan Document Provisions Related to SECURE Act of 2019.

- **3.1 Timing of Distributions if Participant Dies Before Distributions Begin.** Section 2.5.7(b) of the Base Plan Document is replaced in its entirety with the following:
 - (b) Timing of Distributions if Participant Dies Prior to January 1, 2020 and Before Distributions Begin.
 - (1) If the Participant dies prior to January 1, 2020 and before required distributions begin, the Participant's entire interest will be distributed, or begin to be distributed, no later than as follows:
 - (i) If the Participant's surviving spouse is the Participant's sole Designated Beneficiary, then, except as provided in the Adoption Agreement, distributions to the surviving spouse will begin by December 31 of the calendar year immediately following the calendar year in which the Participant died, or by December 31 of the calendar year in which the Participant would have attained age seventy and one-half (70 1/2), if later.
 - (ii) If the Participant's surviving spouse is not the Participant's sole Designated Beneficiary, then, except as provided in the Adoption Agreement, distributions to the Designated Beneficiary will begin by December 31 of the calendar year immediately following the calendar year in which the Participant died.
 - (iii) If there is no Designated Beneficiary as of the date of the Participant's death who remains a Beneficiary as of September 30 of the year immediately following the year the Participant died, the Participant's entire interest will be distributed by December 31

of the calendar year containing the fifth anniversary of the Participant's death.

- (iv) If the Participant's surviving spouse is the Participant's sole Designated Beneficiary and the surviving spouse dies after the Participant but before distributions to the surviving spouse begin, this Subsection 2.5.7(b), other than Subsection 2.5.7(b)(1)(i), will apply as if the surviving spouse were the Participant.
- (2) For purposes of this Subsection 2.5.7(b) and Subsections 2.5.7(f) and (g), distributions are considered to begin on the Participant's Required Beginning Date unless Subsection 2.5.7(b)(1)(iv) applies. If Subsection 2.5.7(b)(1)(iv) applies, distributions are considered to begin on the date distributions are required to begin to the surviving spouse under Subsection 2.5.7(b)(1)(i). If distributions under an annuity contract purchased from an insurance company irrevocably commence to the Participant before the Participant's Required Beginning Date (or to the Participant's surviving spouse before the date distributions are required to begin to the surviving spouse under Subsection 2.5.7(b)(1)(i)), the date distributions are considered to begin is the date distributions actually commence.
- (3) If the Adoption Agreement permits Participants and Beneficiaries to elect the 5-year rule, the election must be made no later than the earlier of:
 - (i) September 30 of the calendar year in which the distribution would be required under Subsections 2.5.7(b)(1)(i) or (ii), or
 - (ii) By September 30 of the calendar year that contains the fifth (5th) anniversary of the Participant's (or if applicable, the surviving spouse's) death.

If this election is made, then the Participant's entire interest must be distributed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.

- **3.2 Required Minimum Distributions On or After Date Required Distributions Begin.** Section 2.5.7(f) of the Base Plan Document is replaced in its entirety with the following:
 - (f) Required Minimum Distributions On or After Date Required Distributions Begin When Participant has Died Prior to January 1, 2020.
 - (1) Participant Survived by Designated Beneficiary. If the Participant died prior to January 1, 2020 and on or after the date required distributions begin and there is a Designated Beneficiary, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the longer of the remaining Life Expectancy of the Participant or the remaining Life Expectancy of the Participant's Designated Beneficiary, determined as follows:
 - (i) The Participant's remaining Life Expectancy is calculated in accordance with the Single Life Table found in Treasury Regulations section 1.401(a)(9)-9, Q&A-1, using the age of the Participant in the year of death, reduced by one for each subsequent year.
 - (ii) If the Participant's surviving spouse is the Participant's sole Designated Beneficiary, the remaining Life Expectancy of the surviving spouse is calculated using the Single Life Table found in Treasury Regulations section 1.401(a)(9)-9, Q&A-1, for each Distribution Calendar Year after the year of the Participant's death using the surviving spouse's age as of the spouse's birthday in that year. For Distribution Calendar Years after the year of the surviving spouse's death, the remaining Life Expectancy of the surviving spouse is calculated using the age of the surviving spouse as of the spouse's birthday in the calendar year of the spouse's death, reduced by one for each subsequent calendar year.
 - (iii) If the Participant's surviving spouse is not the Participant's sole Designated Beneficiary,

the Designated Beneficiary's remaining Life Expectancy is calculated under the Single Life Table using the age of the Beneficiary in the year following the year of the Participant's death, reduced by one for each subsequent year.

- (2) No Designated Beneficiary Survives. If the Participant died prior to January 1, 2020 and on or after the date required distributions begin and there is no Designated Beneficiary as of the Participant's date of death who remains a beneficiary as of September 30 of the year after the year of the Participant's death, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account balance by the Participant's remaining Life Expectancy under the Single Life Table calculated using the age of the Participant in the year of death, reduced by one for each subsequent year.
- **3.3 Distribution Amount If Participant Dies Before Date Required Distributions Begin.** Section 2.5.7(g) of the Base Plan Document is replaced in its entirety with the following:
 - (g) Distribution Amount If Participant Dies Before Date Required Distributions Begin When Participant Died Prior to January 1, 2020.
 - (1) Participant Survived by Designated Beneficiary. If the Participant died prior to January 1, 2020 and before the date required distributions begin and there is a Designated Beneficiary, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the remaining Life Expectancy of the Participant's Designated Beneficiary, determined as provided in Subsection 2.5.7(f). If permitted in the Adoption Agreement, a Participant or Designated Beneficiary may elect to receive distributions over the five (5) year period described in Subsection 2.5.7(b)(3).
 - (2) No Designated Beneficiary Survives. If the Participant died prior to January 1, 2020 and before the date distributions begin and all Designated Beneficiaries, if any, as of the Participant's date of death are no longer Beneficiaries as of September 30 of the year following the year of the Participant's death, distribution of the Participant's entire interest must be completed by December 31 of the calendar year containing the fifth (5th) anniversary of the Participant's death.
 - (3) Death of Surviving Spouse Before Distributions to Surviving Spouse Are Required to Begin. If all of the following are true, this Subsection 2.5.7(g) will apply as if the surviving spouse were the Participant:
 - (i) The Participant dies before the date distributions begin,
 - (ii) The Participant's surviving spouse is the Participant's sole Designated Beneficiary, and
 - (iii) The surviving spouse dies before distributions are required to begin to the surviving spouse under Subsection 2.5.4(b)(1).
- **3.4 Required Beginning Date.** Section 2.5.7(i)(4)(i) and 2.5.7(i)(4)(ii) of the Base Plan Document are replaced in their entirety with the following:
 - (4) Required Beginning Date.
 - (i) General rule. For Participants born prior to July 1, 1949 the Required Beginning Date of a Participant is the first day of April of the calendar year following the calendar year in which the Participant attains age seventy and one-half (70-1/2). For Participants born after June 30, 1949 the Required Beginning Date of a Participant is the first day of April of the calendar year following the calendar year in which the Participant attains age seventy-two (72).

- (ii) Non-5-Percent Owner Exception. If so elected in the Adoption Agreement, the Required Beginning Date of a Participant born prior to July 1, 1949 who is not a 5-Percent Owner shall be the first day of April of the calendar year following the later of the calendar year in which the Participant retires or the calendar year in which the Participant attains age seventy and one-half (70-1/2). If the Adoption Agreement indicates that the Required Beginning Date of a Participant who is not a 5-Percent Owner is the first day of April of the calendar year in which the Participant attains age seventy and one-half (70-1/2) then for a Participant born after June 30, 1949 who is not a 5-Percent Owner the Required Beginning Date shall be the first day of April of the calendar year following the later of the calendar year in which the Participant retires or the calendar year in which the Participant retires or the calendar year in which the Participant attains age seventy-two (72).
- **3.5 Eligible Designated Beneficiaries.** Section 2.5.7(i)(5) is added to the Base Plan Document as follows:
 - (5) Eligible Designated Beneficiary. This Plan section 2.5.7(i)(5) will be interpreted in accordance with Code section 401(a)(9)(E), as amended by the SECURE Act of 2019, and all applicable regulations and guidance issued related to such Act. In the event that a provision of this Plan conflicts with such requirements, such requirements will govern.

An Eligible Designated Beneficiary is a Beneficiary who, as of the date of the Participant's death is:

- (i) the surviving spouse of the Participant;
- (ii) a child of the Participant who has not reached majority;
- (iii) disabled as defined in Code section 72(m)(7);
- (iv) a chronically ill individual as defined in Code section 401(a)(9)(E)(IV); or
- (v) not more than ten (10) years younger than the Participant and not otherwise an Eligible Designated Beneficiary.
- **3.6 Death of Participant On or After January 1, 2020.** Section 2.5.7(k) is added to the Base Plan Document as follows:
 - (k) Death of Participant On or After January 1, 2020. This Plan section 2.5.7(k) will be interpreted in accordance with Code section 401(a)(9)(E), as amended by the SECURE Act of 2019, and all applicable regulations and guidance issued related to such Act. In the event that a provision of this Plan conflicts with such requirements, such requirements will govern.
 - (1) Participant Survived by Designated Beneficiary who is not an Eligible Designated Beneficiary. If the Designated Beneficiary is not also an Eligible Designated Beneficiary, the Participant's entire interest in the Plan will be distributed by December 31 of the calendar year containing the tenth (10th) anniversary of the Participant's death.
 - (2) Participant Survived by Designated Beneficiary who is also an Eligible Designated Beneficiary. If the Designated Beneficiary is also an Eligible Designated Beneficiary and if permitted in the Adoption Agreement, a Participant or Designated Beneficiary may elect to receive distributions over the five (5) year period described in Subsection 2.5.7(b)(3).
 - (3) No Designated Beneficiary Survives. If all Designated Beneficiaries, if any, as of the Participant's date of death are no longer Beneficiaries as of September 30 of the year following the year of the Participant's death, distribution of the Participant's entire interest must be completed by December 31 of the calendar year containing the fifth (5th) anniversary of the Participant's death.

- (4) Non-Person Designated Beneficiary. If the Designated Beneficiary is not also an Eligible Designated Beneficiary, the Participant's entire interest in the Plan will be distributed by December 31 of the calendar year containing the tenth (10th) anniversary of the Participant's death.
- (5) Multi-Beneficiary Trust. If under its terms a Multi-Beneficiary Trust (1) is to be divided immediately upon the death of the Participant into separate trusts for each beneficiary of such trust or (2) no individual other than Disabled or Chronically III Eligible Designated Beneficiary has a right to the Participant's interest in the Plan until after the death of all such Disabled or Chronically III Eligible Designated Beneficiary, the requirements under Section 2.5.7(k) will be applied to each Designated Beneficiary separately.
 - (i) A "Disabled or Chronically III Eligible Designated Beneficiary" is a Designated Beneficiary that is either (1) disabled as defined in Code section 72(m)(7) or (2) a chronically ill individual as defined in Code section 401(a)(9)(E)(IV).
 - (ii) A "Multi-Beneficiary Trust" is a trust which (1) has more than one beneficiary, (2) all of the beneficiaries of which are treated as Designated Beneficiaries for purposes of determining the distribution period pursuant to this paragraph, and (3) at least one of the beneficiaries is a Disabled or Chronically III Eligible Designated Beneficiary.
- (6) Death of Eligible Designated Beneficiary. If the Eligible Designated Beneficiary dies prior to the full distribution of the Participant's interest in the Plan, the remainder of the interest must be distributed to the beneficiary of the Eligible Designated Beneficiary no later than December 31 of the calendar year containing the tenth (10th) anniversary of the Eligible Designated Beneficiary's death.
- (7) Attainment of Majority of an Eligible Designated Beneficiary. A Designed Beneficiary who is Eligible Designated Beneficiary due to their being a child of the Participant who has not reached majority will cease to be an Eligible Designated Beneficiary upon reaching majority. Upon ceasing to be an Eligible Designated Beneficiary, the Designated Beneficiary's entire interest in the Plan will be distributed within ten (10) years of the date they ceased to be an Eligible Designated Beneficiary.

July 27, 2022 Agenda Item: 5A ATTACHMENT 1

DATAIR TRUST AGREEMENT DEFINED CONTRIBUTION PRE-APPROVED PLAN BASE PLAN DOCUMENT #20

DATAIR TRUST AGREEMENT DEFINED CONTRIBUTION PRE-APPROVED PLAN BASE PLAN DOCUMENT #20

ARTICLE I - INTRODUCTION	1
1.1 Creation and Title.	1
1.2 Subject to the Requirements of the Plan.	1
1.3 Effective Date. 1.4 Purpose.	1 1
1.5 Plan Administration.	1
	•
ARTICLE II - TRUSTEE	2
2.1 Acceptance of Trust.	2
2.2 Trustee Capacity - Co-Trustees.	2
2.3 Resignation, Removal, and Successors.	2
2.4 Consultations.	2
2.5 Rights, Powers, and Duties.2.6 Right of Trustee to Contributions.	2 4
2.7 Trustee Indemnification.	5
2.8 Changes in Trustee Authority.	5
2.9 Right to Compensation.	5
ARTICLE III - FIDUCIARY DUTIES	6
3.1 Standard of Conduct.	6
3.2 Individual Fiduciaries.	6
3.3 Incorporation of Plan Provisions Regarding Fiduciaries.	6
ARTICLE IV - TRUST ASSETS	7
4.1 Trustee Is Exclusive Owner.	7
4.2 Investments.	7
4.3 Administration of Trust Assets.	8
4.4 Segregated Funds.	9
4.5 Investment Control Option.	9
ARTICLE V - AMENDMENT AND TERMINATION	11
5.1 Amendments.	11
5.2 Manner of Amending.	11
5.3 Limitations On Amendments.	11
5.4 Trustee Powers Pending Final Distribution.	11
5.5 Delegation to Sponsor.	11
ARTICLE VI - MISCELLANEOUS	12
6.1 No Reversion to Employer.	12
6.2 Persons Dealing With Trustee Protected.	12
6.3 Notices.	12
6.4 Governing Law.	12

	July 27, 2022 Agenda Item: 5A ATTACHMENT 1		
6.5 Severability of Provisions.	12		
6.6 Gender and Number.	12		
6.7. Qualification Under Internal Payanua Lawa	12		

ARTICLE I

INTRODUCTION

- **1.1 Creation and Title.** The Employer and the undersigned Trustee(s) hereby create a Trust to be used with the Plan Donner Summit Public Utility District Money Purchase Pension Plan (the "Trust"). All terms used herein have the same meaning as described in the DATAIR Defined Contribution Pre-Approved Base Plan Document #20.
- **1.2 Subject to the Requirements of the Plan.** All Plan provisions are hereby incorporated by reference and apply to the Trustee in the same manner as if they were stated in this Trust Agreement. This Trust is subject to the terms of the Plan, and the Plan shall control in the event of any conflict between the Plan document and this Agreement.
- **1.3 Effective Date.** The provisions of this Trust shall be effective as of the Effective Date specified in the most recent Adoption Agreement unless modified herein.
- **1.4 Purpose.** The Plan and this Trust are established to provide retirement benefits to Eligible Employees in accordance with the Plan and the Adoption Agreement.
- **1.5 Plan Administration.** The Plan is administered by the Plan Administrator. The Trustee shall act according to the Plan Administrator's directions unless such directions are inconsistent with the terms of the Plan or applicable law.

ARTICLE II

TRUSTEE

2.1 Acceptance of Trust. The Trustee, by the execution of this Trust Agreement, agrees to act in accordance with the express terms and conditions hereof and the Plan.

2.2 Trustee Capacity - Co-Trustees.

- (a) The Trustee may be a bank, trust company, or other corporation possessing trust powers under applicable state or federal law or one or more individuals or any combination thereof.
- (b) The term "Trustee" as used herein also includes any person holding the assets of a custodial account, an annuity contract, or other contract that is treated as a qualified trust pursuant to Code section 401(f), and references to the Trust Fund shall be construed to apply to such custodial account, annuity contract, or other contract.
- (c) At any time that a group of individuals is acting as Trustee, the number of such persons who shall act in such capacity from time to time shall be determined by the Employer. The Employer shall appoint such persons, who may or may not be Participants or Employees of the Employer. When there are two or more Trustees, they may allocate specific responsibilities, obligations, or duties among themselves by their written agreement. An executed copy of such written agreement shall be delivered to, and retained by, the Plan Administrator. Unless otherwise elected by the Employer in the Adoption Agreement, any action taken by the Trustees shall be taken at the direction of a majority of such Trustees, or, if the number of such Trustees is two, by unanimous consent.
- **2.3 Resignation, Removal, and Successors.** Any Trustee may resign at any time by delivering to the Employer a written notice of resignation to take effect at a date specified therein, which shall not be less than thirty days after the delivery thereof; the Employer may waive such notice. The Trustee may be removed by the Employer with or without cause, by tendering to the Trustee a written notice of removal to take effect at a date specified therein.

Upon such removal or resignation of a Trustee, the Employer shall either appoint a successor Trustee who shall have the same powers and duties as those conferred upon the resigning or discharged Trustee, or, if a group of individuals is acting as Trustee, determine that a successor shall not be appointed, and the number of Trustees shall be reduced by one.

In the event that a removal, resignation or incapacity of a Trustee results in no remaining Trustees and the Employer will not, or cannot, appoint a successor Trustee in a timely manner, a majority of the Participants in the Plan will have the authority, but not an obligation, to appoint a successor Trustee.

2.4 Consultations. The Trustee shall be entitled to advice of counsel, which may be counsel for the Plan or the Employer, in any case in which the Trustee shall deem such advice necessary. The Trustee shall not be liable for any action taken or omitted in good faith reliance upon the advice of such counsel.

With the exception of those powers and duties specifically allocated to the Trustee by the express terms of the Plan, it shall not be the responsibility of the Trustee to interpret the terms of the Plan, and the Trustee may request, and is entitled to receive, guidance and written direction from the Plan Administrator on any point requiring construction or interpretation of the Plan documents.

The Trustee shall be entitled to rely and act on the information provided by the Employer and/or Plan Administrator and shall have no duty or responsibility to further verify or question such provided information.

- 2.5 Rights, Powers, and Duties. The rights, powers, and duties of the Trustee are:
- (a) The Trustee shall have exclusive authority, discretion, and responsibility for the management and control of the assets of the Trust Fund in accordance with the provisions of the Plan and any amendments thereto, but the Employer may limit the exclusive authority, discretion, and responsibility of the Trustee by written direction delivered to the Trustee. The duties of the Trustee under the Plan shall be determined solely by the express provisions hereof, and no other further duties or responsibilities shall be implied. Subject to the terms of the Plan, the Trustee shall be fully protected and shall incur no liability in acting in reliance upon the written instructions or directions of the Employer, the Plan Administrator, a duly designated investment manager, or any other named Fiduciary.
- (b) The Trustee shall have all powers necessary or convenient for the orderly and efficient performance of its duties hereunder, including but not limited to those specified in this Section. The Trustee shall have the power generally to do all acts, whether or not expressly authorized, that the Trustee in the exercise of its fiduciary responsibility may deem necessary or desirable for the protection of the Trust Fund and the assets thereof.
- (c) The Trustee shall have the power to collect and receive any and all moneys and other property due the Plan and to give full discharge and release therefor; to settle, compromise, or submit to arbitration any claims, debts, or damages due to or owing to or from the Trust Fund; to commence or defend suits or legal proceedings whenever, in the Trustee's judgment, any interest of the Trust Fund requires it; and to represent the Trust Fund in all suits or legal proceedings in any court of law or equity or before any other body or tribunal. Claims that are not subject to arbitration are ERISA claims, claims involving participants and claims that affect the qualification of the Plan.
- (d) The Trustee shall cause any Life Insurance Policies or assets of the Trust Fund to be registered in its name as Trustee and shall be authorized to exercise any and all ownership rights regarding these assets, subject to the terms of the Plan.
 - (i) The requirements of paragraph (d) of this Section will not fail to be satisfied merely because securities of a plan are held in the name of a nominee or in street name, provided such securities are held on behalf of the Plan by:
 - (A) A bank or trust company that is subject to supervision by the United States or a State, or a nominee of such bank or trust company;
 - **(B)** A broker or dealer registered under the Securities Exchange Act of 1934, or a nominee of such broker or dealer; or
 - **(C)** A "clearing agency," as defined in section 3(a)(23) of the Securities Exchange Act of 1934, or its nominee.
 - (ii) Where a corporation described in section 501(c)(2) of the Internal Revenue Code holds property on behalf of the Plan, the requirements of paragraph (d) of this Section are satisfied with respect to such property if all the stock of such corporation is held in trust on behalf of the Plan by one or more trustees.
 - (iii) If the assets of an entity in which the Plan invests include Plan assets by reason of the Plan's investment in the entity, the requirements of paragraph (d) of this Section are satisfied with respect to such investment if the indicia of ownership of the Plan's interest in the entity are held in trust on behalf of the Plan by one or more trustees.
- (e) The Trustee may temporarily hold cash balances and shall be entitled to deposit any funds received in a bank account in the name of the Trust Fund in any bank selected by the Trustee, including the banking department of a corporate Trustee, if any, pending disposition of such funds in accordance with the Plan. Any such deposit may be made with or without interest.

- (f) The Trustee shall pay the premiums and other charges due and payable at any time on any Life Insurance Policies as the Plan Administrator may direct, provided funds for such payments are then available in the Trust. The Trustee shall be responsible only for such funds and Life Insurance Policies that it actually receives as Trustee and shall have no obligation to make payments other than from such funds and cash values of Life Insurance Policies.
- (g) If the whole or any part of the Trust Fund shall become liable for the payment of any estate, inheritance, income, or other tax that the Trustee shall be required to pay, the Trustee shall have full power and authority to pay such tax out of any moneys or other property in its possession for the Account of the person whose Plan interest is so liable. Prior to making any payment, the Trustee may require such releases or other documents from any lawful taxing authority as it shall deem necessary. The Trustee shall not be liable for any nonpayment of tax when it distributes an interest hereunder on instructions from the Plan Administrator.
- (h) The Trustee shall keep a full, accurate, and detailed record of all transactions of the Trust, which the Employer and the Plan Administrator shall have the right to examine at any time during the Trustee's regular business hours. As of the close of each Plan Year, the Trustee shall furnish the Plan Administrator with a statement of account setting forth all receipts, disbursements, and other transactions effected by the Trustee during the year. The Plan Administrator shall promptly notify the Trustee in writing of his approval or disapproval of the statement of account.

The Plan Administrator's failure to provide written disapproval of the statement within sixty (60) days after receipt shall be considered an approval. Except as otherwise required by law, the Plan Administrator's approval shall be binding on all matters embraced in any statement to the same extent as if the statement of the Trustee had been settled by judgment or decree of a court of competent jurisdiction under which the Trustee, Employer, and all persons having or claiming any interest in the Trust Fund were parties; provided, however, that the Trustee may have its account judicially settled if it so desires.

- (i) The Trustee is hereby authorized to execute all necessary receipts and releases to any parties concerned; and shall be under a duty, upon being advised by the Plan Administrator that the proceeds of any Life Insurance Policies are payable, to give reasonable assistance to the Beneficiary designated therein to collect such sums as may appear to be due and upon payment, transfer such sums to the Beneficiary.
- (j) If, at any time, as the result of the death of the Participant, there is a dispute regarding the person to whom payment or delivery of moneys or property should be made by the Trustee, or regarding any action to be taken by the Trustee, the Trustee may postpone such payment, delivery, or action, retaining the funds or property involved, until such dispute shall have been resolved in a court of competent jurisdiction, the Trustee shall have been indemnified to its satisfaction, or it has received written direction from the Plan Administrator.
- (k) Anything in this instrument to the contrary notwithstanding, the Trustee shall have no duty or responsibility with respect to the determination of matters pertaining to the eligibility of any Employee to become or remain a Participant under the Plan, the amount of benefit to which any Participant or Beneficiary shall be entitled under the Plan, or the size and type of any Life Insurance Policy to be purchased from any Insurer for any Participant under the Plan; all such responsibilities being vested in the Plan Administrator.
- **2.6 Right of Trustee to Contributions.** Unless provided with written direction to the contrary, by the Employer or the Plan Administrator, the Trustee shall have the duty to require any contribution to be made or to determine whether contributions delivered to the Trustee by the Employer comply with the provisions of this Agreement. In the event the Employer assigns the duty to another Fiduciary Trustee shall be notified in writing of the Named Fiduciary to whom such responsibility has been assigned. In this case, the Trustee shall be accountable only for funds actually received by the Trustee.

If the Plan has two or more Named Fiduciaries or Trustees assigned this duty, it may be allocated to a single Named Fiduciary or Trustee. The Employer or Plan Administrator may also provide that a Named Fiduciary may direct the Trustee as to this responsibility or may appoint an Investment Manager to take on

this duty. To the extent the nature and scope of the Trustee's responsibilities are specifically limited in the Plan documents, Trust Agreement or by written direction, it will be the responsibility of the Named Fiduciary with the authority to hire and monitor Trustees to assure that all Trustee responsibilities with respect to the management and control of the Plan's assets (including collecting delinquent contributions) have been properly assigned to a Trustee, Investment Manager or Named Fiduciary.

The Plan must make systematic, reasonable and diligent efforts to collect delinquent Employer Contributions. The steps necessary to discharge this duty to collect contributions will depend on the available facts. In determining what collection actions to take, a Named Fiduciary must weigh the value of the Plan assets involved, the likelihood of a successful recovery, and the expenses expected to be incurred. Among other factors, the Named Fiduciary may take into account the Employer's solvency in deciding whether to expend Plan assets to pursue a claim.

2.7 Trustee Indemnification. The Employer shall indemnify and hold harmless the Trustee, its affiliates, and their respective directors, managers, officers, employees, agents, and other representatives for and from the assertion or occurrence of any liability to a Participant or Beneficiary for any action taken or omitted by the Trustee pursuant to any written direction to the Trustee from the Employer or the Plan Administrator. Such indemnification obligation of the Employer shall not be applicable to the extent that any such liability is covered by insurance. The Trustee shall have not duty to take any action other than the express responsibilities under this Plan unless the Trustee and the Employer agree to further responsibilities in an additional written document.

The Trustee shall not be liable to the Employer for any act, omission, or determination made in connection with the Plan except for its gross negligence or willful misconduct.

- **2.8 Changes in Trustee Authority.** If a successor Trustee is appointed, neither an Insurer nor any other person who has previously had dealings with the former Trustee shall be chargeable with knowledge of such appointment or change until furnished with written notice. Until such notice, the Insurer and any other such party shall be fully protected in relying on any action taken or signature presented that would have been proper in accordance with information previously received.
- **2.9 Right to Compensation.** The Trustee may receive reasonable compensation as may be agreed upon between the Trustee and the Employer. The Trustee is entitled to reimbursement for all proper expenses incurred in carrying out their duties under the Plan including, but not limited to, reasonable legal, accounting, and actuarial expenses.

ARTICLE III

FIDUCIARY DUTIES

- **3.1 Standard of Conduct.** The duties and responsibilities of the Trustee with respect to the Plan shall be carried out (a) in a non-discriminatory manner; (b) for the exclusive benefit of Participants and their Beneficiaries; (c) by defraying the reasonable expenses of administering the Plan; (d) with the care, skill, prudence, and diligence under the circumstances then prevailing that a prudent person acting in a like capacity and familiar with such matters would use in the conduct of an enterprise of a like character and with like aims; (e) by diversifying the investments of the Plan to minimize the risk of large losses, unless under the circumstances it is clearly prudent not to do so; and (f) in accordance with the documents and instruments governing the Plan insofar as such documents and instruments are consistent with the provisions of the Act.
- **3.2 Individual Fiduciaries.** At any time that a group of individuals is acting in the same Fiduciary capacity, the number of such persons who shall act in such capacity from time to time shall be determined by the Employer. Such persons shall be appointed by the Employer and may or may not be Participants or Employees of the Employer. Unless otherwise elected by the Employer in the Adoption Agreement, any action taken by a group of individuals acting in such capacity shall be taken at the direction of a majority of such persons, or, if the number of such persons is two (2), by unanimous consent.
- **3.3 Incorporation of Plan Provisions Regarding Fiduciaries.** The provisions of Sections 3.3.1 through 3.3.11 of the Plan are hereby incorporated by reference, and such provisions shall apply to the Trustee in the same manner as they apply to any other Fiduciary.

ARTICLE IV

TRUST ASSETS

- **4.1 Trustee Is Exclusive Owner.** All assets held by the Trustee, whether in the Trust Fund or Segregated Funds, shall be owned exclusively by the Trustee, and no Participant or Beneficiary shall have any individual ownership thereof. Participants and their Beneficiaries shall share in the assets of the Trust, its net earnings, profits and losses, only as provided in the Plan.
- **4.2 Investments.** The Trustee shall invest and reinvest the Trust Fund without distinction between income and principal in one or more of the following ways, as the Trustee shall from time to time determine:
- (a) The Trustee may invest the Trust Fund or any portion thereof in obligations issued or guaranteed by the United States of America or of any instrumentality thereof, or in other bonds, notes, debentures, mortgages, preferred or common stocks, options to buy or sell stocks or other securities, mutual fund shares, limited partnership interests, commodities, real estate or any interest therein, or in such other property, real or personal, as the Trustee shall determine.
- (b) The Trustee may cause the Trust Fund or any portion thereof to be invested in a common trust fund established and maintained by a national or other bank regulated by the Federal Deposit Insurance Corporation, for the collective investment of fiduciary funds, even though the bank is acting as the Trustee or investment manager, provided such common trust fund is a qualified trust under the applicable Code section, or corresponding provisions of future federal internal revenue laws, and is exempt from income tax under the applicable Code section. In the event any assets of the Trust Fund are invested in such a common trust fund, the Declaration of Trust creating such common trust fund, as it may be amended from time to time, shall be incorporated into the Plan by reference and made a part thereof.

Further, all or any portion of the assets subject to this Trust Agreement may be invested in any collective investment fund maintained exclusively for the investment of assets of (1) exempt, qualified employee benefit trusts and (2) collective investment funds consisting exclusively of assets of such qualified trust. The assets so invested shall be subject to all the provisions of the instrument establishing such collective investment fund, as such instrument may be amended from time to time. Such instrument, as amended from time to time, is hereby incorporated and made a part of this Trust Agreement and shall control notwithstanding any contrary provision of this Trust Agreement or the Plan.

- (c) The Trustee may deposit any portion of the Trust Fund in savings accounts in federally insured banks or savings and loan associations or invest in certificates of deposit issued by any such bank or savings and loan association. The Trustee may retain, without liability for interest, any portion of the Trust Fund in cash balances pending investment thereof or payment of expenses.
- (d) The Trustee may buy and sell put and call options, covered or uncovered, engage in spreads, straddles, ratio writing and other forms of options trading, including sales of options against convertible bonds, and sales of Standard & Poor futures contracts, and trade in and maintain a brokerage account on a cash or margin basis.
- (e) The Trustee may invest any portion or all of the assets of the Trust Fund that are attributable to the vested and nonforfeitable interest in the Accounts of a Participant in the purchase of group or individual Life Insurance Policies issued on the life of the Participant or someone in whom the Participant has an insurable interest and for the benefit of the Participant with the consent of the Participant, subject to the following conditions, as they are represented by the Plan Administrator:
 - (i) If ordinary Life Insurance Policies are used, the aggregate life insurance premiums must be less than one-half of the aggregate Employer Contributions and Forfeitures allocated to the Participant's Account at any particular time, without regard to Trust earnings, capital gains and

- losses; for purposes of this Plan, the term "Ordinary Life Insurance Policies" shall mean Policies with both nondecreasing death benefits and nonincreasing premiums.
- (ii) The aggregate premiums paid for Life Insurance Policies on the life of any Participant or someone in whom the Participant has an insurable interest that are either term, universal or any other contracts which are not ordinary whole life Policies shall not at any time exceed twenty-five percent (25%) of the aggregate amount of Employer Contributions and Forfeitures that have been allocated to the Accounts of such Participant.
- (iii) The sum of one-half of the aggregate premiums for ordinary whole Life Insurance Policies and all premiums for other Life Insurance Policies shall not at any time exceed twenty-five percent (25%) of the aggregate amount of Employer Contributions and Forfeitures that have been allocated to the Accounts of such Participant.
- (iv) If the Plan permits in-service distributions to a Participant prior to his Normal Retirement Date, the amount that may be distributed to the Participant may be applied to the purchase of Life Insurance Policies.
- (f) The Trustee may invest the Trust Fund or any portion thereof to acquire or hold Qualifying Employer Securities or Real Property, provided that the portion so invested shall not exceed the amount allowed as an investment under ERISA section 407. There shall be no limit on the acquisition of Qualifying Employer Securities in an individual account balance plan in which Participants may direct the Trustee to buy Qualifying Employer Securities on their behalf.
- (g) If permitted, the Participant may direct the purchase of life insurance on his life, on the joint lives of the Participant and someone in whom the Participant has an insurable interest, or on the life of someone in whom the Participant has an insurable interest. The amount that can be used to pay life insurance premiums shall be determined according to this Section.
- **4.3 Administration of Trust Assets.** Subject to the limitations expressly set forth here, and in the Plan, the Trustee shall have the following powers and authority in connection with the administration of the assets of the Trust:
- (a) To hold and administer all contributions made by the Employer to the Trust Fund and all income or other property derived therefrom as a single Trust Fund, except as otherwise provided in the Plan;
- (b) To manage, control, sell, convey, exchange, petition, divide, subdivide, improve, repair, grant options, sell upon deferred payments, lease without limit as determined for any purpose, compromise, arbitrate, or otherwise settle claims in favor of or against the Trust Fund, institute, compromise, and defend actions and proceedings, and to take any other action necessary or desirable in connection with the administration of the Trust Fund;
- (c) To vote any stock, bonds, or other securities of any corporation or other issuer; otherwise consent to or request any action on the part of any such corporation or other issuer; to give general or special proxies or powers of attorney, with or without power of substitution; to participate in any reorganization, recapitalization, consolidation, merger, or similar transaction with respect to such securities; to deposit such stocks or other securities in any voting trusts, or with any protective or like committee, or with the trustee, or with the depositories designated thereby; to exercise any subscription rights and conversion privileges or other options and to make any payments incidental thereto; and generally to do all such acts, execute all such instruments, take all such proceedings and exercise all such rights, powers, and privileges with respect to the stock or other securities or property constituting the Trust Fund as if the Trustee were the absolute owner thereof;
- (d) To apply for and procure, at the election of any Participant, Life Insurance Policies on the life of the Participant or someone in whom the Participant has an insurable interest; to exercise whatever rights and privileges may be granted to the Trustee under such Policies, and to cash in, receive, and collect such Policies or the proceeds therefrom as and when entitled to do so under the provisions thereof;

- (e) To make, execute, acknowledge, and deliver any and all documents of transfer and conveyance and any and all other instruments that may be necessary or appropriate to carry out the powers herein granted;
- (f) To register any investment held in the Trust in the Trustee's own name or in the name of a nominee for the benefit of the Plan, or to hold any investment in bearer form, provided that the books and records of the Trustee shall at all times show that all such investments are part of the Trust;
- (g) To borrow money for the purposes of the Plan in such amounts and upon such terms and conditions as the Trustee deems appropriate;
- (h) To commingle the assets of the Trust Fund with the assets of other similar trusts that are exempt from income tax, whether sponsored by the Employer, an affiliate of the Employer, or an unrelated employer, provided that the books and records of the Trustee shall at all times show the portion of the commingled assets that are part of the Trust; and
- (i) To do all acts, whether or not expressly authorized, that the Trustee may deem necessary or proper for the protection of the property held hereunder.
- **4.4 Segregated Funds.** Unless otherwise determined by the Trustee to be imprudent, the Trustee shall invest and reinvest each Segregated Fund without distinction between income and principal. Such accounts shall be held for the benefit of the Participant for whom such Segregated Fund is established in accordance with the terms of the Plan, and the Participant's Segregated Account shall be credited with any interest earned in connection with such accounts. If the Trustee determines that an alternative investment is appropriate, the Trustee may invest the Segregated Fund in any manner permitted with respect to the Trust Fund and such Segregated Fund shall be credited with the net income or loss or net appreciation or depreciation in value of such investments. No Segregated Fund shall share in any Employer Contributions or Forfeitures, any net income or loss from, or net appreciation or depreciation in value of, any investments of the Trust Fund, or any allocation for which provision is made in the Plan that is not specifically attributable to the Segregated Fund.
- **4.5 Investment Control Option.** If the Employer elects in the Adoption Agreement to permit Participants to direct the investment of their 414(k) Accounts and Rollover Accounts, each Participant may elect to transfer funds that do not exceed the balances in his Trustee-directed Accounts to a Controlled Account and exercise investment control of those funds by appropriate direction to the Trustee.

To the extent that the funds to be transferred from a Participant's Account include his share of an Employer Contribution or other funds that have not yet been received by the Trustee, such transfer shall not occur until the Trustee receives such funds. Funds transferred to a Controlled Account on behalf of the Participant shall be thereafter be invested by the Trustee in such property, real or personal (other than collectibles), as the Participant shall direct from time to time in writing, unless policies adopted pursuant to Section 3.4.6 of the Plan permit electromechanical communications in this context. The Participant may not direct the Trustee to make distributions or loans to himself, nor to make loans to the Employer, and the Trustee may limit the investment alternatives available to the Participant in a uniform and nondiscriminatory manner.

A Participant shall make such election by giving notice thereof to the Trustee as the Trustee deems necessary, and such notice shall specify the source and amount of such funds to be transferred. Any such election shall be at the absolute discretion of the individual Participant and shall be binding upon the Trustee unless the Trustee has actual knowledge that such transaction is a prohibited transaction under the Code or the Act. Upon any such election being made, the amount of such funds to be transferred shall be deducted from his Trustee-directed Accounts as appropriate and added to a Controlled Account of the Participant. All earnings thereafter received with respect to such transferred funds, as well as any appreciation or depreciation in his investments, shall be added to or deducted from his Controlled Account.

If the Employer elects to follow ERISA section 404(c), the Plan Administrator or appropriate Fiduciary shall ensure that the Plan provides Participants with the minimum options and information required by ERISA section 404(c) and the regulations thereunder.

If a Participant elects to transfer funds from his Trustee-directed Accounts to a Controlled Account as of a date other than a Valuation Date, the Trustee may defer such transfer until the next succeeding Valuation Date or may make such transfer, provided that the Trustee determines that the assets in the Trust Fund are such that it is feasible and practical to make such transfer. As of the date of such transfer, adjustments to the Participant's Accounts shall be made as if such date is a Valuation Date.

As of any Valuation Date, the Participant may elect to have all or any portion of any cash contained in his Controlled Account transferred back to the general assets of the Trust Fund, in which case the Trustee shall again invest such cash as part of the general assets of the Trust Fund. A Participant shall make such election by giving notice to the Trustee, in the manner that the Trustee deems necessary, and the notice shall specify the amount of cash to be transferred. The amount of such funds so transferred shall be deducted from the Participant's Controlled Account and added to the appropriate Account of the Participant. Any such election shall be at the absolute discretion of the individual Participant and shall be binding upon the Trustee.

The Trustee shall not have any investment responsibility with respect to a Participant's Controlled Account. In the event that a Participant elects to have any such funds transferred to a Controlled Account and invested in particular securities or assets pursuant to this Section, the Trustee shall not be liable for any loss or damage resulting from the investment decision of the Participant.

ARTICLE V

AMENDMENT AND TERMINATION

- 5.1 Amendments. The Employer may at any time or times amend this Trust, in whole or in part.
- **5.2 Manner of Amending.** Each amendment of this Trust shall be made by delivery to the Trustee of a copy of the Employer resolution that sets forth such amendment.

5.3 Limitations On Amendments.

No amendment to this Trust shall:

- (1) Directly or indirectly operate to give the Employer any interest whatsoever in the assets of the Trust or a custodial account or to deprive any Participant or Beneficiary of his vested and nonforfeitable interest in the assets of the Trust as then constituted, or cause any part of the income or corpus of the Trust to be used for, or diverted to purposes other than the exclusive benefit of Employees or their Beneficiaries; or
- (2) Increase the duties or liabilities of the Trustee without the Trustee's prior written consent. The Employer will not be considered to have an individually designed plan merely because the Employer amends administrative provisions of the Trust or custodial account document (such as provisions relating to investments and duties of Trustees) so long as the amended provisions are not in conflict with any other provision of the Plan and do not cause the Plan to fail to qualify under Code section 401(a).
- **5.4 Trustee Powers Pending Final Distribution.** Sections 3.8.4 through 3.8.6 of the Plan provide for the termination or partial termination of the Plan and the withdrawal of an Employer from participation in the Plan. Until final distribution of the assets of the Trust, the Trustee shall continue to have all the powers provided under the Plan and this Trust as are necessary for the orderly administration, liquidation, and distribution of the assets of the Trust.
- **5.5 Delegation to Sponsor.** The Employer expressly delegates authority to the Plan Sponsor the right to amend any part of this Trust on its behalf to the extent necessary to preserve the qualified status of the Plan. For purposes of amendments by the Plan Sponsor, the Plan Document Provider shall be recognized as the agent of the Plan Sponsor. If the Plan Sponsor does not adopt the amendments made by the Plan Document Provider, the Plan shall no longer be identical to or a minor modifier of the Mass Submitter plan. The Plan Sponsor shall submit a copy of the amendment to each Employer who has adopted the Plan after first having received a ruling or favorable determination from the Internal Revenue Service that the Plan as amended satisfies the applicable requirements of the Code.

ARTICLE VI

MISCELLANEOUS

- **6.1 No Reversion to Employer.** Except as specifically provided in the Plan, no part of the corpus or income of the Trust shall revert to the Employer or be used for, or diverted to, purposes other than for the exclusive benefit of Participants and their Beneficiaries.
- **6.2 Persons Dealing With Trustee Protected.** No person dealing with the Trustee shall be required or entitled to see to the application of any money paid or property delivered to the Trustee, or determine whether the Trustee is acting pursuant to the authorities granted to the Trustee hereunder or to authorizations or directions herein required. The certificate of the Trustee that the Trustee is acting in accordance with the Plan shall protect any person relying thereon.
- **6.3 Notices.** Any notice or direction to be given in accordance with this Trust shall be deemed to have been effectively given if hand delivered to the recipient or sent by certified mail, return receipt requested, to the recipient at the recipient's last known address. At any time that a group of individuals is acting as Trustee or in a Fiduciary capacity, notice to the Trustee or such Fiduciary may be given by giving notice to any one or more of such individuals.
- **6.4 Governing Law.** The provisions of this Trust shall be construed, administered, and enforced in accordance with the provisions of the Act and, to the extent applicable, the laws of the state specified in the Adoption Agreement. All contributions to the Trust shall be deemed to take place in such state.
- **6.5 Severability of Provisions.** In the event that any provision of this Trust shall be held to be illegal, invalid, or unenforceable for any reason, said illegality, invalidity or unenforceability shall not affect the remaining provisions, but shall be fully severable and the Trust shall be construed and enforced as if said illegal, invalid or unenforceable provisions had never been inserted herein.
- **6.6 Gender and Number.** Whenever appropriate, words used in the singular shall include the plural, and the masculine gender shall include the feminine gender and vice versa.
- **6.7 Qualification Under Internal Revenue Laws.** The Employer intends that the Trust qualify under the applicable Code provisions. Until advised to the contrary, the Trustee may assume that the Trust is so qualified and is entitled to tax exemption under the Code. If the Plan of the Employer fails to attain or retain qualification, the Employer's Plan shall no longer participate in the Prototype Plan and shall be considered an individually designed plan.

The Employer and the Trustee(s) hereby adopt the foregoing Trust on this 31st day of July, 2022.

Employer: Donner Summit Public Utility District	
Cathy Preis President	Cathy A. Preis Trustee
	Philip Gamick Trustee

Donner Summit Public Utility District Money Purchase Pension Plan LOAN POLICY

Pursuant to the terms of Donner Summit Public Utility District Money Purchase Pension Plan (the "Plan"), the Plan Administrator has adopted a participant loan program as part of such Plan and Trust. All loans granted or renewed on or after the 1st day of January, 2022 will be made in accordance with the provisions specified in the Plan and under this Loan Policy. The Plan intends this loan program to comply with all applicable requirements of the Internal Revenue Code and the Department of Labor. Violating the terms of this Policy may cause a loan to be treated as a taxable distribution from the Plan.

Administration of Program.

Donner Summit Public Utility District (the "Loan Administrator") is responsible for the administration of this loan program. All loan requests and other inquiries should be delivered to:

Donner Summit Public Utility District 53823 Sherritt Lane Soda Springs, CA 95728 (530) 426-3456

Application Procedure.

The Loan Administrator will make loan applications available to any individual who has a vested interest under the Plan. An eligible individual, as defined within this Policy, may apply for a loan by returning a paper loan application.

Loan applications will be processed as soon as administratively feasible following receipt of the completed loan application.

The Loan Administrator will review the loan application for completeness. Incomplete applications will be returned and must be resubmitted for consideration. All loan applicants must meet the requirements of this Policy for consideration and approval. If the loan applicant fails to meet the requirements of this Policy and receives a loan disbursement, the loan will be treated as a "deemed distribution" and reported as taxable to the IRS.

Basis for Approvals.

Loans are available to all actively employed participants.

Loans will be available without regard to any individual's race, color, religion, sex, age, or national origin.

Each application is reviewed on a nondiscriminatory basis. However, its approval will depend on the applicant's creditworthiness.

If an applicant defaulted on a previous loan, the loan application will be denied until the deemed loan has been repaid or offset.

In addition, if a participant submits a loan application at a time when a decision concerning a domestic relations order is pending or the Plan Administrator is on notice that divorce is in progress, the loan request will be placed on hold until the order is finalized or the determination period expires.

Once the loan is approved, a Promissory Note will be generated and issued to the applicant. The applicant must sign the Promissory Note to acknowledge and document receipt of the loan disbursement from the Plan and to affirm such applicant's obligation to make the required repayments.

Spousal Consent.

If a married participant submits a loan application for an amount greater than \$5,000, spousal consent must be obtained no earlier than 90 days before the date on which the loan is processed and secured. The consent must be in writing and properly executed on the loan application and Promissory Note. The participant's spouse is required to acknowledge consent to the participant's pledge and assignment of 50% of the participant's vested interest in the Plan; and to the effect such consent will have on the benefit that would otherwise be paid under the Plan, in the event of a default. The spouse must sign the consent in the presence of a Notary Public or the Plan Administrator who can attest to the spouse's willingness to consent and endorse the loan documents. A new consent must be obtained if the amount of a loan is increased or if a loan is renegotiated, extended, renewed or otherwise revised.

Terms of the Loan.

Subject to the limitations on the amount of any loan, a participant may request a loan for any purpose.

In addition, a participant may rollover a loan from another employer sponsored plan into this plan. The loan must be in good standing (payments are not delinquent and loan has not been deemed distributed). The participant must contact the Loan Administrator to complete the necessary forms and to provide documentation from the other plan regarding the terms and status of the loan. The Loan Administrator retains the ability to deny a rollover in of a loan if insufficient information is provided to ensure the loan has been administered in compliance with all IRS and Department of Labor requirements.

Loan Amounts.

The Loan Administrator will determine the available loan amount at the time the loan request is approved. The maximum amount of any loan is the lesser of 50% of the participant's vested interest minus any existing loan balance, or \$50,000 reduced by the participant's highest outstanding loan balance in the previous twelve months even if all or a portion of this amount has been repaid.

To calculate the maximum loan amount, the participant's vested interest in all plans established or maintained by the employer or a related employer of the employer will be considered. Notwithstanding, the Plan limits the minimum amount of any loan to \$1,000.

An individual may have no more than 5 loans outstanding at any one time. Loans that have been deemed distributed will be counted as outstanding loans until they have been either repaid or offset. Subject to the limitations on the amount and number of loans permitted, refinancing of an existing loan is permitted to change the payment schedule, to change the payment terms, and to replace an existing loan.

Sources for a Loan - Accounts and Investment Options.

The participant will select, on a form provided by the Loan Administrator, the accounts and investment fund or funds from which the amount necessary to grant the loan will be taken. However, the loan will not be taken from any insurance policies.

Loans may be taken from all vested participant accounts.

The loan will be transferred to a segregated account. During the term of the loan, this segregated account will be maintained, and each scheduled principal and interest repayment will be made to this segregated account, until the entire loan is paid in full. This segregated account will not share in any gains or losses credited to the Plan that do not directly relate to the loan.

Interest Rate and Fees.

Interest will be charged on each loan. From time to time, the Loan Administrator will review the interest rate charged for loans, with the intention of providing the Plan with a return commensurate with the interest rates that a commercial lender would charge for loans made under similar circumstances. The interest rate for a loan will take into account the creditworthiness of the participant and the terms of the loan.

The interest rate on the loan will be the prime rate as used by Any federal reserve bank adjusted by 1 percentage point(s). The prime rate will be determined

Once the interest rate is determined, the amount of the loan will be amortized according to the selected repayment terms. Each repayment will include both principal and interest until the loan is no longer outstanding in the Plan.

To cover the added administrative costs associated with processing and maintaining a loan under the Plan, the participant will be charged a \$95.00 loan processing fee, and annually, a loan maintenance fee of \$35.00, until the loan is paid in full. In addition, the loan maintenance fee will be charged for all years in which the loan remains in default.

Security for a Loan.

All loans must be adequately secured with at least 50% of the participant's vested account balance in the Plan. The security interest will be determined and measured at the time the loan is granted. The participant must secure each loan with an irrevocable pledge and assignment of at least 50% of such participant's vested account balance under the Plan.

Repayment Terms.

With limited exceptions, the Internal Revenue Code requires a loan to be repaid through level installment payments at least quarterly, over a period not to exceed 5 years.

Under this Loan Policy, a loan is required to be repaid within 5 years, starting from the payment date outlined in the Promissory Note.

However, if the loan application is for a residential loan and the Loan Administrator confirms that there is sufficient documentation that the entire proceeds of the loan will be used to acquire a dwelling unit that will be used as the participant's principal residence, within a reasonable time, then this residential loan must be repaid within 30 years of the original date of the loan.

A principal residence is a house, apartment, condominium, or mobile home (not used on a transient basis) established and used as the participant's principal dwelling unit.

Loans are to be repaid based on substantially level amortization over the term of the loan with payments made each pay period.

Loan payments will be made through payroll deduction.

Early Payoff.

The participant may elect, in writing to the Loan Administrator, to pay off an entire outstanding loan balance in full prior to its due date.

Leave of Absence or Suspension Period.

Loan payments may be suspended for up to one year if the participant is on an approved leave of absence from the employer either without pay or with a pay rate that will not cover the scheduled installment payments. When the participant's leave has ended, the participant may request that the missed payments be reamortized over the remaining term of the loan or be added to the final payment. The loan must be repaid according to its original repayment term or within 5 years as provided by Internal Revenue Code section 72(p)(2)(B).

Special Provisions for Military Service.

The Loan Administrator may temporarily suspend loan repayments, if the participant is not actively employed due to a qualified military leave or because the participant is performing service in the uniformed services (as defined in chapter 43 of title 38 United States Code). In addition, the period of military suspension will extend the original loan term.

Once the military service has ended, loan repayments must resume. The loan must be repaid in full by the end of its original term plus the period of military service.

For example, if the loan was due in 5 years, and the military leave was for 18 months, then the Loan Administrator would extend the 5 year term by the length of the military leave. The final installment payment would be due within 6 years and 6 months of the date the loan was originally issued.

In addition, upon receipt of proper notice of active military service, if the interest rate for an outstanding loan exceeds 6%, the Loan Administrator will reduce the interest rate on such loan to 6% during the period of military service.

Default and Offset.

A loan is in default when a scheduled installment payment has not been received by the scheduled due date. If the participant fails to arrange for the repayment of the missed payment, in a manner that is reasonably acceptable to the Loan Administrator, the remaining principal and accrued interest on the loan will be declared due and payable. The missed payment must be received by the end of the "cure period". The end of the cure period will be the last day of the calendar quarter following the calendar quarter in which the scheduled installment payment was due. After this date, the Loan Administrator will notify the participant in writing that the loan is in default.

The outstanding loan (including accrued interest) will become taxable and treated as a "deemed distribution". A deemed distribution means the entire outstanding loan balance is immediately due and the Loan Administrator will report and process the outstanding loan amount as a taxable distribution from the Plan. The defaulted loan (outstanding principal plus accrued interest thru the date the loan was deemed distributed) will be reported as taxable income on IRS Form 1099-R. It will be subject to federal and state income taxes, and a 10% additional tax on early distributions if the default occurs before age 59-1/2. This amount will not be eligible for rollover to another employer plan or IRA.

The participant is still under an obligation to the Plan to repay the loan. Therefore, the Promissory Note will remain outstanding and interest will continue to accrue. This outstanding loan obligation will be offset against the participant's vested account balance when the participant requests a distribution of their account balance.

A participant whose loan is in good standing (not deemed distributed) may elect to take an in-kind distribution of the promissory note (offset) provided they are able to take a distribution of their account balance(s) that were used to fund the loan without requiring a distribution of non-loan amounts. This will result in full outstanding amount of the loan (including interest accrued thru the distribution date) becoming taxable. The amount will be reported on IRS Form 1099-R and will be eligible for rollover to another employer plan or IRA.



DEPARTMENT OF THE TREASURY INTERNAL REVENUE SERVICE WASHINGTON, D.C. 20224

Plan Description: Non-Standardized Pre-Approved Profit Sharing/Money Purchase/CODA

FFN: 31799858520-001 Case: 201900821 EIN: 36-3328521

Letter Serial No: Q703729a Date of Submission: 12/31/2018

DATAIR EMPLOYEE BENEFIT SYSTEMS INC 735 NORTH CASS AVENUE WESTMONT, IL 60559

Contact Person: Janell Hayes Telephone Number: 513-975-6319

In Reference To: TEGE:EP:7521

Date: 06/30/2020

Dear Applicant:

In our opinion, the form of the plan identified above is acceptable for use by employers for the benefit of their employees under Internal Revenue Code (IRC) Section 401.

We considered the changes in qualification requirements in the 2017 Cumulative List of Notice 2017-37, 2017-29 Internal Revenue Bulletin (IRB) 89. Our opinion relates only to the acceptability of the form of the plan under the IRC. We did not consider the effect of other federal or local statutes.

You must provide the following to each employer who adopts this plan:

- . A copy of this letter
- . A copy of the approved plan
- . Copies of any subsequent amendments including their dates of adoption
- . Direct contact information including address and telephone number of the plan provider

Our opinion on the acceptability of the plan's form is a determination as to the qualification of the plan as adopted by a particular employer only under the circumstances, and to the extent, described in Revenue Procedure (Rev. Proc.) 2017-41, 2017-29 I.R.B. 92. The employer who adopts this plan can generally rely on this letter to the extent described in Rev. Proc. 2017-41. Thus, Employee Plans Determinations, except as provided in Section 12 of Rev. Proc. 2020-4, 2020-01 I.R.B. 148 (as updated annually), will not issue a determination letter to an employer who adopts this plan. Review Rev. Proc. 2020-4 to determine the eligibility of an adopting employer, and the items needed, to submit a determination letter application. The employer must also follow the terms of the plan in operation.

Except as provided below, our opinion doesn't apply to the requirements of IRC Sections 401(a)(4), 401(l), 410(b), and 414(s). Our opinion doesn't apply to IRC Sections 415 and 416 if an employer maintains or ever maintained another qualified plan for one or more employees covered by this plan. For this purpose, we will not consider the employer to have maintained another defined contribution plan provided both of the following are true:

- . The employer terminated the other plan before the effective date of this plan
- . No annual additions have been credited to any participant's account under the other plan as of any date within the limitation year of this plan

Also, for this purpose, we'll consider an employer as maintaining another defined contribution plan, if the employer maintains any of the following:

. A welfare benefit fund defined in IRC Section 419(e), which provides post-retirement medical benefits allocated to separate accounts for key employees as defined in IRC Section 419A(d)

DATAIR EMPLOYEE BENEFIT SYSTEMS INC

FFN: 31799858520-001

Page: 2

. An individual medical account as defined in IRC Section 415(I)(2), which is part of a pension or annuity plan maintained by the employer

. A simplified employee pension plan

Our opinion doesn't apply to Treasury Regulations Section 1.401(a)-1(b)(2) requirements for a money purchase plan or target benefit plan where the normal retirement age under the employer's plan is lower than age 62.

Our opinion doesn't constitute a determination that the plan is an IRC Section 414(d) governmental plan. This letter is not a ruling with respect to the tax treatment to be given contributions which are picked up by the governmental employing unit within the meaning of IRC Section 414(h)(2).

Our opinion doesn't constitute a determination that the plan is an IRC Section 414(e) church plan.

Our opinion may not be relied on by a non-electing church plan for rules governing pre-ERISA participation and coverage.

Our opinion applies to the requirements of IRC Section 410(b) if 100 percent of all non-excludable employees benefit under the plan.

Employers who choose a safe harbor allocation formula and a safe harbor compensation definition may also rely on this opinion letter for the non-discriminatory amounts requirement under IRC Section 401(a)(4).

If this plan includes a cash or deferred arrangement (CODA) or otherwise provides for contributions subject to IRC Sections 401(k) and/or 401(m), the employer may rely on the opinion letter regarding the form of the non-discrimination tests of IRC Sections 401(k)(3) and 401(m)(2), if the employer uses a safe harbor compensation definition. For plans described in IRC Sections 401(k)(12) or (13) and/or 401(m)(11) or (12), employers may rely on the opinion letter regarding whether the plan's form satisfies the requirements of those sections unless the plan provides for the safe harbor contribution to be made under another plan. For SIMPLE plans described in IRC Sections 401(k)(11) and 401(m)(10), employers may also rely on the opinion letter regarding whether the plan's form satisfies the requirements of those sections.

The provisions of this plan override any conflicting provision contained in the trust or custodial account documents used with the plan, and an adopting employer may not rely on this letter to the extent that provisions of a trust or custodial account that are a separate portion of the plan override or conflict with the provisions of the plan document. This opinion letter does not cover any provisions in trust or custodial account documents.

An employer who adopts this plan may not rely on this letter when:

- . the plan is being used to amend or restate a plan of the employer which was not previously qualified
- the employer's adoption of the plan precedes the issuance of the letter
- . the employer doesn't correctly complete the adoption agreement or other elective provisions in the plan
- . the plan is not identical to the pre-approved plan (that is, the employer has made amendments that cause the plan not to be considered identical to the pre-approved plan, as described in Section 8.03 of Rev. Proc. 2017-41)

Our opinion doesn't apply to what is contained in any documents referenced outside the plan or adoption agreement, if applicable, such as a collective bargaining agreement.

Our opinion doesn't consider issues under Title I of the Employee Retirement Income Security Act (ERISA) which are administered by the Department of Labor.

If you, the pre-approved plan provider, have questions about the status of this case, you can call the telephone number at the top of the first page of this letter. This number is only for the provider's use.

July 27, 2022 Agenda Item: 5A ATTACHMENT 1

DATAIR EMPLOYEE BENEFIT SYSTEMS INC

FFN: 31799858520-001

Page: 3

Individual participants or adopting eligible employers with questions about the plan should contact you.

You must include your address and telephone number on the pre-approved plan or the plan's adoption agreement, if applicable, so that adopting employers can contact you directly.

If you write to us about this plan, provide your telephone number and the best time to call if we need more information. Whether you call or write, refer to the letter serial number and file folder number at the top of the first page of this letter.

Let us know if you change or discontinue sponsorship of this plan.

Keep this letter for your records.

Sincerely Yours,

Khin M. Chow

Director, EP Rulings & Agreements

Klein M. Chow

Letter 6186 (June-2020) Catalog Number 72434C

July 27, 2022 Agenda Item: 5A ATTACHMENT 2

SUMMARY PLAN DESCRIPTION FOR DONNER SUMMIT PUBLIC UTILITY DISTRICT MONEY PURCHASE PENSION PLAN

Donner Summit Public Utility District Money Purchase Pension Plan

Summary Plan Description

Table of Contents

DESCRIPTION	PAGE
INTRODUCTION	1
GENERAL INFORMATION	2
PARTICIPATION IN YOUR PLAN What types of contributions are available in the plan? Who may participate? Who is considered an employee? What are the eligibility requirements to become a participant in the plan? What compensation will be used for my contributions in the plan? Is there a limit on compensation for plan purposes? Does plan compensation include monies paid to me during an absence or after my	4 4 4 4 5 5 5
employment ends? How is service determined? What is a year of service? What is a break in service for eligibility purposes? What is a break in service for vesting purposes?	6 6 6
CONTRIBUTIONS	7
YOUR CONTRIBUTIONS TO THE PLAN: Does the plan accept rollovers? What are profit sharing contributions? Are there requirements to receive a profit sharing contribution? Are there any allocation requirements waived for profit sharing contributions? How is the profit sharing contribution determined? When can I expect the profit sharing contributions to be allocated? When can I expect the employer contributions to be deposited? When is a plan top heavy? How will the plan operate in top heavy years?	7 7 7 7 7 7 7 7
VESTING Is any of my service excluded for vesting purposes? How is my vested percentage calculated? What vesting schedule applies to my other accounts? Does my vested percentage change for any reason? What happens if I terminate employment before I am fully vested? What happens to my forfeited amounts if I am rehired into a position covered by the plan? If I am rehired into a position covered by the plan, how is my vesting service calculated?	8 8 8 8 8 9
INVESTMENT ACCOUNTS What is the value of my account? How are my accounts invested?	10 10 10

July 27, 2022 Agenda Item: 5A ATTACHMENT 2

Does my plan offer life insurance as an investment? May I take a loan from my accounts? Where can I learn about the plan expenses?	10 10 10
DISTRIBUTIONS Does the plan allow for hardship distributions? Does the plan allow for in-service distributions? What is my normal retirement age? When will I receive my normal retirement benefits? When will my beneficiary receive my benefits if I die? When will I receive my benefits upon termination? Does the plan have disability benefits? How might divorce or a Qualified Domestic Relations Order affect my benefits? How will I receive my distribution? Will the plan automatically distribute any of my benefits? What is a required minimum distribution? How will my distributions be taxed? Who may I name as my beneficiary?	11 11 11 11 11 11 12 12 12 12 13 13
OTHER IMPORTANT INFORMATION Are my benefits protected? Can the plan be amended or terminated? Does Pension Benefit Guaranty Corporation Insurance apply to this plan? What are the claims for benefit procedures under this plan? What is the waiting period for a non-disability claim? What is the waiting period for a disability claim? What will I receive if the claim is denied in part or whole? What happens to my disability claim if I need to provide information? How do I appeal a claim denial?	14 14 14 14 14 14 15
PARTICIPANT RIGHTS UNDER ERISA Receive information about your plan and your benefits Actions by plan fiduciaries Enforcing your rights Assistance with your questions	16 16 16 16
APPENDIX 1 - LOAN POLICY	18

SUMMARY PLAN DESCRIPTION FOR DONNER SUMMIT PUBLIC UTILITY DISTRICT MONEY PURCHASE PENSION PLAN

INTRODUCTION

Effective January 1, 1981, Donner Summit Public Utility District established the Donner Summit Public Utility District Money Purchase Pension Plan for the exclusive benefit of all eligible employees and their beneficiaries with the intention to provide a measure of retirement security for your future.

This Summary Plan Description reflects the plan options as of January 1, 2022.

This Summary Plan Description is a brief description of your plan and your rights and benefits under the plan and is not intended to cover every plan provision. This Summary Plan Description is not meant to interpret or change the provisions of your plan. A copy of your plan is on file at your employer's office and may be read by you, your beneficiaries, or your legal representatives at any reasonable time. This plan is subject to the provisions of the Employee Retirement Income Security Act of 1974 (ERISA). If you have any questions regarding either your plan or this Summary Plan Description, you should ask your plan administrator. If any discrepancies exist between this Summary Plan Description and the actual provisions of the plan, the plan shall govern.

GENERAL INFORMATION

Plan Name: Donner Summit Public Utility District Money Purchase Pension

Plan

Employer: Donner Summit Public Utility District

53823 Sherritt Lane Soda Springs, CA 95728

(530) 426-3456

Employer Tax ID: 94-1602556

Three Digit Plan Number: 001

Type of Plan: Profit Sharing Plan

Administration Type: Plan Administrator

Plan Administrator: Donner Summit Public Utility District

53823 Sherritt Lane Soda Springs, CA 95728

(530) 426-3456

Plan Administrator ID Number: 94-1602556

Legal Agent: Donner Summit Public Utility District

53823 Sherritt Lane Soda Springs, CA 95728

(530) 426-3456

Service of legal process may also be made upon a plan trustee

or the plan administrator as listed herein.

Trustees: Cathy A. Preis

53823 Sherritt Lane Soda Springs, CA 95728

(530) 426-3456

Philip Gamick 53823 Sherritt Lane Soda Springs, CA 95728

(530) 426-3456

Funding Arrangement: Trust

Trust Tax ID Number: 68-6090198

July 27, 2022 Agenda Item: 5A ATTACHMENT 2

Plan Year: July 1st to June 30th

Limitation Year: July 1st to June 30th

Anniversary Date: June 30th

Valuation Date: The last day of the plan year

PARTICIPATION IN YOUR PLAN

In order to take advantage of the opportunities provided by your plan you must participate in the plan. There may be certain restrictions to your eligibility and participation. The following is information about how you can participate in the plan.

What types of contributions are available in the plan?

The following contribution types are available in the plan:

1. Employer Non-elective Contribution: This type of contribution is also known as a profit sharing contribution. Your employer may, at its discretion, make a profit sharing contribution to the plan.

Who may participate?

As an employee of Donner Summit Public Utility District, you may participate in the plan once you have met the eligibility requirements.

Who is considered an employee?

An employee is an individual who performs services for the employer as a common law employee, a self-employed individual who is treated as an employee, or a leased employee.

What are the eligibility requirements to become a participant in the plan?

There may be different eligibility and entry date requirements for each contribution type under the plan. Meeting all the eligibility requirements for one contribution type does not automatically make you eligible for other contributions under the plan. You may begin participating under the plan once you have satisfied the eligibility requirements and reached your entry date. The following describes the eligibility requirements and entry dates that apply. You should contact the plan administrator if you have questions about the timing of your plan participation.

Non-Elective Contributions:

Excluded Employees:

The following individuals are not eligible for purposes of Non-Elective contributions under the plan:

- 1. Union employees
- 2. Non-resident aliens

Eligibility Age Conditions:

You must attain age 18 to be eligible to receive non-elective contributions under the plan. However, you will not actually participate for purposes of receiving a non-elective contribution until you have reached the entry date as described below.

Eligibility Service Conditions:

You must complete one (1) year of service to be eligible to receive non-elective contributions under the plan. However, you will not actually participate for purposes of receiving a non-elective contribution until you have reached the entry date as described below.

If you are an eligible employee you will have met the service requirement as soon as they have met the hours of service requirement regardless of the eligibility computation period. However, you will not actually become a participant for purposes of non-elective contributions until you have reached the entry date as described below.

Entry Date:

For the purpose of receiving a non-elective contribution, your entry date will be the date you satisfy the eligibility requirements.

What compensation will be used for my contributions in the plan?

Compensation is defined as your total compensation that is subject to income tax and paid to you by the employer. Amounts paid to you after you terminate employment may or may not be included as compensation used to calculate your contributions as described below. If you are a self-employed individual, your compensation will be equal to your earned income from the employer. The following describes the adjustments to compensation that may apply for the contribution types permitted under the plan.

Non-Elective Contributions:

Compensation used to calculate your non-elective contributions under the plan will be determined as follows:

- * be based on W-2 wages.
- * exclude compensation due to cafeteria plan deferrals under section 125.
- * include compensation for your first year of participation from your entry date as a participant.

Is there a limit on compensation for plan purposes?

The IRS limits the amount of compensation that may be taken into account for each participant for each plan year. For 2022, that limit is \$305,000. For future years, the limit is subject to cost-of-living increases as published by the IRS.

<u>Does plan compensation include monies paid to me during an absence or after my employment ends?</u>

Usually, only the amounts paid to you while you are an employee are considered plan compensation (described above). However, the plan may consider certain types of pay as plan compensation, though paid during an absence or after you leave employment.

If you are totally and permanently disabled, compensation under your plan will not include disability-related salary continuation payments for Non-Elective contributions.

If you are not actively working for the employer due to military service, but are receiving compensation (differential wage payments) as if you were working for Donner Summit Public Utility District, those payments are included as compensation under your plan for Non-Elective contribution purposes.

If you are not actively working for the employer due to military service, but are receiving salary continuation payments (which is compensation that does not meet the definition of differential wage payments) as if you were working for Donner Summit Public Utility District, those payments are included as compensation under your plan for Non-Elective contribution purposes.

Payments you receive after terminating employment might be considered plan compensation for non-elective contribution purposes, if they meet the definition of "post-severance compensation." To be considered post-severance compensation, the payment must be one that you would have received had employment continued, such as your salary or wages. Post-severance compensation does not include severance pay, or other amounts you receive only because your employment ended.

To be included in plan compensation, post-severance compensation must be paid to you by the later of the end of the limitation year in which your employment ends, or within 2-1/2 months after the date your employment ends.

Payments for unused accrued sick, vacation, or other leave that you would have been able to use if your employment had continued are included in your plan's post-severance compensation.

How is service determined?

Eligibility Service:

You are credited with the actual hours you work, and for hours for which you are paid but not at work, such as paid vacation or paid sick leave.

If records of your hours are not maintained, as a backup method of crediting you with hours of service, you will be credited with 10 hours each day in which you work at least one hour.

Vesting Service:

You are credited with the actual hours you work, and for hours for which you are paid but not at work, such as paid vacation or paid sick leave. You cannot earn more than one year of vesting service during a plan year.

What is a year of service?

Eligibility Purposes:

You will earn a year of service for eligibility purposes if you are credited with 1000 hours. The first eligibility computation period is the 12-month period ending on the first anniversary of your employment commencement date. Subsequent eligibility computation periods will switch to plan years beginning with the first plan year commencing prior to the first anniversary of your employment commencement date.

Vesting Service Purposes:

You will earn a year of service for vesting service purposes if you are credited with 1000 hours in the vesting service computation period. The vesting service computation period will be the plan year.

What is a break in service for eligibility purposes?

When you fail to complete more than 500 hours during the eligibility computation period, you incur a break in service. However, in certain circumstances, your plan is required to credit you with 500 hours, even though you didn't actually work 500 hours. This is primarily if you take time off to have, adopt or care for a child for a period immediately following the birth or adoption. You will receive this credit only for the purpose of determining whether you have incurred a break in service and not for receiving additional credit for a contribution or for vesting.

What is a break in service for vesting purposes?

When you fail to complete more than 500 hours during the plan year, you incur a break in service. If you have incurred a break in service, your vesting percentage will not increase for the period in which the break occurs.

However, in certain circumstances, your plan is required to credit you with 500 hours, even though you didn't actually work that number of hours. This is primarily if you take time off to have, adopt or care for a child for a period immediately following the birth or adoption. You will receive this credit only for the purpose of determining whether you have incurred a break in service and not for receiving additional credit for a contribution or for vesting.

CONTRIBUTIONS

Does the plan accept rollovers?

Direct Rollover contributions are permitted from all employees who do not fall into an excluded class of employees.

What are profit sharing contributions?

Your employer may contribute a profit sharing contribution to the plan each year in an amount, if any, as it may determine.

Are there requirements to receive a profit sharing contribution?

To be eligible to receive an allocation of profit sharing contributions, you must complete 1000 hours of credited service in the plan year.

Are there any allocation requirements waived for profit sharing contributions?

There are no allocation condition waivers, you must meet all the allocation conditions described in the previous question to receive a profit sharing contribution.

How is the profit sharing contribution determined?

Your share of the profit sharing contribution is determined each year as a percentage of compensation or a dollar amount per participant. Your plan creates a separate employee classification group for each eligible employee.

When can I expect the profit sharing contributions to be allocated?

The profit sharing contributions made by your employer will be allocated to your profit sharing contribution account as of the last day of the plan year.

When can I expect the employer contributions to be deposited?

The employer contributions to the trust are normally paid directly to the Trust either during the plan year or after the close of the plan year (within the time during which the employer has to file its federal tax return).

When is a plan top heavy?

The plan becomes top heavy if more than 60% of the account balances are attributable to "key employees". Key employees are certain highly compensated officers or owner/shareholders.

Each year, the plan administrator will make a top heavy determination.

How will the plan operate in top heavy years?

If your plan is top heavy, all participants must receive a minimum contribution for such plan year, except participants who are covered by a collectively bargained agreement (union employees).

This top heavy contribution is based on the amount of contribution that the key employees receive and may be zero.

VESTING

Vesting is the non-forfeitable balance of your employer contribution account(s) that you will be entitled to receive after your employment with the employer ends. If you terminate employment before you meet the requirements for retirement, the distribution from your employer contribution account(s) will be limited to the vested portion. Your vesting percentage grows with your years of vesting service.

Is any of my service excluded for vesting purposes?

No, all years of service with your employer except those excluded due to a break in service will be included in determining your vested account balance.

How is my vested percentage calculated?

If you leave employment due to termination, you are entitled to a percentage of your employer contribution accounts along with earnings, based on the following schedules:

Employer Profit Sharing Contributions:

Vestina	Schedule

Years of Vesting Service	Percent Vested
Less than 2	0%
2 but less than 3	20%
3 but less than 4	40%
4 but less than 5	60%
5 but less than 6	80%
6 or more	100%

Employer Top-heavy Contributions:

Vesting Schedule

suring Correctains	
Years of Vesting Service	Percent Vested
Less than 2	0%
2 but less than 3	20%
3 but less than 4	40%
4 but less than 5	60%
5 but less than 6	80%
6 or more	100%

What vesting schedule applies to my other accounts?

The following contributions along with earnings associated with these accounts will be 100% vested and nonforfeitable at all times:

1. Rollovers

Does my vested percentage change for any reason?

You will become 100% vested in all of your accounts if you are still employed when you die, or become disabled.

Additionally, you will become 100% vested in all of your accounts when you reach your normal retirement age.

What happens if I terminate employment before I am fully vested?

The non-vested portion of your account will be forfeited and used to offset employer contributions at the

plan administrator's discretion in a nondiscriminatory and uniform manner.

The non-vested portion of your account that is forfeited may also be used to offset plan expenses or restore forfeited account balances of rehired employees.

The forfeiture takes place as of the end of the plan year in which you receive the final (complete) distribution of your distributable benefit or the end of the plan year of your 5th consecutive break in service. If you elect to receive less than your entire vested account balances from employer contributions, the portion forfeited will be prorated based on the portion of your total account balance distributed.

What happens to my forfeited amounts if I am rehired into a position covered by the plan?

If you were not vested (that is, 0% vested), when you severed employment, and you rejoin the plan before incurring a 5-year break in service, the amounts you forfeited will be restored as of your rehire date.

If you were partially vested (more than 0% but less than 100%), and received a distribution of your vested amounts, the forfeited amount may be restored. However, to restore the forfeiture, you must repay the full amount of your distribution of employer contributions by five (5) years after your rehire date, or if earlier, the date you incur a 5-year break following the date of the distribution.

If you are entitled to a restoration of your account balance that was forfeited, the plan will first use any forfeitures arising in the year of restoration. If that amount is not enough, the employer will make an additional contribution specifically allocated to your account.

If I am rehired into a position covered by the plan, how is my vesting service calculated?

If you were fully (100%) vested at the time your employment ended, you will resume participation and be 100% vested immediately, on your rehire date. This means that the vesting service you earned prior to severing employment (pre-break) will be added to the vesting service you earn after reemployment (post-break).

If you were not fully vested when your employment ended, the length of your break in service determines how your vesting service will be calculated and when you resume participation in the plan.

If your break in service is less than 5 years, your pre-break vesting service will be added to your post-break vesting service. Thus, your total years of vesting service are counted toward vesting in the employer contributions credited to your account after you return, and the pre-break non-vested employer account remaining in the plan, if you did not receive a distribution.

However, if you received a distribution from your employer account, and you would like to have your total years of vesting service (pre-break plus post-break) count toward vesting in your pre-break non-vested employer account, you must repay the full amount of your distribution by the earlier of: five (5) years after your rehire date, or the date you incur a 5-year break following the date of the distribution.

If your break in service is five years or more and you were not fully vested (less than 100%) when you ended your employment, when you are reemployed you will no longer have a vested interest in any prebreak non-vested employer account balance.

However, all your service (pre-break plus post-break) counts toward vesting in employer contributions credited after you are reemployed.

INVESTMENT ACCOUNTS

The money you deposit, if any and any employer contributions are held in a trust, and placed into investment accounts, which are credited with gains and losses at each valuation date.

What is the value of my account?

The value of each of your accounts is established as of the valuation date under your plan. The valuation date is the last day of the plan year for pooled accounts.

As of the valuation date:

- 1. contributions may be added to your accounts (see "Contributions");
- 2. distributions you have received since the prior valuation date will be subtracted from your accounts;
- 3. plan expenses may be subtracted from your accounts; and
- 4. interest and/or dividends, if any, will be added to your accounts.

Also, current market values will be reflected in your accounts as of the valuation date. Depending on stock and/or bond market conditions, the value of your accounts may increase or decrease from one valuation date to the next.

How are my accounts invested?

The plan will determine the investments to be used for all accounts and will direct the deposit of all contributions.

Please note that the trustee is considered the owner of all the assets held in the trust. The trustee, as owner of the securities and other trust property, has the exclusive right to vote the stock in the trust and exercise any other rights of ownership. As a plan participant, you merely have a beneficial interest in the trust and may not exercise the rights of ownership, as can the trustee.

Does my plan offer life insurance as an investment?

No. Life insurance policies are not available as a plan investment.

May I take a loan from my accounts?

Your plan permits loans. See Appendix 1 - Loan Policy attached to this SPD.

Where can I learn about the plan expenses?

Reasonable administrative expenses of the plan and trust may be paid by the plan to the extent not paid by the employer. For more information on plan expenses, refer to your copy of the plan's expense policy, provided by the plan administrator.

DISTRIBUTIONS

Does the plan allow for hardship distributions?

Hardship distributions are not permitted.

Does the plan allow for in-service distributions?

An in-service distribution is one that you receive while you are still employed by the employer. The primary purpose of the plan is to provide benefits to you upon your retirement; however, your plan permits you to request an in-service distribution.

You may request an in-service distribution of all or a portion of all of your accounts if you have attained age 59½.

You may request an in-service distribution of all or part of your rollover account at any time.

Distributions of amounts transferred from a money purchase plan, target benefit plan or defined benefit plan can be distributed upon attainment of age 62.

What is my normal retirement age?

You will reach the plan's normal retirement age when you reach age 65.

Your normal retirement date is the actual date normal retirement age is attained.

When will I receive my normal retirement benefits?

Payment of your benefits from employer contribution accounts will begin as soon as practicable following the date of your retirement, based on the preceding valuation date.

Payment of your benefits from your rollover account will begin as soon as practicable following the date of your retirement, based on the preceding valuation date.

When will my beneficiary receive my benefits if I die?

Payment of your benefits from employer contribution accounts will begin as soon as practicable following the date of your death, based on the preceding valuation date.

Payment of your benefits from your and rollover account will begin as soon as practicable following the date of your death, based on the preceding valuation date.

When will I receive my benefits upon termination?

Payment of your benefits from employer contribution accounts will begin as soon as practicable following your distribution determination date.

Your distribution determination date is your date of termination.

Payment of your benefits from your and rollover account will begin as soon as practicable following your distribution determination date.

Your distribution determination date is the date of your termination.

Does the plan have disability benefits?

You will be considered disabled if you suffer from a medically determinable physical or mental disability that is expected to result in death or to last a continuous period of 12 months that renders you incapable of performing your job duties. A determination of disability will be made by the plan administrator in a uniform, nondiscriminatory manner on the basis of medical evidence. You will also be considered disabled if the Social Security Administration has determined that you are eligible to receive Social Security disability benefits.

You become entitled to a distribution due to disability as of your actual date of disability, as determined by the plan administrator.

If it is determined you are entitled to a distribution due to disability, payment of your benefits will begin as soon as practicable following such determination based on your account value on the preceding valuation date.

How might divorce or a Qualified Domestic Relations Order affect my benefits?

Because your spouse has certain rights under your plan, you should immediately inform the plan administrator of any changes in your marital status.

In general, contributions made by you or your employer to this plan are not subject to alienation. This means they cannot be sold, used as collateral for a loan, given away or otherwise transferred. They are not subject to the claims of your creditors. However, they may be subject to claims under a Qualified Domestic Relations Order (QDRO).

A Domestic Relations Order is a court-issued decree or an order that allocates all or any portion of your plan benefits to your (former) spouse, your child, or other dependent. It is the plan administrator's responsibility to determine if a Domestic Relations Order is qualified (is a QDRO), as defined by law.

Distributions pursuant to a Qualified Domestic Relations Order are permitted on or after the date a Domestic Relations Order is determined to be a Qualified Domestic Relations Order, even if you are employed and have not attained the "earliest possible retirement age" (as defined below).

For QDRO purposes, the "earliest possible retirement age" means the earlier of these two dates:

- 1. the date you are entitled to a distribution; or
- 2. the later of:
 - A. the date you reach age 50; or
 - the earliest date you could begin receiving benefits under the plan if you separated from service.

Participants and beneficiaries can obtain, from the plan administrator, without charge, a copy of the plan's procedures governing Qualified Domestic Relations Orders.

How will I receive my distribution?

You may elect a lump sum distribution.

Will the plan automatically distribute any of my benefits?

The plan will make a mandatory distribution if your vested account balance is \$5,000 or less. The distribution will be made as soon as administratively feasible after the last day of the plan year. Any account balance that is from a rollover that you have transferred into the plan will not be taken into account in the determination of your total vested account balance for purposes of the mandatory distribution threshold.

If you do not provide payment instructions, the plan will automatically roll your distribution over to an IRA if your account balance is greater than \$1,000. If your account balance is less than \$1,000, your vested

account balance will be paid directly to you or your beneficiary. Any account balance that is from a rollover that you have transferred into the plan will not be taken into account in the determination of your total account balance for purposes of the automatic rollover threshold. The plan administrator will notify you if the automatic rollover provisions apply to your distribution. After receiving this notice, you will have an opportunity to decide whether you wish to receive your distribution directly in cash or roll it into an eligible retirement plan or IRA.

The automatic rollover will be invested in an investment product designed to preserve principal and provide a reasonable rate of return, whether or not such return is guaranteed, consistent with liquidity.

What is a required minimum distribution?

Under certain circumstances, the law requires that your distributions begin no later than April 1 of the year following the date you reach age 70-1/2 (the date six months after your 70th birthday) if you are a greater than 5% owner born before July 1, 1949. If you are not a greater than 5% owner, these distributions will be delayed until you retire. If you were born after June 30, 1949, age 70-1/2 is replaced with age 72 where it appears above. Your plan administrator will contact you if you are affected by this requirement.

How will my distributions be taxed?

The benefits you receive from the plan will be subject to ordinary income tax in the year in which you receive the payment, unless you defer taxation by a "rollover" of your distribution into another qualified plan or an IRA. Also, in certain situations, your tax may be reduced by special tax treatment such as "10-year forward averaging."

VERY IMPORTANT NOTE: Under most circumstances, if you receive a distribution from this plan, twenty percent (20%) of your distribution will be withheld for federal income tax purposes, unless you instruct the trustees of this plan to transfer your distribution DIRECTLY into another qualified plan or an IRA. You must give these instructions to the trustees no more than 180 days before the date you receive the payment. Also, the trustees must wait at least 30 days after receiving your instructions before making the payment, to allow you time to change your decision, unless you waive the waiting period in writing.

In addition to ordinary income tax, you may be subject to a 10% tax penalty if you receive a "premature" distribution. If you receive a distribution upon terminating employment before age 55 and you don't receive the payment as a life annuity, you will be subject to the 10% penalty unless you roll over your payment. But, there is no penalty for payments due to your death or disability.

As the rules concerning "rollovers" and the taxation of benefits are complex, please consult your tax advisor before making a withdrawal or requesting a distribution from the plan. As required by law, the plan administrator will provide you with a brief explanation of the rules concerning "rollovers."

Who may I name as my beneficiary?

The plan requires that your spouse be your primary beneficiary and receive 100% of your account balance on your death (see vesting section). You may name someone other than your spouse as your primary beneficiary only if your spouse gives written consent to your choice of beneficiary. A notary public or plan representative must witness your spouse's signature on the consent form. You have a right to designate your primary and contingent beneficiary or beneficiaries at any time by completing a beneficiary form that is provided to you or is acceptable to the plan administrator.

It is important that you keep your designation of beneficiary up-to-date. If you fail to designate a beneficiary, or if your beneficiary designation is not valid, or if all of your beneficiaries fail to survive you, then your benefits will be paid The participant's spouse; if no spouse, the participant's children; if no children, the participant's estate.. If you get divorced, your ex-spouse will be treated as having predeceased you and your benefits will be paid to your contingent beneficiary unless you make a post-divorce designation naming your ex-spouse as a beneficiary.

OTHER IMPORTANT INFORMATION

Are my benefits protected?

Except for the requirements of a Qualified Domestic Relations Order, your plan benefits are not subject to claims, indebtedness, execution, garnishment or other similar legal or equitable process. Also, you cannot voluntarily (or involuntarily) assign your benefits under this plan.

Can the plan be amended or terminated?

The employer has reserved the right to amend or terminate the plan. However, no amendment can take away any benefits you have already earned. If your plan is terminated, you will be entitled to the full amount in your account as of the date of termination, regardless of the percent you are vested at the time of termination.

Does Pension Benefit Guaranty Corporation Insurance apply to this plan?

The benefits provided by this plan are not insured by the Pension Benefit Guaranty Corporation (PBGC). Such insurance is only required under Title IV of the Employee Retirement Income Security Act (ERISA) for defined benefit pension plans.

What are the claims for benefits procedures under this plan?

When you request a distribution of all or any part of your account, you will contact the plan administrator who will provide you with the proper forms to make your claim for benefits. Your claim for benefits will be given a full and fair review.

What is the waiting period for a non-disability claim?

The plan administrator will notify you or your beneficiary of the denial within a reasonable period of time, but not later than ninety (90) days of the date your claim for benefits was received. The plan administrator may extend this deadline by up to ninety (90) days if there are special circumstances beyond the control of the plan that require additional time to process the claim. If a delay occurs, you or your beneficiary will be notified in writing of the reason for the delay and a date by which a final decision will be given (not more than one hundred and eighty (180) days after the receipt of your claim.)

What is the waiting period for a disability claim?

If the claim is for disability benefits, the plan administrator will notify you or your beneficiary within a reasonable period of time, but not later than forty-five (45) days after the date your claim was received. The plan administrator may extend this deadline by up to thirty (30) days if there are special circumstances beyond the control of the plan that require additional time to process the claim. If a delay occurs, you or your beneficiary will be notified in writing before the end of the initial forty-five (45) day period.

If the plan administrator determines that, due to matters beyond the control of the plan, a decision cannot be made within that time, the period may be extended for up to an additional thirty (30) days. You or your beneficiary will be provided with notification prior to the expiration of the first thirty (30) day extension period of the circumstances requiring the extension and the date that the plan expects to make a decision.

After receipt of an extension notice the participant or beneficiary will have one hundred and eighty (180) days to appeal such denial. Upon receipt of such appeal, the plan administrator must act within forty-five (45) days.

What will I receive if the claim is denied in part or whole?

The initial denial letter and any appeal denial letter will be provided to your or your beneficiary with the following information:

- 1. the standards on which the determination is being made
- 2. the unresolved issues that prevent the plan administrator from making the decision
- 3 the additional information that would be needed to allow the plan administrator to make the decision
- 4. reference to the plan provision(s) on which the denial is based
- 5. an explanation of the denial when the claim included a disability determination by the Social Security Administration or other third party disability payer, or any views of health care professionals, medical, or vocational professionals providing treatment or advice (regardless of whether the advice was relied upon)
- 6. an explanation of the scientific or clinical judgment for the determination if the denial is based on a medical necessity, experimental treatment, or similar exclusion or limit, or statement that the explanation is available free of charge upon request
- 7. all internal rules, guidelines, protocols, standards, or other similar criteria that were relied upon in denying the claim or a statement that such criteria do not exist
- 8. a statement that you are entitled to receive, upon request and free of charge, relevant documents

What happens to my disability claim if I need to provide information?

The plan administrator will notify you or your beneficiary if additional information is needed from you in order to complete the claim. You will have 45 days to respond with the needed information. The plan administrator's timing to process the claim will not be counted during this period.

How do I appeal a claim denial?

You or your beneficiary may file a written appeal of the claim denial within sixty (60) days (forty-five (45) days for disability claims) to the plan administrator. You may submit new information relating to the claim. The employer may hold a hearing or otherwise review facts as it deems necessary and shall make a decision, which shall be binding upon both parties.

The decision of the employer shall be made within sixty (60) days (forty-five (45) days for disability claims) after the receipt by the plan administrator of the notice of appeal, unless special circumstances require an extension of time for processing, in which case a decision of the employer shall be rendered as soon as possible but not later than one hundred twenty (120) days (ninety (90) days for disability claims) after receipt of the request for review.

In the case of an appeal denial letter, the letter must describe any contractual limitation period for a lawsuit and the expiration date for that limitation period along with a statement that the limitation period may not expire before the conclusion of the Plan's internal appeals process.

You may request a free of charge access to, copies of any information, records, and/or documents used to deny the claim.

Once you have exhausted the administrative remedies for claim, or the plan fails to establish or follow claims procedures consistent with plan regulations, it is your right to challenge the decision under section 502(a) of Employer Retirement Income Security Act of 1974 (ERISA) or other applicable law.

PARTICIPANT RIGHTS UNDER ERISA

As a participant in Donner Summit Public Utility District Money Purchase Pension Plan, you are entitled to certain rights and protections under the Employee Retirement Income Security Act of 1974 (ERISA). ERISA provides that all plan participants shall be entitled to:

Receive information about your plan and your benefits:

- 1. Examine, without charge, at the plan administrator's office all documents governing the plan and a copy of the latest annual report filed by the plan with the U.S. Department of Labor.
- 2. Obtain copies of all plan documents and other plan information upon written request to the plan administrator (the administrator may make a reasonable charge for the copies).
- 3. Receive a summary of the plan's annual financial report. The plan administrator is required by law to furnish each participant with a copy of this summary annual report.
- 4. Obtain a statement telling you whether you have a right to receive a benefit at normal retirement age and if so, what your benefits would be at normal retirement age if you stop working under the plan now. If you do not have a right to a benefit, the statement will tell you how many more years you have to work to get a right to a benefit. This statement must be requested in writing and is not required to be given more than once a year. The plan must provide the statement free of charge.

Actions by Plan Fiduciaries:

In addition to creating rights for plan participants, ERISA imposes duties upon the people who are responsible for the operation of the plan. The people who operate your plan, called "fiduciaries" of the plan, have a duty to do so prudently and in the interest of you and other plan participants and beneficiaries.

No one, including your employer may fire you or otherwise discriminate against you in any way to prevent you from obtaining a retirement benefit or exercising your rights under ERISA.

Enforcing your rights:

If your claim for a benefit is denied in whole or in part, you have the right to know why this was done and to obtain copies of documents relating to the decision without charge, and to appeal any denial, all within certain time schedules.

Under ERISA, there are steps you can take to enforce the above rights. For instance, if you request written materials from the plan and do not receive them within 30 days, you may file suit in a federal court. In such a case, the court may require the plan administrator to provide the materials and pay you up to \$110 a day until you receive the materials, unless the materials were not sent because of reasons beyond the control of the administrator.

If you have a claim for benefits which is denied or ignored, in whole or in part, you may file suit in a state or federal court. In addition, if you disagree with the plan's decision or lack thereof concerning the qualified status of a domestic relations order or a medical child support order, you may file suit in federal court.

If it should happen that plan fiduciaries misuse the plan's money, or if you are discriminated against for asserting your rights, you may seek assistance from the U.S. Department of Labor, or you may file suit in a federal court. The court will decide who should pay court costs and legal fees. If you are successful, the court may order the person you have sued to pay these costs and fees. If you lose, the court may order you to pay these costs and fees, for example, if it finds your claim is frivolous.

Assistance with your questions:

If you have questions about your plan, you should contact the plan administrator. If you have any questions about this statement or your rights under ERISA, or if you need assistance in obtaining documents from the plan administrator, you should contact the nearest office of the Employee Benefits Security Administration, U.S. Department of Labor, listed in your telephone directory or the Division of Technical Assistance and

July 27, 2022 Agenda Item: 5A ATTACHMENT 2

Inquiries, Employee Benefits Security Administration, U.S. Department of Labor, 200 Constitution Avenue N.W., Washington, D.C. 20210. You may also obtain certain publications about your rights and responsibilities under ERISA by calling the publications hotline of the Employee Benefits Security Administration.

APPENDIX 1 - LOAN POLICY

Pursuant to the terms of Donner Summit Public Utility District Money Purchase Pension Plan (the "Plan"), the Plan Administrator has adopted a participant loan program as part of such Plan and Trust. All loans granted or renewed on or after the 1st day of January, 2022 will be made in accordance with the provisions specified in the Plan and under this Loan Policy. The Plan intends this loan program to comply with all applicable requirements of the Internal Revenue Code and the Department of Labor. Violating the terms of this Policy may cause a loan to be treated as a taxable distribution from the Plan.

Administration of Program.

Donner Summit Public Utility District (the "Loan Administrator") is responsible for the administration of this loan program. All loan requests and other inquiries should be delivered to:

Donner Summit Public Utility District 53823 Sherritt Lane Soda Springs, CA 95728 (530) 426-3456

Application Procedure.

The Loan Administrator will make loan applications available to any individual who has a vested interest under the Plan. An eligible individual, as defined within this Policy, may apply for a loan by returning a paper loan application.

Loan applications will be processed as soon as administratively feasible following receipt of the completed loan application.

The Loan Administrator will review the loan application for completeness. Incomplete applications will be returned and must be resubmitted for consideration. All loan applicants must meet the requirements of this Policy for consideration and approval. If the loan applicant fails to meet the requirements of this Policy and receives a loan disbursement, the loan will be treated as a "deemed distribution" and reported as taxable to the IRS.

Basis for Approvals.

Loans are available to all actively employed participants.

Loans will be available without regard to any individual's race, color, religion, sex, age, or national origin.

Each application is reviewed on a nondiscriminatory basis. However, its approval will depend on the applicant's creditworthiness.

If an applicant defaulted on a previous loan, the loan application will be denied until the deemed loan has been repaid or offset.

In addition, if a participant submits a loan application at a time when a decision concerning a domestic relations order is pending or the Plan Administrator is on notice that divorce is in progress, the loan request will be placed on hold until the order is finalized or the determination period expires.

Once the loan is approved, a Promissory Note will be generated and issued to the applicant. The applicant must sign the Promissory Note to acknowledge and document receipt of the loan disbursement from the Plan and to affirm such applicant's obligation to make the required repayments.

Spousal Consent.

If a married participant submits a loan application for an amount greater than \$5,000, spousal consent must be obtained no earlier than 90 days before the date on which the loan is processed and secured. The consent must be in writing and properly executed on the loan application and Promissory Note. The participant's spouse is required to acknowledge consent to the participant's pledge and assignment of 50% of the participant's vested interest in the Plan; and to the effect such consent will have on the benefit that would otherwise be paid under the Plan, in the event of a default. The spouse must sign the consent in the presence of a Notary Public or the Plan Administrator who can attest to the spouse's willingness to consent and endorse the loan documents. A new consent must be obtained if the amount of a loan is increased or if a loan is renegotiated, extended, renewed or otherwise revised.

Terms of the Loan.

Subject to the limitations on the amount of any loan, a participant may request a loan for any purpose.

In addition, a participant may rollover a loan from another employer sponsored plan into this plan. The loan must be in good standing (payments are not delinquent and loan has not been deemed distributed). The participant must contact the Loan Administrator to complete the necessary forms and to provide documentation from the other plan regarding the terms and status of the loan. The Loan Administrator retains the ability to deny a rollover in of a loan if insufficient information is provided to ensure the loan has been administered in compliance with all IRS and Department of Labor requirements.

Loan Amounts.

The Loan Administrator will determine the available loan amount at the time the loan request is approved. The maximum amount of any loan is the lesser of 50% of the participant's vested interest minus any existing loan balance, or \$50,000 reduced by the participant's highest outstanding loan balance in the previous twelve months even if all or a portion of this amount has been repaid.

To calculate the maximum loan amount, the participant's vested interest in all plans established or maintained by the employer or a related employer of the employer will be considered. Notwithstanding, the Plan limits the minimum amount of any loan to \$1,000.

An individual may have no more than 5 loans outstanding at any one time. Loans that have been deemed distributed will be counted as outstanding loans until they have been either repaid or offset. Subject to the limitations on the amount and number of loans permitted, refinancing of an existing loan is permitted to change the payment schedule, to change the payment terms, and to replace an existing loan.

Sources for a Loan - Accounts and Investment Options.

The participant will select, on a form provided by the Loan Administrator, the accounts and investment fund or funds from which the amount necessary to grant the loan will be taken. However, the loan will not be taken from any insurance policies.

Loans may be taken from all vested participant accounts.

The loan will be transferred to a segregated account. During the term of the loan, this segregated account will be maintained, and each scheduled principal and interest repayment will be made to this segregated account, until the entire loan is paid in full. This segregated account will not share in any gains or losses credited to the Plan that do not directly relate to the loan.

Interest Rate and Fees.

Interest will be charged on each loan. From time to time, the Loan Administrator will review the interest rate charged for loans, with the intention of providing the Plan with a return commensurate with the interest rates that a commercial lender would charge for loans made under similar circumstances. The interest rate for a loan will take into account the creditworthiness of the participant and the terms of the loan.

The interest rate on the loan will be the prime rate as used by Any federal reserve bank adjusted by 1 percentage point(s). The prime rate will be determined

Once the interest rate is determined, the amount of the loan will be amortized according to the selected repayment terms. Each repayment will include both principal and interest until the loan is no longer outstanding in the Plan.

To cover the added administrative costs associated with processing and maintaining a loan under the Plan, the participant will be charged a \$95.00 loan processing fee, and annually, a loan maintenance fee of \$35.00, until the loan is paid in full. In addition, the loan maintenance fee will be charged for all years in which the loan remains in default.

Security for a Loan.

All loans must be adequately secured with at least 50% of the participant's vested account balance in the Plan. The security interest will be determined and measured at the time the loan is granted. The participant must secure each loan with an irrevocable pledge and assignment of at least 50% of such participant's vested account balance under the Plan.

Repayment Terms.

With limited exceptions, the Internal Revenue Code requires a loan to be repaid through level installment payments at least quarterly, over a period not to exceed 5 years.

Under this Loan Policy, a loan is required to be repaid within 5 years, starting from the payment date outlined in the Promissory Note.

However, if the loan application is for a residential loan and the Loan Administrator confirms that there is sufficient documentation that the entire proceeds of the loan will be used to acquire a dwelling unit that will be used as the participant's principal residence, within a reasonable time, then this residential loan must be repaid within 30 years of the original date of the loan.

A principal residence is a house, apartment, condominium, or mobile home (not used on a transient basis) established and used as the participant's principal dwelling unit.

Loans are to be repaid based on substantially level amortization over the term of the loan with payments made each pay period.

Loan payments will be made through payroll deduction.

Early Payoff.

The participant may elect, in writing to the Loan Administrator, to pay off an entire outstanding loan balance in full prior to its due date.

Leave of Absence or Suspension Period.

Loan payments may be suspended for up to one year if the participant is on an approved leave of absence from the employer either without pay or with a pay rate that will not cover the scheduled installment payments. When the participant's leave has ended, the participant may request that the missed payments be reamortized over the remaining term of the loan or be added to the final payment. The loan must be repaid according to its original repayment term or within 5 years as provided by Internal Revenue Code section 72(p)(2)(B).

Special Provisions for Military Service.

The Loan Administrator may temporarily suspend loan repayments, if the participant is not actively employed due to a qualified military leave or because the participant is performing service in the uniformed services (as defined in chapter 43 of title 38 United States Code). In addition, the period of military suspension will extend the original loan term.

Once the military service has ended, loan repayments must resume. The loan must be repaid in full by the end of its original term plus the period of military service.

For example, if the loan was due in 5 years, and the military leave was for 18 months, then the Loan Administrator would extend the 5 year term by the length of the military leave. The final installment payment would be due within 6 years and 6 months of the date the loan was originally issued.

In addition, upon receipt of proper notice of active military service, if the interest rate for an outstanding loan exceeds 6%, the Loan Administrator will reduce the interest rate on such loan to 6% during the period of military service.

Default and Offset.

A loan is in default when a scheduled installment payment has not been received by the scheduled due date. If the participant fails to arrange for the repayment of the missed payment, in a manner that is reasonably acceptable to the Loan Administrator, the remaining principal and accrued interest on the loan will be declared due and payable. The missed payment must be received by the end of the "cure period". The end of the cure period will be the last day of the calendar quarter following the calendar quarter in which the scheduled installment payment was due. After this date, the Loan Administrator will notify the participant in writing that the loan is in default.

The outstanding loan (including accrued interest) will become taxable and treated as a "deemed distribution". A deemed distribution means the entire outstanding loan balance is immediately due and the Loan Administrator will report and process the outstanding loan amount as a taxable distribution from the Plan. The defaulted loan (outstanding principal plus accrued interest thru the date the loan was deemed distributed) will be reported as taxable income on IRS Form 1099-R. It will be subject to federal and state income taxes, and a 10% additional tax on early distributions if the default occurs before age 59-1/2. This amount will not be eligible for rollover to another employer plan or IRA.

The participant is still under an obligation to the Plan to repay the loan. Therefore, the Promissory Note will remain outstanding and interest will continue to accrue. This outstanding loan obligation will be offset against the participant's vested account balance when the participant requests a distribution of their account balance.

A participant whose loan is in good standing (not deemed distributed) may elect to take an in-kind distribution of the promissory note (offset) provided they are able to take a distribution of their account balance(s) that were used to fund the loan without requiring a distribution of non-loan amounts. This will result in full outstanding amount of the loan (including interest accrued thru the distribution date) becoming taxable. The amount will be reported on IRS Form 1099-R and will be eligible for rollover to another employer plan or IRA.

July 27, 2022 Agenda Item: 5A ATTACHMENT 3

RESOLUTION NO. 2022-28

A RESOLUTION OF THE BOARD OF DIRECTORS OF THE DONNER SUMMIT PUBLIC AMENDING AND RESTATING THE MONEY PURCHASE PENSION PLAN

WHEREAS, the Donner Summit Public Utility District (Employer) has the power to restate the Plan and desires to restate the plan with an effective date of January 1, 2022.

NOW, THEREFORE, THE BOARD OF DIRECTORS OF DONNER SUMMIT PUBLIC UTILITY DISTRICT DOES HEREBY RESOLVE AS FOLLOWS:

- 1. The form of Plan presented to the Board is a Profit Sharing Plan as authorized under Code sections 401(a) and 501(a). This restatement shall be effective January 1, 2022.
- 2. The form of the Plan presented to the Board is a restatement intended to meet the requirements as listed in IRS Notice 2017-37 as required in IRS Revenue Procedure 2016-37.
- 3. The Plan presented to the Board is hereby adopted and approved and that the proper officers of the Employer are hereby authorized and directed to execute and deliver to the Plan Administrator one or more counterparts of the Plan.
- 4. The Summary Plan Description presented to the Board is hereby approved. The proper Officers of the Employer shall act as soon as possible to notify employees of the Employer of the restatement of the Plan by delivering to each employee a copy of the Summary Plan Description.
- 5. The Loan Policy presented to the Board with this Plan is hereby adopted and approved with an effective date of January 1, 2022 for delivery to each Participant.

PASSED AND ADOPTED by the Board of Directors of Donner Summit Public Utility District, this 27th day of July 2022, by the following vote:

AYES: NOES: ABSENT: ABSTAIN:	
	DONNER SUMMIT PUBLIC UTILITY DISTRICT
	By: Cathy Preis President, Board of Directors
ATTEST:	1 resident, beard of bilectors
By: Alex Medveczky Secretary of the Board	